

This item was submitted to Loughborough's Institutional Repository (<u>https://dspace.lboro.ac.uk/</u>) by the author and is made available under the following Creative Commons Licence conditions.



For the full text of this licence, please go to: http://creativecommons.org/licenses/by-nc-nd/2.5/

Holocene environmental change in coastal Denmark: interactions

between land, sea and society

by

Jonathan Paul Lewis

A doctoral thesis Submitted in partial fulfilment

of the requirements for the award of

Doctor of Philosophy of Loughborough University

21st July, 2011

© by Jonathan Lewis 2011

Table of Contents

Table of contents	1
List of Figures	5
List of Tables	12
Abstract	14
Acknowledgements	15

Chapter	1.	Introduction, background and aims	.17
1.1	Intro	oduction	.17
1.2	The	modern Baltic Sea, Kattegat and inner Danish coastal waters	.20
1.3	Sea	Level and Salinity change during the Littorina Sea Stage	.25
1.3.	1	The Ancylus Lake-Littorina Sea transition (7,850-6,550 BC)	.25
1.3.	2	The Littorina Sea (~6,550 BC – present)	.27
1.4	Driv	vers of Salinity Change in the Baltic	.29
1.5	Pala	eosalinity records from the Baltic Sea, Kattegat and coastal waters	.30
1.6	Cult	ural change in Denmark	.36
1.6.	1	The Late Mesolithic and the accumulation of Stone Age shell middens	. 39
1.6.2	2	Salinity decline at the Mesolithic-Neolithic transition; a long outstanding issu	e .45
1.6.	3	The Pitted Ware and Single Grave Culture (~2,800-2,400 BC)	.51
1.6.4	4	The Bronze Age	.52
1.6.: mid	5 dens	The Iron Age (~500 BC-AD 1050) and the re-appearance of coastal shell	.52
1.7	Proj	ect aims and proxies	.54
1.7.	1	Proxies	.56
Chapter	2.	Study sites and methodology	. 59
2.1	Intro	oduction	. 59
2.2	Stuc	ly Area	. 59
2.2.	1	Site selection	. 59
2.2.2	2	Site descriptions	.64
2.3	Mat	erial and Methods	.71
2.4	Field	d work, core treatment and sub-sampling	.72

2.4	4.1 Kilen	72
2.4	1.2 Norsminde	76
2.4	4.3 Korup Sø	76
2.5	Core stratigraphy, dating and chronology	77
2.5	5.1 ¹⁴ C dating; Kilen, Norsminde and Korup Sø	77
2.5	5.2 210 Pb and 137 Cs dating; Kilen	77
2.5	5.3 Core Correlation	78
2.6	Laboratory Analysis	79
2.6	5.1 Physical analyses: Kilen and Norsminde	79
2.7	Biological Proxies	
2.7	7.1 Diatom analysis: Kilen, Norsminde & Korup Sø	
2.7	7.2 Molluscan analysis: Kilen, Normsinde and Korup Sø	
2.7	7.3 Foraminifera analysis; Kilen and Norsminde	
2.7	7.4 Pollen and plant macrofossil analysis: Kilen, Norsminde and Koro	up Sø83
2.8	Chemical analyses	
2.8	3.1 Sedimentary Pigments: Kilen	
2.9	δ^{18} O & δ^{13} C isotope analysis of single species of foraminifera: Kilen	
2.10	Numerical analyses	
2.1	10.1 Statistical zonation of sedimentary and biological datasets	
2.10	Quantitative reconstructions of salinity: the MOLTEN salinity tra87	nsfer function
2.1	10.3 Multivariate analysis and environmental-species relationships	
2.11	Summary	90
Chapter	r 3. Results: Kilen	92
3.1	Introduction	92
3.2	Lithology	
3.3	Chronology and Age Model	95
3.4	Physical Parameters	100
3.5	Diatoms	
3.6	Sedimentary pigments	111
3.7	Macrofossils	116
3.7	7.1 Molluscs	116
3.7	7.2 Plant macrofossils	
3.7	7.3 Foraminifera	
3.8	Isotopes from <i>Elphidium excavatum</i> fo. <i>selseyensis</i>	126

3.9	Other organisms	
3.10	Statistical analyses: DCA	
3.11	Synthesis of results and site discussion	
3.12	Summary and conclusions	
Chapter	4. Results: Norsminde	
4.1	Introduction and aims	
4.2	Core Lithology	
4.3	Age Model	161
4.4	Physical parameters	
4.5	Diatoms	166
4.5.	.1 Diatom Preservation	171
4.6	Foraminifera	171
4.7	Molluscs	
4.8	Pollen	
4.9	Plant Macrofossils	
4.10	Statistical analysis	
4.10	0.1 Trends in species datasets	
4.10	0.2 Environmental-species relationships	
4.11	Synthesis of Results and site discussion	
4.12	Summary and conclusions	
Chapter	5. Results: Korup Sø	
5.1	Introduction and aims	
5.2	Core Lithology	
5.3	Age Model	
5.4	Sediment accumulation rate	
5.5	Biological Analyses	
5.5.	.1 Diatoms	
5.5.	.2 Molluscs	
5.5.	.3 Pollen	
5.6	Statistical analysis	
5.6.	.1 Trends in species datasets; DCA and PCA analyses	
5.6.	.2 Environmental-Species relationships	
5.7	Synthesis of results and discussion of site	
5.8	Summary	
Chapter	6. Discussion	

6.1	Introduction	.260
6.2	Interactions between environmental variables and their impact upon biological	
commu	nities	.261
Diato	om-salinity model	.263
6.2.1	Diatom preservation	.268
6.3 l	Environmental change in coastal Denmark and interactions between land, sea and	t
society	; ~7,600-1,700 BC	.271
6.3.1	The marine transgression and the early Mesolithic Maglemose (9,800-6,800	
BC) a	and Kongemose (6,800-5,400 BC) cultures	.276
6.3.2	From Ertebølle (~5,400-3,900 BC) to Funnel Beaker Culture (TBK; ~3,900-	
2,800) BC) and the Mesolithic-Neolithic transition (~3,900 BC)	.277
6.3.3	The Pitted Ware and Single Grave cultural period (~2,800-2,400 BC)	. 292
6.4 I	Environmental change in the Limfjord region; ~1,700 BC – present	. 295
6.4.1	The Bronze Age (1,700-500 BC)	.295
6.4.2	The Iron Age shell midden accumulation period (~500 BC-AD 1050)	.295
6.4.3	The Limfjord during the Historical Period (AD 1050-present)	. 302
6.4.4	The limnic phase of Kilen (AD 1856-present)	. 303
6.5	Conclusions	. 304
Chapter 7	. References	. 309

Appendix A	
------------	--

List of Figures

FIGURE 1.1 - MAP OF THE BALTIC SEA, WITH REGIONAL MODERN DAY AVERAGE SURFACE SALINITIES AND PRESENT CURRENT CIRCULATION SYSTEM FOR THE NORTH SEA, SKAGERRAK AND KATTEGAT (AFTER GYLLENCREUTZ *ET AL.*, 2005 AND LONGVA AND THORSNES, 1997A). SITES ANALYSED IN THIS THESIS ARE INDICATED BY RED DOTS. THE BLUE DOT SIGNIFIES THE LOCATION OF CORE MD99-2286 COLLECTED FROM THE SKAGERRAK (GYLLENCREUTZ *ET AL.*, 2005). ABBREVIATIONS: A = LILLEBÆLT, B = STOREBÆLT, C = ØRESUND, CNSW = CENTRAL NORTH SEA WATER, AW = ATLANTIC WATER, NCC = NORWEGIAN COASTAL CURRENT, TBC = TAMPEN BANK CURRENT, SJC = SOUTH JUTLAND CURRENT, NJC = NORTH JUTLAND CURRENT, BW = BALTIC WATER, BC = BALTIC CURRENT, CNSW = CENTRAL NORTH SEA WATER, SNSW = SOUTHERN NORTH SEA WATER. MAP DOWNLOADED FROM HELCOM:

FIGURE 1.2 - SUMMARY OF BALTIC SEA LEVEL, SALINITY AND TOTAL ORGANIC CARBON (TOC) HISTORY. A= TRADITIONAL DIVISION OF THE BALTIC SEA'S DEVELOPMENT STAGES (CF. HYVÄRINEN, 1988). B= CURRENT BALTIC SEA STAGE DIVISIONS (AFTER ANDRÉN ET AL., 2000; BERGLUND ET AL., 2005; ANDRÉN ET AL., 2007B). C= REPRESENTATIVE DIATOM ASSEMBLAGES FROM OPEN WATERS (AFTER ANDRÉN ET AL., 2000). T. = THALASSIOSIRA (COLUMN C). D= SALINITY STAGES OF THE BALTIC SEA'S HISTORY. E= TOTAL ORGANIC CARBON (TOC) CURVE FROM THE GOTLAND BASIN (ANDRÉN ET AL., 2000). F= BLEKINGE SEA LEVEL CURVE BASED ON SHORE DISPLACEMENT (FROM BLEKINGE SE SWEDEN, BERGLUND ET AL., 2005; ANDRÉN ET AL., 2007B); L1-L5 REFER TO LITTORINA TRANSGRESSIONS AS RECORDED IN COASTAL SEDIMENTS FROM THE BLEKINGE REGION OF SOUTH-EASTERN SWEDEN (AFTER BERGLUND ET AL., 2005). G. SUMMARY OF MAJOR MARINE EVENT AND MAIN SEDIMENT SOURCE AND TRANSPORT PATHWAYS INTO THE SKAGERRAK (CORE MD99-2288) OVER THE LAST 12,000 YEARS (AFTER GYLLENCREUTZ AND KISSEL, 2006). 1. GLACIAL MARINE SEDIMENT FROM MELTWATER DISCHARGE (SOURCE: SOUTHERN NORWAY AND SW SWEDEN). 2. GLACIAL MARINE SEDIMENT FROM MELTWATER OUTFIOW (SOURCE: BALTIC SEA VIA THE VANERN BASIN). 3. MARINE SEDIMENT (SOURCE: EASTERN NORTH SEA VIA THE NORTH JUTLAND CURRENT). 4. MARINE SEDIMENT (SOURCE: CHIEflY FROM THE SOUTHERN NORTH SEA AND THE ATLANTIC OCEAN, TRANSPORTED BY THE NORTH AND SOUTH JUTLAND CURRENT, BUT ALSO FROM THE BALTIC SEA AND REWORKED COASTAL SEDIMENTS IN SWEDEN AND NORWAY TRANSPORTED BY THE BALTIC CURRENT AND CURRENTS ALONG THE COASTS OF WESTERN SWEDEN AND SOUTHERN NORWAY). 5. MARINE SEDIMENT (SOURCE: CHIEfIY FROM THE SOUTHERN NORTH SEA AND THE ATLANTIC OCEAN, TRANSPORTED BY THE NORTH AND SOUTH JUTLAND). IRD = ICE RAFTED DEBRIS. NS = OPENING OF NÄRKE STRAIT, OS = CLOSING OF OTTEID STENSELVA STRAIT. EC = OPENING OF THE ENGLISH CHANNEL (ALONG WITH INCREASED FLOW OF ATLANTIC WATER, OPENING OF THE DANISH STRAITS AND BEGINNING OF THE MODERN CIRCULATION SYSTEM). 8.2K = LARGE SCALE COOLING EVENT RECORDED IN THE GREENLAND ICE CORES (ALLEY ET AL., 1997; ROHLING AND PÄLIKE, 2005). S-K SHIFT = SKAGERRAK-KATTEGAT HYDROGRAPHIC SHIFT, MS = BEGINNING OF MODERN SEDIMENTATION IN THE SKAGERRAK. BOTTOM CURRENT VELOCITY IS BASED ON MEDIAN SORTABLE SILT IN THE MD99-2288 CORE WITH DOTTED LINE REFERRING TO LESS RELIABLE INFERENCE DUE TO LOW VOLUMES OF SILT (GYLLENCREUTZ AND KISSEL, 2006). * (A/B PSL) = ABOVE/BELOW PRESENT SEA LEVEL...24 FIGURE 1.3 - LOCATION MAP FOR DANISH ARCHAEOLOGICAL SITES REFERENCED IN THE TEXT.

FIGURE 1.4 - SUMMARY OF THE GEOLOGICAL AND CULTURAL PERIODS (INCLUDING SHELL MIDDEN ACCUMULATION PHASES AND SHELLFISH COMPOSITION DATA) REFERRED TO IN THE TEXT, ALONG WITH REGIONAL SEA LEVEL AND CLIMATE DATA FOR THE BALTIC SEA AREA. THE SEA LEVEL CURVE IS FROM BLEKINGE, SOUTHEASTERN SWEDEN (BERGLUND

FIGURE 1.5 – Δ^{15} N AND Δ^{13} C VALUES FOR BONE COLLAGEN FROM HUMAN AND DOGS FOR THE MESOLITHIC AND NEOLITHIC PERIOD. OPEN SYMBOLS REFER TO INDIVIDUALS WITH AMBIGUOUS CULTURAL BELONGING, BEING FOUND AT THE MESOLITHIC-NEOLITHIC TRANSITION. ONE-SIDED ERROR BARS REPRESENT MARINE RESERVOIR CORRECTION AND BLACK ARROWS INDICATE UNCERTAIN RESERVOIR EFFECT. THE DOTTED LINE FOR CARBON ISOTOPES (I.E. AT 20 ‰) MARKS THE LIMIT ABOVE WHICH THERE IS AN INDICATION OF A NON-NEGLIGIBLE MARINE DIET, THOUGH INDIVIDUALS BELOW THIS LINE MAY HAVE CONSUMED SMALL QUANTITIES OF MARINE FOOD. THE DOTTED LINE FOR NITROGEN ISOTOPES (I.E. AT 9.5 ‰) MARKS THE LIMIT ABOVE WHICH INDIVIDUALS MUST HAVE CONSUMED AQUATIC FOOD REGULARLY, THROUGH THE SAME MAY APPLY TO INDIVIDUALS BELOW THE LINE IF SUBSTANTIAL PARTS OF THIS DIETARY COMPONENT DERIVED FROM LOW TROPHIC LEVEL ORGANISMS SUCH AS SHELLFISH FIGURE 2.1 – MAP OF DENMARK WITH THE LOCATION OF THE THREE SITES ANALYSED IN THIS THESIS (KILEN, NORSMINDE FJORD AND KORUP SØ), PLUS OTHER IMPORTANT SITES REFERENCED IN THE TEXT. PRESENT DAY ISOSTATIC ADJUSTMENT (DOTTED GREY LINES;

AFTER EKMAN AND MÄKINEN, 1996) AND ISOBASES FOR THE HIGHEST LEVEL OF THE LITTORINA SEA IN DENMARK (BOLD NUMBERS AND BLACK DOTTED LINES; IN METRES AFTER MERTZ, 1924; CHRISTENSEN, 2001A), MODERN DAY ANNUAL MEAN SURFACE SALINITIES (SOLID WHITE LINES; ‰) AND SEA CURRENT SYSTEMS FOR THE KATTEGAT AND SKAGERRAK REGION (AFTER GYLLENCREUTZ ET AL., 2006) ARE ALSO INCLUDED. ABBREVIATIONS: CNSW = CENTRAL NORTH SEA WATER, AW = ATLANTIC WATER, NCC = NORWEGIAN COASTAL CURRENT, SJC = SOUTH JUTLAND CURRENT, NJC = NORTH JUTLAND CURRENT, BW = BALTIC WATER, BC = BALTIC CURRENT. NUMBERED SITES REFER TO SELECTED IMPORTANT ARCHAEOLOGICAL LOCALITIES AND SETTINGS. SHADED AREA HIGHLIGHTS THE AREAS IN WHICH MIDDLE AND LATE STONE AGE SHELL MIDDEN (IN DANISH: KØKKENMØDDING) SITES ARE FOUND. MESOLITHIC ERTEBØLLE AND EARLY NEOLITHIC SITES (BLUE DOTS): 1. ØSTER JØLBY, 2. AGGERSUND, 3. BROVST (INCLUDES KONGEMOSE LAYERS), 4. NØRHOLM, 5. BJØRNSHOLM BAY AND ÅLE, 6. ERTEBØLLE LOCUS CLASSICUS, 7. HJERK NOR, 8. KRABBESHOLM, 9. VISBORG AND HAVNØ, 10. MEILGÅRD, 11. NEDERST, 12. VÆNGESØ, 13. NORSMINDE FJORD AND KALVØ (INCLUDES ERTEBØLLE, FUNNEL BEAKER CULTURE (TBK), SINGLE GRAVE CULTURE AND IRON AGE SITES), 14. MOSEGÅRDEN, 15. KLINTESØ, 16. SØLAGER, 17. VEDBÆK (INCLUDES KONGEMOSE, ERTEBØLLE AND TBK SITES), 18. LOLLIKHUSE, 19. ØLBY LYNG, 20. MULLERUP (IMPORTANT MAGLEMOSE SITE), 21. KONGEMOSE LOCUS CLASSICUS. SUBMERGED ERTEBØLLE SITES (GREY DOTS): 22. TYBRIND VIG, 23. MØLLEGABET. SINGLE GRAVE AND PITTED WARE CULTURAL SITES (GREEN DOTS): 24. KAINSBAKKE AND KIRIAL BRO. 25-34 (YELLOW DOTS) IRON AGE SHELL MIDDEN SITES. V. LOCATION OF VEJLERNE WETLANDS (RYVES ET AL., 2004). STAR - COPENHAGEN (CAPITAL CITY). MAP MODIFIED FIGURE 2.2 - MAP OF THE KILEN SEDIMENTARY BASIN WITH BATHYMETRY (ORIGINALLY PRODUCED BY THORKILD HØY, 1989). INSETS: A. LOCATION MAP OF KILEN AND THE

LIMFJORD, B. CURRENT CONFIGURATION OF KILEN AND THE LIMFJORD. C. PRESENT

CORING LOCATION IS MARKED WITH A BLACK STAR
 FIGURE 2.3 – INSET A: LOCATION MAP FOR NORSMINDE FJORD. NORSMINDE FJORD BATHYMMETRY (LEFT) AND PRESENT DAY HYDROLOGICAL CATCHMENT (RIGHT). THE RED STAR INDICATES THE CORING SITE. THE RECLAIMED PART OF THE FJORD, NAMED 'KYSING FJORD', IS ALSO INDICATED. BATHYMMETRIC MAP FROM ELLEGAARD AND SKOVGAARD, 2007, CATCHMENT HYDROLOGCIAL MAP MODIFIED FROM ÅRHUS AMT, NATUR OG MILJØ 1998
BATHYMMETRY (LEFT) AND PRESENT DAY HYDROLOGICAL CATCHMENT (RIGHT). THE RED STAR INDICATES THE CORING SITE. THE RECLAIMED PART OF THE FJORD, NAMED 'KYSING FJORD', IS ALSO INDICATED. BATHYMMETRIC MAP FROM ELLEGAARD AND SKOVGAARD, 2007, CATCHMENT HYDROLOGCIAL MAP MODIFIED FROM ÅRHUS AMT, NATUR OG MILJØ 1998
RED STAR INDICATES THE CORING SITE. THE RECLAIMED PART OF THE FJORD, NAMED 'KYSING FJORD', IS ALSO INDICATED. BATHYMMETRIC MAP FROM ELLEGAARD AND SKOVGAARD, 2007, CATCHMENT HYDROLOGCIAL MAP MODIFIED FROM ÅRHUS AMT, NATUR OG MILJØ 1998
 'KYSING FJORD', IS ALSO INDICATED. BATHYMMETRIC MAP FROM ELLEGAARD AND SKOVGAARD, 2007, CATCHMENT HYDROLOGCIAL MAP MODIFIED FROM ÅRHUS AMT, NATUR OG MILJØ 1998
SKOVGAARD, 2007, CATCHMENT HYDROLOGCIAL MAP MODIFIED FROM ÅRHUS AMT, NATUR OG MILJØ 1998
 NATUR OG MILJØ 1998
 FIGURE 2.4 – NORSMINDE FJORD GEOLOGY, TOPOGRAPHY AND ARCHAEOLOGICAL MAP (MODIFIED FROM ANDERSEN, 1989). NUMBERED SITES REFER TO KEY ARCHAEOLOGICAL SITES: 1. NORSMINDE KØKKENMØDDING, 2. FREDERIKS ODDE, 3. KALVØ, 4. NORSLUND, 5. FLYNDERHAGE. INSET A: SECTION THROUGH THE NORSMINDE KITCHEN MIDDEN WITH MESOLITHIC AND NEOLITHIC LAYERS INDICATED (PHOTO COURTESY OF SØREN ANDERSERN, MOSEGARD MUSEUM). 69 FIGURE 2.5 - THE DJURSLAND PENINSULA WITH THE LOCATION OF KORUP SØ AND KOLINDSUND. THE PRESENT DAY COASTLINE (DOTTED BLACK LINE) AND THE ESTIMATED COASTLINE DURING THE <i>LITTORINA</i> SEA STAGE (DARK BLUE SHADING) ARE INDICATED (MODIFIED AFTER JESSEN, 1920; RASMUSSEN AND RICHTER, 1991). SITES NOT IN ITALICS REFER TO IMPORTANT ARCHAEOLOGICAL SITES REFERENCED IN THE TEXT. 71 FIGURE 2.6 - KILEN CORING DETAILS. THE FINAL SEDIMENTARY SEQUENCE (I.E. AFTER CORE CORRELATION) INCLUDES ALL CORE SECTIONS SHADED YELLOW. CORES WERE CORRELATED (WHERE POSSIBLE) USING THE SEDIMENTARY PHYSICAL PARAMETERS (INCLUDED IN THIS DIAGRAM FOR ALL CORE SECTIONS AND EXPRESSED PERCENTAGES; SEE TEXT FOR MORE DETAILS). 75 FIGURE 2.7 - MAP OF MOLTEN/DEFINE SITES INCLUDING BOTH WATER CHEMISTRY AND DIATOM COUNTS. RED DOTS INDICATE MOLTEN SITES AND BLACK DOTS INDICATE DEFINE SITES. INSET A: VARIANCE PARTITIONING OF THE FULL MOLTEN/DEFINE DATASET (I.E. 341 SITES) USING CCA. (FROM ANDRÉN <i>ET AL.</i>, 2007A). 89 FIGURE 3.1 - KILEN LITHOLOGICAL PROFILE. 95 FIGURE 3.2 - AGE-DEPTH MODEL AND ²¹⁰PB STRATIGRAPHY (BLUE LINE AND INSET) FOR THE UNDED AND COUNT AND ADD ADD ADD ADD ADD ADD ADD ADD ADD
 (MODIFIED FROM ANDERSEN, 1989). NUMBERED SITES REFER TO KEY ARCHAEOLOGICAL SITES: 1. NORSMINDE KØKKENMØDDING, 2. FREDERIKS ODDE, 3. KALVØ, 4. NORSLUND, 5. FLYNDERHAGE. INSET A: SECTION THROUGH THE NORSMINDE KITCHEN MIDDEN WITH MESOLITHIC AND NEOLITHIC LAYERS INDICATED (PHOTO COURTESY OF SØREN ANDERSERN, MOSEGARD MUSEUM). 69 FIGURE 2.5 - THE DJURSLAND PENINSULA WITH THE LOCATION OF KORUP SØ AND KOLINDSUND. THE PRESENT DAY COASTLINE (DOTTED BLACK LINE) AND THE ESTIMATED COASTLINE DURING THE <i>LITTORINA</i> SEA STAGE (DARK BLUE SHADING) ARE INDICATED (MODIFIED AFTER JESSEN, 1920; RASMUSSEN AND RICHTER, 1991). SITES NOT IN ITALICS REFER TO IMPORTANT ARCHAEOLOGICAL SITES REFERENCED IN THE TEXT. 71 FIGURE 2.6 - KILEN CORING DETAILS. THE FINAL SEDIMENTARY SEQUENCE (I.E. AFTER CORE CORRELATION) INCLUDES ALL CORE SECTIONS SHADED YELLOW. CORES WERE CORRELATED (WHERE POSSIBLE) USING THE SEDIMENTARY PHYSICAL PARAMETERS (INCLUDED IN THIS DIAGRAM FOR ALL CORE SECTIONS AND EXPRESSED PERCENTAGES; SEE TEXT FOR MORE DETAILS). 75 FIGURE 2.7 - MAP OF MOLTEN/DEFINE SITES INCLUDING BOTH WATER CHEMISTRY AND DIATOM COUNTS. RED DOTS INDICATE MOLTEN SITES AND BLACK DOTS INDICATE DEFINE SITES. INSET A: VARIANCE PARTITIONING OF THE FULL MOLTEN/DEFINE DATASET (I.E. 341 SITES) USING CCA. (FROM ANDRÉN <i>ET AL.</i>, 2007A). 89 FIGURE 3.1 - KILEN LITHOLOGICAL PROFILE. 95 FIGURE 4.2 - AGE-DEPTH MODEL AND ²¹⁰ PB STRATIGRAPHY (BLUE LINE AND INSET) FOR THE WATER DATASET (I.E. ADD DATO DUATED DUATED
SITES: 1. NORSMINDE KØKKENMØDDING, 2. FREDERIKS ODDE, 3. KALVØ, 4. NORSLUND, 5. FLYNDERHAGE. INSET A: SECTION THROUGH THE NORSMINDE KITCHEN MIDDEN WITH MESOLITHIC AND NEOLITHIC LAYERS INDICATED (PHOTO COURTESY OF SØREN ANDERSERN, MOSEGARD MUSEUM)
 FLYNDERHAGE. INSET A: SECTION THROUGH THE NORSMINDE KITCHEN MIDDEN WITH MESOLITHIC AND NEOLITHIC LAYERS INDICATED (PHOTO COURTESY OF SØREN ANDERSERN, MOSEGARD MUSEUM). FIGURE 2.5 - THE DJURSLAND PENINSULA WITH THE LOCATION OF KORUP SØ AND KOLINDSUND. THE PRESENT DAY COASTLINE (DOTTED BLACK LINE) AND THE ESTIMATED COASTLINE DURING THE <i>LITTORINA</i> SEA STAGE (DARK BLUE SHADING) ARE INDICATED (MODIFIED AFTER JESSEN, 1920; RASMUSSEN AND RICHTER, 1991). SITES NOT IN ITALICS REFER TO IMPORTANT ARCHAEOLOGICAL SITES REFERENCED IN THE TEXT. 71 FIGURE 2.6 - KILEN CORING DETAILS. THE FINAL SEDIMENTARY SEQUENCE (I.E. AFTER CORE CORRELATION) INCLUDES ALL CORE SECTIONS SHADED YELLOW. CORES WERE CORRELATED (WHERE POSSIBLE) USING THE SEDIMENTARY PHYSICAL PARAMETERS (INCLUDED IN THIS DIAGRAM FOR ALL CORE SECTIONS AND EXPRESSED PERCENTAGES; SEE TEXT FOR MORE DETAILS). FIGURE 2.7 - MAP OF MOLTEN/DEFINE SITES INCLUDING BOTH WATER CHEMISTRY AND DIATOM COUNTS. RED DOTS INDICATE MOLTEN SITES AND BLACK DOTS INDICATE DEFINE SITES. INSET A: VARIANCE PARTITIONING OF THE FULL MOLTEN/DEFINE DATASET (I.E. 341 SITES) USING CCA. (FROM ANDRÉN <i>ET AL.</i>, 2007A). 89 FIGURE 3.1 - KILEN LITHOLOGICAL PROFILE. 95 FIGURE 3.2 - AGE-DEPTH MODEL AND ²¹⁰PB STRATIGRAPHY (BLUE LINE AND INSET) FOR THE
 MESOLITHIC AND NEOLITHIC LAYERS INDICATED (PHOTO COURTESY OF SØREN ANDERSERN, MOSEGARD MUSEUM). 69 FIGURE 2.5 - THE DJURSLAND PENINSULA WITH THE LOCATION OF KORUP SØ AND KOLINDSUND. THE PRESENT DAY COASTLINE (DOTTED BLACK LINE) AND THE ESTIMATED COASTLINE DURING THE <i>LITTORINA</i> SEA STAGE (DARK BLUE SHADING) ARE INDICATED (MODIFIED AFTER JESSEN, 1920; RASMUSSEN AND RICHTER, 1991). SITES NOT IN ITALICS REFER TO IMPORTANT ARCHAEOLOGICAL SITES REFERENCED IN THE TEXT. 71 FIGURE 2.6 - KILEN CORING DETAILS. THE FINAL SEDIMENTARY SEQUENCE (I.E. AFTER CORE CORRELATION) INCLUDES ALL CORE SECTIONS SHADED YELLOW. CORES WERE CORRELATED (WHERE POSSIBLE) USING THE SEDIMENTARY PHYSICAL PARAMETERS (INCLUDED IN THIS DIAGRAM FOR ALL CORE SECTIONS AND EXPRESSED PERCENTAGES; SEE TEXT FOR MORE DETAILS). 75 FIGURE 2.7 - MAP OF MOLTEN/DEFINE SITES INCLUDING BOTH WATER CHEMISTRY AND DIATOM COUNTS. RED DOTS INDICATE MOLTEN SITES AND BLACK DOTS INDICATE DEFINE SITES. INSET A: VARIANCE PARTITIONING OF THE FULL MOLTEN/DEFINE DATASET (I.E. 341 SITES) USING CCA. (FROM ANDRÉN <i>ET AL.</i>, 2007A). 89 FIGURE 3.1 - KILEN LITHOLOGICAL PROFILE. 95 FIGURE 3.2 - AGE-DEPTH MODEL AND ²¹⁰PB STRATIGRAPHY (BLUE LINE AND INSET) FOR THE
 ANDERSERN, MOSEGARD MUSEUM). 69 FIGURE 2.5 - THE DJURSLAND PENINSULA WITH THE LOCATION OF KORUP SØ AND KOLINDSUND. THE PRESENT DAY COASTLINE (DOTTED BLACK LINE) AND THE ESTIMATED COASTLINE DURING THE <i>LITTORINA</i> SEA STAGE (DARK BLUE SHADING) ARE INDICATED (MODIFIED AFTER JESSEN, 1920; RASMUSSEN AND RICHTER, 1991). SITES NOT IN ITALICS REFER TO IMPORTANT ARCHAEOLOGICAL SITES REFERENCED IN THE TEXT. 71 FIGURE 2.6 - KILEN CORING DETAILS. THE FINAL SEDIMENTARY SEQUENCE (I.E. AFTER CORE CORRELATION) INCLUDES ALL CORE SECTIONS SHADED YELLOW. CORES WERE CORRELATED (WHERE POSSIBLE) USING THE SEDIMENTARY PHYSICAL PARAMETERS (INCLUDED IN THIS DIAGRAM FOR ALL CORE SECTIONS AND EXPRESSED PERCENTAGES; SEE TEXT FOR MORE DETAILS). 75 FIGURE 2.7 - MAP OF MOLTEN/DEFINE SITES INCLUDING BOTH WATER CHEMISTRY AND DIATOM COUNTS. RED DOTS INDICATE MOLTEN SITES AND BLACK DOTS INDICATE DEFINE SITES. INSET A: VARIANCE PARTITIONING OF THE FULL MOLTEN/DEFINE DATASET (I.E. 341 SITES) USING CCA. (FROM ANDRÉN <i>ET AL.</i>, 2007A). 89 FIGURE 3.1 - KILEN LITHOLOGICAL PROFILE. 95 FIGURE 3.2 - AGE-DEPTH MODEL AND ²¹⁰PB STRATIGRAPHY (BLUE LINE AND INSET) FOR THE UNDER DATASET DELEMENT DI MULTION DI MULTION DI MODEL DI MULTION DI MULTION DI MULTION DI MULTION DI MULTION DI MODEL AND PLATOR DI MULTION DI MULTION DI MODEL AND DIATOR DI MODEL DI MODEL DI MULTION DI MULTION DI MULTION DI MODEL AND PLATOR DI MULTION DI MULTION DI MULTION DI MODEL AND PLATOR DI MULTION D
 FIGURE 2.5 - THE DJURSLAND PENINSULA WITH THE LOCATION OF KORUP SØ AND KOLINDSUND. THE PRESENT DAY COASTLINE (DOTTED BLACK LINE) AND THE ESTIMATED COASTLINE DURING THE <i>LITTORINA</i> SEA STAGE (DARK BLUE SHADING) ARE INDICATED (MODIFIED AFTER JESSEN, 1920; RASMUSSEN AND RICHTER, 1991). SITES NOT IN ITALICS REFER TO IMPORTANT ARCHAEOLOGICAL SITES REFERENCED IN THE TEXT. 71 FIGURE 2.6 - KILEN CORING DETAILS. THE FINAL SEDIMENTARY SEQUENCE (I.E. AFTER CORE CORRELATION) INCLUDES ALL CORE SECTIONS SHADED YELLOW. CORES WERE CORRELATED (WHERE POSSIBLE) USING THE SEDIMENTARY PHYSICAL PARAMETERS (INCLUDED IN THIS DIAGRAM FOR ALL CORE SECTIONS AND EXPRESSED PERCENTAGES; SEE TEXT FOR MORE DETAILS)
 KOLINDSUND. THE PRESENT DAY COASTLINE (DOTTED BLACK LINE) AND THE ESTIMATED COASTLINE DURING THE <i>LITTORINA</i> SEA STAGE (DARK BLUE SHADING) ARE INDICATED (MODIFIED AFTER JESSEN, 1920; RASMUSSEN AND RICHTER, 1991). SITES NOT IN ITALICS REFER TO IMPORTANT ARCHAEOLOGICAL SITES REFERENCED IN THE TEXT. 71 FIGURE 2.6 - KILEN CORING DETAILS. THE FINAL SEDIMENTARY SEQUENCE (I.E. AFTER CORE CORRELATION) INCLUDES ALL CORE SECTIONS SHADED YELLOW. CORES WERE CORRELATED (WHERE POSSIBLE) USING THE SEDIMENTARY PHYSICAL PARAMETERS (INCLUDED IN THIS DIAGRAM FOR ALL CORE SECTIONS AND EXPRESSED PERCENTAGES; SEE TEXT FOR MORE DETAILS)
ESTIMATED COASTLINE DURING THE <i>LITTORINA</i> SEA STAGE (DARK BLUE SHADING) ARE INDICATED (MODIFIED AFTER JESSEN, 1920; RASMUSSEN AND RICHTER, 1991). SITES NOT IN ITALICS REFER TO IMPORTANT ARCHAEOLOGICAL SITES REFERENCED IN THE TEXT. 71 FIGURE 2.6 - KILEN CORING DETAILS. THE FINAL SEDIMENTARY SEQUENCE (I.E. AFTER CORE CORRELATION) INCLUDES ALL CORE SECTIONS SHADED YELLOW. CORES WERE CORRELATED (WHERE POSSIBLE) USING THE SEDIMENTARY PHYSICAL PARAMETERS (INCLUDED IN THIS DIAGRAM FOR ALL CORE SECTIONS AND EXPRESSED PERCENTAGES; SEE TEXT FOR MORE DETAILS)
INDICATED (MODIFIED AFTER JESSEN, 1920; RASMUSSEN AND RICHTER, 1991). SITES NOT IN ITALICS REFER TO IMPORTANT ARCHAEOLOGICAL SITES REFERENCED IN THE TEXT. 71 FIGURE 2.6 - KILEN CORING DETAILS. THE FINAL SEDIMENTARY SEQUENCE (I.E. AFTER CORE CORRELATION) INCLUDES ALL CORE SECTIONS SHADED YELLOW. CORES WERE CORRELATED (WHERE POSSIBLE) USING THE SEDIMENTARY PHYSICAL PARAMETERS (INCLUDED IN THIS DIAGRAM FOR ALL CORE SECTIONS AND EXPRESSED PERCENTAGES; SEE TEXT FOR MORE DETAILS)
IN ITALICS REFER TO IMPORTANT ARCHAEOLOGICAL SITES REFERENCED IN THE TEXT. 71 FIGURE 2.6 - KILEN CORING DETAILS. THE FINAL SEDIMENTARY SEQUENCE (I.E. AFTER CORE CORRELATION) INCLUDES ALL CORE SECTIONS SHADED YELLOW. CORES WERE CORRELATED (WHERE POSSIBLE) USING THE SEDIMENTARY PHYSICAL PARAMETERS (INCLUDED IN THIS DIAGRAM FOR ALL CORE SECTIONS AND EXPRESSED PERCENTAGES; SEE TEXT FOR MORE DETAILS)
 FIGURE 2.6 - KILEN CORING DETAILS. THE FINAL SEDIMENTARY SEQUENCE (I.E. AFTER CORE CORRELATION) INCLUDES ALL CORE SECTIONS SHADED YELLOW. CORES WERE CORRELATED (WHERE POSSIBLE) USING THE SEDIMENTARY PHYSICAL PARAMETERS (INCLUDED IN THIS DIAGRAM FOR ALL CORE SECTIONS AND EXPRESSED PERCENTAGES; SEE TEXT FOR MORE DETAILS)
CORRELATION) INCLUDES ALL CORE SECTIONS SHADED YELLOW. CORES WERE CORRELATED (WHERE POSSIBLE) USING THE SEDIMENTARY PHYSICAL PARAMETERS (INCLUDED IN THIS DIAGRAM FOR ALL CORE SECTIONS AND EXPRESSED PERCENTAGES; SEE TEXT FOR MORE DETAILS)
CORRELATED (WHERE POSSIBLE) USING THE SEDIMENTARY PHYSICAL PARAMETERS (INCLUDED IN THIS DIAGRAM FOR ALL CORE SECTIONS AND EXPRESSED PERCENTAGES; SEE TEXT FOR MORE DETAILS)
 (INCLUDED IN THIS DIAGRAM FOR ALL CORE SECTIONS AND EXPRESSED PERCENTAGES; SEE TEXT FOR MORE DETAILS)
SEE TEXT FOR MORE DETAILS)
 FIGURE 2.7 - MAP OF MOLTEN/DEFINE SITES INCLUDING BOTH WATER CHEMISTRY AND DIATOM COUNTS. RED DOTS INDICATE MOLTEN SITES AND BLACK DOTS INDICATE DEFINE SITES. INSET A: VARIANCE PARTITIONING OF THE FULL MOLTEN/DEFINE DATASET (I.E. 341 SITES) USING CCA. (FROM ANDRÉN <i>ET AL.</i>, 2007A). FIGURE 3.1 - KILEN LITHOLOGICAL PROFILE. 95 FIGURE 3.2 – AGE-DEPTH MODEL AND ²¹⁰PB STRATIGRAPHY (BLUE LINE AND INSET) FOR THE
DIATOM COUNTS. RED DOTS INDICATE MOLTEN SITES AND BLACK DOTS INDICATE DEFINE SITES. INSET A: VARIANCE PARTITIONING OF THE FULL MOLTEN/DEFINE DATASET (I.E. 341 SITES) USING CCA. (FROM ANDRÉN <i>ET AL.</i> , 2007A)
DEFINE SITES. INSET A: VARIANCE PARTITIONING OF THE FULL MOLTEN/DEFINE DATASET (I.E. 341 SITES) USING CCA. (FROM ANDRÉN <i>ET AL.</i> , 2007A)
DATASET (I.E. 341 SITES) USING CCA. (FROM ANDRÉN <i>ET AL.</i> , 2007A)
FIGURE 3.1 - KILEN LITHOLOGICAL PROFILE
FIGURE 3.2 – AGE-DEPTH MODEL AND ²¹⁰ PB STRATIGRAPHY (BLUE LINE AND INSET) FOR THE
WHEN PROPERTY AND A CORRESPONDED OF CENT (1, 1) THE POWER WATCH
KILEN PROFILE. MODEL CREATED BY JESPER OLSEN (AARHUS UNIVERSITY) USING
OXCAL V.4.1. DOTTED LINES REFER TO DEPTHS FOR WHICH ALLOWANCES FOR SHIFTS IN
THE ACCUMULATION RATE WERE MADE BASED ON CHANGES IN THE CARBONATE CURVE
(SEE FIGURE 3.5)
FIGURE 3.3 – PERCENTAGE VALUES FOR SELECTED TAXA IN THE KILEN POLLEN RECORD
SPANNING ~4,180-3,640 BC. THE ULMUS CURVE (SHADED GREY) EXHIBITS A DECLINE
BETWEEN 1,338 AND 1,362 CM BPSL WITH THE AGE ~3,900 BC, WHICH IS IN GOOD
AGREEMENT WITH THE GENERAL DATING OF THE CLASSIC MID-HOLOCENE 'ELM
DECLINE' IN DENMARK (ANDERSEN AND RASMUSSEN, 1993). ANALYST: PETER
RASMUSSEN (GEUS)
FIGURE 3.4 - DEPTH PROFILES OF ²¹⁰ PB AND ¹⁵⁷ CS FOR KILEN
FIGURE 3.5 - KILEN SEDIMENTARY PHYSICAL PARAMETERS, INCLUDING OVERALL
ACCUMULATION RATES (CM YR ⁺), WATER CONTENT AND DRY MASS EXPRESSED AS
PERCENTAGES OF THE TOTAL WET WEIGHT OF THE SEDIMENT, PERCENTAGES OF THE
DRY MASS THAT ARE ORGANIC MATTER, CARBONATE AND MINEROGENIC MATTER (FOLLOWING LOSS ON IGNETION). IN HIS ELEVES ($MG GMS^2 MD^{-1}$) FOR ORGANIC MATTER
(FOLLOWING LOSS-ON-IGNITION), PLUS FLUXES (MG CM ⁻⁺ YR ⁻⁺) FOR ORGANIC MATTER,
CARBONATE, MINEROGENIC MATTER AND DRY MASS. A MAGNIFICATION OF KIL-S4 IS
PROVIDED IN FIGURE 3.6. (NB. ACCUMULATION RATE AND FLUX MAY BE ERRONEOUS
BETWEEN ~AD 800-1920 DUE TO UNCERTAINTIES WITH THE AGE-DEPTH MODEL (SEE
SECTIONS 3.5 AND 3.11)
TIGURE 3.0 - KILEN SEDIMENTART THI SICAL PARAMETERS FOLUSSING ON KIL-54. FOR FULL DETAILS SEE EICHDE 2.5 CADTION (ND A LITHOLOCY COLUMN IS OMITTED IN THIS
FIGURE DUE TO THE ARSENCE OF ANY LITHOLOGICAL DESCRIPTION FOR THIS SECTION
OF THE PROFILE. ALSO NOTE THAT THE SCALES FOR THE INDIVIDUAL PARAMETERS IN

THIS FIGURE HAVE BEEN ADJUSTED ACCORDINGLY (FROM FIGURE 3.5) FOR
PRESENTATION PURPOSES)102
FIGURE 3.7 – KILEN PERCENTAGE DIATOM RECORD (TAXA >3 %). NUMBERS IN BRACKETS
REFER TO THE SALINITY OPTIMA IN THE MOLTEN TRAINING SET USED FOR THE
QUANTITATIVE RECONSTRUCTION OF SALINITY. ANALYST: JONATHAN LEWIS 105
FIGURE 3.8 - KILEN DIATOM ASSOCIATED METRICS. * SAMPLE SCORES FOR DCA ANALYSIS ON
REDUCED DIATOM DATASET (I.E. BETWEEN ~5,430 BC-AD 800; LOWER 82 SAMPLES) 106
FIGURE 3.9 - KILEN DIATOM RECORD FOR THE UPPERMOST SECTION OF THE PROFILE (KIL-D4-
KIL-D6; ~AD 1830-PRESENT)107
FIGURE 3.10 - SEDIMENTARY PIGMENT RECORD FROM KILEN EXPRESSED AS
CONCENTRATIONS; NMOL G OM. ANALYST: JONATHAN LEWIS112
FIGURE 3.11 - KILEN SEDIMENTARY PIGMENT RECORD EXPRESSED AS ACCUMULATION RATES;
NMOL $CM^{-2} YR^{-1}$
FIGURE 3.12- KILEN MOLLUSCAN RECORD (AS CONCENTRATION PER 50 ML FRESH SEDIMENT,
WITH SQUARE ROOT TRANSFORMATION APPLIED). OPHISTOBRANCHIA FRAGMENTS AND
MYTILUS SP. FRAGMENTS EXPRESSED AS PRESENCE/ABSENCE DATA. NUMBERS IN
BRACKETS REFER TO MINIMUM SALINITY TOLERANCES ACCORDING TO SORGENFREI,
(1958). ANALYST: KAJ STRAND PETERSEN (GEUS)
FIGURE 3.13 - KILEN PLANT MACROFOSSIL RECORD (CONCENTRATION PER 50 ML FRESH
SEDIMENT, UNLESS INDICATED AS PRESENCE/ABSENCE DATA BY BLACK DOTS).
ANALYSTS: PETER RASMUSSEN (GEUS) AND JONATHAN LEWIS
FIGURE 3.14 – KILEN PERCENTAGE FORAMINIFERA RECORD AND $\Delta^{18}O_{CALCITE}$ AND $\Delta^{13}C_{CALCITE}$
ISOTOPES BASED ON TEST OF THE BENTHIC FORAMINIFERAL SPECIES ELPHIDIUM
EXCAVATUM FO. SELSEYENSIS. (ANALYSTS – FORAMINIFERA PETER KRISTENSEN AARHUS
UNIVERSITY, ISOTOPES: JONATHAN LEWIS)
FIGURE 3.15 - ADDITIONAL ORGANISMS PRESENT AT KILEN. OSTRACODS AND DAPHNIA SPP.
(CLADOCERA) ARE EXPRESSED AS CONCENTRATION DATA PER 50 ML FRESH SEDIMENT.
CHRYSOPHYTE CONCENTRATIONS ARE EXPRESSED AS NUMBER PER GRAM OF DRY
WEIGHT. ALL OTHER ORGANISMS ARE EXPRESSED AS PRESENCE/ABSENCE DATA 128
FIGURE 3.16 - SUMMARY DIAGRAM OF THE MAJOR CHANGES OCCURRING AT KILEN (~5,500 BC-
AD 2007). REGIONAL (POLLEN-BASED) TEMPERATURE CURVE FROM LAKE
TREHORNINGEN (ANTONSSON AND SEPPA, 2007) AND (MEAN HIGH WATER) SEA LEVEL
CURVE FROM THE SOUTHERN NORTH SEA (BEHRE, 2007). *DCA AXIS 1 REDUCED DATASET
SAMPLE SCORES EXCLUDES THE UPPERMOST 28 DIATOM SAMPLES (SEE TEXT FOR
DETAILS). †M NN ASL = METRES RELATIVE TO GERMAN ORDNANCE DATUM ABOVE
PRESENT SEA LEVEL
FIGURE 3.17 - A. PALAEO-SHORELINES OF THE NORTH SEA (IN YEARS BC) AS PROPOSED BY
BEHRE, (2007). B. MAP OF DENMARK AND THE WESTERN BALTIC DURING THE LATE
ATLANTIC ERTEBØLLE PERIOD (~5,400-3,900 BC) INDICATING PRESENT SHORELINE
(DOTTED LINE) AND PROPOSED PALAEO-SHORLINES (SOLID LINE). MAP RE DRAWN AND
MODIFIED FROM ENGHOFF ET AL., (2007) AFTER HELLESEN AND TUXEN, (1988). NOTE THE
MORE OPEN LIMFJORD WITH OPENINGS TO BOTH THE NORTH AND WEST
FIGURE 3.18 - SALINITY AND PRODUCTIVITY CHANGES IN THE KILEN PROFILE BETWEEN
~4,500-2,500 BC (COVERING THE MEOSLITHIC-NEOLTIHIC TRANSITION). SEA LEVEL FROM
BEHRE, (2007) FOR SOUTHERN NORTH SEA. *M NN ASL = METRES REALTIVE TO GERMAN
ORDNANCE DATUM ABOVE SEA LEVEL
FIGURE 3.19 – SUMMARY DIAGRAM OF THE UPPER SECTION (~AD 300-PRESENT; TOP TIUCM) OF
THE KILEN PROFILE AND ASSOCIATED DATING PROBLEMS. *GREY SHADING = TOTAL
PLANT MACKOFOSSIL CONCENTRATION EXCLUDING AMBIGUOUS REMAINS (I.E.
DECIDUOUS LEAF FRAGMENTS, BUDS, WOUD, BAKK AND TWIGS), BLACK SHADING =
IUIAL PLANT MACKUPUSSIL CUNCENTRATION. I = APPKUXIMATE DEPTH OF AD 1856 USING A STRAIGHTEODWARD EVERADOL ATION OF THE STORP (12703) STRAID DEVICE
USING A STRAIGHTFORWARD EXTRAPOLATION OF THE 210PB/13/CS STRATIGRAPHY. $2 =$
APPROALWATE DEPTH OF AD 1825 USING A STRAIGHTFORWARD EXTRAPOLATION OF THE 210 DD / ¹³⁷ CC STD ATICD ADUX, A AD 19150 UDDEDMOST DADIOCADDON DATE (DEDTUATE AC
rd/COSTRATIONAPHT. AAK-1210U = UPPEKMUST KADIUCAKBUN DATE (DEPTH 459-460 CM DDSL AD 672 804 (05.4.97). (170)
UVI DESL, AD $0/2-094$ (93.4 %);

FIGURE 3.20 - COMPARISON OF DI-SALINITY AND DI-TOTAL NITROGEN ESTIMATES WITH MONITORING DATA FOR SALINITY, TN AND TP, MEASURED OVER THE PERIOD AD 1972-FIGURE 4.1 - LITHOLOGY OF THE NORSMINDE PROFILE BASED ON THE ORIGINAL DESCRIPTIONS (TABLE 4.1) BY PETER RASMUSSEN (GEUS), USING A MODIFIED TROELS-SMITH SYSTEM FOR CHARACTERISATION OF UNCONSOLIDATED SEDIMENTS (TROELS-FIGURE 4.2 - NORSMINDE AGE MODEL (PRODUCED BY JESPER OLSEN, AARHUS UNIVERSITY, FIGURE 4.3 - SEDIMENT PHYSICAL PARAMETERS, OVERALL ACCUMULATION RATE AND FLUX RATES FOR ORGANIC. CARBONATE AND MINEROGENIC MATTER FOR THE NORSMINDE FIGURE 4.4 - NORSMINDE DIATOM RECORD; ~7,000-6,870 BC. DATA EXPRESSED AS FIGURE 4.5 - NORSMINDE PERCENTAGE DIATOM RECORD. (>3 % ABUNDANCE IN ONE OR MORE SLIDES). VALUES IN BRACKETS REFER TO SALINITY OPTIMA IN THE MOLTEN SALINITY FIGURE 4.6 - NORSMINDE DIATOM ASSOCIATED METRICS. FULL MODEL REFERS TO DIATOM-INFERRED SALINITY INCLUDING FRAGILARIA S.L. AND FOR THE REDUCED MODEL FRAGILARIA S.L. WERE REMOVED FOR RECONSTRUCTION OF SALINITY. CHRYSOPHYTES EXPRESSED AS PRESENCE/ABSENCE DATA.....169 FIGURE 4.7 - NORSMINDE FORAMINIFERA RECORD (ANALYST PETER KONRADI, (GEUS). DATA EXPRESSED AS CONCENTRATIONS PER 20 ML OF WET SEDIMENT......173 FIGURE 4.8 - NORSMINDE MOLLUSCAN RECORD AND OTHER MARINE ANIMALS (I.E. ZOOPLANKTON, ECHINODERMS AND FISH). THE VALUES IN BRACKETS REFER TO THE LOWEST MEAN SALINITY INHABITED BY THE TAXA (ACCORDING TO SORGENFREI, 1958). ANALYSTS KAJ STRAND PETERESEN AND PETER RASMUSSEN, GEUS). ALL MOLLUSC SPECIES, ZOOPLANKTON, ECHINODERM AND FISH BONE DATA ARE EXPRESSED AS FIGURE 4.9 - NORSMINDE PERCENTAGE POLLEN RECORD (SELECTED TAXA). WG = WET GROUND TAXA, S/F = SWAMP AND FEN TAXA, CULT. = CULTIVATED TAXA. (ANALYST PETER RASMUSSEN GEUS). DATA ARE EXPRESSED AS PERCENTAGES OF THE TERRESTRIAL POLLEN SUM (TPS) WITH THE EXCEPTION OF CULTIVATED TAXA WHICH ARE EXPRESSED AS PRESENCE/ABSENCE DATA, NB, SWAMP AND FEN TAXA ARE NOT INCLUDED IN THE TPS BUT ARE EXPRESSED AS PERCENTAGES RELATIVE TO THE TPS... 177 FIGURE 4.10 - NORSMINDE POLLEN GROUPS, AQUATIC TAXA AND POLLEN ASSOCIATED METRICS (I.E. PCA AXIS 1 AND 2 SCORES AND POLLEN SUM. (ANALYST PETER RASMUSSEN GEUS, PCA ANALYSIS JONATHAN LEWIS). POLLEN GROUPS AND AQUATIC GROUP TAXA ARE EXPRESSED AS PERCENTAGES RELATIVE TO THE TERRESTRIAL POLLEN SUM (I.E. FIGURE 4.11 - NORSMINDE PLANT MACROFOSSIL RECORD. GROUPS ARE ORGANISED ACCORDING TO POLLEN GROUPINGS. (ANALYST: PETER RASMUSSEN GEUS). DATA EXPRESSED AS CONCENTRATION PER 50 ML WET SEDIMENT (BARS) OR AS PRESENCE/ABSENCE (CIRCLES). *= GRAY SILHOUETTE REFERS TO PLANT MACROFOSSIL CONCENTRATION EXCLUDING UNDIFFERENTIATED LEAF, BUD, WOOD AND TWIG FIGURE 4.12 - ENVIRONMENTAL PREDICTOR-SPECIES RESPONSE RELATIONSHIPS FOR NORSMINDE FJORD (6,760-2,020 BC). LENGTH OF AXIS 1 GRADIENT REFERS TO DCA ANALYSIS ON REDUCED DIATOM DATASET (IF DCA AXIS 1 GRADIENT LENGTH > 1.5 S.D. THEN UNIMODAL CCA EMPLOYED TO EXPLORE TRENDS. IF DCA AXIS 1 <1.5. S.D. THEN LINEAR RDA TECHNIQUE USED). SUBSET 1 = ENTIRE DATASET (39 SAMPLES). SUBSETS 2-21 = MOVING WINDOW ANALYSES (20 SAMPLES PER WINDOW). SHADINGS: RED = ANTHROPOGENIC INDICATOR, BLACK = SEDIMENTARY REGIME/CATCHMENT EROSION, GREEN: VEGETATION CHANGE PREDICTORS, GREY = CLIMATIC AND BLUE = FJORD DEPTH AND LEVEL OF CONNECTION WITH KATTEGAT......192

FIGURE 4.13- SUMMARY DIAGRAM OF THE MAJOR CHANGES OCCURRING AT NORSMINDE FJORD (~7,000-1,700 BC). *= GRAY SILHOUETTE REFERS TO PLANT MACROFOSSIL
CONCENTRATION EXCLUDING UNDIFFERENTIATED LEAF, BUD, WOOD AND TWIG
FRAGMENTS. FULL MODEL, REFER TO DI-SALINITY INCLUDING ALL FOSSIL TAXA,
REDUCED MODEL REFERS TO DIATOM-INFERRED SALINITY EXCLUDING FRAGILARIA S.L.
FROM THE FOSSIL DATASET. REGIONAL SEA LEVEL CURVE FROM BLEKINGE, SE SWEDEN
(BERGLUND ET AL., 2005) AND POLLEN-BASED TEMPERATURE RECORD FROM LAKE
TREHÖRNINGEN, SWEDEN (ANTONSSON AND SEPPÄ, 2007). †M A/B/ PSL = METRES ABOVE/BELOW PRESENT SEA LEVEL
FIGURE 4.14 - DI- RECONSTRUCTIONS WITH ADDED LOESS SMOOTH (0.25 SPAN) FOR
NORSMINDE FJORD BETWEEN ~6,600-2,100 BC
FIGURE 5.1 - SUMMARY OF KORUP SØ LITHOLOGICAL PROFILE (BASED ON ORIGINAL
DESCRIPTION BY HARALD KROG AND BENT ODGAARD (GEOLOGICAL SURVEY OF
DENMARK, NOW GEOLOGICAL SURVEY OF DENMARK AND GREENLAND). FOR COLOUR
CHANGES, SEE TABLE 5.1
FIGURE 5.2 - KORUP SØ AGE MODELS. CORRELATED DATES REFER TO THE STRATIGRAPHIC
POLLEN LINKS (TABLE 5.3) IDENTIFIABLE IN BOTH WELLS AND INCORPORATED INTO THE
DEPOSTIONAL MODELS (J. OSLEN, P. RASMUSSEN, UNPUB.). ANALYSES PERFORMED AS
PART OF THIS THESIS ARE ALL FROM CORE-1983 SEDIMENTS (WELL NO. 1)
FIGURE 5.3 – SEDIMENT ACCUMULATION RATES AT KORUP SØ (FOR BOTH THE1982 AND 1983
WELLS). LITHOLOGY COLUMN INCLUDED FOR THE 1983 WELL (SEE TABLE 5.1 AND
FIGURE 5.1 FOR KEY)
FIGURE 5.4 - KORUP SØ PERCENTAGE DIATOM RECORD (>3% ABUNDANCE). TAXA SORTED BY
SALINITY OPTIMA IN DIATOM TRAINING SET (VALUES QUOTED IN BRACKETS). ERROR
BARS FOR DI-SALINITY RECONSTRUCTION PLOTTED IN FIGURE 5.5
FIGURE 5.5 - DIATOM ASSOCIATED METRICS
FIGURE 5.6 KORUP SØ MOLLUSCAN RECORD (FROM DATA ORIGINALLY PUBLISHED IN
PETERSEN, 1993). TAXA SORTED BY MINIMUM SALINITY BASED ON VALUES STATED IN
SORGENFREI, 1958. MOLLUSC ABUNDANCES AND TOTAL CONCENTRATION EXPRESSED AS
CONCENTRATION PER L-1 (WITH SQUARE ROOT TRANSFORMATION APPLIED TO THE DATA
FOR PRESENTATION PURPOSES). DOTTED LINES INDICATE OPTIMALLY SPLIT SUB-ZONES.
IT IS IMPORTANT TO NOTE THAT SUB-ZONES HAVE ADDED TO AID INTERPRETATION, AND
ARE THEREFORE NOT NECESSARILY STATISTICALLY SIGNIFICANT SPLITS (SEE SECTION
2.10.1)
FIGURE 5.7 - KORUP SØ PERCENTAGE POLLEN RECORD (SELECTED TAXA). SOLID ZONE LINES
INDICATE SIGNIFICANT ZONES (DETERMINED USING THE BROKEN STICK METHOD;
BENNETT, 1996). ANALYST: HARALD KROG, GEUS. (ABBREVIATIONS: WG = WET GROUND).
DATA ARE EXPRESSED AS PERCENTAGES OF THE TERRESTRIAL POLLEN SUM (TPS) WITH
THE EXCEPTION OF CULTIVATED TAXA WHICH ARE EXPRESSED AS PRESENCE/ABSENCE
DATA. NB. SWAMP AND FEN TAXA ARE NOT INCLUDED IN THE TPS BUT ARE EXPRESSED
AS PERCENTAGES RELATIVE TO THE TPS
FIGURE 5.8- MAJOR PLANT GROUPINGS PERCENTAGE POLLEN DIAGRAM FOR KORUP SØ AND
AQUATIC SPECIES (INCLUDING AQUATIC PLANTS, PEDIASTRUM, DINOFLAGELLATES AND
HYSTRIX; ALL RECORDED DURING PALYNOLOGICAL ANALYSES AND EXPRESSED
RELATIVE TO THE TERRESTRIAL POLLEN SUM. NB. THE PERCENTAGE TERRESTRIAL
POLLEN SUM IS ONLY MADE UP OF (ALL) TAXA FROM THE FOLLOWING GROUPS: TREES.
SHRUBS AND HERBS, HALOPHYTES, GRASSES, DRY GROUND HERBS, CULTIVATED, OTHER
HERBS AND WET GROUND. PERCENTAGE ABUNDANCES FOR ALL OTHER TAXA (I.E.
SWAMP AND FEN, AQUATICS, PEDIASTRUM AND DINOFLAGELLATES) ARE CALCULATED
RELATIVE TO THE TERRESTRIAL POLLEN SUM
FIGURE 5.9 - SUMMARY OF 'MOVING WINDOW' RESULTS FROM CCA/RDA ANALYSES
ASSESSING PREDICTOR-RESPONSE (DIATOMS) RELATIONSHIPS THROUGH TIME. LENGTH
OF AXIS 1 GRADIENT REFERS TO DCA ANALYSIS ON REDUCED DIATOM DATASET (IF DCA
AXIS 1 GRADIENT LENGTH > 1.5 S.D. THEN UNIMODAL CCA EMPLOYED TO EXPLORE
TRENDS. IF DCA AXIS 1 <1.5. S.D. THEN LINEAR RDA TECHNIQUE USED). SUBSET 1 =

ENTIRE DATASET (46 SAMPLES). SUBSETS 2-35 = MOVING WINDOW ANALYSES. SHADINGS: RED = ANTHROPOGENIC INDICATOR, BLACK = SEDIMENTARY REGIME/CATCHMENT EROSION, GREEN: VEGETATION CHANGE PREDICTORS, GREY = CLIMATIC AND BLUE = FJORD DEPTH AND LEVEL OF CONNECTION WITH KATTEGAT, BROWN = MOLLUSCS FIGURE 5.10 - SUMMARY DIAGRAM FOR ALL PROXIES ANALYSED AT KORUP SØ, TOGETHER WITH REGIONAL TEMPERATURE (ANTONSSON AND SEPPÄ, 2007) AND SEA LEVEL (BERGLUND ET AL., 2005) RECONSTRUCTIONS. WELL NO. 1 SEDIMENTATION RATE IS BASED ON LATEST AGE MODEL CALCULATIONS (J. OLSEN, P. RASMUSSEN, UNPUB. DATA). FIGURE 6.1- SEDIMENTARY PHYSICAL PARAMETERS FOR ALL THREE SITES (KORUP SØ, NORSMINDE FJORD AND KILEN) PLOTTED ON A COMMON TEMPORAL SCALE (~6,900-1,700 BC). CULTURAL DIVISION ARE BASED ON THOSE GIVEN IN HVASS AND STORGAARD, 1993. FIGURE 6.2- SELECTED PARAMETERS PROVIDING INFORMATION ABOUT SALINITY AT ALL THREE SITES (KILEN, NORSMINDE FJORD, KORUP SØ; ~7,400-1,700 BC). CULTURAL DIVISION ARE BASED ON THOSE GIVEN IN HVASS AND STORGAARD, 1993. REGIONAL SEA LEVEL CURVE FROM BLEKINGE, SE SWEDEN (BERGLUND ET AL., 2005, M A/B PSL = METRES ABOVE/BELOW PRESENT SEA LEVEL.). *INCLUDES THE HIGH SALINITY DEMANDING FORAMINIFER ELPHIDIUM INCERTUM (SHADED BLACK). † = INCLUDES RUPPIA SPP. (SHADED GREY). DI-SALINITY = DIATOMS INFERRED SALINITY, MI-SALINITY = MOLLUSC-FIGURE 6.3 - SELECTED PARAMETERS PROVIDING INFORMATION ABOUT PRODUCTIVITY AND/OR NUTRIENT STATUS AT ALL THREE SITES (KILEN, NORSMINDE FJORD, KORUP SØ; ~6,900-1,700 BC). CULTURAL DIVISION ARE BASED ON THOSE GIVEN IN HVASS AND STORGAARD, 1993. *= MOLLUSC CONCENTRATION DATA SOUARE ROOT TRANSFORMED FIGURE 6.4 - SUMMARY OF KEY CHANGES OCCURRING OVER THE IRON AGE PERIOD (~500 BC-AD 1050) AT THE LIMFJORD SITE KILEN. REGIONAL TEMPERATURE (FROM LAKE TREHÖRNINGEN; ANTONSSON AND SEPPÄ, 2007) AND SEA LEVEL CURVES (SOUTHERN NORTH SEA; BEHRE, 2007) HAVE ALSO BEEN INCLUDED. DCA AXIS 1 REDUCED DATASET EXCLUDES THE UPPERMOST 28 DIATOM SAMPLES (SEE SECTION 3.10 FOR MORE DETAILS). *M NN ASL = METRES IN RELATION TO GERMAN ORDNANCE DATUM ABOVE/BELOW FIGURE 6.5 - COMPARISON OF LIMFJORD PALAEOSALINITY RECORDS FOR THE LAST ~2,500 YEARS (I.E. KRISTENSEN ET AL., 1995; ELLEGAARD, 2000; CHRISTENSEN ET AL., 2004; THIS STUDY). AN UPDATED CHRONOLOGY IS PROPOSED BASED ON THE AMS ¹⁴C DATES FROM TERRESTRIAL MATERIAL IN THIS STUDY. THE BOX IN THE 'KILEN SALINITY' COLUMN COVERS THE PERIOD OF DATING UNCERTAINTY IN THE AGE-DEPTH MODEL FOR THE

List of Tables

TABLE 1.1 – SUMMARY OF THE MAJOR EVENTS LEADING UP TO THE DEVELOPMENT OF THE
MODERN DAY CIRCULATION AND SEDIMENTATION SYSTEM IN THE NORTH SEA,
SKAGERRAK AND KATTEGAT (MODIFIED FROM GYLLENCREUTZ, 2005), ABBREVIATIONS:
IRD = ICE RAFTED DEBRIS, IMZ – ICE MARGINAL ZONE
TABLE 2.1- SUMMARY OF KILEN MONITORING DATA (AVERAGES FOR THE PERIOD 1989-1996, 66
TABLE 2.2 - DETAILS OF ALL PROXIES ANALYSED FOR KILEN. NORSMINDE AND KORUP SØ
ABBREVIATIONS: LT – LITTORINA TRANSGRESSION (7 000-6 600 BC) M-N =MESOLITHIC-
NEOLITHIC TRANSITION (4 500-3 500 BC) PWC = PITTED WARE CULTURE (2 400-2 800 BC)
$PR/R = PRE_ROMAN/ROMAN PERIOD (500 BC = AD 400) LU = I OUGHBOROUGH UNIVERSITY$
GEUS – GEOLOGICAL SURVEY OF DENMARK AND GREENLAND ALL – AARHUS
UNIVERSITY II - IONATHAN I EWIS BS - BETH STAVNGAARD *- INCLUDES ORGANIC
MATTER CARBONATE CONTENT MINEROGENIC MATTER WATER CONTENT AND DRY
MASS 73
TABLE 2 3 ORGANIC MATTER AND CARBONATE HEATING PROGRAMMES FOR LOSS-ON-
IGNITION RESTING PHASES ARE 10 MINUTES LONG UNLESS OTHERWISE SPECIFIED 81
TABLE 2 4 – SUMMARY OF THE STOCK SOLUTIONS USED DURING THE EXTRACTION AND
PREPARATION STAGES OF SEDIMENTARY PIGMENT ANALYSIS (MODIFIED FROM HPI C
LABORATORY HANDBOOK DR SUZANNE MCGOWAN UNIVERSITY OF NOTTINGHAM) 85
TABLE 2.5 - SEPARATION CONDITION USED IN THE ANALYTICAL PROCEDURE (METHOD
MODIFIED FROM CHEN <i>ET AL.</i> 2001) STATIONARY PHASE (COLUMN) CONSISTING OF A
THERMO SCIENTIFIC ODS HYPERSII, COLUMN (205*4.6 MM· 5 µM PARTICLE SIZE) 86
TABLE 3 1 - LITHOLOGICAL DESCRIPTIONS FOR THE KILEN SEDIMENTARY PROFILE
DESCRIPTIONS ARE BASED ON A MODIFIED TROFI S-SMITH SYSTEM (TROFI S-SMITH 1955)
SFE ALSO BIRKS AND BIRKS 1980) 94
TABLE 3.2 - DETAILS OF THE SAMPLES DATED BY AMS ¹⁴ C FROM THE KILEN PROFILE (*
ESTIMATED VALUE) 98
TABLE 3 3 - SUMMARY OF DCA AND PCA RESULTS FOR THE KILEN DIATOM DATASET 130
TABLE 3.5 - SCALARIE OF DEFINING FORMALS FOR TAXA >3 % ABUNDANT IN ONE OR MORE SLIDES
IN THE UPPERMOST SECTION OF THE KILEN DIATOM RECORD (27 SAMPLES: POST AD
1856)
TABLE 3.5 - DCA AXIS 1 SPECIES SCORES FOR TAXA >3 % ABUNDANT IN ONE OR MORE SLIDES
IN THE UPPERMOST SECTION OF THE KILEN DIATOM RECORD (82 SAMPLES: ~5.430 BC-AD
800). HABITAT PREFERENCES ARE BASED ON SNOELIS <i>ET AL.</i> , 1993-1998, UNLESS
OTHERWISE REFERENCED
TABLE 4.1- ORIGINAL LITHOLOGICAL DESCRIPTIONS OF THE SEDIMENTS FROM THE
NORSMINDE PROFILE, USING A MODIFIED TROELS-SMITH SYSTEM (TROELS-SMITH, 1955).
TABLE 4.2 - DETAILS OF ALL AMS ¹⁴ C DATED SAMPLES FOR THE NORSMINDE PROFILE.
SAMPLES IN GREY WERE OMITTED FROM THE FINAL AGE MODEL (SEE TEXT). *=
ESTIMATED VALUE
TABLE 4.3 - SUMMARY OF NORSMINDE FJORD DIATOM AND POLLEN DCA/PCA RESULTS.
ANALYSES PERFORMED ON ENTIRE DATASETS FOR BOTH POLLEN AND DIATOMS AND A
REDUCED DATASET (63 SAMPLES) FOR DIATOMS (SEE TEXT FOR DETAILS; SECTION
4.10.1.). PCA PERFORMED ONLY ON ENTIRE POLLEN DATASET, APPLYING ATCHISONS LOG
RATIO (*) FOR PCA LINEAR DATA
TABLE 4.4- NORSMINDE FJORD DIATOMS DCA AXIS 1 RANKED SPECIES SCORES (TAXA >1 %
ABUNDANCE ONLY). A = FULL DATASET (69 SAMPLES; 6.900-2.000 BC), B= REDUCED
DATASET (63 SAMPLES; 6,600-2,000 BC)
TABLE 4.5 - NORSMINDE FJORD POLLEN PCA AXIS 1 RANKED SPECIES SCORES (TAXA >1 %
ABUNDANCE ONLY) FOR THE ENTIRE DATASET (I.E. 62 SAMPLES)

TABLE 5.1 - ORIGINAL LITHOLOGICAL DESCIPTIONS OF THE SEDIMENTS FROM WELL NO. 1
KORUP SO, (H. KROG AND B. ODGAARD, GEOLOGICAL SURVEY OF DENMARK, NOW
GEOLOGICAL SURVEY OF DENMARK AND GREENLAND)
TABLE 5.2 - DETAILS OF ¹⁴ C DATED SAMPLES FROM THE TWO KORUP SØ BOREHOLES (WELL
NO. 1 & WELL NO.2). *THE DATING RESULTS ARE REPORTED ACCORDING TO
INTERNATIONAL CONVENTION (STUIVER AND POLACH, 1977) AS CONVENTIONAL ¹⁴ C
DATES IN ¹⁴ C YR BP (BEFORE AD 1950) BASED ON THE MEASURED ¹⁴ C/ ¹³ C RATIO
CORRECTED FOR THE NATURAL ISOTOPIC FRACTIONATION BY NORMALIZING THE
RESULT TO THE STANDARD δ^{13} C VALUE OF -25‰ VPDB ANDERSEN <i>ET AL.</i> , 1989) 215
TABLE 5.3 – UNAMBIGUOUS POLLEN STRATIGRAPHIC MARKER LEVELS INCORPORATED INTO
THE DEPOSITIONAL AGE MODEL (J. OLSEN, P. RASMUSSEN, UNPUB. DATA)
TABLE 5.4 - SUMMARY OF DCA/PCA RESULTS FOR THE FULL POLLEN, MOLLUSC AND DIATOMS
DATASETS
TABLE 5.5- SUMMARY OF DCA RESULTS FOR THE REDUCED DIATOM DATASET (50 SAMPLES)
TABLE 5.6 - DCA SPECIES SCORE FOR THE MOLLUSCAN DATASET 235
TABLE 5.7 - SUMMARY OF PCA AXIS 1 AND 2 SPECIES SCORES FOR THE KORUP SO POLLEN
RECORD (TAXA > 1 % ABUNDANCE IN ONE OR MORE SLIDES)

Abstract

In this study a multiproxy approach (including sedimentary techniques, diatoms, molluscs, foraminifera, sedimentary pigments, isotopes, pollen and plant macrofossils) has been adopted to assess environmental change over the last \sim 9,000 years at three Danish coastal sites (Kilen, Norsminde Fjord and Korup Sø). Particular focus has been placed on periods of intense human coastal occupation, identifiable in Denmark's rich coastal archaeological record (i.e. shell midden accumulation periods), to test critically, hypotheses that changes in the marine environment were contemporary with major cultural and societal changes over the last ~9,000 years. For example, it has been proposed that a decrease in salinity was responsible for the widespread oyster decline, apparent in the Neolithic layers of a number of Danish shell middens. This hypothesis, however, remains speculative to date, lacking any high-resolution and quantitative salinity data covering the Mesolithic-Neolithic transition. Inside the agricultural era, two more phases of shell midden accumulation occur (i.e. during the Pitted Ware/Single Grave cultural period and the Iron Age), suggesting that people must have returned to the sea at these times for increased exploitation of its resources. A diatom-based salinity transfer function (WAPLS-C3 model, $r_{boot}^2 = 0.923$, RMSEP= 0.36 square root salinity units) based on a trans-Baltic training set has been applied to fossil diatom datasets from each site for quantitative assessment of salinity change over the study period. The multiproxy results presented in this study demonstrate a close connection between environmental change and human exploitation of marine resources over the Holocene. This relationship, however, is complex, with the individual fjord systems often exhibiting spatially different responses (i.e. variations in the sedimentary regime, salinity, productivity and nutrient status) to changes in key forcing mechanisms such as sea level change, climate change and human impact upon the catchment (following the introduction of agriculture). Environmental hypotheses for cultural change are reviewed on the basis of the evidence presented in this study. Diatomenvironmental relationships have also been modelled (using multivariate techniques) at Korup Sø and Norsminde Fjord using proxy data as 'predictor' variables for changes in the terrestrial and marine environment. These results suggest that a variety of marine, climatic, human and catchment related processes are important in explaining a proportion of the variation in the fossil diatom datasets, but these influences tend to vary temporally throughout the profile (e.g. human impact becomes important after ~3,900 BC).

Key words: coastal, Denmark, Holocene, diatoms, sedimentary pigments, isotopes, palaeosalinity, productivity, environmental change, archaeology.

Acknowledgements

Special thanks go to Dr. David Ryves and Dr. Peter Rasmussen who have been fantastic supervisors over the past four years, always being inspiring, helpful, available, knowledgeable, calming, generous, and able to share a laugh and a joke. Professor John Anderson is also thanked for constructive criticism throughout, readily highlighting contemporary issues, potential problems and offering interesting ideas/solutions. I feel fortunate to have been mentored by such a strong and charismatic team, exercising thorough and high quality science and look forward to the prospects of future research, collaboration and hopefully many publications in this company.

I am also particularly grateful to Dr. Suzanne McGowan at the University of Nottingham geography department for help with the sedimentary pigment lab work and again helpful discussion. Professor Melanie Leng is also thanked for help and support during the isotopic analysis. Professor Karen Luise Knudsen is thanked for much need guidance with foraminifera identification and ecology, providing a truly enjoyable and informative marine micropalaeontology course and for helpful comments throughout the course of this thesis. I am indebted to all people who have put the long hard hours in generating high quality datasets for each site that have been priceless to the palaeoenvironntal interpretations presented in this thesis. These inlcude Dr. Peter Rasmussen (pollen and plant macrofossil analyses), Kaj Strand Petersen (molluscan analysis), Harald Krog (pollen analysis; Korup Sø), Peter Konradi (foraminifera), Beth Stavngaard (sedimentary physical parameters) Peter Kristensen (foraminifera) and MOLTEN, Define and Detect team members (salinity training set, particularly Dr. Kaarina Weckström and Dr. Annemarie Clarke for taxonomic and transfer function help). Of course, all analyses need context and therefore Dr. Jesper Olsen (Aarhus University) is gratefully acknowledged for valuable work and production of the age-depth models for all three sites. Extended acknowledgments go to and Thjobjørn Andersen (Copenhagen University) and Jan Heinemeier (Aarhus University) for their roles in the radiocarbon and lead dating processes. The Quaternary Research Association (QRA) and the Natural Environment Research Council (NERC) are thanked for supplying funds for sedimentary pigment analysis and isotope analysis respectively. Peter Gravesen (GEUS) is also thanked for kindly supplying funds for AMS ¹⁴C dates for the Kilen profile. Professor Helen Rendell, Dr. Ole Bennike, Professor Søren Andersen and Professor Bent Odgaard are thanked for discussion and/or field work support. Thanks also go to everybody at Loughborough University Geography department and the Geological Survey of Denmark and

Greenland (GEUS) for helpful questions and comments throughout and for making it a truly memorable few years.

Lastly, I am indebted to Sally and my family for making the long writing up hours less painful, supporting (and putting up with) me during the stresses, helping out whenever needed (including financial support), patience, care and keeping me happy and upbeat till the end.

Chapter 1. Introduction, background and aims

1.1 Introduction

Paramount to our understanding and management of highly productive and commercially important biomes such as the coastal zone, is a firm grasp of internal dynamics (e.g. the impacts of changes in salinity, nutrient, productivity and temperature upon ecological communities) within local and regional systems, both at present and in the past. Coastal shelves currently account for 20 % of all marine total primary productivity despite only making up 10 % of the total area of the ocean floor (Scourse et al., 2004). However, ecosystem functioning and the impacts of climate change (e.g. sea level change, saltwater intrusion) and human impact (e.g. nutrient enrichment, dredging, overfishing) upon the various different coastal habitats (e.g. estuaries, coral reefs, tropical mangroves, lagoons) remain poorly understood. Estuaries are extremely common shelf habitats, being found on all continents and at all latitudes across the world. An estuary is a partially enclosed body of water formed where freshwater from rivers or streams flows into the oceans, mixing with the seawater (McLusky and Elliot, 2004). They can originate in a variety of ways (e.g. drowned glacial troughs, drowned river valleys) and are often distinguished by the degree of mixing between the fresh and salt water masses (e.g. salt-wedge estuaries; Mississippi River mouth (USA), well mixed tidal estuaries; Severn Estuary (UK), Delaware Bay (USA), partially mixed estuaries; San Francisco Bay (USA), Mersey River, (UK)), though these states can alter throughout the year.

Estuaries tend to be dynamic habitats characterised by continuously fluctuating temperatures, nutrient concentrations and salinities, steep horizontal and vertical salinity gradients, as well as being subject to variations in sedimentation patterns and rates of river input. They are ecologically stressful habitats and therefore only species which can tolerate such variable conditions tend to reach high abundances (Yonge, 1960; Barnes, 1974; Wiley, 1976; Day *et al.*, 1989; Attrillf and Rundle, 2002). Estuaries also have immense societal importance both at present and in the past. Today many of the largest metropolitan areas border estuarine areas (e.g. London, New York, Tokyo, Shanghai) and in the past some of the earliest civilisations developed along them (e.g. coastal occupation extends to as early as ~5,500 BC in the Yangtze Delta and Nile Delta; Stanley and Warne, 1993; Daniel and Chen, 1996; Stanley and Warne, 1997; Zhang *et al.*, 2005). Their very high primary and secondary productivity helps supports commercial fisheries, with a large number of

commercial fish species dependent on estuaries for at least part of their life cycle (Claridge *et al.*, 1986; MacKenzie *et al.*, 2002). In heavily human impacted systems, eutrophication, pollution and over-fishing may also have serious detrimental effects on coastal ecosystems (Cooper and Brush, 1991; Jørgensen and Richardson, 1996; Blaber *et al.*, 2000; Jackson *et al.*, 2001; Conley *et al.*, 2002b; Md. Islam and Tanaka, 2004; Scheffera *et al.*, 2005; Clarke *et al.*, 2006; Lehtonen and Schiedek, 2006; Smith, 2006; Andersen and Conley, 2009).

In glaciated regions, fjords and fjards are the most common types of estuary originating from glacially scoured troughs being drowned by the sea. Fjords tend to be U-shaped overdeepened basins, whereas fjards usually form in extensive continental shelf regions and tend to be shallower, avoiding over-deepening due to isostatic rebound of the continent. Both are particularly characteristic features along Scandinavian coastlines, due to former ice sheets having completely covered the entire Baltic region in the past (Petersen, 1985; Mangerud et al., 1996). For example, in Denmark alone, over 80 estuaries and fjords exist (Conley et al., 2000). The east coast of Denmark and the inner Danish coastal waters are both particularly interesting and important as these systems open out into largely enclosed micro-tidal (tidal range <2m; Davies, 1964, Danish tides as low 20-30 cm) brackish seas (e.g. Kattegat, Limfjord and western Baltic). The salinity of these seas is very closely connected to the degree of connection with North Sea (strongly related to sea level) and the climate over the Baltic catchment area (Svansson, 1975; Rodhe, 1992; Kristensen et al., 1995; Christensen et al., 2004). There is some evidence to suggest that periods of salinity maxima might coincide with high sea level stands (Iversen, 1937; Mikkelsen, 1949), but this remains critically untested (discussed in more detail below).

The ability of estuaries to record both environmental change from both the land and the sea, make them extremely important archives of palaeoenvironmental change. They are natural sediment traps that often enable good temporal resolution to be obtained due to high sedimentation rates, that have been known to exceed 10 m kyr⁻¹ (Jennings and Weiner, 1996; Maslin *et al.*, 1998; Maslin and Swann, 2006). The presence of terrestrial macrofossils in estuarine sediments also enables radiocarbon dating of terrestrial material, thereby avoiding marine reservoir effect problems often encountered when dating marine material (e.g. foraminifera, molluscs) from coastal sediments or deep sea cores (Bard *et al.*, 1987; Bard *et al.*, 1991; Lowe and Walker, 2000; Larsen *et al.*, 2002; Ascough *et al.*, 2005). Such systems also support dense human populations both at present and in the past, offering the rare

opportunity to assess directly how marine changes affected human communities living at or near coastal sites.

The extremely rich Danish archaeological record indicates intense coastal/estuarine habitation and heavy dependence on marine resources by cultures and societies living in Denmark during the Holocene (Rasmussen, 1968; Andersen, 1989; Enghoff, 1989; Andersen and Rasmussen, 1991; Holm, 1991; Andersen, 1995a, 2000; Holm and Bager, 2002; Andersen, 2007; Andersen, 2008a). The presence and availability of marine resources tend to be sensitively regulated by environmental parameters such as salinity, sea-level, sedimentary regime, nutrient status, primary productivity, temperature and oxygen concentrations (e.g. Jensen and Spärck, 1934; Rasmussen, 1958; Yonge, 1960; Holm and Bager, 2002; Bailey and Milner, 2008). This has subsequently led to a number of environmental hypotheses suggesting that changes in the archaeological record (see below) might have been forced by changes in the natural marine environment (e.g. Rowley-Conwy, 1984). A clear understanding of the impacts that sea level and climate change have had upon past Danish estuarine systems (e.g. salinity, nutrient status, ecosystem structure, marine resources) is particularly important for predicting how projected future changes in sea level, climate and nutrient status (e.g. Solomon et al., 2007; HELCOM; BACC, 2006) might affect marine resources and influence people living in coastal regions both today and in future. Understanding the internal dynamic of such systems is also paramount to our understanding of the global nutrient and biogeochemical cycling of major elements (e.g. carbon, nitrogen, iron and silica).

This study uses high resolution, palaeoenvironmental techniques (i.e. diatoms, sedimentary pigments and isotopes) in order to reconstruct changes in salinity, productivity, nutrients and sedimentary conditions over key archaeological events to critically assess environmental hypotheses for cultural change (e.g. decline in oysters, Rowley-Conwy, 1984; see Section 1.6). This chapter introduces the major research aims of the current study and provides background information for, and current understanding of the past and present, Baltic Sea, Kattegat and Danish coastal waters. The history of human exploitation of marine resources evident in the Danish archaeological record is also summarised. Specifically this chapter aims to address the following issues/questions:

1. What is the present day oceanographic system (e.g. currents, salinity gradients) operating in the transitional waters (i.e. Skagerrak, Kattegat, Belt Sea, Øresund and the

adjacent inner Danish coastal waters) between the North Sea and Baltic Sea and how did the present oceanographic systems evolve?

- 2. How has the marine environment of Baltic Sea, Kattegat and inner Danish waters changed over the Holocene, particularly in response to changing sea level and climate?
- 3. What factors influence sea level, salinity and productivity in the modern day system and how have palaeoecological techniques been used to reconstruct long term changes in these parameters?
- 4. How well are Holocene changes in these environmental parameters currently understood for the Danish Baltic region and what are the key outstanding issues/ uncertainties?
- 5. How has human resource exploitation changed/evolved over the Holocene as detailed by the Danish archaeological record?
- 6. What environmental hypotheses have been proposed to explain some of the cultural and societal changes identifiable in the archaeological records?
- 7. What are the specific aims of the present study and what techniques have been chosen (and why) in order to attempt to answer these research questions?

1.2 The modern Baltic Sea, Kattegat and inner Danish coastal waters

Today circulation in the Kattegat and Skagerrak is mainly driven by the input of cold saline water (30-35‰) from the North Sea via the North Atlantic, Central North Sea and North Jutland currents (NJC), counteracted by intermittent inputs of cold brackish water (8-12 ‰) from the Baltic Sea via Baltic currents (Svansson, 1975; Rodhe, 1987; Figure 1.1). The NJC is formed at the North western tip of Jutland, where the South Jutland Current (SJC) meets central North Sea and Atlantic water masses (e.g. the dominant South Trench current (STC) and the lesser Tampen Bank current (TBC)). The SJC is a product of English Channel water masses mixing with southern North Sea water and flows northwards along the sand-dominated coast of west Jutland, amassing high concentrations of suspended particles (Gyllencreutz *et al.*, 2005). The combined NJC then flows into the Skagerrak along the northern coast of Jutland until it mixes with Baltic surface currents at the entrance to the Kattegat and flows off in a northerly direction, along the Swedish and Norwegian coasts becoming the Norwegian coastal current (NCC). Further mixture with the north Atlantic currents forms an anticlockwise gyre in the deep northeastern region of the Skagerrak and results in reduced velocity of currents and sedimentation of fine grained materials at rates of

up to 1 cm yr⁻¹ (Gyllencreutz *et al.*, 2005; Gyllencreutz *et al.*, 2006). Part of the NJC current also enters the Kattegat at depth to counteract the lower salinity Baltic water (or Baltic current) flowing out from the Baltic Sea at the surface through the Danish straits (i.e. the three channels – Store Bælt, Lille Bælt and Øresund, connecting the Baltic Sea to the North Sea through the Kattegat and Skagerrak). This outflow produces a north-south salinity gradient ranging from approximately 30 ‰ in the north to 15 ‰ in the south (Falt, 1982; Jiang *et al.*, 1998; Figure 1.1).

Reconstruction of palaeo-currrent patterns in the Skagerrak and Kattegat indicate that this modern circulation system began approximately 6,550 BC (all years are given as calendar years unless otherwise stated) following a series of hydrographic shifts associated with the opening of the English channel and the Danish straits (Gyllencreutz, 2005; Gyllencreutz *et al.*, 2005; see below). The events leading up to the establishment of the modern circulation pattern and the subsequent evolution of current strength and modern day sedimentation regime are summarised in Figure 1.2 and Table 1.1. This stratigraphy is based on chirp sonar data, mineral magnetic and grain size analyses from the high resolution MD99-2286 core (Figure 1.1) collected from the north-eastern Skagerrak and dated via 27 AMS ¹⁴C dates on mollusc shells or foraminifera tests (with marine reservoir effect correction applied; Gyllencreutz, 2005; Gyllencreutz *et al.*, 2005; Gyllencreutz *et al.*, 2005; Gyllencreutz *et al.*, 2006; Gyllencreutz and Kissel, 2006).

Whilst these major events appear to be correlative across the North Sea, Skagerrak and Kattegat, precise dating has proved somewhat problematic. For example, the opening of the English Channel has been previously placed somewhere between 7,050-5,750 BC (Jelgersma, 1979; Conradsen and Heier-Nielsen, 1995; Lambeck, 1995; Jiang *et al.*, 1997) and the Danish straits between 7,050-6,250 BC (Björck, 1995; Conradsen, 1995; Lambeck, 1999).



Figure 1.1 - Map of the Baltic Sea, with regional modern day average surface salinities and present current circulation system for the North Sea, Skagerrak and Kattegat (after Gyllencreutz *et al.*, 2005 and Longva and Thorsnes, 1997). Sites analysed in this thesis are indicated by red dots. The blue dot signifies the location of core MD99-2286 collected from the Skagerrak (Gyllencreutz *et al.*, 2005). Abbreviations: A = Lillebælt, B = Storebælt, C = Øresund, CNSW = central North Sea water, AW = Atlantic water, NCC = Norwegian coastal current, TBC = Tampen Bank current, SJC = south Jutland current, NJC = north Jutland current, BW = Baltic water, BC = Baltic current, CNSW = central North Sea water, SNSW = southern North Sea water. Map downloaded from HELCOM: http://maps.helcom.fi/website/mapservice/index.html

Marine Event	References	¹⁴ C age BP	Age cal. BP	Age cal. BC
Transgression of the southern North Sea	Stabell and Thiede, 1986; Lambeck, 1995	10,000	11,600-11,300	9,650-9,350
End of IRD dep. in Skagerrak	van Weering, 1982b, 1982a	10,000	11,600-11,300	9,650-9,350
Aker IMZ, Oslo area	Gjessing and Spjeldnaes, 1979; Sørensen, 1979; Gjessing, 1980; Andersen <i>et al.</i> , 1995	9,800-9,600	11,300-10,800	9,650-8,850
Marine limit, Oslo area	Hafsten, 1983	9,700	11,200-10,800	9,250-8,850
Glomma drainage event	Longva and Bakkejord, 1990; Longva and Thoresen, 1991	9,100	10,400-10,200	8,450-8,250
Otteid-Stenselva strait closing	Björck, 1995; Lambeck, 1999	9,100	10,400-10,200	8,450-8,250
English Channel opening	Nordberg, 1991 Conradsen and Heier-Nielsen, 1995 Jiang <i>et al.</i> , 1997 Björklund <i>et al.</i> , 1985 Jelgersma, 1979; Lambeck, 1995	8,000 7,600 7,700 8,000-7,000 8,000-7,000	9,000-8,700 8,500 8,600 9,000-7,700 9,000-7,700	7,050-6,750 6,550 6,650 7,050-5,750 7,050-5,750
Danish straits opening	Björck, 1995 Conradsen, 1995 Lambeck, 1999	8,200 8,000 7,500-7,800	9,300-9,000 9,000-8,700 8,700-8,200	7,350-7,050 7,050-6,750 6,750-6,250
Isolation of Dogger Bank	Lambeck, 1995	8,000	9,000-8,700	7,050-6,750
Skagerrak–Kattegat hydrographic shift	Conradsen, 1995; Conradsen and Heier-Nielsen, 1995	5,500	6,200	4,250
	Jiang <i>et al.</i> , 1997 Nordberg and Bergsten, 1988; Nordberg, 1991	5,100 4,000	5,900 4,600-4,300	3,950 2,650-4,350

Table 1.1 – Summary of the major events leading up to the development of the modern day circulation and sedimentation system in the North Sea, Skagerrak and Kattegat (Modified from Gyllencreutz, 2005). Abbreviations: IRD = ice rafted debris, IMZ – ice marginal zone.



Figure 1.2 - Summary of Baltic Sea level, salinity and total organic carbon (TOC) history. A= Traditional division of the Baltic Sea's development stages (cf. Hyvärinen, 1988). B= Current Baltic Sea stage divisions (after Andrén *et al.*, 2000; Berglund *et al.*, 2005; Andrén *et al.*, 2007b). C= Representative diatom assemblages from open waters (after Andrén *et al.*, 2000). T. = *Thalassiosira* (column C). D= Salinity stages of the Baltic Sea's history. E= Total organic carbon (TOC) curve from the Gotland Basin (Andrén *et al.*, 2000). F= Blekinge Sea level curve based on shore displacement (from Blekinge SE Sweden, Berglund *et al.*, 2005; Andrén *et al.*, 2007b); L1-L5 refer to *Littorina* transgressions as recorded in coastal sediments from the Blekinge region of south-eastern Sweden (after Berglund *et al.*, 2005). G. Summary of major marine event and main sediment source and transport pathways into the Skagerrak (core MD99-2288) over the last 12,000 years (after Gyllencreutz and Kissel, 2006). 1. Glacial marine sediment from meltwater discharge (source: southern Norway and SW Sweden). 2. Glacial marine sediment from meltwater outflow (source: Baltic Sea via the Vanern basin). 3. Marine sediment (source: eastern North Sea via the North Jutland Current). 4. Marine sediment (source: chiefly from the southern North Sea and the Atlantic Ocean, transported by the North and South Jutland current, but also from the Baltic Sea and reworked coastal sediments in Sweden and Norway transported by the Baltic Current and currents along the coasts of western Sweden and southern Norway). 5. Marine sediment (source: chiefly from the southern North Sea and the Atlantic Ocean, transported by the North and South Jutland). IRD = Ice rafted debris. NS = opening of Närke Strait, OS = closing of Otteid Stenselva Strait. EC = opening of the English channel (along with increased flow of Atlantic water, opening of the Danish straits and beginning of the modern circulation system). 8.2k = large scale cooling event recorded in the G

1.3 Sea Level and Salinity change during the *Littorina* Sea Stage

In semi-enclosed seas such as the Baltic Sea, Kattegat and Limfjord, fluctuations in sea level and salinity are common on both short and long timescales, and have important propagative impacts upon other environmental variables such as nutrient cycling and marine productivity (Bianchi et al., 2000; Zillén et al., 2008). They also have clear implications for humans living in the coastal environment and exploiting marine resources both at present (MacKenzie et al., 2002; BACC, 2006) and in the past (Rowley-Conwy, 1984; Enghoff, 1999; Holm and Bager, 2002; Andersen, 2007; Enghoff et al., 2007). For example, successful cod spawning in the Baltic Sea requires regular renewal of oxygenated, saline water from the North Sea, while prolonged periods of oxygen depletion in the deep waters inhibit the successful development of cod eggs (Nissling and Westin, 1997; MacKenzie et al., 2002; Köster et al., 2005). This deep-water hydrographic exchange is driven by sea level and atmospheric pressure differences between the southern Kattegat and western Baltic Sea. Surface salinity, and perhaps to some extent sea level, is further affected by freshwater inputs from the large catchment draining into the Baltic Sea. This section considers how salinity, sea level and hydrography have changed since the beginning of the Littorina Sea stage (~7,850 BCpresent; Figure 1.2), and how such changes have affected the hydrographic and ecological systems of the Baltic Sea, Kattegat and inner Danish coastal waters (e.g. Limfjord).

1.3.1 The Ancylus Lake-Littorina Sea transition (7,850-6,550 BC)

Prior to the *Littorina* Sea stage, the Baltic Sea underwent a series of freshwater and brackish phases (Figure 1.2) caused by complex interplays between ice retreat, global sea level rise and isostatic land adjustment (see Björck, 1995; Andrén *et al.*, 2000; Björck, 2008 and references therein for a full review). Marine waters reached the Skagen area of northern Jutland by 13,050 BC (Knudsen, 1994) and inundated the Kattegat between 11,550-11,050 BC (Petersen, 1985), but the southern Baltic Sea remained freshwater, isolated via a land bridge connecting eastern Jutland to south west Sweden. Brackish-marine conditions in the Baltic Sea were not established until between 6,550-6,150 BC (Andrén *et al.*, 2000; Bennike *et al.*, 2004; Björck, 2008). However, the earliest pulse of saline water through the Great Belt is now believed to have occurred as early as 7,850 BC (Andrén *et al.*, 2000). This, probably combined with several other intermittent intrusions of salt water, resulted in a very weakly

brackish phase, recently termed the Early *Littorina* Sea Stage from 7,850-6,550 BC (Berglund *et al.*, 2005; Björck, 2008) or *Mastogloia* Sea Stage (Sundelin, 1922). The impact of these intermittent intrusions probably only affected the most southern and western regions of the Baltic Sea with only very minor changes in the diatom flora being detectable in the Gotland Basin (Andrén *et al.*, 2000) and southern Baltic coast (Saarse *et al.*, 2009; Witkowski *et al.*, 2009).

The major post-glacial marine transgression beginning at 6,550 BC and complete by 6,150 BC, often termed the Initial *Littorina* transgression, is documented across the entire Baltic Sea, Kattegat and Limfjord (Mörner, 1976; Petersen, 1981; Winn et al., 1988; Petersen, 1993; Björck, 1995; Winn et al., 1998; Andrén et al., 2000; Hoffman and Winn, 2000; Bennike et al., 2004; Berglund et al., 2005; Miettinen et al., 2007; Rasmussen, submitted), and was caused by global sea level rise associated with the final stages of deglaciation of the northern hemisphere ice sheets (Berglund et al., 2005). This eventually drowned the Øresund and Danish Great Bælt, creating a permanent connection between the Skagerrak/Kattegat and Baltic Sea, thus beginning the *Littorina* Sea Stage as we know it today. Also associated with this global sea level rise was the opening of the English Channel (~7,050-5,750 BC; Nordberg, 1991; Conradsen and Heier-Nielsen, 1995; Lambeck, 1995; Table 1.1), the initiation of the Jutland current and the beginning of the modern Skagerrak-Kattegat current circulation system (see Figure 1.1) (Svansson, 1975; Rodhe, 1987; Otto et al., 1990; Rodhe, 1996; Jiang et al., 1998; Gyllencreutz and Kissel, 2006). This transgression also drowned low lying areas of northern Jutland connecting the Limfjord to the Skagerrak, though the exact date for this inundation remains uncertain (Petersen, 1981). It is likely that the Limfjord remained detached from the North Sea to the west, due to the existence of an emergent deposit ~15 km off the current coastline, known as the Jutland Bank (e.g. Leth, 1996). It is believed the Jutland Bank was eventually drowned in the mid-Holocene (~4,250 BC; Leth, 1996) as sea levels continued to rise in the North Sea (cf. Behre, 2007).

The *Littorina* transgression caused the salinity of the southern Baltic to rise up to levels similar to present day (10 ‰), and the regular input of nutrient-rich sea water at depth from the Kattegat created a vertical halocline and initiated cyclic blooms of nitrogen-fixing cyanobacteria (Bianchi *et al.*, 2000; Borgendahl and Westman, 2007). North Sea water entering the Kattegat and Baltic Sea at depth is highly saline, nutrient-rich and oxygenated, and mixes poorly with the overlying freshwater surface inflow from the catchment (mainly

via river runoff and glacial meltwater). The depth of this halocline is both spatially and temporarily variable, being dependent upon weather and climatic factors (e.g. precipitation, atmospheric pressure) that regulates the amount and periodicity of saline water input into the Baltic Sea basin (Gustafsson and Andersson, 2001; BACC, 2006).

A minor regressive phase which has been dated to approximately 6,250-6,150 BC in the Blekinge records southeastern Sweden appears to have interrupted the Littorina transgression (Berglund et al. 2005). This is broadly correlative with the widespread "8.2 k" (~6,250 BC) cooling event (Alley et al., 1997; Rohling and Pälike, 2005) and has been linked to a possible deterioration in weather conditions and alterations to the atmospheric pressure systems (e.g. North Atlantic Oscillation (NAO), Hurrell, 1995) operating over northern Europe (Berglund et al., 2005). For the recent past, it has been demonstrated that sea level variations in the Baltic Sea and Kattegat are closely connected with predominant wind patterns operating over the Baltic Sea area that are in turn governed by atmospheric pressure cells (Hanninen et al., 2000; Andersson, 2002; Hunicke and Zorita, 2006; Hunicke et al., 2008). For example, stronger winter westerly winds associated with a positive NAO can cause a rise in sea level in some areas of the Baltic Sea (Andersson, 2002; Woolf et al., 2003; Jevrejeva et al., 2005) by piling up sea water. It is tenable that atmospheric systems will have impacted Baltic Sea level over much longer timescales and that a global climate anomaly such as the 8.2 k event will have caused disruptions to various regional atmospheric systems. The exact causes of this apparent sea level regression, however, remain largely unknown and un-investigated.

1.3.2 The *Littorina* Sea (~6,550 BC – present)

Sea level during the *Littorina* Sea stage was somewhat dynamic, with the western and southerly areas of the Baltic exhibiting very different sea level curves to those in the northern Baltic Sea. Sites south of the Stockholm-south Finland region and in the Kattegat and Danish/west Swedish coastal waters document a series of step-like sea level transgressions occurring up until ~3,050 BC, with maximum Holocene sea level being reached between 4,750-4,150 BC during the late Atlantic period (Iversen, 1937; Troels-Smith, 1942; Mikkelsen, 1949; Berglund *et al.*, 2005; Björck, 2008). Coincidental with highest sea levels, salinity is believed to have reached its Holocene maximum, as suggested by diatom and molluscan assemblages, isotopic analyses (Mikkelsen, 1949; Punning *et al.*, 1988; Winn *et al.*, 1988; Petersen, 1993; Winn *et al.*, 1998; Andrén *et al.*, 2000) and in natural shell fossil

beds where high-salinity demanding species such as oysters (*Ostrea edulis*, requiring a salinity of at least 23-25 ‰; Jensen and Spärck, 1934; Yonge, 1960) penetrated much further south than their present day limits (Nielsen, 1938).

What caused this series of sea level changes during the early *Littorina* Sea stage, ~6,550-3,050 BC (e.g. five transgressive phases in the Blekinge region; Berglund et al., 2005, Figure 1.2) remains uncertain, but eustatic changes must be predominately responsible, as prior to the Littorina stage, regional isostatic rebound had practically ceased in southern Scandinavia (Mörner, 1979; Björck, 1995). Continued global eustatic sea level rise, caused by the melting of the Northern hemisphere and Antarctic ice sheets (Fairbanks, 1989; Peltier, 2002; Carlson et al., 2007), most likely explains the transgressional peaks prior to ~4,000 BC. For example, the second Littorina transgression (L2 of Berglund et al., 2005) is documented in the Blekinge area of southeastern Sweden by a complex two-stage rapid sea level rise between 5,850-5,350 BC (Berglund et al., 2005; Yu et al., 2007). Yu et al., (2007) attributed the rapid flooding of six elevated basins (between 4.0-7.2 m above present sea level) in the region at 5,550 BC to rapid sea level rise, caused by the abrupt decay of the Labrador section of the Laurentide ice sheet (e.g. Carlson et al., 2007). A similarly rapid sea level rise around this time is broadly correlative across the Baltic Sea (e.g. Bennike *et al.*, 2004) and Kattegat (e.g. Mörner, 1976; Christensen, 1995) as well as being evident in other parts of the world, supporting global sea level rise (Blanchon and Shaw, 1995; Siddall et al., 2003).

After ~4,550 BC, global sea level rise slowed down (Fairbanks, 1989; Bard *et al.*, 1996). In the southern and western Baltic, sea level begins to decrease at 4,250 BC and the magnitude of subsequent transgressions appears to decrease before eventually ceasing by ~3,050 BC (Björck, 2008). Ingólfsson and Hjort, (1999) suggest that between ~5,050-3,050 BC global sea level rise is predominately of Antarctic origin. Alternatively, as indicated above, shorter term variations in atmospheric pressure systems and storminess over northern Europe can also have impacts upon sea level and salinity within the Baltic and Kattegat region (e.g. Hurrell, 1995; Hanninen *et al.*, 2000; Yu *et al.*, 2003) and may offer a more localised explanation for the later more minor variations in sea level.

In the northern Baltic (approximately just north of Stockholm in the west and south Finland in the east), only the initial *Littorina* transgression (L1; Berglund *et al.*, 2005) is evident, with none of the subsequent transgressions being recorded in sedimentary records from the region

(Björck, 1995; Jantunen and Donner, 1996; Miettinen *et al.*, 2007; Björck, 2008). In the more northerly areas of the Baltic Sea, the rate of isostatic readjustment (rise) remained high enough to either exceed or at least keep pace with eustatic sea level rise (Björck, 2008) thereby overriding the changes associated with global sea level rise identifiable in the southern and western Baltic Sea.

By ~3,050 BC, the whole Baltic Sea was in a state of regression as sea level gradually fell to that of present day (Andrén *et al.*, 2000; Berglund *et al.*, 2005). Whilst some evidence exists for several minor, and short lived sea level high stands occurring in the more recent past (e.g. Roman period and Viking Age transgressions; Miller, 1982; Miller and Hedin, 1988), none appear to be as large in magnitude as those from early *Littorina* Sea. Three further stages (Post *Littorina* Sea; ~1,750-550 BC, Recent Baltic Sea; ~550-1950 BC and Present Baltic Sea; AD 1950-present) have been identified in the Baltic Sea history by Andrén *et al.*, (2000), but these are related to changes in core sedimentology, chemistry or biological communities as a response to climate, oxygen levels and nutrient status rather than sea level.

1.4 Drivers of Salinity Change in the Baltic

The salinity of the Baltic Sea, Kattegat and inner Danish coastal waters is influenced by a wide range of interrelated factors including sea level, ocean currents, climate change, atmospheric pressure, and freshwater inputs from the surrounding catchment area. A surface salinity gradient exists both within the Baltic and Kattegat, driven by distance to the source of saline water input (Figure 1.1). In the Kattegat, highest salinities are found in the north (25-30 ‰) where it joins the Skagerrak, and salinity decreases in a southerly direction to 10-15 ‰ to the north of the Øresund and Danish Straits. On the eastern side of the straits, the southwestern Baltic Sea has a present day salinity of ~10 ‰, which decreases in a northerly direction. The Gulf of Bothnia, the northernmost arm of the Baltic Sea, experiences almost freshwater conditions (1-2 ‰).

A vertical halocline also exists in the Kattegat and Baltic Sea, created by the input of highly saline, oxygenated (and nutrient rich) North Sea water entering the Kattegat and Baltic Sea at depth, and overlying freshwater sourced from the catchment via rivers and run-off entering at the surface. The depth of this halocline is governed by relative sea level and local climatic and hydrographic conditions (Kullenberg, 1981; Wulff *et al.*, 1990). The shallowness of the

Øresund and Storebælt sills create further complications, as they limit the amount of North Sea water that can enter the western Baltic Sea. Only through major inflow events will significant amounts of highly saline, oxygenated water enter, which is critical for the maintenance of suitable salinity levels to support the current ecosystem (Hanninen et al., 2000; MacKenzie et al., 2002; BACC, 2006). Unfortunately, this replenishment to the Baltic Sea deep water only occurs at irregular intervals, and therefore periods of lowered salinity and hypoxia (low oxygen conditions) occur between saline pulses (Conley et al., 2002a; Feistel et al., 2003; Zillén et al., 2008). In recent decades, under fears that deep water replenishment of the Baltic Sea was becoming less frequent, much effort has been focussed in trying to understand fully what drives the inflows of saline-oxygenated water into the Baltic Sea at annual-decadal scales (Matthäus and Franck, 1992; Lehmann and Hinrichsen, 2000a, 2000b; Winsor et al., 2001; Feistel et al., 2003; Meier et al., 2006; Matthäus et al., 2008). Regional variations in atmospheric pressure systems are important seasonally, while the North Atlantic Oscillation (NAO) appears to have a key role on a sub-decadal scale (Hurrell, 1995; Hanninen et al., 2000; Lehmann and Hinrichsen, 2000b; Feistel et al., 2006; Matthäus et al., 2008).

1.5 Palaeosalinity records from the Baltic Sea, Kattegat and coastal waters

The classic salinity model for the Kattegat and Baltic Sea is summarised in Figure 1.2. This is characterised by the establishment of fully marine conditions by ~6,550-6,050 BC (Björck, 2008), increasing salinity up until ~4,250 BC (Berglund *et al.*, 2005) where salinity reached maximum Holocene levels (Punning *et al.*, 1988), followed by a gradual decrease since the late Atlantic down to the present day level. This model mirrors sea level change, as studies have generally dealt with salinity indirectly as a response to changes in sea level during the *Littorina* Stage (Mikkelsen, 1949; Miller and Hedin, 1988; Petersen, 1993; Andrén *et al.*, 2000; Berglund *et al.*, 2005). However, as already indicated above, the complexity of salinity dynamics causes variability on very short timescales (e.g. both seasonal and decadal variation associated with atmospheric systems such as the NAO, (Hurrell, 1995; Hanninen *et al.*, 2000; Zorita and Laine, 2000) for which very little palaeoenvironmental data is available.

Studies that have more directly considered salinity have largely been qualitative (Petersen, 1993; Huckriede *et al.*, 1996; Sohlenius *et al.*, 1996; Westman and Sohlenius, 1999), with low temporal resolution, which often does not pick up short-term variation. Where marine shells or sediments have been used for dating, these studies also suffer from chronological uncertainty due to the marine reservoir effect (Heier-Nielsen *et al.*, 1995; Hedenström and Possnert, 2001; Olsen *et al.*, 2009). Only a handful of studies have attempted to quantify salinity change (Punning *et al.*, 1988; Winn *et al.*, 1988; Jiang *et al.*, 1998; Winn *et al.*, 1998; Emeis *et al.*, 2003; Ryves *et al.*, 2004; Widerlund and Andersson, 2006;Burman and Schmitz, 2005 A.L. Clarke *et al.* unpublished data). Some of these are discussed in more detail below, though it is stressed that the aim here is not to review all palaeosalinity reconstructions from the Baltic Sea and Kattegat region, but rather to highlight current knowledge, progress and limitations.

As quantitative reconstruction from biological assemblages require the creation of large, modern-day training sets (Birks, 1998), isotopic signals (including oxygen, carbon and strontium) from shells and tests of marine organisms or from bulk sediment have been more commonly used (Punning et al., 1988; Winn et al., 1988; Winn et al., 1998). The pioneering Baltic Sea isotope work of Winn et al., (1988) on the foraminifera Ammonia beccarii, and Punning et al., 1988 on Cerastoderma glaucum and Lymnaea baltica, produced the first estimates of salinity change for the Holocene, particularly over the Littorina transgression, and shed new light on the establishment of marine conditions in the Baltic Sea. However, probably the most detailed of these isotope-based salinity studies (accompanied by diatom analysis and sedimentological expression of anoxia) comes from the central Baltic Sea using the ¹³C isotopic ratios from sedimentary bulk organic carbon (Emeis *et al.*, 2003). This study contradicts previous salinity estimates (e.g. Punning et al., 1988; Winn et al., 1988; Widerlund and Andersson, 2006) by suggesting that salinity in the Baltic Sea never exceeded present day values even during highest Littorina sea levels (~6,550-4,250 BC; Berglund et al., 2005). According to Emeis et al., (2003), the period between 6,050 BC and 3,050 BC was subjected to pronounced changes in salinity before starting a downward trend at ~3,150 BC. For the late Holocene, Emeis et al., (2003) suggest that the environmental conditions within the Baltic Sea were closely linked to the evolution of climate, with variations in salinity being closely connected to patterns of reconstructed summer temperatures and winter precipitation in the southern Scandinavia. For example, the expansion of Scandinavian glaciers associated with the Neoglacial cooling (since ~2,050 BC; Nesje and Dahl, 1994) represent wetter winter

conditions in the Baltic (1,750-1,150 BC) and therefore greater freshwater input (from rivers and glacial meltwater during spring-summer ablation period) and lower salinities. A renewed salinity increase also coincides with the beginning of the Roman Warm Period temperature increase, and over the Little Ice Age, lower winter temperatures produce parallel step-like changes in salinity (Emeis *et al.*, 2003).

Despite the use of these novel geochemical techniques (e.g. isotopic analysis and diagenetic mineralogy; Huckriede et al., 1996), much of our knowledge concerning palaeosalinity of the Baltic Sea and Kattegat is based on qualitative biological assemblage data, most notably molluscs and diatoms. Molluscs have been the subject of a particularly long history of research in the Baltic Sea (e.g. Lindström, 1886; Munthe, 1894, 1910 with some of the original sea stages being named after the molluscs found present in the characteristic sediments (e.g. the Littorina Sea stage after finds of common periwinkle Littorina littorea; Lindström, 1886). Detailed reviews of the molluscan histories of the Baltic Sea, Kattegat and Danish waters can be found in (Petersen, 1981, 1986, 1993; Petersen, 2004; Petersen et al., 2005). Though here we do stress the importance of these studies, particularly to our early understanding of relative sea-level and salinity changes in the Baltic region, to date molluscan studies have been seriously limited in terms of temporal resolution and the quantification of fossil assemblages in terms of salinity values. As large amounts of sediment are often needed to obtain enough molluscs to produce reliable inferences, it is difficult to identify the more subtle shifts in salinity that might operate on decadal-centennial timescales (e.g. smaller-scale changes in freshwater inputs and atmospheric forcing, including patterns in the NAO). Furthermore, though palaeosalinity estimates have often been inferred from molluscan assemblage data (e.g. Munthe, 1894; Petersen, 1993; Kristensen et al., 1995; Christensen et al., 2004; Rasmussen et al., 2007), salinity values are generally given as a range based upon minimum and maximum salinity tolerances for the species present (e.g. Sorgenfrei, 1958; Funder et al., 2002; Petersen, 2004).

Diatom analysis has the potential to obtain palaeoenvironmental data at far higher resolution than molluscs as they require very small samples (~0.1 g wet weight; Battarbee, 1986; Battarbee *et al.*, 2001; Jones, 2007). Diatom analysis is a widely used tool in palaeosalinity reconstructions (Juggins, 1992; Jiang *et al.*, 1998; Reed, 1998; Parsons *et al.*, 1999; Ryves *et al.*, 2002; Ryves *et al.*, 2004; Wachnicka *et al.*, 2010), due to their sensitivity to changes in salinity, ubiquitous nature, global distribution, and a growing taxonomic and ecological

literature in coastal systems (e.g. Snoeijs *et al.*, 1993-1998; Witkowski, 1994; Witkowski *et al.*, 2000). While poor preservation is an issue complicating diatom analysis in marine and coastal sedimentary environments, many coastal and pelagic sites with good preservation can be found in the Baltic and Kattegat area (especially in deep, anoxic basins, and during periods of lower salinity) (Thulin *et al.*, 1992; Westman and Sohlenius, 1999; Andrén *et al.*, 2000; Ryves *et al.*, 2004; Berglund *et al.*, 2005; Ellegaard *et al.*, 2006). As a result there is a rich history of diatom research from the Baltic Sea tracking salinity change since the retreat of the Weichselian ice sheet and subsequent formation of the Baltic Ice Lake and throughout the Holocene (e.g. Iversen, 1937; Mikkelsen, 1949; Andrén *et al.*, 2000; Berglund *et al.*, 2005). The majority of these studies have remained qualitative, with diatom assemblages being classified with respect to their salinity tolerances (e.g. Figure 1.2) based on halobian systems such as Kolbe, 1927; Hustedt, 1957; van der Werff and Huls, 1958-1974 and the Baltic Sea intercalibration guides (Snoeijs *et al.*, 1993-1998). Again diatom analysis has largely been used either together with sea level reconstructions (Mikkelsen, 1949; Berglund *et al.*, 2005) or to examine salinity in relation to sea level change (Andrén *et al.*, 2000).

The highest abundance of marine diatoms in the Baltic Sea and Kattegat again occurs during the late Atlantic prior to ~4,000 BC (Figure 1.2) and is believed to represent maximum Holocene salinity. Some diatom-studies have even tracked multiple transgressions throughout the early Littorina stage (7,850-3,050 BC) (e.g. Mikkelsen, 1949; Berglund et al., 2005). However, only two studies to date, have attempted to quantify salinity change in the Baltic Sea, Kattegat or inner Danish coastal waters using a diatom-based transfer function. Jiang, (1996) developed a modern environment-diatom training set from 39 surface samples collected from the Kattegat and Skagerrak and found that the diatom distribution was heavily associated with summer surface salinity (explaining ~30 % of the variation in the diatom dataset). A subset of 19 sites was applied to a core (Skagen-3) collected from northern Jutland (Figure 1.3), which demonstrated that the salinity of the Kattegat and Skagerrak is primarily driven by the climate of the circum Baltic area but complicated by variations in local current patterns and hydrodynamics of the Baltic Sea. Whilst this study has provided important information concerning palaeocurrents in the Skagerrak-Kattegat region (e.g. Table 1.1), its geographic location makes it unrepresentative of salinity conditions and changes in the Baltic Sea, Kattegat and inner Danish coastal waters. Furthermore the training set is based on deep sea sites from the Skagerrak and Kattegat and therefore cannot be used for sedimentary reconstructions from lower salinity waters such as the Baltic Sea or in the coastal

waters of Denmark due to low species representation in the model (particularly the absence of wide range of benthic diatoms common in fjord and estuarine environments).

The other quantitative diatom-salinity reconstruction is from the Limfjord, where Ryves et al., 2004 applied a diatom-based salinity transfer function to three fossil diatom sequences from Vejlerne (Han Vejle, Glombàk and Lund Fjord; Figure 1.3) covering the last ~150 years. The diatom-salinity model was produced from 27 brackish lakes and 9 fjords across Jutland spanning a salinity range of 0.2-31.1 ‰. The fjord samples were selected from the larger MOLTEN data set (Clarke et al., 2003; Clarke et al., 2006; Andrén et al., 2007a), added in order to extend the salinity gradient into fully marine conditions. Salinity was most important environmental variable uniquely explaining 6% the variation in the lake-fjord diatom dataset, with depth and total phosphorus accounting for 3.9% and 3.5% of the total variation respectively. The final diatom-salinity model was a two component weighted averaging-partial least squares (WAPLS) model with a relatively high predictive power (r_{iack}^2) = 0.887, RMSEP $= 0.246 \log \text{ salinity } \%$, under internal validation). In all cores, the species coverage present in the model dataset was very good (ranging between 83-100% for all samples) and on the whole salinity predictions appear to be reasonable and potentially accurate (accompanied by, and generally in agreement with qualitative multiproxy inferences from cladoceran and foraminiferal data; Amsinck et al., 2003), highlighting the potential for using diatom-salinity models in Danish coastal environments. However, the model did appear to struggle at the high end of the salinity gradient (i.e. towards fully marine conditions) where low abundances and poor preservation of diatoms become more of a problem (e.g. $r^2 = 0.33$ -0.74 (positive) relationship between salinity and diatom dissolution as measured by the diatom **F** index). The inclusion of both fjord and lake sites in the training set might also have weakened the model, dismissing potentially important natural differences between these two types of site which might also drive diatom distributions (e.g. stratification, natural seasonal variations in salinity, ice cover, silica cycling biological structure, presence of mussels in fjords but not in lakes).

The long-term palaeosalinity history of the Limfjord is also poorly understood due to the lack of records extending back further than the last ~2,500 years. The few longer-term studies that do exist suffer from poor chronological control (i.e. dating problems associated with marine reservoir effect and/or sediment reworking) and low temporal resolution (Petersen, 1981; Andersen, 1992; Heier-Nielsen, 1992). For example, Petersen, (1981) has estimated that the
marine transgression of the Limfjord occurred between ~7,000-6,000 BC (from 14C dating of a peat layer and several marine molluscs from overlying layers at Vust, northern Limfjord), though the chronology of this study has been seriously questioned (e.g. Christensen, 2001b) and new, preliminary evidence also exists suggesting that the Limfjord was inundated earlier than ~7,000 BC (P. Rasmussen et al. unpublished data). Burman and Schmitz, 2005 attempted to quantitatively estimate salinity and temperature during the Mesolithic period using δ^{18} O and δ^{13} C analysis, performed on two Littorina littorea (common periwinkle) present in the Ertebølle locus classicus shell midden (Andersen and Johansen, 1986). Using end member estimates from modern L. littorea shells sampled from the Limfjord, Burman and Schmitz, 2005 suggest that salinity (31 ‰) and summer surface water temperature (22°C) in the central Limfjord region were higher than today in the late Mesolithic (5,100-4,100 BC), suggesting greater water exchange/connection with the North Sea/Skagerrak than at present. This is in good accord with estimates that the Limfjord was more exposed during the mid-Holocene (e.g. sea level 3 m higher than present day by ~5,000 BC; Petersen, 1981) with an opening into the Skagerrak existing in the north (e.g. Hellesen and Tuxen, 1988). Aside from these spot estimates, salinity change within the Limfjord remains uncertain and largely un-investigated up until the late pre-Roman period (~500 BC).

During the late Holocene (~2,500 years), several palaeoenvironmental studies have demonstrated that the Limfjord has been subject to large shifts in salinity associated with variable connection with the North Sea and Skagerrak (Kristensen *et al.*, 1995; Christensen *et al.*, 2004; Ryves *et al.*, 2004). For example, a shift to brackish conditions is believed to have occurred some time during the late pre-Roman period as the Limfjord became totally closed off from the North Sea (Kristensen *et al.*, 1995; Christensen *et al.*, 2004), probably due to lowering of sea level (e.g. Behre, 2007). A return to high salinity conditions occurred during the Germanic or Viking period, before the Limfjord was closed off again sometime between ~AD 1100-1200. With the exception of several marine breakthroughs during severe storms possibly associated with the Little Ice Age (Kristensen *et al.*, 1995), the Limfjord remained closed up until ~AD 1825, when it was breached for the final time through Agger Tange and has remained open since.

The palaeoenvironmental records are also supplemented by documentary evidence (e.g. Petersen, 1976; Kristensen *et al.*, 1995; Holm and Bager, 2002) and archaeofaunal analyses (Enghoff, 1999). For example, the much debated Snorres Saga suggests that the Danish King

Knud the Great returned through the Agger sound from an expedition to England in AD 1027, indicating that the west was open in the late Viking period (Petersen, 1976). An opening in the North is also likely to have existed in AD 1061, through which Norwegian King Harald Haardraade is thought to have escaped from a superior Danish fleet led by King Sven Estridsson (Petersen, 1976). Documentary fishing records from the Limfjord have also enabled detailed information concerning the herring industry in the Limfjord throughout historical times (post ~AD 1200; Rasmussen, 1968; Holm and Bager, 2002). The herring fishery prospered in the Limfjord following the switch to more brackish conditions after ~AD 1100-1200. On several occasions (between AD 1200-1825), the herring fishery was suppressed due to marine breaches following storm events (see above; Kristensen et al., 1995; Holm and Bager, 2002), prior to its final decline in AD 1825 (Holm and Bager, 2002). Archaeofaunal records from Iron Age sites situated within the Limfjord have extended the fishing records back to the fifth century, when high salinity conditions supported large plaice and eel fisheries (Enghoff, 1999). Whilst the late Holocene history of the Limfjord (i.e. post ~AD 1200) is relatively well constrained (i.e. due to documentary evidence), unfortunately the longer term studies extending back to the pre-Roman period (i.e. Kristensen et al., 1995; Ellegaard, 2000; Christensen et al., 2004) are again subject to dating uncertainties, associated with the marine reservoir effect. Therefore, improved chronological control is needed in order to fully understanding the timings of earlier openings and closings of the Limfjord and to assess how climate and sea level change might have driven these changes (e.g. Christiansen et al., 1990; Tanner, 1993; Behre, 2007).

1.6 Cultural change in Denmark

Reconstructing long-term environmental change in the Baltic Sea is important not only in terms of ecological and hydrographic variability, but also in relation to humans inhabiting the coastal zone. The historical and archaeological record along the Danish coast is especially rich, and attests the importance of marine resources to societies living in coastal regions of Denmark throughout the Holocene (Andersen, 1995a; Enghoff, 1995; Holm and Bager, 2002; Andersen, 2007; Enghoff *et al.*, 2007). Prehistoric cultures and societies (e.g. Kongemose, Ertebølle) were often heavily dependent upon the dynamic Baltic Sea, yet our knowledge of this complex interplay remains fragmentary. For example, changing sea levels throughout the *Littorina* Sea stage (e.g. Iversen, 1937; Berglund *et al.*, 2005 and see above) is likely to have forced changes in important environmental parameters such as substrate, sediment

accumulation rate, salinity, temperature, productivity and nutrient status at both local (e.g. within individual fjords) and regional scales (e.g. the entire Baltic basin). This is likely to have disrupted ecosystem structure and functioning within coastal waters and potentially affected the presence and availability of important marine resources (e.g. Rowley-Conwy, 1984). Unfortunately, changes in these environmental parameters in response to sea level (and climate change or human impact) remain poorly understood to date.

The estuaries and the inner coastal waters of Denmark are very attractive settlements for a variety of reasons. Firstly, they are highly productive systems, supporting both high primary (e.g. macrophyte and algal communities) and secondary productivity (e.g. zooplankton, bacteria) which is extremely important for supporting good commercial fishing industries. Furthermore resources within estuaries are often easily accessible and tend to vary seasonally rather than spatially. For example, eel grass, oyster and mussel beds commonly reside in the inner fjords and estuaries as they offer penetration of solar energy required for them to grow as well protection from damaging ultraviolet light. Inflowing rivers and streams also offer settlers a source of fresh water and alternative fresh water resources for a diverse diet. Estuaries can also offer protection from storms, be easily defensible, offer good trade routes both along rivers and fjords and provide easy access to the open seas. The attractiveness of such sites was clearly identified and exploited by communities living in Denmark throughout the Holocene (Andersen, 1995a; Enghoff, 1995; Enghoff, 1999; Andersen, 2007; Enghoff et al., 2007). Archaeological evidence is diverse, ranging from single finds and small shell deposits through to larger sites and settlements such as burial mounds, graves and cemetries (Albrethsen and Brinch Petersen, 1977; E. Brinch Petersen, unpub.), submerged sites (Andersen, 1980, 1987) and large shell middens (Andersen, 2000, 2007; Andersen, 2008a). In the more recent past (last ~1,000 years) this is further supported by historical and documentary evidence (Holm and Bager, 2002; Poulsen et al., 2007).



Figure 1.3 - Location map for Danish archaeological sites referenced in the text.

This study particularly focuses on reconstructing natural environmental change in coastal Denmark over the mid-Holocene with three important phases of shell midden ('køkkenmøddinger' in Danish, Figure 1.4) accumulation evident in the archaeological records:

- 1. Ertebølle-Funnel Beaker culture (~5,400-2,800 BC) with focus on the Mesolithic-Neolithic transition (~3,900 BC).
- 2. The period of the Pitted-Ware Culture and Single Grave Culture (~2,800-2,400 BC).
- 3. The Iron Age (500 BC-AD 1050).

Denmark is one of the classic areas for studies of prehistoric shell middens, a tradition that extends back into the early 19th century (e.g. Madsen et al., 1900; Andersen, 2000; Fischer and Kristensen, 2002 and references therein). Danish shell middens are defined as a deposit that contains over 50 % shells or shell fragments and forms an undisturbed horizon at least 10 m^2 (Andersen, 2000, 2007). Shell bearing sites that fail to meet both of these criteria are simply referred to as shell deposits or shell heaps. Shell midden sites tend to be found within close 'economic' distance to natural shell banks and with good fishing access (Andersen, 2000). The shell middens are always found together with coastal settlement sites and were likely used for a wide range of daily activities (including processing, consuming and disposal of shell fish and other food). Over the last ~150 years a large number of shell middens (e.g. Andersen, 2000, 2007) have been discovered and detailed (large-scale) excavations have taken place in order to reconstruct changes in the practices and behaviour of past cultures and societies living in coastal areas of Denmark over the Holocene. The wealth of information now available from these shell middens (and other coastal sites) provides an excellent and rare opportunity to assess (using modern palaeoenvironmental techniques) how environmental change might have influenced (or even forced changes in) former coastal dwelling cultures and societies, which relied heavily upon marine resources.

1.6.1 The Late Mesolithic and the accumulation of Stone Age shell middens

The oldest known coastal sites in Denmark belong to the Maglemose culture (9,000-6,800 BC), though these are few in number and are largely confined to the Øresund (Andersen, 1995a; Pedersen *et al.*, 1997). Evidence for complex flint artefacts used to make arrow heads and various wooden fishing tools such as nets from pine bark and wooden spears indicate that the Maglemose culture heavily relied on fishing (e.g. pike, perch) and hunting (e.g. aurochs, elk, wild boar) to support their communities (Jensen, 2001; Schmölcke *et al.*, 2006). This culture mainly inhabited inland sites often positioning themselves on wind-sheltered plateaus in close proximity to rivers and lakes (Grøn, 1995), where they exploited fish from these freshwater sources. No coastal sites extending back into the earliest Mesolithic and late Palaeolithic have been found in Denmark to date, an absence most likely attributed to geographical or geological factors (i.e. below present day sea level having been drowned by the *Littorina* transgression; Christensen, 1995).

Far more numerous in Denmark are sites from the Kongemose culture (~6,800-5,400 BC, named after the type site 'Kongemose'), which commonly inhabited coastal areas and showed far greater affinity to marine resources than the preceding Maglemose culture (e.g. Andersen, 1970a; Brinch Petersen, 1973; Andersen, 1985; Price, 1985; Larsson, 1990; Brinch Petersen, 1993; Enghoff *et al.*, 2007). Again, the Kongemose were skilled hunters and fishers that produced complex hunting tools such as arrowheads, scrapers, drills, awls and toothed blades from flint (e.g. Larsson, 1990; Vang Petersen, 1999). Their economy was based around the hunting of elk, aurochs, wild boar, red deer, roe deer, small fur animals as well as exploiting marine resources, with the earliest shell middens being dated to the late Kongemose period (Andersen, 1970a).

The Kongemose was succeeded by another hunter-fisher-gatherer culture named the Ertebølle, after excavation of the type-site 'Ertebølle' situated near Ertebølle, in the Limfjord region of northern Jutland (Madsen *et al.*, 1900, Figure 1.3). The Ertebølle people commonly lived along (or near) the coastline of fjords and estuaries (e.g. large number of Ertebølle shell middens; Andersen, 2000, 2007), relied heavily on a wide range marine resources (including fish, molluscs, marine birds and mammals), exhibiting a predominately marine diet (e.g. Tauber, 1981; Bratlund, 1991; Andersen, 1995a; Enghoff, 1995; Andersen, 2007; Enghoff *et al.*, 2007; Andersen, 2008a). They often amassed high population densities at coastal localities and again developed various complex tools in which to exploit these resources (Andersen, 1995a; Enghoff *et al.*, 2007). Whilst they appear to have adopted pottery from their close neighbours, the Ertebølle culture did not adopt agricultural practices until much later (i.e. introduction of agriculture in Denmark ~3,900 BC, but in northern Germany ~4,600 BC; Hartz *et al.*, 2002 and this 'delay' has sparked over 150 years of intense debate (Fischer and Kristensen, 2002 and papers therein).



Figure 1.4 - Summary of the geological and cultural periods (including shell midden accumulation phases and shellfish composition data) referred to in the text, along with regional sea level and climate data for the Baltic Sea area. The sea level curve is from Blekinge, southeastern Sweden (Berglund *et al.*, 2005) and the (annual mean) temperature record is from Lake Trehörningen, western Sweden (Antonsson and Seppä, 2007). The Baltic Sea stages (sea level column) are based on the stratigraphy suggested by Andrén *et al.*, 2000; Berglund *et al.*, 2005; Andrén *et al.*, 2007b. The climatic periods (temperature column) follow divisions in Snowball *et al.*, 2004. The percentage shellfish composition data is based on data from Figure 3 in Andersen, 2007. The quoted values for the Iron Age midden at Store Nor, Norsminde Fjord are also from the reference. The date for the Single Grave Culture midden at Kalvø is from Andersen, 1983. Abbreviations: BIL = Baltic ice lake, LG = late-glacial, YD = Younger Dryas. *m a/b psl = metres above/below present sea level.

The number of (stratified) shell middens increases dramatically in the Ertebølle period, becoming common features of the late Mesolithic coastal landscape, only exceeded in size, number and volume during the Iron Age (e.g. Andersen, 2007). Whilst the majority of Stone Age middens belong to the Mesolithic Ertebølle culture (particularly the period ~4,600-4,400 BC; Andersen, 2007), some contain Neolithic layers (belonging to the Funnel Beaker Culture (TBK); e.g. Andersen, 1989) and cover the Mesolithic-Neolithic transition, thereby providing important archives of both cultural development and marine environmental change (Andersen, 1989; Andersen and Rasmussen, 1991). Excavation and analysis of a large number of late Stone Age middens has generated substantial information concerning the diets (i.e. through quantification of mollusc, mammal and fish bone compositions), technology and hunting/fishing practices (i.e. analysis of pottery, tools, artefacts and cooking utensils) of the Ertebølle and Funnel Beaker Cultures.

These analyses have been supplemented by state of the art techniques such as δ^{13} C and δ^{15} N isotopic analysis of teeth and bones from humans and dogs (Tauber, 1981; Milner *et al.*, 2004; Fischer *et al.*, 2007) and the analysis of incremental growth lines from the hinges of molluscan shells present within the middens (Milner, 2001a, 2001b, 2002). The former (i.e. isotopes) reveal information concerning the dietary composition (i.e. different group exhibit different isotopic signatures) and the latter enable establishment of shellfish collection patterns (e.g. season of collection) and resource pressure (i.e. the age and size of the molluscs at the time of collection). It is beyond the scope of the present study to provide a detailed review of the archaeological composition of Stone Age shell middens (e.g. differences/similarities between all sites, detailed assessment of pottery, tools and artefacts of these cultures), and therefore in the following section just some of the key characteristics, relevant to this study are highlighted. More detailed reviews can be found in the following references: Price, 1985; Larsson, 1990; Price, 1991; Stafford, 1999; Andersen, 2000; Fischer and Kristensen, 2002; Andersen, 2007; Andersen, 2008a.

One of the most striking characteristics of the late Danish Stone Age middens is the high abundances of large oysters (*Ostrea edulis*) present in the Mesolithic layers (Andersen, 1995a, 2007; Andersen, 2008a; Bailey and Milner, 2008; Bailey and Milner, submitted), sometimes accounting for up to 60-80 % of the total shell composition (e.g. Bjørnsholm Bay and Ertebølle *locus classicus*; Andersen and Rasmussen, 1991). Where midden accumulation persists into the Neolithic (e.g. Bjørnsholm Bay, Norsminde Fjord), there is a subsequent decline in oysters, with other less salt-demanding species such as *Cerastoderma edule*

(cockles) and *Mytilus edulis* (mussels) becoming far more dominant (Andersen, 1989; Andersen and Rasmussen, 1991; Andersen, 2008a; Figure 1.4). One of the most classical examples of this (i.e. oyster decline) is the stratified Norsminde shell midden (occupied between ~4,500-3,200 BC; Figure 2.4). Oysters account for ~40-50 % of the total shell accumulation in the Mesolithic layers, but drop to less than 10 % in the late Mesolithic (after ~4,300 BC) and remain low thereafter (Andersen, 1989; Andersen, 2007; Bailey and Milner, 2008; Bailey and Milner, submitted). Several theories (including environmental, demographic and cultural hypotheses) have been proposed to explain this oyster decline, but to date none offer a completely satisfactory explanation or currently lack the sufficient supporting evidence (e.g. environmental 'salinity' hypothesis) to prove/discard the theory. In this study particular attention is placed on critically testing environmental-hypotheses (i.e. salinity, productivity and sedimentary regime; discussed in more detail below) by reconstructing changes in key environmental parameters over the Mesolithic-Neolithic transition (discussed below in more detail).

Also apparent in shell middens and other coastal sites are a large number of fish bones, particularly in the Mesolithic layers, indicating the importance of fishing to the Ertebølle culture (Enghoff, 1989, 1991, 1995; Enghoff *et al.*, 2007). For example, Enghoff, 1989 analysed almost 9,000 fish bones from the Norsminde shell midden, yielding over 25 different species, almost all of which were brackish or marine taxa (flounder most abundant, 57%). Flounder and gadids (e.g. cod and saith) are generally most abundant in settlements from the Vedbæk area, whilst in one of the northern most Ertebølle settlements (Yderhede) flatfish and spurdog bones dominate the assemblage (Enghoff, 1995). Other sites, however, have demonstrated that fishing was not strictly a marine activity, with many fish species present from freshwater and fluvial sources. For example, 71% of identified fishbones from Ertebølle *locus classicus* midden belonged to freshwater species (mostly roach) with 12% saltwater taxa (gadids most abundant) and 17% migratory fish (eel). At Bjørnsholm Bay, eel bones were most abundant (56%), with saltwater taxa accounting for just 22% of fish bones.

The variability in fish bone composition between sites is somewhat contrasting to the relatively consistent shellfish composition of Ertebølle sites. Whilst fishing was clearly an important activity, particularly in the summer months (Enghoff, 1991, 1995; Enghoff *et al.*, 2007), it appears that the catch (i.e. species exploited) was more strongly governed by localised factors, such as the topography, proximity to freshwater inlets or sites (e.g. rivers or lakes), water depth, protection, and availability of suitable breeding grounds (e.g. eel-grass

beds, macrophytes). For example, it has been suggested that eel might have been readily caught at Bjørnsholm Bay, at the mouths of small streams running into the former fjord inlet, whilst at Norsminde, the shallow waters of the former Kysing Fjord might have provided ideal breeding grounds for flounder (Enghoff, 1989, 1991, 1995).

Whilst fish bones are often abundant in Mesolithic coastal shell middens and deposits, they appear to be very scarce in Neolithic layers (e.g. Bjørnsholm Bay, Mesolithic: Neolithic fish bone ratio, 11,490: 252 bones (Enghoff, 1991, 1995), Norsminde Fjord, all bar one bone were found in the Mesolithic layers; Enghoff, 1989). The reason for this remains uncertain, particularly as species composition and relative proportions (e.g. Bjørnsholm Bay), do not appear to change markedly from the Mesolithic layers (Enghoff, 1986; Enghoff, 1994; Enghoff, 1995; Andersen, 2007), suggesting similar fishing patterns (i.e. methods and targeted species). It is plausible that the fish catch was substantially reduced due to introduction of agriculture and reduced dependence on marine resources (e.g. Tauber, 1981; Milner et al., 2004), though this does not satisfactorily explain the almost entire absence of fish bones from sites such as the Norsminde Fjord shell midden. Site abandonment is ruled out due the clear continuation of numerous middens (and other sites bearing coastal resources) in the Neolithic (including some sites, dated entirely to the Neolithic period). It remains possible that fish bones were deposited elsewhere, but this is hard to believe considering the fact that both molluscs and bones from other animals were deposited on these sites. Perhaps the most likely explanation is accelerated decomposition of fish bones in the Neolithic layers, due to a reduction in shell midden accumulation rates (as marine resources became less important) or changes in the physical environment (e.g. increased aerial exposure, weathering and erosion).

Shell middens (and other coastal sites) remain present for the first ~400-500 years of the Neolithic, but gradually become smaller in size and more sporadic (particularly between ~3,600-3,300 BC; Andersen, 2008a) before practically stopping in the middle Neolithic (Andersen, 2007). This early Neolithic period (i.e. ~3,900-3,600 BC) therefore probably marks a transitional phase from a marine-dependent community in the Ertebølle period, to a predominately terrestrial-based resource economy in the Neolithic period. This is supported by the continued collection of marine resources (most notably cockle and mussels, but also smaller amount of oysters in the early Neolithic) and the presence of some human skeletal remains (dated to the early Neolithic) that still exhibit isotopic signatures indicative of a predominately (or at least substantial) marine diet (Bailey and Milner, 2008). In some early

TBK layers, oysters are present at relatively high abundances (e.g. Krabbesholm and Havnø), though these sites appear to form the exception rather than the normal (Andersen, 2005, 2008b; Nielsen, 2008). After ~3,600 BC the pollen records also generally show strong signs of large scale agricultural impacts (e.g. reduction in arboreal pollen, continuous curves for arable and pastoral grassland plants such as *Plantago lanceolata* and *Rumex acetellosa* and regular presence of cereal pollen; Iversen, 1941; Aaby, 1986; Andersen, 1995b; Rasmussen, 2005).

1.6.2 Salinity decline at the Mesolithic-Neolithic transition; a long outstanding issue

A popular hypothesis for the delay of agriculture is that the marine environment was productive enough during the late Mesolithic not to necessitate agricultural innovation (e.g. Rowley-Conwy, 1984). Higher sea levels in the Ertebølle period (~5,600-4,200 BC; Iversen, 1937; Troels-Smith, 1942; Mikkelsen, 1949; Petersen, 1981; Christensen, 1995; Berglund et al., 2005) is believed to have created a wider connection between the Kattegat and Skagerrak, causing greater water exchange with the North Sea. This resulted in increased inputs of high salinity, oxygenated and more nutrient rich North Sea water, important for maintaining a healthy and productive ecosystem (e.g. by reducing the likelihood of stagnation and hypoxia of deeper waters, particularly in the deeper basins). It also enabled high-salinity demanding organisms (and popular food stuffs) such as oysters to extend into southern areas of Kattegat and penetrate the inner (more human accessible) parts of the fjord (Nordmann, 1903). Some studies have also alluded to a larger tidal amplitude existing in the Mesolithic (Nielsen, 1938; Petersen, 1993; Troels-Smith, 1995; Petersen et al., 2005). In particular, Nielsen, (1938) proposed that larger tides caused greater mixing of water masses and increased the 'delivery' of high salinity water into the innermost part of estuaries and fjords, though this hypothesis is more speculative than fact, lacking any good physical supporting evidence. Furthermore, recent modelling of tidal and tide dependent changes for NW European shelf seas, however, suggest that over the last ~8,000 years (i.e. the period in which sea level has been closest to its present level), tidal changes have been generally small (Uehara et al., 2006). The Ertebølle period also coincides with the mid-Holocene climatic optimum (~5,550-3.550 BC; Snowball et al., 2004; Antonsson and Seppä, 2007; Seppä et al., 2009), meaning higher temperatures might also have been important for increasing the productivity of the marine environment at this time (e.g. longer growing/breeding seasons, reduced freezing and thawing events in winter). Abundant shell fish populations in close proximity to good fishing sources (e.g. marine and freshwater resources available within in the same fjord) offered a reliable, diverse

and rich marine-based diet for the Ertebølle culture. In accord with the palaeodietary evidence (see below, Figure 1.5), it is therefore feasible that the marine biotope was more attractive than agricultural innovation at this time.

There are two main lines of palaeodietary evidence that indicate marine resources were predominant in the Ertebølle diet: 1, δ^{13} C and δ^{15} N isotopic analyses of human skeletons (most notably bones and teeth). 2. Abundance volume and composition (of marine shellfish, fish and mammals see above) of shell middens and other coastal sites. The principle underlying isotope based palaeodietary techniques such as δ^{13} C and δ^{15} N isotopic analysis is that skeletal bones are made from what the animal eats (or drinks) (Hedges et al., 2005). Therefore carbon isotope ratios (from bones and teeth) can reveal the predominant source of food eaten (i.e. terrestrial or marine organism exhibit different isotopic signatures), whilst nitrogen isotope can provide further information concerning the trophic level of organisms exploited (generally quoted as 3-5 ‰ enrichment at each trophic level; Schoeninger et al., 1983). Mesolithic skeletons from Denmark exhibit δ^{13} C values in the expected range for a predominately marine based diet (e.g. generally 10-15 ‰; Tauber, 1981; Milner et al., 2004; Fischer *et al.*, 2007), whilst the δ^{15} N values (average ~14 ‰; Schoeninger *et al.*, 1983; Richards and Hedges, 1999; Fischer *et al.*, 2007) and δ^{13} C: δ^{15} N ratios demonstrate the importance of resources such as shell fish to the Ertebølle culture. In the Neolithic period, there is a shift towards a predominately terrestrial-based (Figure 1.5), with substantial consumption of agricultural resources (Tauber, 1981; Schoeninger et al., 1983; Fischer et al., 2007). Whilst substantial debate has taken place about the real meaning of these isotopic results (e.g. Hedges, 2004; Milner et al., 2004, 2006; Richards and Schulting, 2006b; Richards and Schulting, 2006a), it seems there is a general consensus that over time the proportion of terrestrial protein consumed in the human diet has increased (Bailey and Milner, 2008).



Figure $1.5 - \delta^{15}$ N and δ^{13} C values for bone collagen from human and dogs for the Mesolithic and Neolithic period. Open symbols refer to individuals with ambiguous cultural belonging, being found at the Mesolithic-Neolithic transition. One-sided error bars represent marine reservoir correction and black arrows indicate uncertain reservoir effect. The dotted line for carbon isotopes (i.e. at 20 ‰) marks the limit above which there is an indication of a non-negligible marine diet, though individuals below this line may have consumed small quantities of marine food. The dotted line for nitrogen isotopes (i.e. at 9.5 ‰) marks the limit above which individuals must have consumed aquatic food regularly, through the same may apply to individuals below the line if substantial parts of this dietary component derived from low trophic level organisms such as shellfish (source Fischer *et al.*, 2007).

Shell analyses (i.e. for size, age and seasonality) from the Norsminde shell midden indicate that molluscan populations (particularly *Ostrea edulis*) were subject to less pressure in the Ertebølle period, when they formed a greater part of the cultural diet (Bailey and Milner, 2008; Bailey and Milner, submitted). The age and size of oysters is stable in the Mesolithic layers indicating a consistent collection pattern (in late winter/early spring), taken from a large healthy population, with no signs that exploitation was ever damaging to the availability of this resource. In the Neolithic, however, the collection season lengthens (i.e. collection season extends into the summer) and the age and size at the time of collection fluctuates suggesting variable exploitation, at time even exerting pressure upon the oyster populations.

Bailey and Milner, 2008 suggest several potential reasons for this including exaggeration of the importance of marine resources (i.e. potential inaccuracy associated with the isotope results, see above), reduced availability of marine resources (due to environmental change), increased population in the Neolithic, changes in the yearly resource schedule (associated with the introduction of agricultural resources) and less care taken by Neolithic cultures to preserve marine resources, due to their wider resource base.

It has even been suggested that the oyster decline might have been an important factor behind the neolithisation (i.e. introduction of agriculture) of Denmark (Rowley-Conwy, 1984). Peter Rowley-Conwy suggests, that for the majority of the year, oysters were probably exploited as a supplement or relish (probably forming less than 10% of the total Ertebølle diet at most; Bailey, 1975; Clark, 1975), being far less attractive than the majority of other food stuffs due to their low productivity per hour of work (based on tentative calculations for calories per hour work; Rowley-Conwy, 1984). The high abundance of oysters in the Ertebølle layers of the shell middens (Andersen, 2000, 2007; Andersen, 2008a), however, indicates that at some point during the year exploitation must have been intensive (i.e. late spring/early summer; later confirmed by Bailey and Milner, 2008; Bailey and Milner, submitted). As ovsters are available all year-round (and were likely more accessible during the Ertebølle), have relatively low seasonal variation in fat content compared with other resources and are a good source of carbohydrate, their appeal is likely to have increased with the diminishment of other resources (Rowley-Conwy, 1984). It is never suggested that during the late winter/early spring, that the Ertebølle culture lived entirely on oysters, but rather that due to the reduction in other resources in these months, oysters became a more important part of the diet. When this resource become under pressure and less abundant and/or reliable due to changing environmental conditions (most notably lower salinity), it is hypothesized that humans might have adopted agricultural techniques to fill this gap in the seasonal resource cycle (Rowley-Conwy, 1984).

The decrease in salinity is believed to have been caused by declining sea levels in the late Mesolithic (e.g. Iversen, 1937; Berglund *et al.*, 2005) and a reduction in the tidal amplitude (Rowley-Conwy, 1984). In the majority of the inner Danish coastal waters today, salinity is below the minimum 23-25 ‰ required (Jensen and Spärck, 1934; Yonge, 1960) to support breeding oyster populations, except for in some areas of the Limfjord and in the higher salinity waters of the Kattegat. Nordmann, (1903) suggested that even in the Ertebølle period, when sea levels and temperatures were higher, the extension of oysters into the

southern Kattegat (as far south as northern Zealand) and inner Danish coastal waters meant they were near to their lower salinity tolerance limits and even small magnitude changes in salinity might have serious implications for oyster populations. It is important to note that oysters never completely disappear from the Neolithic layers of Danish shell middens (Andersen, 2007; Andersen, 2008a, but rather appear to become exploited less and subject to greater stress (Bailey and Milner, submitted), perhaps related to reduced spawning years. Cockles are able to continue spawning at lower salinities and temperatures (6 ‰ salinity and 9°C or less temperature) and can survive better in softer muddier substrates (Sorgenfrei, 1958; Petersen, 2004; Bailey and Milner, submitted). However, despite the high abundance of cockles (and mussels) in the Neolithic layers, it is suggested that these taxa were not suitable nutritional substitutes for oysters at this time of year (i.e. late winter/early spring). Cockles and mussels do not reach their peak nutritional levels (i.e. highest meat: shell ratio) until later in the year (i.e. the summer months; Rowley-Conwy, 1984; Bailey and Milner, submitted).

The 'salinity' hypothesis (e.g. Rowley-Conwy, 1984) remains speculative to date, lacking any high quality salinity data from the inner Danish waters over the late Mesolithic/early Neolithic period. Furthermore, whilst sea level is likely to be an important driver of salinity change, in settings such as estuaries and fjords, salinity is also likely to be influenced by a number of other variables (e.g. climate change, terrestrial hydrology, changes in basin morphology, stratification, and marine exchange, all of which might independently be influenced by sea level). In addition to salinity, it has been proposed that changes in temperature and substrate could also explain the decline in oysters in the Danish middens (Bailey and Milner, 2008; Bailey and Milner, submitted). Regional temperature records suggest decreasing temperatures after ~4,000 BC following the mid-Holocene thermal maximum (Snowball et al., 2004; Antonsson and Seppä, 2007; Seppä et al., 2009). As Ostrea edulis only spawns when water temperatures reach ~15°C or above (Yonge, 1960), then decreased temperatures might have suppressed the spawning season, possibly even preventing spawning in some years. Ostrea edulis is also known to struggle with large volumes of fine sediment (Yonge, 1960), preferring harder, sandier substrates, with little vegetation cover (Rasmussen, 1958). Some evidence exists for increased accumulation rates in the Neolithic (e.g. Petersen, 1993), which might have therefore further stressed oyster population particularly in the shallower areas where large volumes of material might have been settling out (both temperature and substrate are discussed in more detail in Chapter 6).

The idea of a rapid Mesolithic-Neolithic transition has also been questioned and a number of sociological arguments have been made for the introduction of agriculture forming part of a long-term social transition, with large regional differences (e.g. Petersson, 1996-97; Stafford, 1999; Johansen, 2006; Larsson, 2007). It is beyond the scope of this thesis to review all (non-environmental) arguments concerning the Mesolithic-Neolithic transition in southern Scandinavia. Therefore, here just some of the key counter arguments are highlighted, focussing on those relating to the interpretation of shell midden evidence.

Petersson, (1996-97) has suggested that current interpretation of the archaeological shell midden evidence might be erroneous, and that even within the shell middens there is evidence for a gradual transition, with regional variability and potential cultural dualism. Petersson draws attention to the mixture of Ertebølle and TBK artefacts in some middens (e.g. one third of the pottery presented in the Norsminde shell midden is in fact Neolithic; Petersson, 1996-97), for which no completely satisfactory explanation has been given. Based on the stratified shell middens (e.g. Norsminde, Bjørnsholm Bay), these mixed cultural layers are generally interpreted as artefacts of geological processes such as re-working and redeposition rather than possibly reflecting cultural dualism (e.g. Petersson, 1996-97 and references therein). Furthermore, the shell middens are dated via ¹⁴C dating on marine shells present (most notably Ostrea edulis and Cerastoderma spp.; Andersen and Johansen, 1986; Andersen, 1989; Andersen and Rasmussen, 1991; Andersen, 2000) and therefore it is argued that the ecological shift has been dated rather than the cultural shift (e.g. Petersson, 1996-97). In sites such as Bjørnsholm Bay, this ecological shift occurs prior to any clear signs of a cultural shift (e.g. pottery) within the midden (Andersen and Rasmussen, 1991) and some 200 years before the transition at Norsminde. Petersson, 1996-97 cites climate change as a possible cause for the ecological shift, but what is dismissed here, is that if environmental change forced the Mesolithic-Neolithic transition, then an ecological shift would be expected to pre-date the cultural shift. The delay at Norsminde is contradictory to the idea of a rapid and regional shift towards agriculture and in better agreement with finds at sites such as Åmosen (Zealand) which indicate a clearer gradual Ertebølle-TBK transformation (Fischer, 1993).

Johansen, 2006 has also critiqued the proposed model for Ertebølle settlement at larger fjords (e.g. Bjørnsholm Bay, Mariager Fjord), suggesting that the Ertebølle culture were not a stable sedentary culture, inhabiting big year round coastal sites, but rather comprised of numerous small groups rotating between sites on a seasonal basis. The 'constellation' settlement model

is based around the findings at larger fjords such as Bjørnsholm Bay and Roskilde Fjord, where one large settlement (containing resources from all seasons) is surrounded by several seasonal smaller extraction camps (Andersen, 1995a; Pedersen et al., 1997). Johansen, 2006 argues that the smaller camps are not seasonal extraction camps, but rather represent the day to day living activities of small groups, being not dissimilar in content to the large 'year round camps'. Johansen also highlights that the evidence for year round occupation of the larger sites is equivocal. Winter resources are poorly represented (i.e. only birds and small fur animals) in the Ertebølle locus classicus and Bjørnsholm Bay middens, with the majority of resources representative of spring to autumn based activities. It is also suggested that the larger middens are palimpsests of countless smaller shell heaps, being returned to on a regular basis due to the large resource base (e.g. large natural fossil beds) present in close proximity to these sites. This is based on evidence within the middens displaying sections of intense exploitation interspaced by periods with little deposition (e.g. Andersen and Johansen, 1986). Whilst there are clearly still many gaps in the archaeological record, high resolution palaeoenvironmental analysis is vitally important for understanding better, the physical environment in which the agricultural transition took place in, and to provide proper environmental context for future debates on this issue.

1.6.3 The Pitted Ware and Single Grave Culture (~2,800-2,400 BC)

During the Pitted Ware and Single Grave cultural period, there is a renewed increase in the number of shell middens and shell banks present along Danish coastlines (Andersen, 2007). There is also a shift from cockle (*Cerastoderma edule*) dominated middens in the earlier Neolithic, back to oyster (*Ostrea edulis*) dominated middens, which might reflect another period of higher salinity (Andersen, 1983, 2007). Whilst the skeletal isotopic analyses from humans suggest that terrestrial resources continue to dominate the diet throughout this period (Tauber, 1981; Bailey and Milner, 2008, Figure 1.5), there is certainly an increase in the importance and exploitation of marine resources. It is possible that a shift in diet is only (or more) apparent in the coastal-dwelling communities, as Pitted Ware cultural sites tend to most commonly be found along the former coastlines, often found near or with Ertebølle remains. This has led to suggestions that sea level was near to Ertebølle levels at this time (Becker, 1951), but sea level (and salinity) change remains relatively poorly understood in this part of the record. Several studies exhibit potential sea level peaks (i.e. transgressions) broadly coinciding with the Pitted Ware Cultural period (e.g. Miller, 1982; Miller and Hedin, 1988; Christensen, 1995), but problems with dating (e.g. marine reservoir) prevent any

confident links being made. The Pitted Ware culture (PWC) is seen as an independent group distributed across southern Sweden and the north/north-east of Denmark (around the Kattegat, on islands and by fjords including the Limfjord; Rasmussen, 1993a). The PWC, is however, synchronous with the Single Grave culture (and possibly the Swedish Battle Axe culture) but is treated separately due to the lack of evidence for continuity between these other cultures (Becker, 1951; Rasmussen, 1993a). Coastal sites belonging to the Single Grave Culture demonstrate that this culture also exploited marine resources and the composition of shell midden and shell sites is often similar in mollusc composition to sites from Pitted Ware Culture. For example a Single Grave culture layer at the top of the Kalvø shell midden, (located on the north-eastern coast of the former island within Norsminde Fjord) contains 68 % oysters, 20 % mussels, 4.5 % cockles and 2.5 % periwinkles (*Littorina* spp.) (Andersen, 1983).

1.6.4 The Bronze Age

Collection of marine shells seems to decrease in the Bronze Age (1,700-500 BC), with no shell middens present, but rather just thin layers or small heaps of shells present at various settlements (e.g. rich Bronze Age burial sites in the Limfjord beach ridges at Hverrehus and Mellemholm; Andersen, 2007 and references therein). Mussels (*Mytilus edulis*) are the most abundant species in Bronze Age deposits, but oysters remain present at most sites but generally in small numbers (e.g. a small heap at Vadgård; Rasmussen, 1993b). Fish, seal and whale bones and fishing tools (e.g. fish hooks) are also found, further suggesting that coastal resources played at least some role in the Bronze Age economy (at least in the coastal regions), though the diet appears to be predominately terrestrial based (Tauber, 1981; Andersen, 2007).

1.6.5 The Iron Age (~500 BC-AD 1050) and the re-appearance of coastal shell middens

In the Iron Age, there is a large increase in the number of shell middens and other settlements containing marine resources (molluscs and fish-bones) or fishing tools present along Danish coastlines (Poulsen, 1978; Andersen, 2007). The majority of these belong to the late pre-Roman/early-Roman period (~300 BC-AD 200; Poulsen, 1978; Andersen, 2007). These shell middens are predominately composed of *Mytilus edulis* (generally most abundant), *Cerastoderma edule* and *Littorina littorea*, but again *Ostrea edulis* shells are regularly present. The larger shell midden sites appear to be confined to the eastern coast of Jutland and the northern coasts of Fyn and Zealand, but with middens extending further south into the

Baltic, to the east coast of Schleswig (Anger, 1973; Harck, 1973; Poulsen, 1978; see Figure 1.3). Some of the Iron Age middens are extremely large, with a cubic content of up to 15,000 m³ and sometimes extend up to 1 km long with just short interruptions (e.g. Norsminde Fjord and Mariager Fjord), thereby far exceeding the size of the largest Stone Age shell middens (Andersen, 2007).

Interestingly, there are no Iron Age shell middens right on the shoreline of the Limfjord, though a number of sites are found several kilometres inland, containing heaps of marine molluscs or smaller shell middens (Mikkelsen, 1994; Andersen, 2007). The reason behind this absence of coastal middens in the Limfjord is uncertain, but evidence exists for both climatic/environmental changes (e.g. lower salinity, higher productivity, higher temperatures; Lamb, 1982; Kristensen et al., 1995; Christensen et al., 2004) and cultural/behavioural changes (e.g. molluscs transported inland prior to shelling; Poulsen, 1978; Ringtved, 1992) occurring in the Iron Age. Kristensen et al., 1995 and Christensen et al., 2004 suggest that the salinity of the Limfjord might have dropped as low as 7-20 ‰ during the pre-Roman/Roman period, indicating either partial connection or complete closure from the North Sea and/or Skagerrak at this time (though these salinity estimates are qualitative and dating remains uncertain; see above). It is believed the western and/or northern entrances to the Limfjord were re-established (or widened) by the Viking period (AD 750-1050) with foraminifera and molluscan records suggesting that high salinities occurred in this cultural period (22-30 ‰; Kristensen et al., 1995; Christensen et al., 2004, probably due to rising sea level (e.g. Viking sea level high stand identifiable in Tanner, 1993; Christensen et al., 2004).

In the Kattegat and western Baltic Sea, it is believed that the marine environment was rich and prosperous during the late pre-Roman/early-Roman period and that salinity probably exceeded its present day levels (Petersen, 1985; Hebbeln *et al.*, 2006; Rasmussen *et al.*, 2007; Zillén *et al.*, 2008). For example, *Ostrea edulis* extends into the southern Kattegat and Baltic Sea, well beyond its present day limits (e.g. Harck, 1973; Poulsen, 1978), suggesting greater exchange of water with the North Sea (and subsequent increased input of high salinityoxygenated water) must have existed at this time (Emeis *et al.*, 2003; Rasmussen *et al.*, 2007). Sea level is the most likely cause for this increased exchange of water, with one or more transgressions identifiable in regional sea level records covering the pre-Roman/Roman periods (e.g. Miller, 1982; Miller and Hedin, 1988; Behre, 2007).

No shell middens are found later than the Viking Age, but archaeofaunal evidence (e.g. fish and marine mammal bones) and documentary records (e.g. fish catches) demonstrate that marine resources remained an important dietary component throughout the Medieval and Historical times (Enghoff, 1999; Holm and Bager, 2002; Poulsen *et al.*, 2007). For example, along the Baltic and Danish Kattegat coastline, cod, haddock and flatfish formed an important part of the economy over the last ~1,500 years (e.g. Enghoff, 1999; Holm and Bager, 2002) and as already indicated above plaice, eel and herring fisheries were present and exploited at different times in the Limfjord (see Section 1.5). It has also been shown that the development and prosperity of these fisheries is often closely related to climatic and environmental conditions (e.g. Kristensen *et al.*, 2002; MacKenzie *et al.*, 2007), again highlighting the need for long-term records of ecosystem response to climatic and environmental change in order to maintain and manage important commercially important marine resources.

1.7 Project aims and proxies

This project aims to reconstruct changes in key environmental parameters (most notably salinity, nutrient status, productivity and the sedimentary regime) at three coastal Danish sites over the Holocene, in order to assess how land, sea and society have interacted through time and assess critically environmental hypotheses for cultural change (e.g. Rowley-Conwy, 1984). To achieve this, three nearshore coastal sites (Kilen, Norsminde Fjord and Korup Sø; Figure 1.1, Figure 1.3), situated in close proximity to important archaeological sites and settings have been chosen for detailed multiproxy palaeoenvironmental analyses. At each site, diatom analysis has been performed for quantitative reconstruction of salinity change (with increased resolution over key periods of archaeological change, particularly shell midden accumulation phases) using a salinity transfer function, based upon a trans-Baltic training set (derived from sites sampled during the MOLTEN and DEFINE project; Andrén et al., 2007a: http://craticula.ncl.ac.uk/Molten/jsp/). At Kilen, isotopic analysis of singlespecies for a species for a sp and high resolution sedimentary pigment analyses for assessment of changes in productivity and/or nutrient status over the study period. Physical sedimentary analyses (including lithological descriptions, organic matter, carbonate and minerogenic matter; Troels-Smith, 1955; Dean, 1974; Birks and Birks, 1980; Bengtsson and Enell, 1986) have also been

performed to provide important information concerning the physical environment, sedimentary regime and substrate at each site over the study period.

In addition to the proxies analysed as part of this thesis, mollusc (all sites), foraminifera (Kilen and Norsminde Fjord), pollen (Norsminde Fjord and Korup Sø) and plant macrofossil (Kilen and Norsminde Fjord) datasets have been used as part of this multiproxy investigation. It is important to note that these additional datasets have not been analysed by the author, but all interpretations are the author's work, unless otherwise referenced. The pollen and plant macrofossil analyses are particularly important as they provide information concerning changes in the terrestrial environment and human impacts upon the catchment. These records are vital for examining interactions between the land, sea and society and are important for verification of the age models (e.g. *Ulmus* decline at ~3,900 BC; Andersen and Rasmussen, 1993). Where possible, site chronologies have been based on AMS ¹⁴C dating of terrestrial plant macrofossil material to avoid problems associated with the marine reservoir effect (Heier-Nielsen *et al.*, 1995; Olsen *et al.*, 2009).

This project specifically aims to address the following questions:

- 1. How did the Danish coastal environment (salinity, nutrients, productivity and sedimentary regime) change during the Holocene at each site?
- 2. How do the changes identifiable in the independent sedimentary records compare with other local and/or regional records of climate and environmental change (e.g. terrestrial and lacustrine records, sea level changes), and can local events be separated from regional (or global) change?
- 3. What were the drivers behind the changes detectable in the environmental records and are they synchronous across coastal Denmark?
- 4. To what extent can events be attributed to the same causal mechanisms or are there different drivers of environmental change detectable in the proxy records from the different sites?)
- 5. Do these events occur within the same time frame as other known and well-dated marine (e.g. sea level), terrestrial or archaeological (e.g. introduction of agriculture) changes?
- 6. To what extent are changes in the societal record (e.g. shell midden accumulation periods, the oyster decline in the Neolithic shell middens) driven by changes in the marine (or terrestrial) environment?

- 7. Can a successful salinity signal be obtained for the inner Danish coastal waters using a diatom-based salinity transfer function and what other environmental variables (e.g. nutrients, sedimentary regime, preservation), are important in explaining variation in the diatom (and foraminifera or molluscan) dataset?
- 8. How have land, sea and society interacted throughout the Holocene?

1.7.1 Proxies

Diatoms are unicellular photosynthetic protists that have dominated marine primary productivity since the mid-Cretaceous (Round *et al.*, 1990; Sinninghe Damste *et al.*, 2004). They currently account for approximately 40% of modern day total primary marine productivity (Treguer *et al.*, 1995; Field *et al.*, 1998), a figure that can increase to 75-90% in highly productive areas such as coastal shelves, estuaries and areas of intense upwelling (Nelson and Smith, 1986; Nelson *et al.*, 1995; Maslin and Swann, 2006). Diatoms are very sensitive indicators of a number of environmental parameters, tend to preserve well in sediments due to their siliceous frustules, are identifiable to species level and are globally distributed (Battarbee, 1986; Battarbee *et al.*, 2001; Jones, 2007). In coastal and estuarine systems they have been used to reconstruct changes in productivity (Muylaert *et al.*, 2000) sea-surface temperatures (Birks and Koç, 2002), sea-level (Dellwig *et al.*, 1999; Zong and Horton, 1999; Gehrels *et al.*, 2006), salinity (Juggins, 1992; Jiang *et al.*, 1998; Parsons *et al.*, 1999; Ryves *et al.*, 2004) and nutrient status (Clarke *et al.*, 2003; Weckström *et al.*, 2004; Clarke *et al.*, 2006; Weckström, 2006).

In order to quantify salinity, a diatom-based salinity transfer function will be applied to the fossil diatom records from each site, based on a trans-Baltic diatom-environmental training set. The training set is derived from sites analysed during three projects (2001-2006): MOLTEN (Monitoring long-term trends in eutrophication and nutrients in the coastal zone), DEFINE (Defining reference conditions for coastal areas in the Baltic Sea for the Water Framework Directive) (http://craticula.ncl.ac.uk/Molten/jsp/; Andrén *et al.*, 2007a) and DETECT (Developing a tool for assessing ecological reference conditions in the coastal zone of the Baltic Sea; http://www.helsinki.fi/bioscience/ecru/projects/detect.htm). In total the MOLTEN project incorporates modern surface sediment diatom assemblages from 206 sites from Denmark (91), Sweden (35), Finland (55) and Netherlands (25). The DEFINE (2004-2006) project extended this training set into the Bothnian Sea (both Swedish and Finnish coasts), the Baltic states, the North German coast and Norway, and the DETECT project added more sites from the Finnish Archipelago Sea. To date, this training set have primarily

been used for monitoring long-term nutrient trends (particularly total nitrogen, TN) in the Baltic Sea over the last ~250 years (Clarke *et al.*, 2003; Weckström *et al.*, 2004; Weckström, 2005; Clarke *et al.*, 2006; Weckström, 2006; Andrén *et al.*, 2007a). The combined training set, however, also offers great potential for quantitative inference of salinity change over the Holocene across the Baltic, from coastal to pelagic areas (i.e. spanning a salinity gradient of 0-31 ‰ (measured at almost all sites), with salinity independently explaining the largest amount of variation (5.2 %) throughout the entire MOLTEN/DEFINE dataset (Andrén *et al.*, 2007a).

Whilst a wide range of statistical techniques are available for examining species-environment relationships, and a series of measures can be taken to reduce the impact of other environmental variables, such influences can never be completely eliminated. Diatom based salinity models have been shown to perform reasonably well (i.e. produce accurate inferences of salinity) in limnic and coastal systems (Juggins, 1992; Gasse et al., 1995; Ryves et al., 2004) but confidence in diatom-based reconstructions can be further increased by using a multiproxy approach (e.g. Pienitz et al., 2000; Birks and Birks, 2006; Ellegaard et al., 2006; P. Rasmussen, D.B. Ryves et al. unpublished data). Mollusc, foraminifera and isotopic analysis (δ^{18} O and δ^{13} C on single species of foraminifera) have been included in this study to provide independent salinity data for comparison with and/or verification of the diatom inferred salinity estimates over the study period. These proxies have been regularly used in coastal and estuarine studies as palaeosalinity indicators (Petersen, 1993; Kristensen et al., 1995; Burman and Schmitz, 2005; Kristensen and Knudsen, 2006) and a large amount of literature is available detailing the salinity requirements/tolerance limits of a large number of mollusc and foraminifera taxa (e.g. Sorgenfrei, 1958; Knudsen, 1971; Alve and Murray, 1995, 1999; Petersen, 2004).

Sedimentary pigments have been used in this study to assess independently marine productivity and nutrient status (Leavitt and Hodgson, 2001; McGowan, 2007) and help identify periods when these variables might be driving changes in the diatom community, thereby potentially skewing diatom inferred-salinity estimates. This technique is increasingly being used for assessment of nutrient status in coastal systems (Chen *et al.*, 2001; Hodgson *et al.*, 2003; Reuss, 2005a; Reuss *et al.*, 2005; Ellegaard *et al.*, 2006; Rabalais, 2007), being much less labour intensive than analysis of microfossils such as diatoms and dinoflagellates. Pigment samples are also relatively cheap and fast to process making it feasible for high temporal resolution to be obtained at suitable sites. Careful storage and treatment procedures

must be followed to avoid degradation of pigments (Jeffrey *et al.*, 1997; Leavitt and Hodgson, 2001; Reuss, 2005b, 2005a), with recommended storage below -20° C, out of natural light and freeze-drying only immediately prior to analysis (Reuss, 2005b). Pigments are produced by a large number of organisms (e.g. bacterial group, higher plants, diatoms, chrysophytes) and under suitable sedimentation conditions can provide information regarding past marine productivity, nutrient status, and (where taxonomically specific pigments are preserved) can provide information regarding relative abundances of particular groups. For example, chlorophyll-*a* is produced by all photosynthetic algae and higher plants and therefore can provide a general overview of marine productivity, whereas pigments such as alloxanthin (Cryptophytes), diatoxanthin (diatoms, dinoflagellates, and chrysophytes), okenone and rhodopinal (purple sulphur bacteria) are specific to these groups. For more details on the affinity of major pigments preserved in sediments from aquatic environments such as lakes, estuaries and coastal environments, readers are referred to Goodwin, 1980; Leavitt, 1993; Jeffrey *et al.*, 1997; Leavitt and Hodgson, 2001; McGowan, 2007.

Interpretation of pigments must be treated with care, and consideration must be given to the depositional environment in which they accumulate. It is well known that only a small percentage (often <5% of all pigments, McGowan, 2007) actually get preserved in sediments with most being degraded in the water column, and therefore absolute abundances of pigments should not be interpreted too strictly (Reuss, 2005a). At sites with poorer preservation, any productivity or environmental record is likely to be overridden by this preservation signal (e.g. Reuss, 2005a; Reuss *et al.*, 2005). Nevertheless, under favourable conditions, sedimentary pigments can produce excellent records of past changes in marine productivity and nutrients status and have been demonstrated to work well in Danish fjord and estuarine systems (Reuss *et al.*, 2005; Ellegaard *et al.*, 2006; P. Rasmussen, D.B. Ryves *et al.* unpublished data).

Chapter 2. Study sites and methodology

2.1 Introduction

In this chapter the study sites and methodologies employed to answer the major research questions outlined in Chapter 1, section 1.7 are described. This chapter aims to address the following questions:

- 1. What criteria must each site meet in order to be able to address the main research questions, outlined in Chapter 1, section 1.7?
- 2. To what extent do the selected sites meet the specified criteria and why are these factors so important?
- 3. What methods have been used to obtain chronological control at all three sites?
- 4. What palaeoenvironmental/palaeoecological techniques were employed to reconstruct changes in marine environment (most notably salinity, nutrients and productivity) of each site over the Holocene?
- 5. What statistical/numerical methods were used to quantify changes in salinity at each site?
- 6. What palaeoenvironmental/palaeoecological techniques were used as proxies for change in the terrestrial environment (i.e. the catchment of the fjord)?
- 7. What statistical/numerical techniques were used explore the major trends in the biological datasets and to assess the interaction between the terrestrial and marine environment over the Holocene?

2.2 Study Area

2.2.1 Site selection

Three sites were analysed as part of this Ph.D. (Figure 2.1), all situated in the nearshore coastal region of Denmark and carefully selected to match several specific criteria:

- 1) The presence of terrestrial macrofossils for AMS 14 C dating.
- 2) Good chronological control.
- 3) High temporal resolution.
- 4) Situated in close proximity to important archaeological sites and settlements.
- 5) Good preservation of fossil diatoms.

As serious problems exist with the marine reservoir effect in Danish coastal regions (e.g. Heier-Nielsen et al., 1995; Olsen et al., 2009), terrestrial plant macrofossils were favoured over marine plants and shells for ¹⁴C dating at both Kilen and Norsminde. The marine reservoir effect skews radiocarbon dates from marine material generally making them appear older than terrestrial samples of equivalent age (typically set at ~400 years; Stuiver and Braziunas, 1993) due to differences in exchange between carbon reservoirs. For the world oceans, it is caused by delays in carbon exchange between the atmospheric and oceanic bicarbonate together with the dilution effect (mixing of 'young' surface waters with very old deep waters in upwelling regions) (Mangerud, 1972). The magnitude of this effect is not the same in all locations and a large amount of research has gone into producing standard corrections errors for different areas (e.g. Marine reservoir correction database; http://radiocarbon.pa.qub.ac.uk/marine/).

Within this database, a number of standard correction dates exist for the Danish coastal regions (including the Limfjord), but temporal variability appears to be a major problem when trying to create accurate age-depth models (Heier-Nielsen *et al.*, 1995; Olsen *et al.*, 2009, submitted; B. Phillipe and J. Olsen, unpub.). For example, comparison between parallel terrestrial and marine dates from three Danish fjords (Horsens Fjord, Isefjord & Skælskør Nor) yielded a reservoir effect of -425 to +620 years (Olsen *et al.*, 2009). Olsen *et al.*, (2009) suggest this non systematic change between terrestrial and marine derived dates is due to variations in local hydrographic and hydrologic configuration (e.g. precipitation affecting residence time), catchment characteristics and connection with inner Danish seas. More recently, a preliminary assessment of the marine reservoir effect for Kilen (Limfjord) has also revealed significant temporal variability (B. Phillipesen, J. Olsen et al. unpublished data). It is therefore clear that in these systems a standard 400 year correction for the reservoir effect cannot be applied, and that the chronologies presented in previous studies using ¹⁴C dates based on marine material might be erroneous (Petersen, 1981; Andersen, 1992; Heier-Nielsen, 1992; Kristensen *et al.*, 1995; Christensen *et al.*, 2004).



Figure 2.1 – Map of Denmark with the location of the three sites analysed in this thesis (Kilen, Norsminde Fjord and Korup Sø), plus other important sites referenced in the text. Present day isostatic adjustment (dotted grey lines; after Ekman and Mäkinen, 1996) and isobases for the highest level of the Littorina Sea in Denmark (bold numbers and black dotted lines; in metres after Mertz, 1924; Christensen, 2001a), modern day annual mean surface salinities (solid white lines; ‰) and sea current systems for the Kattegat and Skagerrak region (after Gyllencreutz et al., 2006) are also included. Abbreviations: CNSW = central North Sea water, AW = Atlantic water, NCC = Norwegian coastal current, SJC = south Jutland current, NJC = north Jutland current, BW = Baltic water, BC = Baltic current. Numbered sites refer to selected important archaeological localities and settings. Shaded area highlights the areas in which middle and late Stone Age shell midden (in Danish: køkkenmødding) sites are found. Mesolithic Ertebølle and early Neolithic sites (blue dots): 1. Øster Jølby, 2. Aggersund, 3. Brovst (includes Kongemose layers), 4. Nørholm, 5. Bjørnsholm Bay and Åle, 6. Ertebølle locus classicus, 7. Hjerk Nor, 8. Krabbesholm, 9. Visborg and Havnø, 10. Meilgård, 11. Nederst, 12. Vængesø, 13. Norsminde Fjord and Kalvø (includes Ertebølle, Funnel Beaker culture (TBK), Single Grave culture and Iron Age sites), 14. Mosegården , 15. Klintesø, 16. Sølager, 17. Vedbæk (includes Kongemose, Ertebølle and TBK sites), 18. Lollikhuse, 19. Ølby Lyng, 20. Mullerup (important Maglemose site), 21. Kongemose locus classicus. Submerged Ertebølle sites (grey dots): 22. Tybrind Vig, 23. Møllegabet. Single grave and Pitted Ware cultural sites (green dots): 24. Kainsbakke and Kirial Bro. 25-34 (yellow dots) Iron Age shell midden sites. V. location of Vejlerne wetlands (Ryves et al., 2004). Star - Copenhagen (capital city). Map modified from Dahl et al., 2003 and Rasmussen et al., 2007.

An advantage of the coastal zone over more open sea localities is the potential presence of terrestrial organic material for dating and therefore problems associated with the marine reservoir effect can be avoided or reduced. Sufficient plant macrofossil material was present at both Norsminde and Kilen and therefore these stratigraphies are based upon AMS ¹⁴C dates from terrestrial plant material. Korup Sø is an exception as the site was sampled and core material processed in the 1980s for molluscan, pollen and diatom analysis (Petersen, 1993, H. Krog, unpub.) with the original stratigraphy being based upon ¹⁴C dates from shell material (see Table 5.2). Too little material remained from the original investigation to produce a stratigraphy based on terrestrial macrofossils. However chronological confidence has been increased (at all three sites), by comparing independently derived age-depth models to the palynological records to test the 'fit' of well-dated stratigraphical markers (e.g. Ulmus decline at ~3,900 BC; Andersen and Rasmussen, 1993 (H. Krog, unpub.). The age-depth models are discussed in more detail in section 2.5. In addition, there are always difficulties encountered when comparing records from different sites or archives (e.g. Lowe and Walker, 1997; Lowe et al., 2001) often sampled at different resolutions and perhaps even dated via different methods. In some situations these difficulties can be overcome to some degree by using incremental age methods (e.g. tree rings), widely spread (i.e. regional) chronostratigraphic events or markers such as tephra (Turney and Lowe, 2001) or well dated ecological markers (e.g. Ulmus decline in Denmark, Andersen and Rasmussen, 1993).

High precision chronological control and high temporal resolution are essential for assessment of ecological and cultural responses to 'rapid' environmental changes (e.g. climate and sea level change) that might happen on decadal to centennial scales. For example, based on the rapidity of changes in the Danish shell middens (e.g. oyster decline and cultural transition from Ertebølle-Funnel Beaker Culture layers), it has been suggested that the Mesolithic-Neolithic transition might have taken place over as little as ~50-200 years (Rowley-Conwy, 1984; Andersen, 2008a). This is based on the clustering of dates between 4,000-3,800 BC for the Mesolithic-Neolithic transition in numerous shell middens (Andersen, 2008a) and the appearance of agricultural indicators in the pollen records after ~3,900 BC. Therefore, it is essential that the 'selected site' (and proxy) offers high enough temporal resolution to examine these events at the necessary temporal level. In this study, a sampling resolution of 25-32 years is maintained for each site in the period ~4,000-3,800 BC.

Close proximity to archaeological sites and settlements is extremely important for 'direct' assessment of how changes in the local and regional marine environment might have forced cultural/societal changes in the past (e.g. marine resource exploitation). Whilst deep sea cores (or sites distant from archaeological settings) might provide an overview of the major regional changes in the marine environment over the Holocene (e.g. Mörner, 1976; Jiang et al., 1998; Gyllencreutz et al., 2006), uncertainties will still exist as to how these changes affected conditions within the inner coastal areas, where the people actually lived and laboured. For example, many regional sea level (Mörner, 1976; Christensen, 1995; Berglund et al., 2005; Behre, 2007) and temperature curves (Seppä and Birks, 2001; Antonsson and Seppä, 2007; Seppä et al., 2009) now exist, but it is still poorly understood how environmental variables such as the sedimentary accumulation rate, salinity and productivity (which subsequently exert pressure on marine resources such as fish and shellfish) respond to rising/falling sea levels (or temperatures). Furthermore, changes in the local marine environment, perhaps driven by terrestrial forcing factors (e.g. reduced run off, natural vegetation restructuring, human impact) might also be important, but are unlikely to be detectable in distant, deep sea cores. Therefore, all sites analysed in this study are located nearshore and within the same marine system (i.e. Kilen within the Limfjord system, Korup Sø connected to the Kolindsund system; see Figure 2.2 and Figure 2.5) as important archaeological sites and settlements (and always less than 50 km away from a major shell midden; see Section 2.2 for more details). Norsminde is particularly advantageous as it contain a high quality sedimentary record (up until ~1,700 BC; Knudsen and Konradi, submitted; Rasmussen, submitted; P. Rasmussen, unpublished data) in the immediate vicinity of palaeosettlements (Andersen, 1983; Andersen, 1989; Enghoff, 1989).

For quantitative reconstruction of salinity using a diatom-based transfer function, presence and good preservation of fossil diatoms were also key pre-requisites for each site. Sites with poor preservation may exhibit a strong taphonomic signal, which will inevitably weaken inferences from the diatom salinity-model or potentially even override diatom response to important environmental changes (in more extreme conditions). As Kilen was the only new site sampled during the course of this thesis, a preliminary diatom preservation assessment was made on all core sections (at ~20 cm resolution) prior to any other core processing. The presence (and state) of diatoms in Norsminde and Korup Sø sediments had been assessed prior to this project and deemed suitable for analysis (D.B. Ryves pers comm., H. Krog, unpub). Preservation was monitored throughout the course of the diatom analysis at all sites

via assessment of diatom valve dissolution (see Section 2.7.1) expressed as the sample **F** index (Flower, 1993; Ryves *et al.*, 2001; Ryves *et al.*, 2009).

2.2.2 Site descriptions

2.2.2.1 Kilen

Kilen (56°30'005"N 08°34'089"E) is today an 'S' shaped shallow eutrophic lake situated in the Limfjord region of northern Jutland in the Danish commune of Struer (Figure 2.1). Kilen is classified as a brackish water lake (~6 ‰), since being almost entirely cut off from the Limfjord by the building of a road and rail embankment in 1856 (AD), connecting the small towns of Bremdal and Knarbjerg to the north with the municipal capital Struer to the south. A connection with the Limfjord is retained via a small stream in the south-east corner called the Kilerkanal that is important for maintaining salinity levels. Since 1952, Kilen has been protected as a nature reserve and today is an important breeding ground for birds. The lake is approximately 5 km long, 1 km wide with a total area of 3.34 km^2 , an average depth of 2.9 m and a max depth of 6.5 m (Windolf *et al.*, 1996; Jensen *et al.*, 2006). The deepest part of the lake is a small hollow situated very close to the southeastern shoreline, deemed unsuitable for palaeoenvironmental analysis after a preliminary core contained a record of less than 2 m before reaching underlying sands and gravels. For this reason a secondary test coring was made upon a plateau area (~4 m deep; see Figure 2.2) in the north-central part of the lake (water depth 3.9 m) and a much longer Holocene record was obtained (see section 2.4.1).



Figure 2.2 - Map of the Kilen sedimentary basin with bathymetry (originally produced by Thorkild Høy, 1989). Insets: A. Location map of Kilen and the Limfjord, B. Current configuration of Kilen and the Limfjord. C. Present fluvial inflows and basic limnological data for Kilen (Jensen *et al.*, 2006). Coring location is marked with a black star.

The Kilen catchment is relatively small (35.2 km²) with Bredkær Bæk (Bredkær Brook), the main freshwater input entering the lake in the southwestern corner (and draining ~15.9 km² of the catchment). In addition, Vasens Bæk (Vasens Brook) and a couple of other minor watercourses also supply freshwater (Figure 2.2C). Despite the freshwater input, brackish conditions (6-7 ‰) are maintained in the lake via water exchange with Veno Bugt through the Kilerkanal in south-eastern corner of the lake. The introduction of a new sluice in 1986 increased the salinity in Kilen from ~2.5-4.5 ‰ (early 1980s) to ~7 ‰ in 1986-87, by improving the efficiency of water exchange. This new sluice system can be artificially regulated (i.e opened or closed) if necessary to protect the inner fish farms (Ringkjøbning Amtskommune, 1988). The catchment consists of agricultural land along its northern and south western shores, a spruce forestry plantation (Klosterhede) to the west and urban land lining the south east (Struer) and north east (Bremdal) shoreline. Substantial nutrient inputs from the surrounding agricultural land and fish farms together with long water retention times (266 days; Jensen et al., 2006) has led to nutrient enrichment of the lake, culminating in Kilen being monitored between 1989-2003 (Table 2.1), as part of the Danish Nationwide Monitoring Programme (Kronvang et al., 1993; Windolf et al., 1996; Jensen et al., 2006).

Reference	Water retention time (days)	Inlet P (annual) (mg L ⁻¹ TP)	Lake P (annual) (mg L ⁻¹ TP)	Lake P (summer) (mg L ⁻¹ TP)	Chlorophyll- a (summer) (µg L ⁻¹)	Secchi depth (summer) (m)
Jensen <i>et al.</i> , 2006	266	0.15	0.168	0.239	154	0.5

Table 2.1- Summary of Kilen monitoring data (averages for the period 1989-1996.

Previous palaeoenvironmental research or archaeological analyses

No previous palaeoenvironmental research has been undertaken at Kilen itself and for the entire Limfjord, only a handful of studies extend back beyond Roman times (Petersen, 1981; Andersen, 1992; Heier-Nielsen, 1992; Burman and Schmitz, 2005). This site was, however, included in a calibration dataset for a diatom-based salinity transfer function (Ryves *et al.*, 2004) which involved analysis of diatoms present in the surface sediment and the measurement of associated environmental conditions (salinity, depth, total nitrogen and total phosphorus) at the time of sampling or time averaged between AD 1996-2000. In terms of archaeology, Kilen does not contain any shell middens from the Mesolithic, Neolithic or Iron

Age, but is situated in relative close proximity to some of the key Limfjord shell middens (e.g. Krabbesholm and Hjerk Nor both less than 50 km away; Figure 2.1). Documentary sources (Holm and Bager, 2002; Poulsen *et al.*, 2007) and archaeofaunal fishbone evidence (Enghoff, 1986; Enghoff, 1991; Enghoff *et al.*, 2007) is also available for Limfjord sites in the close vicinity of Kilen.

2.2.2.2 Norsminde Fjord

Norsminde Fjord (56 °01'018"N, 10 °14'049"E) is an elongated basin situated along the east coast of Jutland about 20 km south of the city of Århus. It was formed as a valley during the last advances of the Weichseilian ice sheet by glacial meltwater erosion and inundated by seawater in the early Holocene ~6,800-6,900 BC (Rasmussen, submitted). Norsminde Fjord is a relatively small estuarine system (~6 km long, 250 m wide) predominately fed by one major inflow (Rævs Å or Rævs River) that drains a catchment area of ~101 km² (Nielsen et al., 1995). Rævs Å (fed by several tributaries) is responsible for draining 85 % of the catchment before entering the fjord in the south-west at its innermost section (Figure 2.3). This creates a salinity gradient within the fjord with salinities of 0 ‰ freshwater occurring at the mouth of the Rævs Å, gradually increasing up to 24 ‰ at the Kattegat entrance (with most of the fjord around 18-20 ‰). The fjord lies in a SW-NE direction, entering Århus Bay and the Kattegat Sea through an artificially-maintained narrow (40-50 m) sluice established in 1964. The fjord has a surface area of $\sim 1.68 \text{ km}^2$, an average depth of 0.6 m with a maximum depth of 2 m (see Figure 2.3) and a tidal range of 0.2 m (Nielsen et al., 1995). It is estimated that during the higher sea level stands of the Atlantic period (7000-4000 BP: see Figure 1.2), the fjord was significantly larger (~10 km long, 2.7-3 km wide and ~500 m at the Kattegat mouth, Andersen, 1989; Rasmussen, submitted; Figure 2.4).

The subsequent lowering of sea level (Figure 1.2), together with the continuous build up of sand banks and beach ridges at its mouth (Andersen, 1989), has led to the isolation of previously claimed land (e.g. 'Kysing Fjord'; Figure 2.3 and Figure 2.4) and a reduction in the overall surface area of the fjord. The majority of the present day catchment is agricultural land, making Norsminde a moderately eutrophic fjord (primary productivity of plankton 200 g C m⁻² yr⁻¹; Jørgensen and Sørensen, 1985) subject to occasional toxic planktonic blooms and suppression of natural seagrass (*Ruppia* spp.) and Charophyte populations by eutrophic macroalgae (e.g. *Ulva lactuca*). Nitrogen and phosphorus inputs have been reduced by 40 %

since the 1980s due to reduced run-off from agricultural land and improved waste water purification (Ellegaard and Skovgaard, 2007), though nutrient loads remain variable, particularly during years of higher precipitation (e.g. 1994). Norsminde is currently being monitored by the Integrated Protection of Surface and Groundwater in Agricultural Regions Plan (AGWAPLAN) and scenarios are in place for further reductions of nutrient inputs by 2012 and improvement of ecological status.



Figure 2.3 – Inset A: Location map for Norsminde Fjord. Norsminde Fjord bathymmetry (left) and present day hydrological catchment (right). The red star indicates the coring site. The reclaimed part of the fjord, named 'Kysing Fjord', is also indicated. Bathymmetric map from Ellegaard and Skovgaard, 2007, catchment hydrological map modified from Århus Amt, Natur og Miljø 1998.

Previous palaeoenvironmental or archaeological analyses

Prior to 1996, no palaeoenvironmental or palaeoecological research had taken place at Norsminde. The site has, however, been subject to extensive archaeological research which began in 1945 and continued intermittently up until 1989 (Andersen, 1989). Norsminde is now one of the most important (and thoroughly researched) Stone Age sites in Denmark, with its shorelines hosting a whole series of Mesolithic and Neolithic aged settlements (e.g. kitchen middens) supplemented by a large number of individual archaeological finds (e.g. artefacts, cooking implements, fishing tools) which clearly document the importance of the fjord and marine environments to Stone Age cultures and societies. It was during the latest major excavation (1972-1989) that the characteristic 'køkkenmødding' containing both Mesolithic and Neolithic layers (see section 1.6.1) was unearthed (Andersen, 1989). Prior to this, 'normal' kitchen midden sites (i.e. without the characteristic change in shell assemblage found across the Mesolithic-Neolithic transition) were uncovered at Flynderhage (Gabrielsen, 1953) and Norslund (Andersen and Malmros, 1966) of Ertebølle age, and at Kalvø, containing layers dated to the Neolithic Single Grave culture (Andersen, 1983).



Figure 2.4 – Norsminde Fjord geology, topography and archaeological map (modified from Andersen, 1989). Numbered sites refer to key archaeological sites: 1. Norsminde køkkenmødding, 2. Frederiks Odde, 3. Kalvø, 4. Norslund, 5. Flynderhage. Inset A: section through the Norsminde kitchen midden with Mesolithic and Neolithic layers indicated (photo courtesy of Søren Andersern, Mosegard Museum).

2.2.2.3 Korup Sø

Korup Sø is situated in the central part of the Djursland peninsula, east Jutland, ~27 km northeast of the city of Århus (Figure 2.5). Today the site is terrestrial, existing as agricultural land positioned ~3 m above sea-level. The history of the site is somewhat complex, involving a series of limnic and brakish-marine phases influenced by eustatic sea level change, glacio-isostatic rebound and artificial draining. Korup Sø is a southward extension of the former Kolindsund fjord system (Figure 2.5), which was probably formed by meltwater and ordinary fluvial erosion during the Late Weichselian and Early Holocene. The Korup Sø basin itself

was likely formed by a block of dead ice left by a retreating glacier (Pedersen and Petersen, 1997). The Littorina transgression eventually drowned Korup Sø thereby connecting it to the Kattegat via the Kolindsund to the east and Randers Fjord to the north-west whereby the northern part of Djursland became an island separate from Jutland (Figure 2.5; Jessen, 1920; Petersen, 1993; Pedersen and Petersen, 1997). It is estimated that the Kolindsund was a ~50 km long strait when the sea level was at its highest with the Korup Sø area being positioned along its southward arm in a relatively central position (Jessen, 1920; Petersen, 1993; Pedersen and Petersen, 1997). Kolindsund and Korup Sø remained brackish-marine until at least the end of the Stone Age but how long this lasted is unknown. Some historical records exist, suggesting that Viking ships might have entered Kolindsund around AD 1060 (Flou et al., 1980). Due to the combination of a declining sea level during the Late Holocene and continued isostatic rebound, the connection between Kolindsund/Korup Sø and the sea was cut, and the former sound was transformed into a lake (area $\sim 1.5 \text{ km}^2$), sometimes named Djurssø. At this stage (of unknown age) the Korup Sø basin became a freshwater lake again (known as Korup Sø). Between AD 1872 and 1880, Kolindsund/Djurssø including Korup Sø was reclaimed and turned into agricultural land (Flou et al., 1980).

The aquatic phase of Korup Sø's history has received much attention being first investigated as early as the 1930's by Johannes Iversen who analysed a sediment sequence for pollen, diatoms and the so-called "Hystrix" (hystrichosphaerids, resting cysts for dinoflagellates). At the base of a 8.5 m long sequence Iversen identified Late Glacial limnic deposits superseded by limnic gyttja, referred to the late Boreal chronozone, followed by brackish-marine deposits (Iversen, 1937). The site was revisited in the 1980s by Harald Krog and Kaj Strand Petersen (both at DGU/GEUS) who analysed pollen, diatoms (H. Krog, unpublished data) and molluscs (Petersen, 1993) in two sediment sequences. Pollen and molluscan results from one of these corings (Well no. 1; Petersen, 1993) are reported in this thesis (see section 5.5.2 and 5.5.3) as well as new diatom analyses from raw material (Harald Krogs counts were unusable and the original slides were unavailable). Whilst no archaeological excavations have taken place from within the Korup Sø basin itself, the site is located close to a number of important archaeological sites and settlements (Figure 2.5), most notably Meilgård, Nederst, Dryholmen and the important Pitted Ware cultural sites of Kainsbakke and Kirial Bro (Troels-Smith, 1942; Rasmussen and Richter, 1991; Rasmussen, 1993a; Enghoff, 1994; Andersen, 2007).


Figure 2.5 - The Djursland peninsula with the location of Korup Sø and Kolindsund. The present day coastline (dotted black line) and the estimated coastline during the *Littorina* Sea stage (dark blue shading) are indicated (modified after Jessen, 1920; Rasmussen and Richter, 1991). Sites not in italics refer to important archaeological sites referenced in the text.

2.3 Material and Methods

This project incorporates a range of physical, biological and chemical techniques for highresolution assessment of coastal environmental change at three sites in Denmark over the Holocene. Table 2.2 provides a complete record of all proxies analysed, indicating clearly the author's contribution. In this section, field work, core treatment and subsampling details are provided together with methodologies for all proxies included in this thesis. All stratigraphic diagrams presented in this thesis, detailing proxy results were produced using C2, version 1.6.2 (Juggins, 1991-2009).

2.4 Field work, core treatment and sub-sampling

2.4.1 Kilen

Two parallel boreholes (~1 m apart and vertically offset by 50 cm, labelled R3 and R4) were cored from the nearshore area of Kilen (see Figure 2.2) between 26th - 29th April 2007. For each borehole 1 m cores sections (15 for each borehole) were individually collected using Russian D-section peat samplers (Jowsey, 1966) with diameters of 10 cm and 5 cm diameter (both with 100 cm chambers). The water depth at the point of coring was 390 cm. A HON-Kajak sampler was also used to collect the very topmost section (390-426 cm below present sea level; bpsl) of the profile, which was sliced up into 1/2 cm intervals on site. The R3 borehole (starting at 400 cm bpsl) consisted of 15 core sections extending down to a maximum depth of 1,700 cm bpsl (Figure 2.6). The uppermost 9 core sections (400-1,200 cm bpsl) were retrieved using the 10 cm diameter Russian peat sampler and the remainder collected using the 5cm diameter sampler. The R4 borehole (starting at 450 cm bpsl) also consisted of 15 core sections extending the depth down to 1,950 cm bpsl. For the R4 borehole, the uppermost 10 core sections (450-1,450 cm bpsl) were collected using the 10 cm diameter Russian sampler and below this (1,450-1,950 cm bpsl) using the 5cm diameter sampler. All core sections were briefly visually assessed on site and any distinguishing features (e.g. shell layers, macrofossils) were recorded before being carefully wrapped up for storage in the GEUS cold store (at 4°C).

Site	Proxy	Analyst	No. of samples	Resolution
Kilen	Diatoms	J. Lewis (LU)	110	Average 68 years (M-N transition ~39 years, PWC ~60 years, PR/P ~70 years, Limnic ~7 years)
(~5,500 BC- present)	Sedimentary pigments	J. Lewis (LU)	777	Average ~10 years
	Isotopes (foraminifera - $\delta^{18}O, \delta^{13}C$)	J. Lewis (LU)	44	Average ~100 years, (M-N transition ~38 years, PWC ~46 years)
	Sedimentary physical parameters*	J. Lewis (LU) , B. Stavngaard (GEUS)	1,249 (JL) 150 (BS)	Average ~9 years
	Foraminifera	P. Kristensen (AU)	15	Average ~378 years
	Macrofossils (plants and other marine organisms)	P. Rasmussen (GEUS), J. Lewis (LU)	611	Average ~12 years
	Molluscs	K.S. Petersen (GEUS)	611	Average ~12 years
Norsminde	Diatoms	J. Lewis (LU)	76	Average ~66 years (LT ~18 years, M-N transition ~45 years, PWC ~43 years)
(~7,000- 1,700 BC)	Molluscs	K.S. Petersen (GEUS)	75	Average ~69 years
	Foraminifera	P. Konradi (GEUS)	77	Average ~65 years
	Pollen	P. Rasmussen (GEUS)	62	Average ~85 years
	Plant macrofossils	P. Rasmussen (GEUS)	74	Average ~70 years
	Sedimentary physical parameters *	B. Stavngaard (GEUS)		Average ~10 years
Korup Sø	Diatoms	J. Lewis (LU)	54	~100 years (M-N transition ~70 years, PWC ~76 years)
(~7,600-2,300 BC)	Molluscs	K.S. Petersen (GEUS)	65	Average 81 years
	Pollen	H. Krog (GEUS)	110	Average ~49 years

Table 2.2 - Details of all proxies analysed for Kilen. Norsminde and Korup Sø. Abbreviations: LT – Littorina transgression (7,000-6,600 BC), M-N =Mesolithic-Neolithic transition (4,500-3,500 BC), PWC = Pitted ware culture (2,400-2,800 BC), PR/R = Pre-Roman/Roman period (500 BC – AD 400), LU = Loughborough University, GEUS = Geological Survey of Denmark and Greenland, AU = Aarhus University. JL = Jonathan Lewis, BS = Beth Stavngaard. *= Includes organic matter, carbonate content, minerogenic matter, water content and dry mass. All core sections were cleaned thoroughly on their outer surface and their lithologies described using a modified Troels-Smith system as described in (Troels-Smith, 1955; Birks and Birks, 1980). The Russian core sections from the R4 borehole were divided up into 2 cm depth intervals and separate $\sim 1 \text{ cm}^3$ aliquots were subsampled for physical analyses (including water, organic matter and carbonate content), microfossil (pollen and diatom), and sedimentary pigments analyses, using a 1 cm³ syringe. The material sub-sampled for physical analyses was placed directly into pre-weighed crucibles for drying (see Section 2.6.1), the diatom/pollen aliquots were freeze-dried to improve preservation (particularly of pollen grains) and the sedimentary pigment samples were placed straight into the freezer and stored at -20°C until ready for analysis. The remainder of the material was retained for wet sieving, primarily to search for appropriate material for AMS ¹⁴C dating but also for other analyses such as plant macrofossils, molluscs, foraminifera and ostracods (Table 2.2). Only two core sections from the R3 borehole (R3-C1; 400-500 cm & R3-C3; 500-600 cm) were used for the final analyses. Based on the distinctive lithological progression at the top of the profile (see Figure 3.1), R3-C1 and the upper half of R3-C3 (500-550 cm) were sliced up into 1cm intervals and the bottom half of R3-C3 (550-600 cm) into 2 cm intervals. The remaining overlapping R3 core sections (R3-C4 – R3-C14) were also sliced up into 2 cm intervals, but only the central 30 cm overlapping sections (with parallel R4 corings) were subsampled for physical (microfossil and pigment) analyses to attempt core correlation between the two boreholes. For diagrammatic representation see Figure 2.6.

In total, 625 samples (1-2cm thick, 610 of which were included in final depth profile), were wet sieved through 2 different mesh sizes (500 μ m & 100 μ m). The volume of sediment for each sample was quantified using the water displacement method, where prior to wet sieving the sediment was dropped in a known volume of water and the displacement volume was recorded. The recorded volumes for Kilen samples were between 1.5-45 ml, mean 21.1 ml. After wet-sieving, the two fractions were transferred into Petri dishes and potential AMS ¹⁴C dating material (see below), molluscs (both fractions), ostracods (fine fraction) and any other plant or animal remains were picked out by hand.



Figure 2.6 - Kilen coring details. The final sedimentary sequence (i.e. after core correlation) includes all core sections shaded yellow. Cores were correlated (where possible) using the sedimentary physical parameters (included in this diagram for all core sections and expressed percentages; see text for more details).

2.4.2 Norsminde

In 1996, two overlapping boreholes, ~1m apart were cored at Norsminde by Peter Rasmussen and Ole Bennike (GEUS). In total 25 cores sections (12 and 13 from boreholes 1 and 2 respectively) were collected using 2 variable-width Russian peat samplers (100 cm chambers, 5 cm and 10 cm diameters), together with a HON-Kajak corer to retrieve the uppermost section of the profile (31 cm). Alternating core segments from each borehole was cored at overlapping depth intervals (~20 cm) to try to account for any depth discrepancies. Cores were subsampled at 2 cm intervals for physical analyses (including water content, organic, carbonate and minerogenic using methods described below) and microfossil analyses (pollen and diatoms) by Beth Stavngaard (GEUS), before being sliced up for macrofossil analysis. The physical analyses were also performed by Beth Stavngaard using methods outlined below and overlapping core segments were correlated using water content and LOI data (P. Rasmussen pers. comm.).

In total, 74 samples (8-10 cm thickness, 79-393cm³ volume; determined by water displacement method) were wet-sieved through 2 different mesh sizes (500 µm & 200 µm) and then analysed for plant macrofossil (and mollusc) remains. All identifiable plant macrofossils were recorded and any terrestrial material deemed suitable for AMS ¹⁴C dating was picked out and frozen for preservation. Molluscan remains at Norsminde were very scarce, but those present were sent to Kaj Strand Petersen (GEUS) for identification (see Section 2.7.2). Any additional identifiable remains such as cladocerans, echinoid remains, barnacles, fish remains or ostracods were also picked out (and/or recorded). The finer fraction was retained for foraminifera analysis (performed by Peter Konradi, GEUS; see Section 2.7.3).

2.4.3 Korup Sø

Two boreholes were drilled in 1982/83 (labelled Well no. 1 and Well no. 2; Petersen, 1993). All analyses involved in this thesis (see Table 2.2 and Section 2.7) have been performed on material from Well No. 1, which extends back to ~7,600 BC (~9,550 BP) with a maximum depth of 880 cm. The age model is discussed in detail in section 5.3. The second borehole (Well no. 2) extended ~3 m further down (1,251 cm maximum depth), offering greater temporal resolution (Petersen, 1993), but shorter longevity than Well No. 1, yielding a maximum age of ~7,400 BC. Core correlation was not attempted between the parallel

76

drillings due to the significant distance between boreholes and the absence of physical sedimentary analyses (i.e. water content, organic, carbonate and minerogenic content). It is important to note that some analyses (molluscs and pollen) were performed on samples from Well no. 2, together with the production of an age model based upon 7 ¹⁴C dates on marine molluscs (see section 5.3) and the pollen stratigraphy.

2.5 Core stratigraphy, dating and chronology

2.5.1 ¹⁴C dating; Kilen, Norsminde and Korup Sø

All samples for ¹⁴C dating were submitted to the AMS ¹⁴C Dating Centre at Aarhus University. The terrestrial plant macrofossil samples were pre-treated by 1M HCl at 80°C for 24 hours, prior to being left for one to seven days in 1M NaOH at 80°C. Samples were then treated in 1M HCl, before being converted to CO₂ by combustion (in sealed evacuated quart tubes containing CuO). Shell samples were subjected to thorough washing and leaching (outer 25% leached by HCl) prior to dehydrated phosphoric acid treatment (100%) for 8 hours at 25° C. Following liberation of CO₂, all samples were converted to graphite using the H₂ reduction method and AMS ¹⁴C measurements were made using an EN tandem accelerator (under the supervision of Jan Heienemeier, Aarhus University). Results were reported as conventional radiocarbon dates (¹⁴C yr BP, before 1950 AD; Stuiver and Polach, 1977) and calibrated into calendar years using the IntCal04 calibration curve (Reimer *et al.*, 2004) with a 400±50 year marine reservoir correction applied to shell samples (from Korup Sø).

2.5.2 ²¹⁰Pb and ¹³⁷Cs dating; Kilen

For dating of the uppermost section of the Kilen profile, a total of 23 freeze-dried samples (all 1cm homogenised slices selected from the Kajak core) were sent to Thjorbjørn Andersen at the Gamma Dating Centre, University of Copenhagen for analysis of ²¹⁰Pb, ¹³⁷Cs and ²²⁶Ra activity. The activity was measured via gammaspectrometry, carried out on a Canberra low-germanium well-detector. ²¹⁰Pb was measured via its gamma-peak at 46.5 keV, ²²⁶Ra via the granddaughter ²¹⁴Pb (peaks at 295 and 352 keV) and ¹³⁷Cs via its peak at 661 keV (T. Andersen, pers. comm.). Dates were calculated using a modified constant rate of supply model (Appleby, 2001). The final Kilen age-depth model was based on 13 AMS ¹⁴C dates, together with a ²¹⁰Pb, ¹³⁷Cs and ²²⁶Ra stratigraphy for the uppermost section of the profile using an OxCal 4.1 depositional model (Ramsey, 2008). To account for changes in accumulation rate, boundaries are inserted at 524, 1,055 and 1,740 cm. The position of the

boundaries are based on large changes in the $CaCO_3$ content (J. Olsen, pers. comm. Figure 3.2).

2.5.3 Core Correlation

Kilen

In an attempt to eradicate any depth discrepancies during the coring process, physical parameters (dry mass/water content (%), organic matter (%), carbonate (%) and minerogenic matter (%); see Section 2.6) analysed on both the master profile and on overlapping sections of the parallel drive, were compared using wiggle-matching for best fit (Dearing, 1986). For Kilen, cores were correlated, where possible, based on organic matter (%) and dry mass/water content (%). This was only possible for the top section of the record (cored depths 390-650 cm), as below core R4-C2 (550-650 cm) only small magnitude (and non-systematic) changes occur in these parameters. The lithology of this section (i.e. 650-1,950 cm bpsl) is marked by homogenous marine clay gyttja (see section 3.2). Despite the presence of some shell layers throughout the profile, these appear randomly distributed, often not even extending across the whole diameter of an individual core, and therefore shell layers were not deemed suitable for cross-correlation.

In total, 13 samples (all terrestrial plant macrofossils; Table 3.2) from the Kilen profile were sent for AMS ¹⁴C dating (at the AMS ¹⁴C Dating Centre, Aarhus University). Preliminary pollen analysis (5 samples) was undertaken by Peter Rasmussen (GEUS) to identify the approximate location of the *Ulmus* (elm) decline (see Chapter 3). In some samples, the absolute number of pollen grains counted was low (282-500 grains) due to very low pollen concentration, probably as a result of the high sedimentation rate (~0.2 cm yr⁻¹). There was, however, sufficient evidence suggesting that the elm decline (~3,900 BC; Andersen and Rasmussen, 1993) fell between 1,337-1,361 cm bpsl. Accordingly the age of the elm decline in Kilen at 1,337-1,361 cm bpsl is dated to 3840-3910 BC and provides an independent check on the age model.

Norsminde

Overlapping core sections from Norsminde Fjord were correlated (by wiggle-matching) by Peter Rasmussen (GEUS) based upon the minerogenic profiles, measured at 2 cm intervals (see Section 2.4.2) on all core sections (Rasmussen, submitted).

Korup Sø

No attempt was made to correlate the core sections from the two Korup Sø wells, based on the large distance between the wells (i.e. 100 m apart), the large differences in sedimentation rate and the absence of data for any sedimentary parameters (i.e. water content, organic, carbonate and minerogenic content).

Age-depth models

Following the core correlation, ¹⁴C dating and calibration processes, age-depth models were produced (by Jesper Olsen, Aarhus University, now at Queens University Belfast) using Oxcal 4.1 (for Kilen and Korup Sø; Ramsey, 2008) and B-peat 1.0 (for Norsminde; Blaauw and Christen, 2005). Both Bpeat and Oxcal incorporates Bayesian statistical modelling (allowing for sedimentation rate changes) to identify the most probable match for the complete sequence of dates by taking into account the stratigraphic information (e.g. changes in the sedimentary physical parameters) in the age calibration (Blaauw and Christen, 2005; Ramsey, 2008, 2009). Following age-depth model production and verification (using elm decline marker for example), sedimentary accumulation rates (expressed as cm yr⁻¹) for each site were calculated over the sediment sequences, together with sediment bulk density measurements (these form the basis for areal accumulation (flux) rates for individual sedimentary (i.e. organic matter, minerogenic matter, carbonate) and biological components (i.e. diatoms and sedimentary pigments).

2.6 Laboratory Analysis

2.6.1 Physical analyses: Kilen and Norsminde

The physical parameters analysed at both Kilen and Norsminde include dry weight (for calculation of water content; %), organic matter (%), carbonate content (%) and minerogenic matter (%). For Kilen, these analyses were performed on all samples from the R4 borehole and core sections R3-C1 and R3-C3 from the R3 borehole. For the remaining core sections from the parallel R3 borehole, only overlapping parts (Figure 2.6) were analysed for purposes of core correlation (see Section 2.5.3; Figure 2.6). Measurements of dry mass (for water content), organic matter and carbonate content for Norsminde Fjord follow the same methodology (outlined below), but were not undertaken by the author (see Table 2.2).

Unfortunately, no physical data exist for Korup Sø sequences, and insufficient material remained on which to perform the analyses.

After subsampling (see Section 2.4), the ~1 cm³ aliquots of sediment were left to dry overnight in a standard laboratory oven at 105°C (in pre-weighed and pre-numbered crucibles). Samples were then cooled and re-weighed with the dry-weights (DW) recorded for calculation of percentage water content (Equation 1). For organic matter (%), samples were ignited in a stepwise manner (see Table 2.3 for ignition program) up to 550°C (Dean, 1974) at which temperature they were held for 60 minutes. Samples were then again left to cool at room temperature before being re-weighed and the remaining ignition residue weights were recorded and percentage organic matter calculated (Equation 2). For carbonate content, the samples were again ignited in a stepwise manner (Table 2.3) up to 925°C (Dean, 1974; Bengtsson and Enell, 1986) and held at the temperature for 15 minutes. Cooled samples were then weighed and recorded for calculation of calcium carbonate (%) content assuming all weight loss is CO₂ from the decomposition of CaCO₃ (Equation 3). The recorded weight of the remaining residue makes up the minerogenic component of the sample (converted to percentage of total weight, as in Equation 4).

> Eq. 1 - WC (%) = $((WW - DW_{105})/WW)*100$ Eq. 2 - OM (%) = $((DW_{105} - DW_{550})/DW_{105})*100$ Eq. 3 - CC (%) = $((DW_{550}-DW_{925})/DW_{105})*2.27*100$ Eq. 4 - MM (%) = 100-(OM+CC)

Where WC = water content, WW = wet weight OM = organic matter, CC = calcium carbonate content, MM = minerogenic matter, DW_{105} = Dry weight after heating to 550°C, DW_{925} = Dry weight after heating to 925°C.

Relative abundances for each component are expressed as percentages of the total wet weight (for water content and dry mass) or dry weight (for organic, carbonate and minerogenic components). Accumulation rates for each component have also been calculated (using the age-depth models, see Section 2.5 (and 3.3, 4.3, 5.3) and are expressed as mg cm⁻² yr⁻¹.

Proxy	Stage	Rate	Temperature	Rest
Organic matter content	1	500 °C/hour	100°C	10 mins
(OM)	2	500°C/hour	300°C	10 mins
	3	500°C/hour	550°C	60 mins
	4	Cool		
Calcium carbonate	1	700 °C/hour	500°C	10 mins
(CC)	2	700 °C/hour	700°C	10 mins
	3	700°C/hour	925°C	15 mins
	4	Cool		

 Table 2.3 Organic matter and carbonate heating programmes for loss-on-ignition. Resting phases are 10 minutes long unless otherwise specified.

2.7 **Biological Proxies**

2.7.1 Diatom analysis: Kilen, Norsminde & Korup Sø

All diatom slides from Kilen were prepared by the author at Loughborough University using the method outlined below (Battarbee and Kneen, 1982; Renberg, 1990). For Norsminde and Korup Sø, some of the diatom slides were prepared by Beth Stavngaard (GEUS), using the same method. Diatom samples were prepared at carefully selected depth intervals from the aliquots of material specifically subsampled for microfossil analysis (see section 2.4). For each sample, approximately 0.01 g (0.1 g for wet samples; several HON-Kajak core samples) of freeze-dried sediment was weighed out into labelled test tubes and left overnight in ~5 ml of concentrated hydrogen peroxide ($30 \ \mbox{M} \ \mbox{H}_2O_2$) at room temperature (~20°C). Samples were then prepared using hot hydrogen peroxide ($30 \ \mbox{M} \ \mbox{H}_2O_2$) and HCl (5-10 %) digestion and washing techniques as described by Renberg, (1990). For absolute diatom concentrations, microspheres were added to aliquots of each sample (Battarbee and Kneen, 1982).

A minimum of 300 valves were counted per sample (together with microspheres) from fields of vision along random transect lines using a Leica Leitz Laborlux S microscope at x1000 and under oil-immersion phase contrast. Diatoms were identified to species level where possible, using a range of taxonomic literature including: van der Werff and Huls, 1958-1974; Hendey, 1964; von Tynni, 1978; Krammer and Lange-Bertalot, 1986-1991; Snoeijs *et al.*, 1993-1998; Witkowski, 1994; Hasle and Syvertsen, 1996; Witkowski *et al.*, 2000; Lange-Bertalot, 2001; Levkov, 2009. Diatoms from the genera *Chaetoceros* and *Rhizosolenia* were recorded but excluded from the total count due to difficulties with identification to species level. This was due to *Chaetoceros* cysts being predominately found in their resting stage and *Rhizosolenia* valves as just remnant spines. Diatom taxonomy was also harmonised to MOLTEN/DEFINE concepts (<u>http://craticula.ncl.ac.uk/Molten/jsp/</u>) for application of the salinity transfer function to fossil diatom data (see below).

In order to assess the taphonomic process of dissolution, diatom valves were placed into one of two dissolution categories where possible; pristine or dissolved (following Ryves *et al.*, 2001). Valves in the pristine category were excellently preserved, showing very little or no signs of dissolution under light microscopy. Alternatively, valves exhibiting clear signs of the dissolution process were placed in the dissolved category. A number of valves could not be assessed accurately and were subsequently placed into an unassessed category. This category typically included valves that were partly covered up by other material (or other diatoms) in the slide, valves appearing in girdle view, or in the case of complete frustules in valve view, difficulties were often encountered assessing the state of the lower valve.

For *Paralia sulcata*, a 4-stage dissolution assessment was made on all valves that appeared in valve view for each site following Ryves *et al.*, (2009). *P. sulcata* was deemed suitable due to its clear 4-stage dissolution process (see appendix A), general robustness (making it a good indicator for particularly poor preservation conditions) and its regular presence in sediments from Danish coastal sites (e.g. Mikkelsen, 1949; Ryves *et al.*, 2004; H. Krog, unpub., A.L.Clarke et al. unpub.). Additionally, the diameters of all *P. sulcata* valves were measured and then placed into one of two different size categories (<20 μ m and >20 μ m) as suggested by McQuoid and Nordberg, 2003. In McQuoid and Nordberg's observations, greater abundances of the smaller valves were found where water columns are vertically mixed and have more upwelling of nutrient-rich high salinity water. In contrast, larger sized valves were more abundant under lower nutrient (phosphorus) conditions, often in stratified waters with low surface water salinity.

2.7.2 Molluscan analysis: Kilen, Normsinde and Korup Sø

For Kilen and Norsminde Fjord, molluscan remains were picked out of all of the samples selected for wet sieving (from both the coarse and fine fractions; see Section 2.4) and were subsequently identified by Kaj Strand Petersen (GEUS). Molluscan nomenclature follows Petersen, (2004). For the Korup Sø profile, all molluscs were picked and identified from thick sediment slices (3-30 cm thick; 166-3,475 cm³), covering the entire sequence (precise sieving details, including mesh sizes remain unknown). The Korup Sø molluscan assemblages have previously been reported in Petersen, (1993).

2.7.3 Foraminifera analysis; Kilen and Norsminde

Foraminifera samples (i.e. remainder of the finer 0.1-0.5 mm fractions following wet sieving and removal of molluscan and plant macrofossil material; see Sections 2.4.1 and 2.4.2) selected for analysis were concentrated by floatation in a heavy liquid (specific weight 1.8 g cm³; cf Knudsen, 1998), prior to counting using standard techniques described in Feyling-Hanssen *et al.*, (1971) and Knudsen, (1998). A minimum of 300 tests were picked identified and counted where possible. Foraminifera nomenclature follows Ellis and Messina, (1949 and supplements).

2.7.4 Pollen and plant macrofossil analysis: Kilen, Norsminde and Korup Sø

Pollen samples were prepared using standard techniques described in Fægri and Iversen, (1995), from ~0.1 g aliquots of freeze-dried material. Kilen and Norsminde samples were counted by Peter Rasmussen (GEUS) using a Leitz Ortholux microscope at x320 or x800 magnification (phase contrast), and Korup Sø samples were counted by Harald Krog (GEUS). For plant macrofossils (Norsminde and Kilen only), all distinguishable remains (present from the wet sieved fractions; see above) were recorded and, where possible, identified by either the author (Kilen only) or Dr. Peter Rasmussen (GEUS; Norsminde and Kilen). Pollen and plant macrofossil nomenclature follows Tutin *et al.*, (1964-1980) (except for *Poaceae* = *Gramineae*). *Triticim* and *Hordeum* (cereal pollen) were distinguished using criteria outlined in Andersen, (1979).

2.8 Chemical analyses

2.8.1 Sedimentary Pigments: Kilen

Pigment analysis was performed by the author at the University of Nottingham, School of Geography. All sedimentary pigment procedures were carried out under subdued lighting to limit the degradation of pigments during the preparation process. Weighed aliquots of freeze-dried samples (~0.2 - 0.4 g) were placed overnight (in a freezer, -20° C) in 500 ml of extraction solvent (80 % methanol, 15 % methanol, 5 % deionised water) before being filtered to remove any remaining solid material. Samples were then dried under a nitrogen stream and dissolved in a known quantity (most commonly 500 µl) of HPLC injection solvent (70 % acetone, 25 % IPR stock solution (Table 2.4), 5 % methanol) and placed in small vials ready for injection into an Agilent 1200 series separation module with Quaternary pump (HPLC

machine). Samples were loaded into an autosampler in batches (normally between ~15-25 samples, always including 2 'Green standards'; see Table 2.4). Each HPLC run lasted 52 minutes which included several separation phases using a modified method from that described in Chen et al., 2001; separation conditions for each phase are summarised in Table 2.5). Eluted pigments were measured by passing through a photo-diode array detector which detects UV-visible spectral data every 2 seconds. Pigment concentrations were calculated using the package Agilent Chemtstation and output as chromatogram reports which were interpreted manually. Pigments were identified based on their retention time and absorbtion spetra compared with published literature (e.g. Jeffrey et al., 1997) and authentic standards. Sedimentary pigment concentrations are expressed as nmol/g organic matter and pigment accumulation rates as nmol cm⁻² yr⁻¹.

Solution	Details	Phase	Method
Extraction solvent	80 % acetone, 15 % methanol, 5 % deionised water	Pigment Extraction phase	This solvent is to be made in small batches and kept in a screw cap bottle. Use HPLC-grade solvents (acetone and methanol) and nanopure deionised water and combine in the stated ratios.
Injection Solution	700 ml acetone, 50 ml methanol, 250 ml IPR stock solution (see below)	Injection phase	In a 1L volumetric flask combine a 70:25:5 mixture of acetone, IPR, and methanol. Transfer the completed solution to a foil wrapped 1L ground glass reagent bottle and store in the freezer. De-gas the solution in a sonicator before using in the HPLC machine.
HPLC solution A	80 % methanol, 20 % of 0.5 M ammonium acetate, aq. pH 7.2	Mobile phase A	Dissolve 7.708 g ammonium acetate in 200 ml deionised (HPLC-grade) H_2O . Add the 200 ml 0.5 M ammonium acetate to a 1 L volumetric flask and top up to 1 litre with HPLC-grade methanol. De-gas for 5 minutes in a sonicator before use.
HPLC solution B	90 % acetonitrile, 10 % deionised water	Mobile phase B	In a 1 L measuring cylinder, measure out 900 ml acetonitrile and top up with 100 ml deionised HPLC-grade H_2O . De-gas for 5 minutes in a sonicator before use.
HPLC solution C	100% HPLC-grade ethyl acetate		De-gas for 5 minutes in a sonicator before use.
Ion-pairing reagent (IPR) stock	To make 250 ml: 1.875 g of tetra butyl ammonium acetate, 19.25 g of sigma grade ammonium acetate, 250 ml volumetric flask with HPLC-grade H ² O	Injection phase: used for production of the injection solution	This solution must be made fresh each time it is used. Weigh out 0.75g of tetra butyl ammonium acetate and 7.7g of Sigma-grade ammonium acetate and rinse into a 100 ml volumetric flask with HPLC- grade H_2O . The chemicals are hygroscopic and temperature sensitive and therefore must not be allowed in contact with the air for any length of time. Reseal the containers and return to the freezer immediately. Swirl the flask to completely dissolve the chemicals. Now add enough H_2O to bring the volume up to the 100 ml mark.
Green standard			Grind a handful of grass or other green plant and a few ml of extraction solvent in a mortar and pestle. Filter the collected material using the procedure outlined in the extraction of pigments section. Dry down the concentrated pigment under nitrogen. Add injection solution to the dry pigments until the colour turns a light green. Label the vial as "Green Standard" and store in the freezer.

Table 2.4 – Summary of the stock solutions used during the extraction and preparation stages of sedimentary pigment analysis. (Modified from HPLC laboratory handbook, Dr. Suzanne McGowan, University of Nottingham).

Time (minutes)	Solvent A (80:20 methanol: 0.5 M ammonium acetate) - %	Solvent B (9:1 acetonitrile: water) - %	Solvent C (ethyl acetate) - %	Flow (ml min ⁻¹)
0	100	0	0	1
4	0	100	0	1
38	0	25	75	1
39	0	25	75	1
43	100	0	0	1
52	100	0	0	1

Table 2.5 - Separation condition used in the analytical procedure (method modified from Chen *et al.*, 2001). Stationary phase (column) consisting of a Thermo Scientific ODS Hypersil column (205*4.6 mm; 5 µm particle size).

2.9 $\delta^{18}O \& \delta^{13}C$ isotope analysis of single species of foraminifera: Kilen

Isotopic analyses were performed at the NERC Isotope Geosciences Laboratory (NIGL), Keyworth, UK. Material was once again prepared using exactly the same method as above (Section 2.4.1), but with the selected isotope samples being immediately filter dried after wet sieving. Isotopic analyses were performed on tests of *Elphidium excavatum* fo. *selseyensis*, chosen due its regular presence and often high abundance throughout the profile, particularly through some of the key periods of archaeological and environmental change (e.g. the Mesolithic-Neolithic transition, ~3,900 BC; see Chapter 3). *Elphidium excavatum* fo. *selseyensis* has also been used in other isotopic studies (e.g. Scheurle and Hebbeln, 2003; Kristensen and Knudsen, 2006; Schweikhardt *et al.*, 2010) which have demonstrated its ability to record changes in salinity and/or temperature sensitively.

From selected samples (44 in total), between 5-20 individuals of *Elphidium excavatum* fo. *selseyensis* were picked out and placed into labelled micropaleontology slides. Tests were cleaned as much as possible by hand during the picking process with any attached matter visible being removed. Specimens from each sample were then individually transferred to small Eppendorf vials until a desired weight was achieved (30-100 µg). Samples were analysed using a GV IsoPrime mass spectrometer plus Multiprep device and Isotope values (δ^{13} C, δ^{18} O) are reported as parts per mille (‰) deviations of the isotopic ratios (13 C/ 12 C, 18 O/ 16 O) calculated to the VPDB scale using a within-run laboratory standard calibrated against NBS standards. Analytical reproducibility of the standard calcite (KCM) is < 0.1 ‰ for δ^{13} C and δ^{18} O (M.J. Leng pers. comm.).

2.10 Numerical analyses

2.10.1 Statistical zonation of sedimentary and biological datasets

All biological (and sedimentary pigment) zones were determined using the package PsimPoll 4.27 (Bennett, 2003-2009) using an optimal splitting method based on an information content measure of total variance. This splitting method was chosen over agglomerative methods (e.g. CONISS, CONSLINK; Grimm, 1987) because of its ability to deal with transitional phases better. Significant splits were identified using a broken stick model (Bennett, 1996). However, it is important to note that when using optimal splitting techniques, the broken stick method can only help with identification of zones that are definitely significant. It cannot, positively determine if a split is non-significant (Bennett, 1996). Therefore, in some cases, additional optimal splits (not deemed significant by the broken stick method) were included as sub-zones, but only if they did not change the preceding significant split boundaries. In these circumstances, additional sub-zones were introduced if they aided the ease of interpretation (subjective measure).

2.10.2 Quantitative reconstructions of salinity: the MOLTEN salinity transfer function

Salinity at each site was inferred quantitatively using a diatom-based transfer function produced by Dr. Annemarie Clarke (Newcastle University, GEUS and APEM Ltd) and Dr. David Ryves (Loughborough University) from a subset of sites sampled during the MOLTEN and DEFINE projects (Clarke *et al.*, 2006; Andrén *et al.*, 2007a; <u>http://craticula.ncl.ac.uk/Molten/jsp/</u>). A combined total of 341 sites (i.e. 217 MOLTEN sites and 124 DEFINE sites; Andrén *et al.*, 2007a) were analysed for diatoms present in surface sediment samples together with measurement of associated environmental conditions (including water depth, Secchi depth (m), salinity (‰), ammonium, nitrate, nitrite, chlorophyll *a*, total nitrogen (TN), total phosphorus (TP), phosphate (PO4-P) and silica (all µg L⁻¹). Surface sediments were collected using a HON-Kajak corer (Renberg, 1991) with the top 1 cm being prepared for diatom analysis. A minimum of 500 valves were counted per sample, excluding *Chaetoceros* cysts and *Rhizosolenia* spines.

Salinity is the single most significant environmental variable explaining the distribution in the full MOLTEN/DEFINE diatom dataset (i.e. all sites, Andrén *et al.*, 2007a; Figure 2.7). The geographic expanse of the dataset, however, meant a number of other environmental gradients were also captured, and that other environmental variables are also important in explaining

small amounts of the variation in the full diatom dataset (most notably depth, total nitrogen (TN), total phosphorus (TP) and exposure; Figure 2.7). The variation explainable by salinity alone was increased by constructing a smaller training set, removing sites more strongly associated with other environmental parameters (e.g. heavily eutrophic sites). The main gradients in the diatom species dataset were explored using a series of cluster and ordination techniques (e.g. detrended correspondence analysis (DCA) and non-metric multidimensional scaling (NMDS); A.L. Clarke et al. unpublished; and see Andrén et al., 2007a). Principal component analysis (PCA) was used for investigation of environmental gradients and canonical correspondence analysis (CCA) and variance partitioning was used for examining relationships between the diatom distributions and the environmental variables (CCA deemed appropriate due to unimodal distribution of species data, as determined by DCA; DCA axis 1 for species dataset >2 SD units). Within CCA, forward selection was applied to identify significant variables (i.e. p < 0.05; 999 permutations), with Bonferroni corrections (for the nth variable to be accepted $p_n = p_1/n$, where $p_1 = 0.05$) until variables fell below the corrected significance value and were subsequently omitted from further analyses. Forward selection identified salinity, depth, TN and TP to be the most important (and significant) variables. The final salinity subset includes 318 diatom species and 210 sites spanning a salinity gradient of ~0-31 ‰ (A.L. Clarke, et al. unpub.), with salinity the most significant variable, explaining 6.58 % percent of the variation in the diatom species data.

Prior to application of the transfer function (for salinity reconstruction) to the fossil diatom datasets, the predictive power of a series of weighted average, weighted average-partial least squares (WAPLS) and modern analogue technique (MAT) models were tested. The predictive power of these models were evaluated using r_{boot}^2 and RMSEP scores, following internal validation (via bootstrapping x1000 cycles) using the modern training set. The WAPLS model performed best and final diatom inferred (DI) salinity reconstructions are based on a component-3 model ($r_{boot}^2 = 0.923$ and RMSEP = 0.36 square root salinity units. The component-3 model was selected because it significantly (p=0.001) improved the RMSEP (2.6 %), compared to the simpler component-1 and component-2 models.



Figure 2.7 - Map of MOLTEN/DEFINE sites including both water chemistry and diatom counts. Red dots indicate MOLTEN sites and black dots indicate DEFINE sites. Inset A: Variance partitioning of the full MOLTEN/DEFINE dataset (i.e. 341 sites) using CCA. (From Andrén *et al.*, 2007a).

2.10.3 Multivariate analysis and environmental-species relationships

Major trends in the biological datasets, relationships between the different proxies and landsea interactions (i.e. modelling predictor-response scenarios) have been explored using a series of multivariate ordination techniques (including detrended correspondence analysis (DCA), principal component analysis (PCA), redundancy analysis (RDA) and canonical correspondence analysis (CCA)) using CANOCO v4.5 (ter Braak and Šmilauer, 2002). For all ordinations using linear methods (i.e. PCA and RDA), as the response (species) data were % abundances Aitchison's log-ratio analysis (Aitchison, 1986) was applied, with samples and species data both centred. For unimodal methods (i.e. DCA, CCA) rare species were always downweighted. Prior to ordination analyses, the frequency distributions of variables included within the environmental predictor datasets were examined using Calibrate 0.81 (Juggins and ter Braak, 1997). In cases where distributions were positively skewed (generally with a few large values), the gradient length was deemed too long, data were log transformed to reduce the effect of high value outliers. Further details are given in the appropriate sections of results chapters.

2.11 Summary

- In this study, three coastal sites (Kilen, Norsminde Fjord and Korup Sø) were selected for palaeoenvironmental analyses on the basis that they matched several important criteria: 1. the presence of terrestrial macrofossils for AMS ¹⁴Cdating and/or a detailed pollen record for verification of the age-depth model, 2. good chronological control, 3. offered high temporal resolution, 4. situated in close proximity to important archaeological sites and settlements, and 5. good preservation of fossil diatoms.
- Kilen is a small brackish lake (presently eutrophic) situated near to the town of Struer at the Limfjord, north-west Jutland. Up until ~AD 1856, it was directly connected to the Limfjord, but has since been cut off by the building of a road and rail embankment. A narrow connection still exists in the south-east corner (termed the Kilerkanal), through which water exchange between Venø Bugt and Kilen occurs, maintaining brackish water salinities (~6 ‰). In 2007, a ~15 m long sediment sequence was retrieved at a water depth of 390 covering the period from ~5,500 BC-present day (dated via AMS ¹⁴C on terrestrial plant material and ²¹⁰Pb/¹³⁷Cs dating techniques, with age model produced in Oxcal v. 4.1).
- Norsminde Fjord is a small, shallow fjord system (average depth 0.6 m, maximum depth 2 m) situated on the east coast of Jutland. Today it has an average salinity of ~18-20 ‰, (gradient from 0 ‰ in the west at the mouth of Rævs Å, to 24 ‰ in the east where it opens into the Kattegat Sea). In 1996, a 1,180 cm core sequence was retrieved covering the time period ~6,900-1,700 BC (dated via AMS ¹⁴C dating using terrestrial plant macrofossils, with age model produced in B-peat 1.0), with the most recent ~3,700 years lost due to a sedimentary hiatus.
- Korup Sø is a former fjord system (previously forming a southward extension of the Kolindsund Fjord) situated in Djursland, eastern Jutland. Today, the Korup Sø basin is agricultural land, lying ~3 m above sea level. In 1982 and 1983, two core sequences were retrieved, the longest of which (in temporal terms and used in this study) covers

the period ~7,600-2,300 BC (880 cm sediment sequence, dated by 14 C dating on marine molluscs with age model produced in Oxcal v. 4.1), with the most recent ~4,300 years again lost due to a sedimentary hiatus.

- A series of palaeoenvironmental techniques have been employed to reconstruct environmental changes over the study period (with higher resolution focus upon the archaeological shell midden accumulation phases). These include sedimentary physical parameters (lithological descriptions, organic matter, carbonate and minerogenic matter; Kilen and Norsminde Fjord), diatoms (all sites), isotopes from the tests of *Elphidium excavatum* fo. *selseyensis* (Kilen), sedimentary pigments (Kilen), foraminifera (Kilen and Norsminde Fjord), molluscs (all sites), pollen (Norsminde Fjord and Korup Sø), plant macrofossils (Kilen and Norsminde Fjord). The methodologies for all these techniques have been outlined in this chapter (and the analysts involved clearly referenced).
- A diatom-based salinity transfer function has been developed from the MOLTEN and DEFINE training set (consisting of 210 sites, 318 species spanning a salinity gradient 0-31 ‰). A three-component WAPLS model performed best under internal validation (via bootstrapping x1000 cycles) and has been applied the fossil diatom data for each site, to quantitatively reconstruct changes in salinity.

Chapter 3. Results: Kilen

3.1 Introduction

In this section results from multiproxy palaeoenvironmental analyses performed on sediment cores from the Limfjord site, Kilen, are presented. The analyses are presented in logical progression, starting with lithology, chronological control and proxies reconstructing the physical and sedimentary environment before progressing onto the biological and chemical remains of past communities living within the fjord over the last ~7,500 years. The chapter finishes with a synthesis and discussion section, integrating the results from the different proxies in order to critically assess how the Kilen environment has changed over the study period. Specifically this chapter aims to address the following questions:

- 1. How has chronological control been achieved at this site and how accurate is the dating?
- How has the physical and sedimentary environment at Kilen changed over the last ~7,500 years?
- 3. How has the salinity, productivity and nutrient status of the fjord changed over the last ~7,500 years?
- 4. How has the Kilen diatom flora changed since the building of the road/rail embankment in AD 1856 and its transition from a marine to limnic environment?
- 5. What are the most important environmental parameters driving the changes in the biological communities and can their contributions be quantified?
- 6. Can these different environmental parameters be completely isolated from one another to produce accurate and independent reconstruction for these variables?
- 7. To what extent do the key environmental variables driving the changes in the biological communities interact (e.g. co-vary or drive one another) and how do these relationships vary over time?
- 8. To what extent do the different proxies employed to reconstruct the natural environment of Kilen agree/disagree with one another?

3.2 Lithology

The Kilen lithological profile is summarised in Figure 3.1 and the lithological descriptions based on a modified Troels-Smith system (Troels-Smith, 1955; see also Birks and Birks,

1980) are provided in Table 3.1. No description is available for the section collected using the HON-Kajak gravity corer (390-425 cm bpsl) as this was sliced up into half centimetre intervals on site, straight from the coring tube. For this section, information from a correlated parallel core is also provided in Table 3.1 to give an indication of the lithology, though exact boundaries must be treated with caution. No overlap is available for the uppermost 10 cm of the profile (390-400 cm bpsl).

Below this, the profile is entirely composed of marine gyttja containing varying amounts of silt, clay, plant matter and shell material. Slight colour changes are identifiable throughout the profile and occasional narrow shell layers are visible (see Table 3.1). The uppermost sections are generally browner in colour probably due to the higher organic content (see Figure 3.5) and occasional intervals containing dy (e.g. top-434 cm, 476-517 cm) and more humified material (e.g. 400-417 in parallel core). Below 434 cm the sediment becomes more clayey and below 524 cm there are higher levels of silt. In all cases, the transition between layers is very gradual (i.e. Limes superior = 0 according to the Troels-Smith system).

The abundance of visible shell fragments increases below 465 cm, with the uppermost section being quite barren (very few shell fragments between 425 and 434 cm and no visible shell fragments between 434 and 465 cm). The shell density is particularly high between 497 and 524 cm with some shells (e.g. cockles) even found sitting in living position. Below this, the shell concentration decreases again but scattered shells (both whole and fragmented) remain present throughout, occasionally in clusters or sometimes even in narrow layers (see Table 3.1, Figure 3.1). Superimposed upon this general trend is another short-lived phase where shell density increases between 1033-1041 cm. Below 1,753 cm, there is a further drop in concentration. Plant fragments are less abundant than shells, generally occurring sporadically throughout the profile (e.g. twig fragment at 540 cm). An exception to this occurs between 476 and 517 cm, where plant material becomes quite abundant for a short period of time.

93

Donth (om hngl)	Lithology Description
<u> </u>	Water
390-425	No description
425-434	Dark brown gyttja containing dy. No shells visible. Limus detrituosus $(Ld)^2$ 3. Sand, silt, clay (SSC) - 1. Limes superior (LS; upper boundary) - 0 (i.e. boundary area >1 cm).
434-443	Dark brown greenish grey clay gyttja (lighter than above layer and more minerogenic). Very few shell fragments. Ld^{1-2} 2. SSC – 2. LS - 0.
443-465	Dark brownish grey clay gyttja (again lighter than above layer). Very few shell fragments. Ld ¹⁻² 2. SSC - 2. LS - 0.
465-476	Dark brownish grey clay gyttja (darker than above layer and more shell fragments). Ld ¹⁻
476-497	² 2. SSC - 2. LS - 0. Dark-brow blackish clay gyttja containing dy and many plant fragments and more dispersed shell fragments. Gets more clayish towards top. Below 486 cm – more pure gyttja. Concentration of plant material higher Ld^{23} . Detritus granosus (Dg) 1. LS = 0.
497-517	Dark brown blackish clay gyttja containing dy, some plant fragments and many shell fragments (including whole shells sitting in living position. <i>Cerastoderma</i> spp. (cockle) in living position at boundary to layer below. Ld^23 . SSC - 1, Dg ++. LS - 0.
517-524	Dark brown blackish clay gyttja (lighter than above layer) containing many dispersed shells and more minerogenic matter. Transitional layer. Ld ¹⁻² 2-3. SSC - 1-2. LS - 0.
524-597	Dark grey homogenous silty clay gyttja with a few scattered shells (whole and fragments). Twig fragment at 504 cm. $Ld^{1}1-2$. SSC - 2-3. Dg +. LS - 0.
597-653	Dark grey (slightly lighter than above layer) homogenous silty clay gyttja. Scattered shells (whole and fragments). Probably more minerogenic material. Ld ¹ 1-2. SSC - 2-3. LS - 0
653-753	Dark grey-brown homogenous silty clay gyttja with a few scattered shells (whole and fragments). Argilla steatodes (As) +, Argilla granosa (Ag) - 3, Ld 1, LS - 0.
753-853	Dark grey-brown homogenous silty clay gyytja with a few scattered shells (whole and fragments). As +, Ag - 2-3, Ld 1-2, LS - 0.
853-953	Dark grey-brown homogenous silty clay gyttja with a few scattered shells (whole and fragments). As $+$, Ag $-$ 3, Ld 1, LS $-$ 0.
953-1,253	Dark grey homogenous silty clay gyttja with a few scattered shells (whole and fragments). Shell fragments more abundant between 1033-1041cm. Shell layers at 1,101-1,103cm and 1,149-1,151cm. Concentration of shells at 994-996 and 1,109-1,111 cm. As $\pm Ag 2-3$ L d $1-2$ L S = 0
1,253-1,453	Dark grey homogenous silty clay gyttja with a few scattered shells (whole and fragments). Slightly more than in cores RC11-RC15 (1,453-1,952 cm bpsl). Shell layers at 1,324-1,326, 1,363-1,365, 1,407-1,408 cm. Pebble at 1,425-1,426 cm $A_{S} + A_{T} = 3$ I d 1 I S = 0
1,453-1,753	Dark grey homogenous silty clay gyttja with a few scattered shells (whole and fragments) $As + Ag - 3Ld + LS - 0$
1,753-1,953	Dark grey homogenous strongly silty clay gyttja with very few scattered shells (whole and fragments). As +, Ag - 3, Ld 1. LS - 0.
Correlated to 400- 417 cm (From core ref. R3 C1).	Dark brown-blackish gyttja containing dy with a little sand, silt and clay. No shells visible. Ld^24 , SSC +++. $LS - 0$.
Correlated to 417- 425 cm (From Core ref R3 C1)	Dark brown gyttja (lighter than above layer – slightly greener) containing dy and a little sand, silt and clay. No shells visible. Ld^23 , SSC – 1. LS – 0.

 Table 3.1 - Lithological descriptions for the Kilen sedimentary profile. Descriptions are based on a modified Troels-smith system (Troels-Smith, 1955; see also Birks and Birks, 1980)



Figure 3.1 - Kilen lithological profile.

3.3 Chronology and Age Model

The Kilen chronology is based on 13 AMS ¹⁴C dates together with a lead (²¹⁰Pb) and caesium (¹³⁷Cs) stratigraphy for the uppermost section of the record (since ~AD 1920). All AMS ¹⁴C dates are derived from terrestrial plant macrofossil material (Table 3.2) thereby eliminating the problematic marine reservoir effect (Heier-Nielsen *et al.*, 1995; Olsen *et al.*, 2009). Preliminary analyses at Kilen suggest an extremely variable reservoir effect of between -150 and +320 years (J. Olsen and B. Phillpesen, unpublished data). All 13 dates were included in the final age model (produced by J. Olsen Aarhus University using Oxcal 4.1, k=10, A=99.4%). Six samples were analysed for pollen (by Peter Rasmussen, GEUS) in order to identify the approximate location of the *Ulmus* (elm) decline (see Figure 3.3). Accordingly the age of the elm decline in Kilen is dated to 3,840-3,910 BC and provides an independent check on the age model.

Unsupported ²¹⁰Pb activity is detectable from background levels down to a depth of 30 cm. The concentration generally decreases exponentially with depth from its surface concentration of 241 bg kg⁻¹ down to \sim 25 cm, after which irregularities occur in the profile (Figure 3.4). Supported ²¹⁰Pb activity is detectable in all samples, remaining almost constant throughout. Sample sediment ages were derived using a constant rate of supply (CRS) model Appleby, 2001 and results are displayed in Figure 3.2. The Pb-derived chronology suggests an average accumulation rate of ~0.35 cm yr⁻¹ over the duration of the lead profile (beginning ~AD 1920; see also Figure 3.5). These results have been verified to some degree by ¹³⁷Cs, detectable in all samples. ¹³⁷Cs levels also decrease with depth down until ~AD 1920 (as indicated by the ²¹⁰Pb stratigraphy). This pre-dates its release into the atmosphere (~1945) suggesting a degree of sediment mixing or downward diffusion of ceasium, and therefore Pb-based ages must be considered as minimum ages as mixing will result in the derivation of 'younger' ages. The maximum ¹³⁷Cs activity occurs at 8 cm dated by ²¹⁰Pb as 1984, which can be considered broadly consistent with fallout from the Chernobyl incident in 1986. Despite some clear signs of sediment mixing and irregularity in lead activity towards the bottom, the exponential decrease with depth in the upper section gives confidence in the results and therefore the dating can be considered to be reasonably accurate (T. J. Andersen, pers. comm.).

Both the AMS ¹⁴C (Figure 3.2, Table 3.2) and ²¹⁰Pb/¹³⁷Cs stratigraphies (Figure 3.2, Figure 3.4) are deemed robust chronologies being independently verified by the *Ulmus* decline (~3,900 BC; Andersen and Rasmussen, 1993) and caesium fallout peaks (e.g. Chernobyl, AD 1986; Figure 3.4). There are, however, dating uncertainties in the section between the top of radiocarbon profile (depth 459-460; ~AD 800) and the ²¹⁰Pb/¹³⁷Cs stratigraphy (depth 420-321 cm bpsl; ~AD 1920). The period is characterised by extremely low accumulation rates (~0.03 cm yr⁻¹), low sedimentary pigment and diatom concentrations and absence of molluscs and foraminifera. This might reflect low productivity and subsequent sediment mixing and poor preservation conditions. Alternatively a sedimentary hiatus might exist in the late Holocene. Broad estimates have been made for the stratigraphic positions of key events such as the closure of the Limfjord (dated to ~1,100-1,200 BC according to Petersen, 1976; Kristensen *et al.*, 1995; Christensen *et al.*, 2004) and the and the building of the road/rail embankment (AD 1856-57) based on the available proxy evidence (see section 3.11). These estimates must, however, be treated with caution until the chronology has been improved for this part of the sequence (i.e. ~AD 800-1920).



Figure 3.2 – Age-depth model and ²¹⁰Pb stratigraphy (blue line and inset) for the Kilen profile. Model created by Jesper Olsen (Aarhus University) using Oxcal v.4.1. Dotted lines refer to depths for which allowances for shifts in the accumulation rate were made based on changes in the carbonate curve (see Figure 3.5).

Laboratory number	Depth (cm bpsl)	Material dated	¹⁴ C age (BP)	Model age (± 2 SD AD/BC)	δ ¹³ C (‰ VPDB)
AAR-12150	459-460	1 leaf fragment, 1 <i>Betula</i> sp. fruit, 2 bud scales, 1 <i>Cyperaceae</i> seed	1315±65	672-894 AD (95.4%)	*-27
AAR-12151	472-473	1 flower from Quercus sp.	1440±65	433-495 AD (7.7%), 504-685 AD (87.7%)	*-27
AAR-12152	523-524	1 Scirpus maritimus/lacustris seed	1880±60	20-12 BC (0.5%), 1BC-257 AD (93.5%) 300-318 AD (1.4%)	*-27
AAR-12142	603-605	1 leaf fragment	2464±40	761-682 BC (25.4%), 671-413 BC (70.0%)	-26.87
AAR-12143	745-747	1 twig fragment with bark	3215±35	1,605-1,576 BC (4.3%), 1,536-1,415 BC (91.1%)	-28.38
AAR-12144	975-977	1 twig fragment with bark; presumably <i>Salix</i> sp.	4132±45	2,874-2,581 BC (95.4%)	-27.27
AAR-12145	1,149-1,151	1 leaf fragment	4635±75	3,634-3,553 BC (11.2%), 3,541-3,311 BC (68.8%) 3,295-3,286 BC (0.6%), 3,275-3,265 BC (0.7%) 3,239-3,105 BC (14.1%)	-28.19
AAR-11463	1,171-1,175	206 leaf fragments, 3 <i>Betula</i> sp. fruits, 1 bud, 1 bud scale	4635±40	3,621-3,611 BC (1.0%), 3,521-3,347 BC (94.4%)	-28.36
AAR-12146	1,279-1,281	1 Alnus sp. twig with bark	4848±42	3,709-3,626 BC (65.2%), 3,598-3,526 BC (30.2%)	-28.61
AAR-12147	1,399-1,401	97 leaf fragments	5225±40	4,228-4,201 BC (6.1%), 4,170-4,127 BC (10.5%) 4,119-4,097 BC (3.1%), 4,080-3,961 BC (75.7%)	-27.52
AAR-12148	1,483-1,487	11 leaf fragments	5260±90	4,331-3,942 BC (94.0%), 3,856-3,844 BC (0.6%) 3,836-3,821 BC (0.8%)	*-27
AAR-12149	1,715-1,717	43 leaf fragments	6120±65	5,222-4,881 BC (93.5%), 4,871-4,848 BC (1.9%)	*-27
AAR-11464	1,923-1,927	25 leaf fragments	6405±60	5,170-4,848 BC (95.4%)	*-25

Table 3.2 - Details of the samples dated by AMS ¹⁴C from the Kilen profile (* estimated value).



Figure 3.3 – Percentage values for selected taxa in the Kilen pollen record spanning ~4,180-3,640 BC. The *Ulmus* curve (shaded grey) exhibits a decline between 1,338 and 1,362 cm bpsl with the age ~3,900 BC, which is in good agreement with the general dating of the classic mid-Holocene 'elm decline' in Denmark (Andersen and Rasmussen, 1993). Analyst: Peter Rasmussen (GEUS).



Figure 3.4 - Depth profiles of ²¹⁰Pb and ¹³⁷Cs for Kilen.

3.4 Physical Parameters

Results of all physical parameters measured at Kilen profile are provided in Figure 3.5. In total 1,369 samples from the Kajak and two boreholes (R3 and R4) were measured for organic, carbonate, minerogenic and dry mass/water contents, with the final profile consisting of 860 samples. The physical sedimentary parameters for Kilen have been divided up into four (statistically significant) zones determined by optimal splitting (see section 2.10.1) and are described below in relation to these zones. The accumulation rate for the Kilen profile and flux rates for the organic, carbonate and minerogenic components are also plotted in Figure 3.5 (calculated using the age model presented in Figure 3.2).

KIL-S1 (~**5,500-3,130 BC**) – At the bottom of the profile the overall sedimentation rate is high (0.5 cm yr⁻¹), with all three components (i.e. organic, carbonate and minerogenic matter) exhibiting their highest fluxes for the marine phase of the profile (i.e. pre-1856). However, these fluxes decrease at ~5,100 BC, causing a drop in the overall accumulation rate (~0.33-0.43 cm yr⁻¹ between 5,100-3,130 BC). After ~5,100 BC, the carbonate and mineogenic fluxes steadily decrease and the organic flux gradually increases again. Only minor changes occur in the relative abundance within the sedimentary matrix, such as a slight rise in the carbonate content between ~5100-5,000 BC and the very gradual increases in both the organic and mineogenic contents after ~5,030 BC.

KIL-S2 (~3,130 BC- AD 30) – This zone begins with another drop in the accumulation rate (down to 0.19 cm yr⁻¹), largely driven by decreases in both the organic and minerogenic fluxes. There is very little change in the sedimentary regime between ~3,120 BC – AD 30, with the exception of a small and gradual decrease in the minerogenic flux (and subsequently the overall accumulation rate; 0.14-0.15 cm yr⁻¹) after ~1,500 BC and an increase in the carbonate flux at the very top of this zone (at ~AD 60). The relative abundance of organic matter in the sedimentary matrix gradually increases between ~3,120-100 BC (~7 % increase). In contrast, the overall carbonate content decreases up until ~200 BC and the minerogenic component decreases after ~800 BC.



Figure 3.5 - Kilen sedimentary physical parameters, including overall accumulation rates (cm yr⁻¹), water content and dry mass expressed as percentages of the total wet weight of the sediment, percentages of the dry mass that are organic matter, carbonate and minerogenic matter (following loss-on-ignition), plus fluxes (mg cm⁻² yr⁻¹) for organic matter, carbonate, minerogenic matter and dry mass. A magnification of Kil-S4 is provided in Figure 3.6. (NB. Accumulation rate and flux may be erroneous between ~AD 800-1920 due to uncertainties with the age-depth model (see sections 3.3 and 3.11)).



Figure 3.6 - Kilen sedimentary physical parameters focussing on Kil-S4. For full details see Figure 3.5 caption. (NB. A lithology column is omitted in this figure due to the absence of any lithological description for this section of the profile. Also note that the scales for the individual parameters in this figure have been adjusted accordingly (from Figure 3.5) for presentation purposes).

KIL-S3 (**~AD 30-1920**) – A further drop in the overall accumulation rate occurs at the beginning of this zone (i.e. at ~AD 30) with both the organic matter and minerogenic flux exhibiting decreases. After ~AD 30, both remain relatively constant (~0.07-0.09 cm yr⁻¹) and there is subsequently, little change in the accumulation up until ~AD 800. The carbonate flux differs slightly as it fluctuates up until ~AD 200, before also becoming constant. Problematic dating and a potential sedimentary hiatus suggests likely errors in the overall accumulation rate (0.03 cm yr-1) and sedimentary flux calculations between ~AD 800-1920. If a continuous accumulation of sediments did continue between ~AD 800-1920 (i.e. no hiatus), then an extremely low accumulation rate must have ensued, but currently this remains equivocal. Large changes occur in the sedimentary matrix in KIL-S3, as increased percentages of both organic matter (between ~AD 20-700) and carbonate content (between ~90 BC-AD 500) result in decreased abundance of minerogenic material between ~90 BC-AD 900. Relative abundances of these sedimentary parameters continue to fluctuate in the upper section of this zone (i.e. ~AD 900-1920), but changes are lower in magnitude.

KIL-S4 (**~AD 1920-present**) – This zone begins with an apparent large, but likely erroneous increase in the overall accumulation rate and fluxes of all three sedimentary components. This date marks the beginning of the lead stratigraphy and therefore increased accumulation rates (and fluxes) are just an artefact of the linear interpolation between the uppermost radiocarbon date and the lowermost ²¹⁰Pb/¹³⁷Cs date (see section 3.3 and 3.11). Nevertheless, in the post AD 1920 period, the accumulation rates remains high but subject to fluctuations. The predominant pattern, however, is a decrease in the accumulation of sediments up until ~AD 1950, followed by a general increase in post ~AD 1950 period. The individual fluxes also drop in this period, but increase again after ~AD 1960 (with more pronounced increases for organic and minerogenic components than for carbonates). In terms of the sedimentary matrix, the organic content of the sediment gradually increases throughout KIL-S4, with the minerogenic content decreasing and the carbonate content remaining relatively constant.

3.5 Diatoms

A total of 110 diatom slides have been counted over the ~7,500 year record with increased resolution over the Mesolithic-Neolithic transition, Pitted Ware or Single Grave cultural

103

period, Roman period and the recent limnic phase (see Table 2.2). 96 genera were recorded with a total of 332 taxa identifiable to species level. The full diatom record and associated DI-salinity reconstruction are provided in Figure 3.7 (and for zones KIL-D4-KIL-D6, Figure 3.9) and diatom-associated metrics are illustrated in Figure 3.8 (and Figure 3.9). These include the total sum of diatoms, the centric/pennate ratio, benthic: planktonic: tychoplanktonic (B:P:T) ratio, DCA axis 1 scores (for entire and reduced datsets; see section 3.10), diatom concentration and valve accumulation rate Hill's N₂ diversity, dissolution **F** index, *Paralia sulcata* dissolution index and *P. sulcata* valve size. Six statistically significant zones were determined via optimal splitting with one additional split added to aid interpretation (i.e. sub-zone KIL-D1b).

The salinity reconstruction presented in Figure 3.7 is inaccurate in the uppermost section of the record (since ~AD 1830), due to increasing numbers of diatoms indicative of enriched nutrient status appearing in the record. This suggests that nutrients (rather than salinity) are the main environmental variable driving the changes in the diatom assemblage. Therefore, a diatom-inferred total nitrogen (DI-TN) reconstruction for the last ~200 years is also presented in Figure 3.9 using a TN-training set based on 91 MOLTEN sites from Denmark and Western Sweden (Juggins, 2004). The final TN-reconstruction utilises a WAPLS-component 2 model ($r^2_{jack-knifing}= 0.74$, RMSE_{jack-knifing}= 0.14). Reconstructions using the MOLTEN weighted averaging model and modern analogue technique (not shown) were also tested and yielded largely similar results. Application of similar TN-models (i.e derived from the MOLTEN training set) in other Danish coastal sites (Roskilde Fjord, Mariager Fjord) over the recent past has yielded relatively accurate reconstructions of past nutrient status (e.g. Clarke *et al.*, 2003; Clarke *et al.*, 2006; Ellegaard *et al.*, 2006).



Figure 3.7 – Kilen percentage diatom record (taxa >3 %). Numbers in brackets refer to the salinity optima in the MOLTEN training set used for the quantitative reconstruction of salinity. Analyst: Jonathan Lewis.



Figure 3.8 - Kilen diatom associated metrics. * Sample scores for DCA analysis on reduced diatom dataset (i.e. between ~5,430 BC-AD 800; lower 82 samples).


Figure 3.9 - Kilen diatom record for the uppermost section of the profile (KIL-D4-KIL-D6; ~AD 1830-present).

KIL-D1a (~**5,500-4,250 BC**) - *Chaetoceros* cysts the most abundant group of species, fluctuating intensely throughout, but supplemented by relatively constant percentages of *Opephora mutabilis* (most abundant benthic; ~3-13 %), *Cocconeis scutellum* agg. (*scutellum* and *parva* varieties), *Paralia sulcata, Dimerogramma minor, Bacillaria paxillifer, Plagiogramma staurophorum, Amphora helenensis* and *Rhopalodia acuminata*. The DI-salinity reconstruction suggests salinity is generally around 23-24 ‰ for the majority of the zone but with occasional fluctuations (e.g. ~25-29 ‰ between ~5,080-4,830 BC). High salinity demanding species such as *Delphineis minutissima* and *Cymatosira belgica* briefly appear in the record between (5,080-4,670 BC). *Cyclotella choctawatcheeana* numbers also fluctuate which might suggest variable nutrient status and/or water depth.

KIL-D1b – (**4,250-2,740 BC**) - *Chaetoceros* cysts continue to fluctuate, but exhibit a general increase from ~18 % at ~4,150 BC to ~33-35 % between ~3,650-3,480 BC (most abundant taxa up until ~3,480 BC). There is also an increase in several other planktonic taxa in the lowermost section of this zone, most notably *Skeletonema costatum*, *C. choctawatcheeana*, *P. sulcata* (tychoplanktonic), *Thalassionema nitzschoides* and *Cyclotella striata*. However, planktonic diatoms decrease again after ~3,520 BC (e.g. decrease in B:P-:T and centric:pennate ratios), largely driven by a ~20 % decrease in *Chaetoceros* cysts (down to ~15 % by 3,330 BC) but also by much smaller reductions of *T. nitzschoides*, *C. striata* and *S. costatum*. *P. sulcata* increases after ~3,540 BC and with it the DI-salinity rises, reaching 26 ‰ by ~3,160 BC. The benthic community remains relatively unchanged with *O. mutabilis* (3-11 %) still the most abundant benthic species, followed by relatively constant numbers of *G. oceanica*, *O. minuta*, *C. scutellum*, R. *acuminata* and *D. minor*.

KIL-D2 (~2,740 BC-AD 20) - *P. sulcata* is the most abundant species throughout, accounting for ~15-60 % of assemblage, and *Chaetoceros* cysts further decrease at the beginning of this zone, and thereafter fluctuate between ~1-12 %. DI-salinity generally increases throughout, with the exception of a drop between ~930-600 BC (25-29 ‰), at which point, *P. sulcata* drops to ~15 % of the assemblage. However by ~600 BC, DI-salinity is increasing again and at ~390 BC, *Cymatosira belgica* (high salinity demanding diatom) begins to increase in abundance. This is followed shortly after by *T. nitzschoides* (~310 BC) and *Delphineis minutissima* (~280 BC). *O. mutabilis* is still the most abundant

benthic species in the lower section of KIL-D2, but numbers drop after ~1,760 BC. *D. minor* numbers also drop at this time and *G. oceanica* becomes the most abundant benthic taxon in most samples between ~1430-0 BC, generally accounting for ~1-8 % of the community. *C. scutellum* (both *parva* and *scutellum* varieties) and *R. acuminata* increase in abundance in the upper section of this zone (after ~930 BC).

KIL-D3 (~**AD 20-1640**) - A large drop in salinity occurs at the onset of this zone as *P*. *sulcata* number fall dramatically and *C. belgica, D. minutissima* and *T. nitzschoides* practically disappear from the record. *Chaetoceros* cysts, *O. mutabilis* and *Cocconeis scutellum* var. *scutellum* all increase in abundance and the DI-salinity fluctuates between ~14-19 ‰ in the period ~AD 50-600. A series of brackish water benthic species also appear in the record early in this zone, most notably *Rhoicosphenia abbreviata, Epithemia turgida, C. placentula* and *Achnanthes amoena* (single peak at ~AD 260; not shown in figure 3.4a). A large peak in *Chaetoceros* cysts also occurs at ~AD 360. DI-salinity briefly rises back up to ~33 ‰ at ~AD 800, driven mainly by a large increase in *P. sulcata* (which accounts for ~54 % of the assemblage at this time). However, the salinity declines in the uppermost sample (~AD 1,430) as conditions turn brackish with *O. mutabilis* (~14 %), *Ellerbeckia arenaria* (12 %, potential oligotrophic indicator), *Chaetoceros cysts* (10%) and *Amphora pediculus* (7%) the most abundant taxa.

KIL-D4 (~AD 1640-1965) (see Figure 3.9) - Salinity is lower in this zone as brackish, nutrient-demanding taxa dominate the assemblage. DI-TN increases up until ~AD 1930, largely driven by the increasing numbers of *C. choctawatcheeana* (most abundant species between ~AD 1890-1960; 23-38 %). However, *C. choctawatcheeana* percentages are decreasing again from after the 1930s. *Chaetoceros* cysts (6-13 %), *Cyclotella meneghinina* (2-7 %) and *Thalassiosira baltica* (1-4 %) are all ever-present in the plankton. *O. mutabilis* (7-23 %) and *A. pediculus* (4-14 %) are the most abundant benthic species with smaller abundancs of *P. delicatulum, C. neothumensis, Diatoma moniliformis, C. placentula* and R. *abbreviata. O. mutabilis* increases after ~AD 1935, but most other benthic taxa exhibit reduced percentages in the post AD 1920 period (e.g. decreasing B:P:T ratio; Figure 3.9).

KIL-D5 (**~AD 1965-1996**) (see Figure 3.9) - *C. choctawatcheeana* continues to decrease, being replaced by *Chaetoceros* cysts as the dominant planktonic species at **~AD 1970**. *O*.

mutabilis is again the most abundant benthic species in KIL-D5 and continues its increase up until ~AD 1978, after which it remains relatively constant as it co-dominates the assemblage with *Chaetoceros* cysts and represents ~36 % of the assemblage. There is also a sharp change in both the centric:pennate and B:P:T ratios in the late 1980s, as in addition to the increase in *Chaetoceros* cysts, several benthic species experience a decrease, most notably *A. pediculus, C. placentula, C. neothumensis, D. moniliformis* and *R. abbreviata.* There is also a reduction in *T. baltica*, while *Staurosira construens* fo. *subsalina* and *S. construens* fo. *venter* appear in the record.

KIL-D6 (~AD 1996-2007) (see Figure 3.9) - *Chaetoceros* cysts experience a major decrease at the beginning of this zone dropping by over 20 % and almost disappearing from the record by the uppermost (most recent; AD 2007) sample. DI-TN (and the centric:pennate ratio) continues to increase suggesting further intensification of eutrophic conditions. A dramatic increase in *C. choctawatcheeana* occurs (~27 % rise), which again replaces *Chaetoceros* cysts as the main component of the plankton. *O. mutabilis* (~19-31%) progressively declines, but still remains the main benthic representative, with smaller amounts of *S. contruens* fo. *subsalina* and *P. delicatulum* also present.

Diatom Preservation

Overall diatom preservation is good for the majority of the core, with an average **F** index of 0.45 for the entire profile (range = 0.14-0.82). Despite this, there are distinct phases when preservation condition appears to either improve/deteriorate. **F** index values gradually rise (improve) between ~5,500-3,500 BC, with exceptional preservation conditions persisting between ~4,200-3,600 BC (lowermost section of KIL-D1b). This phase coincides with lower DI-salinities (Figure 3.7) and higher productivity (see below, Figure 3.10, Figure 3.11). Exceptional preservation is further supported by the substantial numbers of the weakly silicified diatom *S. costatum* between 4,100-3,650 BC) and the increase in numbers of stage 1 classified *P. sulcata* valves. The **F** index fluctuates between ~0.3-0.8 in the succeeding phase, prior to a period of poorer preservation between ~1,760 BC-AD 110 (**F** index = 0.2-0.3). This period of poorer preservation coincides with highest DI-salinities, which might be expected under fully marine conditions. **F** index values increase again in the early Roman period (after ~AD 190) and become very good (**F** index =~0.5-0.7) in the uppermost section of the profile (post ~AD 1830).

In almost all samples the majority of *P. sulcata* valves fall into dissolution stages 1 or 2, with stage 2 generally being the most common category. Two exceptions to this occur at ~930 BC and ~AD 1830, but the latter of these is based on too fewer *P. suclata* valves (5 valves) for an accurate assessment. At ~930 BC, the **F** index score (~0.7) suggests that the overall preservation in this sample is good and therefore poorer *P. sulcata* preservation appears an anomaly. This is perhaps related to dissolution in water column, specific habitat characteristics or being washed in from external sources (e.g. older sediments from the catchment or deeper waters of Limfjord/North Sea).

3.6 Sedimentary pigments

The Kilen sedimentary record is presented as concentration (nmol g/OM) in Figure 3.10 and as pigment accumulation rates in Figure 3.11. The final sedimentary pigment profile consists of 777 samples, giving a temporal resolution of ~4-58 years (average 9 years). Twelve identifiable pigments were present in the profile (see Figure 3.10), with one undifferentiated pigment (labelled carotenoid X) and three ultraviolet (UV) absorbing compounds (labelled compound A, B and C). The uppermost section of the profile is absent (~1800-present; depth 390-424 cm bpsl), due to unsuitable treatment and storage conditions of pigment samples from the HON-Kajak core (upper 35 cm of profile). Superimposed upon the major trends described below, are regular minor fluctuations in pigment abundances, which occur throughout the entire profile. Sedimentary pigment affiliation (quoted in brackets after the first mention) are based on affiliations quoted in (Leavitt and Hodgson, 2001; McGowan, 2007). Chlorophylls refer to chlorophyll *a*, chlorophyll *b*, pheophytin-*a* and pheophytin-*b*, and carotenoids refer to all other pigments excluding compounds A, B and C (UV-absorbing pigments).



Figure 3.10 - Sedimentary pigment record from Kilen expressed as concentrations; nmol g OM. Analyst: Jonathan Lewis



Figure 3.11 - Kilen sedimentary pigment record expressed as accumulation rates; nmol cm⁻² yr⁻¹.

KIL-SP1 (~5,500-4,170 BC) - At the bottom of the profile, the overall abundance of pigments is relatively high, with diatoxanthin (from diatoms, dinoflagellates and chrysophytes), lutein (higher plants, green algae, euglenophytes), β-carotene (most algae and plants), pheophytin-a (chlorophyll a derivative) and UV-absorbing compound B (primarily cyanobacteria) the most abundant pigment types. Carotenoid X and the UVabsorbing pigments are both at their highest levels for the entire profile. The majority of pigments are either stable or increasing between ~5,500-5,200 BC, but most tend to exhibit a brief fall in concentrations somewhere between ~5,200-4,800 BC. When plotted as accumulation rates, there is a particularly noticeable decrease after ~5,000 BC, but this decrease is relatively short lived and by ~4,800 BC the accumulation rates (and concentrations) of most pigments are increasing again. There are particularly notable increases in okenone (purple sulphur bacteria), chlorophyll a (higher plants, all photosynthetic algae), pheophytin-a and pheophytin-b (chlorophyll b derivative) at this time. Later in the zone (~4,500-4,400 BC) all pigments begin to increase in concentration, which culminate in a high productivity phase (including high accumulation rates) that lasts throughout the succeeding zone (i.e. KIL-SP2; see below). The UV pigments are generally present in lower abundances after ~5,060 BC, but subject to fluctuations and high concentrations in occasional samples.

KIL-SP2 (~4,170-3280 BC) - The overall concentration of sedimentary pigment are high in KIL-SP2, with almost all most pigments (except for the UV-absorbing compounds) abundant throughout, following their increases at the top of the previous zone (KIL-SP1). The UV pigments practically disappear at the onset of this zone (by ~4,000 BC), and for the remainder of the profile only reappear sporadically. Lutein, diatoxanthin and β carotene remain the most abundant pigments, with diatoxanthin present at its highest concentrations for the entire profile. The highest accumulation rates for all chlorophylls and carotenoids (for the entire profile) occur between ~4,300-3,200 BC suggesting this period is the most productive in the maritime section of the Kilen record presented here (i.e. pre-AD 1856).

KIL-SP3 (~3,280-10 BC) – The UV-absorbing pigments (compounds A and B) reappear briefly in small concentrations between 3,250-3,150 BC, but all other pigments decrease in the lowermost section of KIL-SP3 (~3,300-3,000 BC) as overall productivity decreases. Thereafter pigment concentrations remain relatively low, but subject to fluctuations

throughout, largely driven by the abundance of β -carotene (and the pheophytins). Noticeable increases in β -carotene (and overall concentration) occur between ~2,750-2,090 BC and ~1,500-900 BC, with a period of low concentration between phases (i.e. β -arotene <10 nmol g OM; ~2,090-1,500 BC). During the latter of these phases some other pigments also exhibit short term increases in concentrations, most notably diatoxanthin (~1,490-1,180 BC) and lutein (~1,360-1,180 BC). After ~500 BC all pigments begin to increase in abundance. The accumulation of pigments is low after 3,100 BC, with very minor rises visible during the periods of elevated concentrations (listed above).

KIL-SP4 – (~ **10 BC - 810 AD**) – Overall pigment concentrations are generally high, with several pigments, most notably alloxanthin (cryptophytes), lutein, canthaxanthin (colonial bacteria, herbivore tissue) exhibiting large increases in abundance near the beginning of this zone and remaining high throughout. Most other pigment gradually increase, peaking towards the middle-upper section of this zone (~AD 300-810). Diatoxanthin is an exception, exhibiting a decrease in abundance at the onset of KIL-SP4 and remaining relatively low throughout. In this zone, the pigment accumulation rates suggest that the apparent increase in productivity might be exaggerated due to a decrease in the sediment accumulation rate (Figure 3.5). The total pigment accumulation rate actually suggests only minor changes in marine productivity in this zone, with overall productivity increasing only slightly after ~AD 200. Minor increases in the accumulation of alloxanthin and canthaxanthin occur slightly earlier (~200 BC-AD 30).

KIL-SP5 (~AD 810-1800) – Almost all pigments decrease in concentration at the beginning of this zone, with the exception of canthaxanthin and pheophytin-*b*, both of which are abundant throughout. The high levels of Pheophytin-*b* and high pheophytin-*a*/chlorophyll *a* ratio suggest increased degradation of pigments. The rest of the pigments remain at lower concentrations up until ~AD 1,500 before increasing again in the uppermost section of the pigment profile (i.e. after ~AD 1800). Most notable increases occur in alloxanthin, lutein, canthaxanthin and echinenone. Diatoxanthin concentrations drop to their lowest levels for the entire profile in KIL-SP5. The accumulation rates for all pigments also drop to their lowest levels for the entire profile in this zone, particularly between ~AD 810-1,100. β -carotene, lutein and both pheophytins (*a* and *b*) exhibit the highest accumulation rates. It is important to highlight that pigment accumulation rates

must be treated with caution and may even be erroneous in this section of the record due to dating uncertainties (see section 3.3 and 3.11).

3.7 Macrofossils

A total of 611 samples were analysed for plant and animal macrofossils at Kilen at an average resolution of 12 years (max. 57.6 years, min. 2.2 years). The volume of sediment analysed ranged between $1.5-45 \text{ cm}^3$ (average 21 cm³) taken from 1-2 cm thick slices (see section 2.4). Macrofossils present included molluscs, terrestrial and aquatic plant material, ostracods, cladocerans, echinid remains, fish bones and scales, charcoal, crustacea and foraminifera. Due to the relatively low abundance of macrofossils present in the cores (particularly plant material), zonation is based on the entire macrofossil dataset (species abundance >3%) rather than being split into its different components (e.g. molluscs, plant macrofossils). Foraminifera were excluded for the zonation process due to the low number of samples analysed (15 in total). Presence/absence data (see Figure 3.12-Figure 3.15) were also removed. Ten statistically significant zones have been determined using an optimal splitting technique (see section 2.10.1).

3.7.1 Molluscs

The Kilen molluscan assemblage is presented in Figure 3.12. All species are expressed as concentration per 50 ml of wet sediment with the exception of *Ophistobranchia* spp. fragments and *Mytilus* spp. fragments, which are expressed as presence/absence data. Overall, the molluscan concentration (maximum concentration 125 per 50 ml at depth 504 cm bpsl; ~AD 1692) was low throughout the profile, with a total of 424 samples (out of 611) containing molluscan remains. A total of 33 genera were recorded, yielding 38 species (17 gastropods and 21 bi-valves).



Figure 3.12- Kilen molluscan record (as concentration per 50 ml fresh sediment, with square root transformation applied). *Ophistobranchia* fragments and *Mytilus* sp. Fragments expressed as presence/absence data. Numbers in brackets refer to minimum salinity tolerances according to Sorgenfrei, (1958). Analyst: Kaj Strand Petersen (GEUS).

KIL-MM1 (~**5,500-4,450 BC**) - *Corbula gibba* and *Spisula subtruncata* are the most abundant and consistently present species throughout. There are also fairly regular occurrences of *Hinia reticulata* and *Abra alba* and several appearances by *Bittium reticulatum* and *Rissoa albella*. The mollusc concentration is comparatively high (only exceeded in KIL-MM7), particularly in the lowermost section (up until ~5,050 BC), due to the slightly higher concentrations of *C. gibba* and *S. subtruncata*. The high salinity demanding *Tellimya ferruginosa* is also present in some of the lowermost samples, but temporarily disappears from the record around ~5,090 BC.

KIL-MM2 (~4,450-4,105 BC) - Molluscan diversity and concentration decrease as *S. subtrunctata* and *C. gibba* practically disappear from the record. *A. alba* is the only consistently present mollusc in this zone, supplemented by small and sporadic occurrences of high salinity demanding species, most notably *B. reticulatum* but with individual peaks of *Acanthocardia echinata* and *T. ferruginosa*. The latter of these achieves its highest abundance for the entire profile at ~4,440 BC.

KIL-MM3 (~4,105-2,855 BC) - Mollusc concentration and diversity increase, with *A*. *alba* almost ever-present and common occurrences of *H. reticulata*, *Mysella bidentata*, *C. gibba* and *S. subtruncata*. A distinct peak in molluscan concentration occurs at ~3,590 BC where *M. bidentata* reaches its highest levels for the entire profile. *B. reticulatum* is also present in numerous samples in this zone which together with occasional finds of *T. ferruginosa* suggests relatively high salinity. Fragments of *Ophistobranhcia* spp. (marine sea slug) also become more regular.

KIL-MM4 (~2,855-2,030 BC) - *A. alba* is again the most consistent mollusc species, continuing at similar concentrations to the previous zones (i.e. KIL-MM2 and KIL-MM3). This is supplemented by *Abra* spp. (undifferentiated to species level) particularly in the lower section of this zone (up until ~2,460 BC), after which it is only present in a few samples and only at very low concentrations. All other species present in this zone make sporadic appearances, including the first recordings of the brackish-water edible cockle *Cerastoderma edule*, in an otherwise medium-high salinity fauna.

KIL-MM5 (2,030-890 BC) - *A. alba* decreases in abundance in this zone and even disappears for a short period between ~1,600-1,180 BC. *T. ferruginosa* and *M. bidentata*

are present in several samples in the lower section of this zone and there are also occasional finds of brackish water species such as undifferentiated mussel (*Mytilus* spp.) and cockle (*C. edule* and *Cerastoderma* spp.) fragments and *C. edule* throughout. Overall, the molluscan concentration drops in this zone, particularly between (1,530-1,230 BC), where concentrations fall to their lowest pre-Viking (AD 750-1,050) levels (see KIL-MM8).

KIL-MM6 (890-10 BC) - Mollusc concentration increases again in this zone, due to increased concentrations of *A. alba* and *B. reticulatum*. Despite this, the molluscan fauna remains relatively limited, with just several samples containing undifferentiated fragments of *Cerastoderma* and *Mytilus* species and supplemented by occasional appearances from several other species. The upper section of this zone (~280-40 BC), is almost devoid of molluscs, as *B. reticulatum* and *A. alba* disappear. After ~40 BC, *S. plana* and *Retusa truncata* make their first appearances in the profile and *T. ferruginosa* reappears briefly at ~25 BC.

KIL-MM7 (10 BC- AD 650) - There is a large increase in mollusc concentration and diversity in this zone, driven by an increase/arrival of a number of species indicative of low (e.g. ~ 0-10 ‰; *Macoma baltica, M. edulis, Ceratoderma glaucum* and later *Theodoxus fluviatalis*) and medium salinity (e.g. 10-20 ‰; *Hydrobia ventrosa, Hydrobia ulvae, Littorina saxatalis, C. gibba, Scrobicularia plana, R. albella* and *Parvicardium ovale*). Molluscan concentration is variable with two distinct phases of higher concentrations (~10 BC-AD 260 and ~AD 460-650) separated by a short phase (~AD 260-460) where molluscan concentration drops very low.

KIL-MM8 (~AD 650-910) - This zone is marked by a drop in both molluscan concentration and diversity, driven by the disappearance of most of the low-medium salinity tolerant species characteristic of KIL-MM7. *B. reticulatum* becomes the most abundant species, suggesting increased salinity, supported by occurrences of *Tapes* spp. and *Gibbula tumida* (not shown in Figure 3.12). *B. reticulatum* is also the only species consistently present up until ~AD 830, after which molluscs remains almost disappear from the record. Some undifferentiated mussel fragments, however, do remain present up until ~AD 670.

KIL-MM9 (**AD** ~910-1930) - There are very few molluscs present in this zone, with just one sample (at ~AD 1400) containing some unidentifiable bi-valve fragments and another sample containing *Mytilus* (mussel) fragments (~AD 1490), but only identifiable to the genus level.

KIL-MM10 (AD 1930-2007) - This zone is completely devoid of molluscs.

3.7.2 Plant macrofossils

Plant material is generally scarce throughout the record with only 154 samples (out of 611) containing any plant material distinguishable to genus/species level (maximum concentration 171 per 50 ml). This number rises to 213 samples if samples containing undifferentiated fragments of wood, deciduous leaves, bark, buds and twigs are included. The main purpose of plant macrofossil analysis at Kilen was to find terrestrial material suitable for AMS ¹⁴C dating (for which sufficient material was found; see above and Table 3.2). However, all identifiable terrestrial and marine plant macrofossil material was recorded and the results are illustrated in Figure 3.13 and summarised below in relation to the statistically determined macrofossil zones (labelled KIL-MP1 – KIL-MP10).

KIL-MP1 (~**5,500-4,450 BC**) - Some general tree taxa remains are present belonging to *Betula, Quercus* and *Alnus* trees, and several samples contain undifferentiated deciduous leaf, bark or twig fragments. There are also some sporadic finds of moss stems and one sample containing *Juncus* seeds (~5,320 BC),

KIL-MP2 (~4,450-4,105 BC) - KII-MP2 is almost devoid of plant macrofossils, with the exception of a few levels containing deciduous leaf fragments, an individual *Alnus* female cone at ~4,420 BC and an undifferentiated bud scale fragment at ~4,290 BC.

KIL-MP3 (~4,105-2,855 BC) - *Betula* fruits are commonly present in the record after ~2,700 BC, together with a few *Betula* catkin scales. Moss stems are also sporadically present throughout and *Juncus* seeds are occasionally recorded between ~3,500-3,400 BC.



Figure 3.13 - Kilen plant macrofossil record (concentration per 50 ml fresh sediment, unless indicated as presence/absence data by black dots). Analysts: Peter Rasmussen (GEUS) and Jonathan Lewis.

KIL-MP4 (~2,855-2,030 BC) - *Betula* fruits disappear again by ~2,900 BC, and in the phase that follows, the sediment is again almost devoid of macrofossils, except for a few scattered tree remains (e.g. deciduous leaf fragments, alder fruit and female flower).

KIL-MP5 (2,030-890 BC) - Macrofossil concentration remains low in both the lower (~2,030-1,080 BC) and uppermost (after ~1,530 BC) sections, with just occasional *Betula* fruits present throughout and a few moss stems at ~1,045 BC. There is, however, a slight increase in plant macrofossils between ~1,530-1,080 BC as deciduous leaf fragments become more common, and some remains from wet ground taxa (*Juncus* seeds and moss stems) appear in the record.

KIL-MP6 (890-10 BC) - Throughout the majority of this zone, just a few scattered finds of deciduous leaf fragments are recorded. At the very top (after ~75 BC), some swamp/reed taxa appear, most notably *Sphagnum* leaves and *Juncus* seeds.

KIL-MP7 (10 BC- AD 645) - Small amounts of swamp/reed taxa also continue to appear in the record early in KII-MP7 with finds of *Scirpus* spp. (*maritimus/lacustris*) seeds and cf. *Rumex maritimus* fragments. However, the most pronounced change in this zone is the large expanse of fruits from the aquatic horned pondweed *Zannichellia palustris*, This expanse occurs at the onset of this zone and is regularly present (and at fairly consistent concentrations; generally 3-8 per 50 ml, max. 18.75 at AD ~260) up until ~AD 270. Above this (i.e. after ~AD 270), it becomes more sporadic, causing the overall plant macrofossil concentration to drop briefly.

KIL-MP8 (**~AD 645-910**) - *Z. palustris* disappears after the first sample in KIL-MP8 (**~AD 650**), leaving this zone containing just several recordings of *Betula* fruits (**~AD 795-**825) and some individual remains of several taxa (including *Cyperaceae* seed, *Ranunculus sceleratus* nutlet, moss stems, *Potamogeton* fruitstone and *Chara* spp. oospores). Some undifferentiated leaf and bud remains are also present.

KIL-MP9 (AD ~910-1930) - *Chara* spp. oospores appear at ~ AD 940 and fluctuate thereafter and *Z. palustris* is recorded several times in the lowermost section (~AD 1000-1320). After ~AD 1830, wetland taxa increase in abundance, most notably *Juncus* spp. seeds at ~AD 1830 and slightly later *Ranunculus* sect. *Batrachium* (water crowfoot) seeds

and moss stems (both at ~1860 AD). The overall plant macrofossil concentration also increases at the top of this zone (i.e. after ~AD 1830).

KIL-MP10 (**AD 1930-2007**) - *Chara* spp. oospores remain relatively abundant in the early part of KIL-MP10 (see below) but disappear from the record after ~AD 1970. Wet ground, swamp and reed taxa continue to increase in abundance, with *Juncus* spp., moss stems and *Sphagnum* leaves fairly regularly recorded throughout. *Betula* fruits are present in a few of the uppermost samples (after ~AD 1994) and deciduous leaf fragments become more regular between ~AD 1980 and 1994. Occasional seeds from the macrophyte *Ranaunculus* sect. *Batrachium* continue up until ~AD 1983.

3.7.3 Foraminifera

The foraminifera assemblages present in 15 samples from Kilen are presented in Figure 3.14. The results are displayed as percentages with a minimum of 300 foraminifera counted per sample (by Peter Kristensen, Aarhus University). Fourteen species from seven genera were identified in total.

KIL-MF1 (~5,500-4,450 BC, 5 samples) - *Elphidium incertum* (~24-28 %) is initially high but decreases in abundance (down to ~8-12 %) after ~5,250 BC. *Haynesina germanica* numbers generally fluctuate (~5-28 %), briefly being the most abundant species at ~5,250 BC. *Ammonia beccarii* and *Elphidium excavatum* fo. *selseyensis* gradually increase throughout reaching ~46 % and 27 % respectively by the uppermost sample (~4,510 BC). *Elphidium magellanicum* is also initially quite abundant (~17 %), but briefly drops to ~2 % at ~5,250 BC, before fluctuating between ~5-11 % thereafter. Foraminfera diversity is greatest in this lowermost zone.

KIL-MF2 (~4,450-4,105 BC, 1 sample) - High numbers of *E. incertum* (~32 %), *E. excavatum* (~28 %) and *H. germanica* (~28 %) indicate high salinity at ~4,210 BC. *A. beccarii* accounts for just 6 % of the foraminiferal assemblage in this sample.



Figure 3.14 – Kilen percentage for a minifera record and $\delta^{18}O_{calcite}$ and $\delta^{13}C_{calcite}$ isotopes based on test of the benthic for a miniferal species *Elphidium excavatum* fo. *selseyensis*. (Analysts – For a minifera Peter Kristensen Aarhus University, Isotopes: Jonathan Lewis)

KIL-MF3 (~4,105-2,855 BC, 4 samples) - At ~3,920 BC (lowermost sample of KIL-MF3) *A. beccarii* is the dominant species accounting for ~60 % of the assemblage. *E. excavatum* (26-66 %) and *H. germanica* (8-44 %) are the next most abundant species, briefly replacing *A. beccarii* between 3,650-3,400 BC, which drops to less than 4 % of the assemblage. There are also minor increases of *E. incertum* and *Elphidium magellanicum* in this period. However, by ~3,150 BC (upper sample), *A. beccarii* (~38 %) has recovered, and again co-dominates the assemblage (with *E. excavatum*; 30 % and *H. germanica*; 26 %).

KIL-MF4 (~2,855-2,030 BC, 2 samples) – The lowermost sample (~2,680 BC) is dominated by a mixture of *A. beccarii* (~23 %), *H. germanica* (~20 %) and *E. excavatum* (~36 %), but with elevated percentages of *E. incertum* (14 %) and *E. magellanicum* (6 %). *A. beccarii* (~35 %) and *E. excavatum* (52 %) increase substantially in the upper sample, as *H. germanica* decreases to 11 % of the assemblage and *E. magellanicum* and *E. incertum* almost disappear from the record (both <1 %).

KIL-MF5 (2,030-890 BC, 2 samples) - At ~1,590 BC (lower sample) *A. beccarii* completely dominates, accounting for ~80 % of the assemblage. *E. excavatum* is the next most abundant species in both samples, but exhibits a decrease of ~8 % between the two (i.e. between ~1,585-930 BC), whilst *H. germanica* numbers are initially very low (2 % at ~1,585 BC), but increases slightly in the upper sample (reaching ~5 % at ~930 BC).

KIL-MF6 (890-10 BC, 1 sample) - *A. beccarii* (~65 %) is still the dominant species (at ~220 BC), followed by *E. excavatum* (17 %) and *H. germanica* (9 %). There is also an increase in *Elphidium williamsonii* (~5 %), whereas *E. incertum* disappears from the record.

No foraminifera samples have been counted in macrofossil zones 7-10. However, a preliminary assessment of 138 samples (foraminifera presence/absence recorded) from these zones indicates a series of phases in KIL-MF7 where foraminifera are absent. These phases occur between ~AD 12-100, ~AD 165-225 and ~AD 315-520. Towards the middle of KIL-MF8 (after ~AD 755), foraminifera disappear and remain absent throughout KIL-MF9 and KIL-MF10.

3.8 Isotopes from *Elphidium excavatum* fo. *selseyensis*

The $\delta^{18}O_{\text{calcite}}$ and $\delta^{13}C_{\text{calcite}}$ results determined from tests of the benthic foraminifer *Elphidium excavatum* fo. *selseyensis* are illustrated in Figure 3.14. In total 44 measurements were made for both $\delta^{18}O_{\text{calcite}}$ and $\delta^{13}C_{\text{calcite}}$, covering the period ~4,670-390 BC. Isotopic values are uncorrected for any vital effect that may take place during synthesis of the foraminiferal test, which can skew the isotopic composition out of equilibrium with the surrounding water (Grossman, 1987). It is assumed that the vital effect for any single species will be consistent and stationary over the timescale investigated in this study (Kristensen and Knudsen, 2006).

The $\delta^{18}O_{\text{calcite}}$ isotopic values are highest at the bottom of the profile and exhibit an erratic but overall decreasing trend up until ~2,740 BC, falling by ~1.3 ‰ over this period. Soon after ~2,740 BC, the isotopic values increase again and thereafter fluctuate between ~ 0.8-1.2 ‰ up until ~810 BC, before experiencing a large drop in the uppermost isotope sample (390 BC). In this setting, salinity, rather than temperature is likely to be the most important factor influencing the δ^{18} O water composition of the seawater (K. Knudsen pers. comm.). This is due to the much larger magnitude shifts in salinity over the study period, compared to temperature (e.g. salinity gradient from brackish to fully marine; Kristensen et al., 1995; Christensen et al., 2004, temperature changes ~3°C over the entire Kilen profile; Antonsson and Seppä, 2007). Whilst the modern calcite-salinity and calcite-temperatures relationships are unknown for Elphidium excavatum fo. selsevensis in the Limfjord, estimates on modern periwinkle shells (Littorina littorea) exhibit a 0.25 ‰ change for every salinity unit (i.e. 1 ‰) and a 0.22 ‰ per °C change in temperature (Burman and Schmitz, 2005). Therefore the $\delta^{18}O_{calcite}$ profile suggests that salinity decreases up until ~2,740 BC (i.e. less negative values infer higher salinity), prior to a relatively sharp increase at ~2,680 BC, and thereafter remains relatively constant for most of the upper section of the isotope record (~2,740-810 BC).

The $\delta^{13}C_{\text{calcite}}$ isotopic curve also exhibits an erratic pattern in the lower section of the record, with large fluctuations up until ~3,600 BC. In this phase, values generally fluctuate between -1 to -2.5 ‰. Most of the lower $\delta^{13}C_{\text{calcite}}$ values (i.e. < -2 ‰ occur in individual samples (e.g. 3,630 BC, 3,710 BC and 3,980 BC), with the exception of a distinct phase between 4,510-4,330 BC. This ensues after a ~1.5 ‰ decrease in the $\delta^{13}C_{\text{calcite}}$ isotopic

composition shortly after ~4,670 BC (i.e. immediately above the lowermost isotope sample which yielded a $\delta^{13}C_{calcite}$ isotopic ratio of -1.09 ‰). The $\delta^{13}C_{calcite}$ isotopic composition becomes more stable for a brief period between 3,600-3,155 BC, prior to a ~1.5 ‰ drop at ~3,040 BC, where it falls to its lowest value for the entire profile. However, this minimum appears to be very short lived, and in the sample immediately above (i.e. by ~2,850 BC), isotopic values increase again, a trend that largely continues until ~2,100 BC. Above this, the sampling resolution is very low, but it appears that the $\delta^{13}C_{calcite}$ values remain relatively high (between -1.2 and -0.4 ‰) for the upper part of the isotopic record (i.e. ~390 BC).

3.9 Other organisms

Any additional remains present in either the diatom slides or wet sieved samples were recorded and results are illustrated in Figure 3.15. In numerous diatom slides, small numbers of chrysophytes were found as well as occasional recordings of the marine silicoflagellate *Distephanus speculum* and the ebridian species *Ebria tripartita*. Additional macro-organisms (present in the wet sieved samples) include ostracods, fish (bones and scales), echinoids (spines), barnacles (*Balanus* spp.) and zooplankton remains (*Daphnia* spp. ephippia).

Chrysophytes, expressed as concentrations, after methods described in Battarbee and Kneen, (1982), are present for much of the profile appearing in 62 of the 110 diatom samples. Highest concentrations are found in the uppermost, more bracksih section of the profile (after ~AD 1930), but two other phases of apparent elevated concentration occur between ~4,150-3,330 BC and in the late pre-Roman/early-Roman period (~280 BC-AD 260). *Distephanus speculum* is regularly present up until ~2, 380 BC, with the exception of a ~600 year disappearance between ~3,330-2,740 BC. After ~2,380 BC, it is only recorded four more times, all of which occur between ~310 BC-AD 600.



Figure 3.15 - Additional organisms present at Kilen. Ostracods and *Daphnia* spp. (Cladocera) are expressed as concentration data per 50 ml fresh sediment. Chrysophyte concentrations are expressed as number per gram of dry weight. All other organisms are expressed as presence/absence data.

Ostracods (expressed as concentrations per 50 ml wet sediment in Figure 3.15) are consistently present in the record up until ~AD 1030. There are four apparent phases of slightly elevated concentrations; 5,500-4,550 BC, 3,910-3,400 BC, 2,170-1,580 and ~65 BC-260 AD, with highest concentration occurring during the latter of these phases. It must, however, be noted that even during these four phases, absolute numbers of ostracods are still relatively low (maximum 32 specimens per sample), and therefore results must be treated with caution. Furthermore, without identification to species level it is impossible to derive any information regarding salinity (and/or temperature) from just their presence within the profile (as species differ widely in their salinity and temperature requirements). Nonetheless, in conjunction with the other proxies, they may be able to offer some insight into calcite preservation and/or aquatic productivity.

Like ostracods, undifferentiated fish bones provide little information concerning the environment and are always low in abundance, but again can potentially provide some insight into changing calcite preservation conditions. Periods where fish bones are more regular include ~5,500-5,170 BC, ~4,150-3,330, ~2,250-1,725 BC, ~120 BC-AD 1085 and ~AD 1720-1990. Echinoid remains are fairly regular up until ~AD 240, after which they disappear from the record. *Daphnia* spp. ephippia (zooplankton) only appear in the uppermost section of the record, first appearing at low concentrations around ~AD 1600. They exhibit a substantial increase in the early 1930s (KIL-MP10) and thereafter remain both common and relatively abundant up until ~AD 1996. *Daphnia* spp. eggs disappear again in the very recent past (i.e. post-AD 1996).

3.10 Statistical analyses: DCA

Major trends in the diatom dataset were explored using detrended correspondance analysis (DCA) and principal component analysis (PCA) using Canoco, v4.5 (ter Braak and Smilauer, 2002). For the entire diatom dataset (i.e. 110 samples), DCA (unimodal) was deemed the most appropriate technique, due to the axis 1 gradient length exceeding 2 SD units (Table 3.3). The first DCA axis is responsible for explaining 26.8 % of the variation in the Kilen diatom dataset, with very little variance in comparison being explained by any of the additional axes (Table 3.3). DCA axis 1 appears to be driven by salinity, which is perhaps expected for this site when considering the substantial shifts in salinity evident over the last \sim 7,500 years. The majority of species appear to be strongly associated with axis 1, with those present in the uppermost (limnic) section of the profile exhibiting the lowest (i.e. most negative) score on DCA axis 1. Most notably (i.e. >3 % abundance in one or more slides) these include Cyclotella choctawatcheeana, Cyclotella meneghiniana, Ctenophora pulchella, Thalassiosira baltica and Rhoicosphenia abbreviata. At the other end of the gradient (i.e. positive values along axis 1) are the high-salinity tolerant species such as Cymatosira belgica, Delphineis minutissima, Actinoptychus senarius and Paralia sulcata.

DCA axis 1 sample scores are illustrated in Figure 3.8 and Figure 3.9 as they provide a summary of major compositional shifts in the diatom assemblage throughout the profile. The shifts in the DCA axis 1 sample scores tie in with the major salinity shifts from the DI-reconstruction. This adds further support for salinity as a major factor driving the

change in the diatom assemblages. Only minor compositional changes occur in the lower section (~5,500-2,740 BC) of the profile, but after ~2,750 BC, DCA sample scores begin to increase as higher-salinity demanding species begin to dominate the profile. Highest DCA axis 1 sample scores generally occur between ~400-40 BC, prior to the major drop at the beginning the Roman period (KIL-D3; ~AD 20). A brief return to higher scores occurs at ~AD 800, where *P. sulcata* dominates the profile again, before dropping down to lowest sample scores for the entire profile in the uppermost, limnic section of the record.

Dataset	No. of samples	No. of species	Total inertia	Axis	1	2	3	4	Technique
Diatoms	110	516	2.026	Eigenvalues	0.543	0.076	0.056	0.042	DCA
(%)				Length of gradient	3.028	1.791	1.223	1.062	
				% variance explained	26.8	3.7	2.8	2.1	
	27	169		Eigenvalues	0.379	0.137	0.069	0.054	PCA
				% variance explained	37.9	13.7	6.9	5.3	
	82	477	1.345	Eigenvalues	0.208	0.115	0.047	0.039	DCA
				Length of gradient	1.956	1.555	1.005	1.032	
				% variance explained	15.4	8.6	3.5	2.9	

Table 3.3 - Summary of DCA and PCA results for the Kilen diatom dataset

As the whole-core variation is driven by the contrast between two very different aquatic ecosystems (limnic and marine, the diatom dataset was subsequently split into two smaller datasets to further explore the major trends within both the limnic and marine phases at Kilen. The limnic phase (i.e. post AD 1856) is most likely to be represented by the uppermost 27 slides (see section 3.3 and 3.11 for dating uncertainties) and a preliminary DCA on this reduced dataset suggested linear analyses (i.e. axis 1 = <1.5) were the more appropriate. Following PCA analysis (applying Aitchison's log ratio analysis; Aitchison, 1986), axis 1 explained 37.9 % of the variation with a further 13.7 % explainable by axis 2. PCA axis 1 sample scores (Figure 3.9) exhibit an overall increasing trend, which accelerates in the 1970s/early 1980s, resulting in highest overall scores occurring at the very top of the profile (~AD 1985-2007). The major gradient in the upper section of the profile appears to be predominately driven by nutrients, but possibly also influenced by water depth, sediment type and the abundance of macrophytes.

Species scoring lowest (i.e. most negative) on PCA axis 1 are mainly benthic species, which are most abundant in the pre-1920 period. These include *Amphora pediculus, Cocconeis neothumensis, C. placentula* agg., *R. curvata, D monoliformis, Ctenophora pulchella* and *E. sorex* (Table 3.4). These species gradually decline between ~AD 1920-1980 and almost disappear from the record after ~AD 1985, likely due to increasing salinity and/or increased nutrient status (as indicated by increasing salinity, nitrogen and phosphours concentrations; Ringkjøbning Amtskommune, 1988). The planktonic diatom, *Thalassiosira baltica* provides an exception to this by scoring negatively on PCA axis 1, but exhibits a similar trend to the low scoring benthics and practically disappears from the record at ~AD 1970, possibly being outcompeted by *Chaetoceros* spp. (24-53 % between ~AD 1980-1994).

Species scoring highest on PCA axis 1 are generally more brackish diatoms, most notably O. mutabilis and S. construens fo. subsalina, but to a lesser extent Chaetoceros resting cysts. These diatoms reach their maximum abundances between ~AD 1980-1994 (particularly ~AD 1988-1994 for *Chaetoceros* cysts) following the substantial reduction in both the abundance and diversity (e.g. reduction in Hills N₂; Figure 3.9) of the benthic community. The diatom-salinity model performs poorly in this section of the record (see Section 3.11), but salinity data measured between AD 1972-1987 shows that salinity increased after AD 1984, due to increased water exchange (i.e. greater input of higher salinity water) through the Kilerkanal, following the installation of a new sluice system (Ringkjøbing Amtskommune, 1988). The disappearance of diatoms such as C. neothumensis and C. placentula agg., both of which are commonly present throughout the marine phase of the profile, suggests that PCA axis 1 does not represent a salinity gradient. The new sluice system (built in AD 1984) increased phosphorus-release from the sediments following marine water input, causing regular decreases in oxygen concentration of the bottom waters (Ringkjøbing Amtskommune, 1988). Furthermore, DI-TN continued to gradually increase (as does the planktonic: benthic and centric: pennate ratios) between ~AD 1970-2000, which combined with elevated phosphorus concentrations, probably forced the reduction in the benthic community, by enabling planktonic diatoms (and other organisms) to bloom, resulting in increased water turbidity and reduced light penetration (Cederwell and Elmgren, 1990; Bonsdorff et al., 1991; Cooper, 1995a; Clarke et al., 2003; Ellegaard et al., 2006). This might suggest a degree of co-variation between nutrients and salinity, with increasing nutrient status responsible for

the elevated abundance of planktonic diatoms (and supressed benthic community) and *Chaetoceros* spp. perhaps outcompeting *C. choctawatcheeana* as the dominant component of the phytoplankton under higher salinities (i.e. between ~AD 1970-1994).

Increased numbers of *O. mutabilis* (and *S. construens* fo. *subsalina*) might also result from sandier sediments and/or decreased benthic competition following a reduction in the benthic community after ~AD 1980. Other benthic plants (e.g. submerged macrophytes) might also have been affected by increased numbers of phytoplankton (i.e. being shaded out), leading to a reduction in the availability of habitats for some species of diatom (e.g. epiphytic diatoms such as *R. abbreviata* and *C. placentula*). Furthermore, it is also clear that *O. mutabilis* has an affinity for, or at least is able to tolerate relatively high nutrient conditions being relatively abundant (7-40 %) in all slides in the upper section of the Kilen diatom record, again supporting co-variation between salinity and nutrient concentration in the upper section.

Taxa	PCA axis 1
Opephora mutabilis	0.82
Staurosira construens fo. subsalina	0.64
Chaetoceros resting cysts	0.29
Cyclotella choctawatcheeana	-0.01
Cyclotella meneghiniana	-0.17
Planothidium delicatulum agg.	-0.36
Epithemia sorex	-0.62
Ctenophora pulchella	-0.64
Diatoma moniliformis	-0.72
Rhoicosphenia curvata	-0.79
Thalassiosira baltica	-0.79
Cocconeis placentula agg.	-0.84
Cocconeis neothumensis	-0.85
Amphora pediculus	-0.96

Table 3.4 - PCA axis 1 species scores for taxa >3 % abundant in one or more slides in the uppermost section of the Kilen diatom record (27 samples; post AD 1856).

For the reduced dataset, spanning the marine section of the profile (i.e. 86 samples between ~5,500 BC-AD 800), species scoring lowest (most negative) are the high salinity species (*C. belgica, D. minutissima, P. sulcata* and *A. senarius*), which are particularly abundant between ~400-0 BC. The majority of other species (present at >3 % abundance) score positively, with absolute highest scores exhibited by species (*Chaetoceros* cysts, *Bacillaria paxillifer, S. costatum, Achnanthes amoena*) which are most abundant during the lower DI-salinity phases (e.g. KIL-D1b ~AD 50-600). This again suggests that salinity

is the most important variable driving DCA axis 1 and the associated variation in the diatom assemblage. Inclusion of floristically unusual the sample at depth 437 cm bpsl (~AD 1,430) does not affect the main gradient of change, with salinity still being the main driving factor. Species abundant in this particular sample (*E. areanaria*, *E. frickei* and *A. pediculus*), however, appear to score very high on axis 1.

The DCA sample scores (Figure 3.7) again provide a good summary of the major compositional shifts in the diatom record, with highest DCA sample scores associated with lower salinity periods, and sample scores decreasing as salinity increases. Sample scores fluctuate (between 1 and 2) at the bottom of the record, before decreasing after ~3,500 BC, driven mainly by the increase in higher salinity diatoms, but also by the decrease in planktonic diatoms (most notably *Chaetoceros* cysts) up until ~2,700 BC. Sample scores generally decrease up until the pre-Roman period (~500-0 BC), at which point almost fully marine conditions are inferred by the diatom record (i.e. highest DI-salinities; Figure 3.7) and DCA sample scores reach their lowest values. A shift back to higher scores occurs in the Roman/early Germanic period (~AD 0-600), where DI-salinity suggests more brackish-marine conditions (i.e. 14-19 ‰), before another low score in the sample at ~AD 800 (DI-salinity of ~33 ‰).

DCA axis 2 species scores are also provided in Table 3.5, and habitat preferences stated in Snoeijs *et al.*, 1993-1998, Sabbe and Vyverman, 1995, Hasle and Syvertsen, 1996 are given. Highest sample scores are generally recorded between ~5,500-0 BC prior to a major drop at ~0 AD/BC, after which low values prevail up until ~AD 600. This axis explains 8.6 % of the variation in the diatom dataset and is most likely driven by substrate and habitat preference, with epiphytic species being most abundant between ~AD 0-600 BC scoring lowest. This period coincides with the regular appearance of *Z. palustris* in the plant macrofossil record (Figure 3.13), likely indicating increased presence of macrophytes providing greater habitat availability for epiphytic diatoms. For taxa >3 % abundant, pelagic or sediment-dwelling (e.g. epipsammic, epilithic and epipelic) diatoms score highest (see Table 3.5), but the majority of taxa also appear to score positively on DCA axis 2. This is probably due to reduced macrophyte populations during the higher salinity (more marine) phase of the Kilen profile (i.e. ~5,500-0 BC), which likely favoured sediment-dwelling (benthic) diatoms.

Таха	DCA axis 1	Taxa , habitat	DCA axis 2
Chaetoceros resting cysts agg.	2.66	Skeletonema costatum, pelagic	3.57
Melosira moniliformis	2.57	Opephora minuta, epipsammic, Sabbe and Vyverman, 1995	2.50
Bacillaria paxillifer	2.33	Dimmergramma minor, in or on sediments	2.49
Epithemia turgida var westermanii	2.31	Plagiogramma staurophorum, in or on sediments	2.17
Staurosirella pinnata var pinnata	2.15	Bacillaria paxillifer, epipelic and epilthic	2.16
Skeletonema costatum	2.10	Thalassionema nitzschoides, pelagic	2.14
Achnanthes amoena	2.07	Cyclotella striata, pelagic, Hasle and Syvertsen, 1996	2.08
Opephora mutabilis	1.77	Fallacia sp. 1, unknown, probably epipelic	1.96
Dimmergramma minor	1.72	Rhopalodia acuminata, epipelic and epilithic	1.94
Navicula cf. oestrupei	1.70	Staurosirella pinnata var. pinnata, epiphytic, Hendey, 1964	1.85
Cyclotella striata	1.61	Thalassiosira decipiens, pelagic, Hasle and Syvertsen, 1996	1.67
Cyclotella choctawatcheeana	1.59	Paralia sulcata, in or on sediments or pelagic	1.62
<i>Fallacia</i> sp. 1	1.58	Cyclotella choctawatcheeana, perlagic, sometime epilithic	1.62
Staurosira elliptica agg. MOLTEN	1.47	Fragilaria gedanensis, epipsammic	1.51
Rhoicosphenia abbreviata	1.43	Chaetoceros cyst agg., pelagic	0.97
Thalassiosira decipiens	1.37	Grammatophora oceanica, epiphytic	0.94
Cocconeis placentula agg.	1.31	Actinoptychus senarius, in and on sediments, occasionally in plankton	0.92
Thalassionema nitzschoides	1.20	Cocconeis neothumensis, epipsammic	0.39
Rhopalodia acuminata	1.13	Staurosira elliptica agg. MOLTEN, epipsammic	0.32
Cocconeis scutellum var scutellum	1.08	Opephora mutabilis, epipsammic	0.25
Plagiogramma staurophorum	1.04	Cymatosira belgica, epipsammic, Hendey, 1964	0.24
Cocconeis neothumensis	1.02	Delphineis minutissima, epipsammic, Hendey, 1964	0.21
Grammatophora cf oceanica	0.95	Amphora helenensis, epipelic	-0.02
Opephora minuta	0.90	Cocconeis scutellum var. parva, epiphytic	-0.04
Amphora helenensis	0.85	Cocconeis scutellum var. scutellum, epiphytic	-0.64
Cocconeis scutellum var parva	0.73	Navicula perminuta, epilithic	-0.69
Navicula perminuta	0.58	Cocconeis placentula agg., epiphytic or epilithic	-1.38
Fragilaria gedanensis	0.49	Rhoicosphenia abbreviata, epiphytic or epilithic	-1.72
Paralia sulcata	-0.27	Navicula cf. oestrupei, unknown, probably epilithic	-2.18
Actinoptychus senarius	-0.42	Epithemia turgida var. westermanii, epiphytic or epilithic	-2.32
Delphineis minutissima	-0.86	Achnanthes amoena, epiphytic or epilithic	-2.75
Cymatosira belgica	-1.38	Melosira moniliformis, epiphytic or epilithic	-3.52

Table 3.5 - DCA axis 1 species scores for taxa >3 % abundant in one or more slides in the uppermost section of the Kilen diatom record (82 samples; ~5,430 BC-AD 800). Habitat preferences are based on Snoeijs *et al.*, 1993-1998, unless otherwise referenced.

If the dataset is further reduced to exclude the brackish water phase between ~AD 50-600 (75 samples), DCA axis 2 appears to be either driven by nutrient status or productivity. Planktonic species more abundant in KIL-D1b (e.g. *Skeletonema costatum, C. choctawatcheeana, C. striata, Rhizolsolenia* spp.; ~4,250-2,740 BC) generally score lowest, with benthic taxa, more abundant in other parts of the sequence scoring highest (most notably *C. belgica, D. minutissima, C. scutellum, D. minor* and *G. oceanica*). Despite excluding the major salinity shift after ~0 AD/BC, salinity still drives DCA axis 1.

3.11 Synthesis of results and site discussion

This section provides a synthesis of results derived from the independent proxy records analysed for the Kilen sedimentary profile (described above) and a site discussion, assessing how the marine/limnic environment has changed over the last ~7,500 years at Kilen. A summary of all proxy results is provided in Figure 3.16 plotted against regional environmental parameters such as sea level (Behre, 2007) and temperature (Antonsson and Seppä, 2007). Based on relatively good agreement between the independent proxy zones (see Figure 3.16), the Kilen profile has been split up into five environmental phases (labelled KIL-E1-KIL-E5). Environmental phase markers have been positioned in places where two or more independent proxy zone markers (determined via optimal splitting) fall within close temporal proximity to one another. The environmental history of Kilen is discussed below in relation to these five phases.

KIL-E1 (~**5,500-4,200 BC)** - *The late Atlantic phase* - In the lowermost phase, DIsalinities (range 18-29 ‰, average 24 ‰) are generally lower than the present day salinity of the Limfjord (average 25 ‰; 27 ‰ for the Struer region), which suggests a reduced connection with the North Sea and/or the Skagerrak. This might be considered contradictory to other studies (e.g. Petersen, 1981; Andersen, 1992) which suggest the Limfjord was more open during the late Atlantic period, due to higher sea levels (4-6 m; Petersen, 1981). This is believed to have created entrances to both the North Sea, in the west, and to the Skagerrak, in the north (Figure 3.17b).



Figure 3.16 - Summary diagram of the major changes occurring at Kilen (~5,500 BC-AD 2007). Regional (pollen-based) temperature curve from Lake Trehörningen (Antonsson and Seppä, 2007) and (mean high water) sea level curve from the southern North Sea (Behre, 2007). *DCA axis 1 reduced dataset sample scores excludes the uppermost 28 diatom samples (see text for details). †m NN asl = metres relative to German ordnance datum above present sea level.



Figure 3.17 - A. Palaeo-shorelines of the North Sea (in years BC) as proposed by Behre, (2007). B. Map of Denmark and the western Baltic during the late Atlantic Ertebølle period (~5,400-3,900 BC) indicating present shoreline (dotted line) and proposed palaeo-shorlines (solid line). Map re drawn and modified from Enghoff *et al.*, (2007) after Hellesen and Tuxen, (1988). Note the more open Limfjord with openings to both the north and west.

The history of the Limfjord, in terms of the timings of former opening and closings of these entrances, is poorly constrained, particularly before ~AD 1200. It is believed that the northern entrance remained open until sometime between ~500-0 BC (Petersen, 1976, 1981; Kristensen *et al.*, 1995; Christensen *et al.*, 2004; discussed in more detail below). According to Burman and Schmitz, (2005) this resulted in almost fully marine salinity (30-33 ‰) in the central and northern areas of the Limfjord during the late Mesolithic. Based on foraminiferal evidence from Vilsund, Andersen, (1992) suggested that the west was open ~8,000 ¹⁴C years BP and remained open up until late pre-Roman times (0 AD/BC). The Vilsund site is situated almost equidistant from both openings, probably making it unsuitable for isolation of the predominant marine inputs during the Holocene. Furthermore, all these studies are based on molluscan or foraminferal analyses, dated using marine shells, and therefore most likely subject to chronological problems due to the site and time-variable marine reservoir effect (Heier-Nielsen *et al.*, 1995; Olsen *et al.*, 2009).

Today, a salinity of ~27 ‰ exists in the Struer region due to a narrow opening (~1 km wide) to the North Sea in the western part of the Limfjord (Figure 2.2). For salinities to be lower than ~27 ‰ in the Struer area between ~5,400-2,800 BC, it is likely that the western part the Limfjord was still either totally closed off, or at least only connected to the North Sea via a narrow connection, perhaps located even further north than its current latitude (e.g. Figure

3.17). The emergent Jutland bank (Leth, 1996), might have largely blocked off the western entrance to the North Sea, prior to its drowning sometime around ~4,200 BC (~6,200 yrs BP; Leth, 1996; Gyllencreutz, 2005). The fluctuations in salinity in this early phase (e.g. higher salinity between ~4,960-4,830 BC) are probably related to short term variations in sea level, varying degrees of exposure with the Skagerrak and North Sea and/or variable changes in the strength of the North Sea and Skagerrak current systems.

More localised factors (e.g. precipitation, surface run-off, vegetation change) might also be important for explaining fresher conditions than today before ~2,800 BC. Unfortunately, the absence of a full pollen record (and poor plant macrofossil record) means information concerning changes in the terrestrial environment is limited. More localised sediment banks or beach ridges might have reduced Kilen's connection to the Limfjord during this early period in this section, though no evidence exists for this to date. It is also possible that the salinity model is under-estimating salinity in the lowermost section of the record. The representation of fossil taxa in the model is generally low (40-66 %) between ~5,500-2,800 BC, due to both the high diatom diversity (~70-114 species per sample and high Hills N₂ diversity; Figure 3.8) and relatively low number of training set sites at the high end of the salinity gradient. The high numbers of *Chaetoceros* cysts present between ~5,500-3,480 (8-43 %) are also partly responsible for the low species representation. Chaetoceros spp. are predominately a marine genus, but include species that can also live in low salinity waters (Rines and Hargraves, 1988). They are present in all parts of the Kilen profile, including the limnic phase, are unfortunately extremely difficult to differentiate between species and therefore have limited values as salinity indicators.

The common occurrence of high salinity-demanding species in the molluscan record (Figure 3.12) and foraminiferal record (Figure 3.14) might also indicate higher salinity than the diatom inferences suggest. *Bittium reticulatum* and *Tellimya ferruginosa* are present in the molluscan record throughout this phase (i.e. ~5,500-2,800 BC), supplemented by occasional appearances by *Rissoa parva* and *Achanthocardia echinata*. In the foraminiferal record, the presence of *E. incertum* (8-32 %) *E. magellanicum* (5-17 %) and *Haynesina depressula* (2-4 %) suggest salinities of at least 25 ‰ (e.g. Alve and Murray, 1999), particularly before ~4,200 BC. However, in both cases the most abundant taxa are generally medium-high salinity demanding species (i.e. *A. alba, M. bidentata, C. gibba, S. subtruncata* for molluscs, *A. beccarii, E. excavatum* fo. *selseyensis* and *H. germanica* for foraminifera). This might

suggest the higher salinity-demanding species (listed above) were living in the more exposed outer reaches of the fjord. The isotope record also suggests that salinity was actually highest in this lowermost phase, and decreasing thereafter up until ~2,740 BC. There might, however, be some dissolution issues affecting the isotopic signal, particularly after ~3,200 BC (see below; KIL-E3).

Despite the above possibilities, the bulk of evidence suggests lower salinity than today in this section of the record (i.e. between ~5,500-2,700 BC) due to a likely narrow more northerly opening or complete closure of the west at this time (e.g. Figure 3.17). The relatively low percentages of the heavily silicified marine diatom *Paralia sulcata*, which is often over-represented in coastal sediments (Ryves *et al.*, 2004) might also support this. Relative abundances of this diatom appear to increase markedly during other phases of high salinity in the Kilen profile (e.g. ~2,700-0 BC). Furthermore, DI-salinities appear to agree with other published records (e.g. Kristensen *et al.*, 1995; Christensen *et al.*, 2004) in the uppermost section of the profile (~2,500-1,050 BC), which suggests that the diatom salinity model is accurately predicting salinity change. However, our understanding of the history and timings of marine inputs into the Limfjord will remain fragmentary until long-term, well dated records are available from the northern Limfjord.

High pigment concentrations, accumulation rates and organic matter content between ~5,500-5,000 BC suggest productivity is relatively high at the bottom of the profile. The slightly elevated pheophytin-*a*/chlorophyll *a* ratio might also suggest higher rates of degradation of pigments and therefore productivity might even be under-estimated at the very bottom of the profile. There is, however, a brief decrease in pigment abundances and accumulation rates between ~5,000-4,800 BC, which coincides with increased DI-salinities (28-29 ‰; see above). A decrease in the overall accumulation rate and the organic, carbonate and minerogenic flux rates also occurs between ~5,100-5,000 BC. This suggests that this site is more productive and accumulation of sediments is faster when there is reduced exposure to the Skagerrak and North Sea, but with a regular input of marine water. Linkages between sedimentation, productivity, salinity and the configuration of the Limfjord are also evident in other parts of the profile (see below). After ~4,800 BC, pigment abundances appear to gradually increase again, culminating in a high productivity phase between 4,300-3,150 BC (discussed in more detail in KIL-E2).

High abundance of UV-radiation absorbing pigments are also present in this lowermost phase, suggesting increased depth penetration of UV-radiation (Leavitt *et al.*, 1997). These pigments are primarily produced by benthic sediment dwelling organisms (e.g. cyanobacteria) as photoprotectants when subjected to increased UV-exposure (Leavitt *et al.*, 1997; McGowan, 2007). This might suggest shallower water existed within the fjord or reduced levels of plankton (i.e. benthic diatom dominated system, ~60-80 % between ~5,500-4,200 BC) and dissolved organic carbon (DOC), important in aquatic systems for absorbing or shading out UV radiation. Shallower water is, however, deemed less likely due to the depth of the Kilen basin, with sediment being deposited between ~1,500-1,900 cm bpsl at the same time as sea level reached close to its present day levels (e.g. Petersen, 1981; Behre, 2007). Reduced DOC might be related to lower input from terrestrial sources or from reduced production from phytoplankton and bacterioplankton, with the latter again likely to be influenced by shallower water and subsequent reduced habitat availability for planktonic organisms.

KIL-E2 (~4,200-3,100 BC) - *High productivity phase* - A substantial increase in marine productivity occurs in this phase, marked by higher concentrations and fluxes of organic matter and increased accumulation of sedimentary pigments and diatoms (including an increased presence of planktonic taxa). There is also a striking disappearance of the UV-absorbing pigments at ~4,000 BC, most likely due to internal loading of DOC from phytoplankton blooming (Mannino and Harvey, 2000) and shading by phytoplankton reducing the UV-penetration within the fjord. This productivity increase is likely to be the result of complex interactions between salinity and/or nutrients, most likely caused by changes in the marine environment, rather than changing inputs from terrestrial sources. As the beginning of this productivity increase pre-dates the introduction of agriculture (~3,900 BC; Andersen and Rasmussen, 1993) it is unlikely to be human induced. There is, however, a secondary increase in sedimentary pigment concentrations around 3,700 BC, which might suggest a later role for terrestrial input of nutrients following human activities (e.g. induced by forest clearances) as evident in numerous terrestrial pollen records (Iversen, 1941; Aaby, 1986; Andersen, 1995b; Rasmussen, 2005).

The beginning of this high productivity phase (i.e. ~4,300 BC) broadly coincides with the strengthening of the south Jutland current between ~4,250-2,850 BC, which ultimately causes key hydrographic changes in the North Sea and Skagerrak-Kattegat region (Conradsen and Heier-Nielsen, 1995; Leth, 1996; Jiang *et al.*, 1997; Jiang *et al.*, 1998; Gyllencreutz *et al.*,

2005; Gyllencreutz *et al.*, 2006). The beginning of this hydrographic shift appears to be synchronous with the final drowning of the Jutland Bank at ~4,250 BC (Leth, 1996; Gyllencreutz and Kissel, 2006), which enabled increased flow of North Sea water into the Skagerrak. Increased grain size of sediments in Skagen and the Skagerrak, higher energy conditions (i.e. stronger currents) and the greater transport of robust North Sea diatoms (e.g. *Delphineis surirella* in the Skagen-3 core; Jiang *et al.*, 1997; Jiang *et al.*, 1998) all support strengthening of the Jutland current (most notably the southern component) around this time (Conradsen and Heier-Nielsen, 1995; Jiang *et al.*, 1997; Jiang *et al.*, 1998; Gyllencreutz, 2005; Gyllencreutz *et al.*, 2006). The shift has been linked to a climatic cooling event at ~4,200 BC (Conradsen and Heier-Nielsen, 1995; Jiang *et al.*, 1997), but a more recent revision by Gyllencreutz, (2005), suggests a continuous hydrographic transition lasting over ~2,000 years.

The Jutland current is an erosive current, flowing over sandy sediments off western Denmark, and is responsible for eroding, transporting and re-depositing large amounts of sediment (Eisma and Kalf, 1987; Longva and Thorsnes, 1997). An increase in the strength of this current is likely to result in greater erosion along the western coast and increased sediment deposition along the northern coast of Jutland (e.g. Skagen; Jiang *et al.*, 1997; Jiang *et al.*, 1998). Sediment accretion might have even intermittently reduced the northern opening of the Limfjord, temporarily reducing the marine connection and subsequently lowering salinity (e.g. 4,100-3,900 BC; Figure 3.18). Care must be taken with the interpretation of salinity change in this phase of the record due to difficulties with isolating a clear direct salinity signal from other parameters (e.g. productivity, nutrients) which are also exhibiting intense changes (e.g. Anderson, 2000).

The direct influence that greater exposure to, or shelter from, the North Sea and Skagerrak has on the trophic status of the Limfjord is less clear than for salinity. It is possible that nutrients are retained longer when the Limfjord's connection with the North Sea and Skagerrak is reduced, causing higher productivity and enabling nutrient-demanding planktonic organisms (e.g. diatoms; *Chaetoceros* cysts, *C. choctawatcheeana*, *C. striata*, *S. costatum*) and bacterioplankton to bloom. This might in turn induce greater production of DOC and reduce UV-light penetration. In systems such as the Baltic Sea, phytoplanktonic blooming can lead to stagnation of bottom waters due to hypoxia (Sohlenius and Westman, 1998; Sohlenius *et al.*, 2001; Zillén *et al.*, 2008; Weckström *et al.*, in review). A prolonged period of high

productivity in the Limfjord might be sustainable due to a regular input of oxygenated water through the northern and possibly western openings. Alternatively, regular renewal events might be required, possibly associated with short term sea level fluctuations and phases of increased erosion (e.g. climatic cooling and storm events). Renewal events might be separated by sediment accretion phases, reducing the Limfjord-marine connection, and again enabling phytoplanktonic blooming and high turnover of plants and algae.

This could explain both prolonged high productivity and the fluctuating DI-salinities between 4,100-3,150 BC (Figure 3.18). For example, lower DI-salinity exists between $\sim 4,100-3,900$ BC BC ($\sim 20-22 \%$; Figure 3.18), followed by numerous fluctuations between $\sim 3,900-3,150$ BC, with salinity reaching as high as 27-28 ‰ in some samples. This also suggests significant inputs of saline water through the west. The low salinity between 4,100-3,900 BC is particularly interesting, occurring just prior to the Mesolithic-Neolithic transition and potentially limiting marine resources important to Mesolithic/early Neolithic cultures and societies (e.g. oyster decline; Rowley-Conwy, 1984; Andersen, 1989, discussed in more detail in section 3.11). There also appears to be exceptional preservation in this section of the record, a likely response to reduced marine influence, lower salinity and high sediment accumulation rates. This is demonstrated by consistently high **F** index scores and relatively high abundances of the weakly silicified diatom species *Skeletonema costatum* (Figure 3.7).

Deeper water might provide an alternative explanation for a reduction in UV- penetration (and increased planktonic diatoms), but is deemed unlikely with respect to the current evidence. The most plausible explanation for increased water depth would be higher sea levels in the Skagerrak and North Sea after ~4,300 BC (Behre, 2007; Pedersen *et al.*, 2009) and therefore greater input of saline water into the Limfjord. Under such circumstances, a prolonged and probably substantial increase in salinity would be expected, this is not apparent in any of the salinity records presented here (i.e. diatom, isotopes, foraminifera, or molluscan records). According to the southern North Sea level curve, sea level is rising at this point, but the low DI-salinities suggests that that the western Limfjord had not been breached at this time (possibly until ~3,700 BC or later). In the other regional sea level records (e.g. Mörner, 1976; Christensen, 1995; Berglund *et al.*, 2005), sea level decreases shortly after ~4,300 BC associated with the late Atlantic regression (or late Mesolithic regression; L3, 4,200-3,800 BC in Berglund *et al.*, 2005). There is also no evidence in the Skagen and Skagerrak record for increased water depth at this time (Conradsen and Heier-Nielsen, 1995).


Figure 3.18 - Salinity and productivity changes in the Kilen profile between ~4,500-2,500 BC (covering the Meoslithic-Neoltihic transition). Sea level from Behre, (2007) for southern North Sea. *m NN asl = metres realtive to German ordnance datum above sea level.

In terms of terrestrial change, the increased regularity of *Betula* fruits in the plant macrofossil record after ~3,850 BC probably corresponds with the early Neolithic *Betula* expansion as recorded in the Danish pollen diagrams (e.g. Iversen, 1941; Aaby, 1986; Rasmussen, 2005). This is probably a consequence of human-induced land-use changes associated with the introduction of agriculture (Iversen, 1941). The palynological analyses spanning ~4,180-3,640 BC suggest a mixed deciduous forest dominated by *Corylus, Quercus* and *Alnus*, with *Ulmus* probably being quite abundant initially. The low percentages of *Tilia, Fraxinus* and *Pinus* pollen suggest that these tree taxa were either very sparse or were not growing in the immediate vicinity of Kilen. Agricultural indicators are also sparse in the short record. *Plantago lanceolata* appears in the record at ~3,900 BC, but a continuous curve does not

begin until at least ~3,640 BC. *Poaceae* increases slightly after ~3,900 BC, but no cultivated cereal crops are present in the record (Figure 3.3).

The absolute disappearance of the UV-absorbing pigments broadly ties in with the *Ulmus* decline (Figure 3.3) and the introduction of agriculture, which might also suggest a more important role for externally sourced DOC. A reduction in forest density and a more open catchment might have increased inputs of DOC into the Kilen basin from terrestrial sources. An increase in minerogenic matter might be expected with greater input of terrestrial material, but there is no change in the minerogenic flux rate at this time.

A secondary increase in productivity is identifiable in the sedimentary pigment record at \sim 3,700 BC, which coincides with a substantial increase in the accumulation rate (\sim 0.7 cm yr⁻¹ increase from 0.36-0.43 cm yr⁻¹). This is caused by a small increase in both the organic and minerogenic flux rate. In the palynological record, elevated *Poaceae* pollen after \sim 3,900 BC, and an appearance by *Plantago lanceolata* at \sim 3,640 BC, probably marks the first signs of major agriculture-based land-use changes in the Kilen area. This interpretation must, however, be treated with caution due to the limited pollen analyses, but would be in good agreement with other Danish palynological studies (Aaby, 1986; Rasmussen, 2005). Nevertheless, it remains plausible that increased input of nutrients (and sediment) from the surrounding catchment, following agricultural land-use change and forestry removal, caused, or at least contributed to the rise in marine productivity after \sim 3,700 BC.

KIL-E3 (~3,100-0 BC) - *Maximum marine phase* - DI-salinity begins to rise after ~3,330 BC and reaches its modern day levels for the Struer area (27 ‰) by ~2,680 BC and almost fully marine conditions (>30 ‰) by ~1,430 BC. The rise broadly coincides with increased sea level in the southern North Sea (Behre, 2007; Figure 3.16) and suggests that a widening (or opening) of the western Limfjord occurred somewhere between ~3,330-2,680 BC. The relatively good agreement between DI-salinity change and sea level rise in the southern North Sea (Behre, 2007) and western Jutland (Gehrels *et al.*, 2006; Pedersen *et al.*, 2009) continues throughout this phase, suggesting that the connection with the North Sea is driving salinity between ~3,100-0 BC. The southern North Sea record is included in Figure 3.16 as it is one of the most comprehensive sea level records for the region, being based on 118 sea level tie points from the German North Sea coast (Behre, 2007), and forms an extension of the Calais-

Dunkirk system, which is applicable throughout the entire southern North Sea (Dubois, 1924; Bateman, 1981; Ervynck *et al.*, 1999).

A substantial drop in the accumulation rate occurs at ~3,100 BC, caused by marked decreases in the flux of organic and minerogenic matter. The intensified marine conditions that ensued (between $\sim 3,150-0$ BC) probably lowered the accumulation rate ($\sim 0.15-0.18$ cm yr⁻¹), due to increased energy (from stronger currents and increased exposure) in the system, greater transportation, re-working and re-deposition of sediments and poorer preservation (e.g. lower plant macrofossil and mollusc concentration, lower diatom F index). Greater distance to the coring location might also partly explain the decreased concentration of terrestrially derived material (e.g. plant macrofossils), likely due to deeper water associated with increasing sea levels. The reduction in organic matter and accumulation rates of sedimentary pigments suggest a decline in marine primary productivity also occurred around this time. In addition, there is little change in the pheophytin-a/chlorophyll a ratio, which suggests poorer preservation is not responsible for this drop in pigment concentrations. After ~3,100 BC, the sedimentary pigment concentration and accumulation rates generally remain relatively low for the remainder of this phase, but subject to minor fluctuations, perhaps related to variable inputs of high salinity oxygenated water from the North Sea, or from changing trophic status, perhaps due to nutrients being washed in from land following land-use changes (e.g. forest clearance for agricultural purposes and settlements).

The foraminiferal samples in this phase are dominated by *E. excavatum* and *A. beccarii* after ~1,590 BC, with relatively low numbers of the higher salinity species (e.g. *E. incertum, E. magellanicum*) present in the lower part of the sequence (Murray, 1991; Alve and Murray, 1999; Knudsen and Konradi, submitted). However, the foraminiferal record might be heavily affected by dissolution, with lower concentrations of foraminifera between ~3,000-200 BC, generally smaller tests and higher quantities of coarser grained material (probably sand and quartz) present within the wet sieved samples. The $\delta^{18}O_{calcite}$ isotope signal suggests that salinity increases at ~2,680 BC, after which it remains relatively stationary up until ~800 BC, prior to a substantial drop in the uppermost isotope sample. An increase at ~2,680 BC is consistent with DI salinity, but after this it becomes contradictory, by suggesting that salinity was lower between ~2,680-400 BC period than it was in the pre-3,200 BC period. This may be the result of alteration of the isotopic signal due to increased dissolution of the foraminiferal tests under intensified marine conditions. This is in contrast to observations at

both Horsens Fjord and Norsminde Fjord, where foraminifera appear to preserve better under higher salinity conditions (Kisum, 2007; Knudsen and Konradi, submitted; D.B. Ryves *et al.* unpublished data). It is, however, plausible that preservation only deteriorates when near fully marine conditions are reached, with foraminiferal preservation at Kilen being very good between ~5,500-3,100 BC, when DI-salinity ranges between ~18-25 ‰. Factors such as increased current energy, greater transport of material and coarser sediments associated with greater exposure to the North Sea might all be affecting foraminiferal preservation.

After increasing up until ~1,500 BC, a sea level regression phase begins which culminates in a sustained period (~1,500-500 BC) of lower sea levels in the southern North Sea (Behre, 2007). DI-salinity for the Kilen record decreases slightly later than this, but salinities of ~24-28 ‰ are inferred between ~930-600 BC, which actually coincides with the lowest reconstructed sea levels for the last ~4,000 years (i.e. labelled regression phases 2-3 and Dunkirk 1a in the Calais-Dunkirk system; Behre, 2007).

This regressive phase is followed by a sharp rise in sea level rise between $\sim 400-150$ BC, which caused the abandonment of settlements along the clay districts of northern Germany (Behre, 2007). A high sea level stand is also identifiable in the Skagen region at ~500 BC (Clemmensen *et al.*, 2001). In the late pre-Roman period, sea level in the southern North Sea reached its highest pre-Roman Iron Age levels and probably marks the maximum connection of the Limfjord with the North Sea and Skagerrak. The Kilen diatom record is dominated by high salinity demanding species such as P. sulcata, C. belgica and D. minutissima between ~390-0 BC, which drives exceptionally high reconstructed salinities, even exceeding present day average salinities for fully marine conditions (i.e. > 35 %) in several samples (35-43 ‰). Salinities up to ~38 ‰ are often found in open oceans (e.g. Pacific Ocean, Sargasso Sea in the North Atlantic) and can exceed in this range in highly evaporative basins (e.g. Persian Sea, Red Sea). Even during this high sea level stand, the salinity of the Limfjord is unlikely to exceed 35 ‰ and therefore salinity has been over-estimated, probably resulting from a mathematical artefact incorporated into the deshrinking tool of the transfer function (e.g. Yuan, 2005). Almost fully marine salinities in the late pre-Roman times are also consistent with foraminiferal, dinoflagellate and mollusc records form the Bjørnsholm Bay region (Kristensen et al., 1995; Ellegaard, 2000; Christensen et al., 2004) and high salinity inferred from diatoms and molluscs at Horsens Fjord and Tempelkrog (D.B.Ryves et al. unpublished data).

KIL-E4 (**~AD 0-1856**) - *Limfjord opening and closing phase* - This is a dynamic phase subject to large scale shifts between marine and brackish water conditions, driven by the degree of connection to the North Sea and/or the Skagerrak. A marked shift to brackish conditions occurs at the onset of this phase, with DI-salinity falling by ~10 ‰ between ~AD 0-50. The high salinity diatoms such as *C. belgica* and *D. minutissima* practically disappear from the record, and *P. sulcata* abundance drops substantially, as brackish-marine species such as *O. mutabilis, Chaetoceros* resting cysts and *Cocconeis scutellum* become more abundant in the record. Between ~AD 0-630, more brackish water taxa appear in the mollusc record (e.g. *Mytilus edulis, Cerastoderma glaucum, H. ventrosa, H. ulvae, Cerastoderma* spp.) and foraminifera become more sporadic. *Zannichellia palustris* also becomes commonly present in the plant macrofossil record, which is likely to have provided an important habitat for epiphytic diatoms (see section 3.10 and Table 3.5), which also increase in this phase.

There is also some evidence for increased marine productivity with increased concentrations of certain sedimentary pigments (most notably alloxanthin, lutein and canthaxanthin; after ~AD 40), increased mollusc and ostracod concentrations and an increase in the organic and carbonate content of the sediment. The majority of other pigments exhibit increases in abundance slightly later in this phase, with the exception of diatoxanthin (from diatoms, dinoflagellates and chrysophytes), which decreases at the onset of this zone and thereafter remains relatively low throughout. A clear difference between the high relative abundances and the low flux rate of organic matter and accumulation of sedimentary pigments suggests, that overall productivity is actually being overestimated in the early Roman period. Exaggerated concentrations probably result from the drop in accumulation rate driven by substantial decrease in the minerogenic flux and to a lesser extent the organic flux. Consistent with the decrease in diatoxanthin is a drop in the diatom accumulation rate in the early Roman period (Figure 3.8).

The cause of this salinity decline is likely to be a closing, or at least severe narrowing, of both the northern (Kristensen *et al.*, 1995; Christensen *et al.*, 2004) and western openings of the Limfjord, probably due to accretion of sediments and a lowering of sea level (e.g. Behre, 2007). The salinity decrease at Kilen (~AD 0-50) begins later than the drop in sea level in the southern North Sea (~150 BC-AD50), suggesting that temporal sedimentary accretion due lower sea level conditions (causing reduced currents and lower energy) played an important role in the eventual closure of the Limfjord. Alternatively, this may be due to dating

discrepancies between profiles, slight time lags between regions or variation in other factors that might also drive salinity change at Kilen (and the Limfjord). The reduction in the accumulation rate and flux of minerogenic material probably results a reduced input of coarse grained material, eroded form the west of Denmark and being redeposited in the lower energy waters of the Limfjord when the Limfjord entrances were open (or wider).

Shortly after ~AD 600, salinity appears to increase again causing the lower salinity-indicating molluscs and Z. palustris to disappear from the macrofossil record, and subsequently the reappearance of B. reticulatum in several samples between ~AD 590-830. DI-salinity at ~AD 800 suggests a return to almost fully marine conditions, perhaps due to a re-opening of the northern entrance (e.g. Kristensen et al., 1995) together with an opening to the west (see section 6.4.2.2). This coincides with a period of higher sea levels in the Limfjord during the late Iron Age (Christiansen et al., 1990; Tanner, 1993). The diatom sampling resolution in this section of the record and the absence of molluscs after ~AD 830, prohibits detailed examination of salinity change throughout the Germanic and Viking periods (AD 400-1050). Sedimentary pigments remain relatively abundant in sediments up until ~AD 780, after which most exhibit a marked drop in concentration, with the exception of the pheophytins. An increase in the ratio of the pheophtyin-a: chlorophyll a after ~AD 750 suggests greater degradation of pigments. A further drop in the concentration of diatoxanthin is consistent with the generally low concentration of diatoms found between depth ~427-450 cm bpsl (following preliminary scanning of diatom samples; results summarised in Figure 3.19), but dating of sediments become particularly problematic in this section of the profile.

It is unclear how long after ~AD 800 (uppermost radiocarbon date) the slow accumulation phase or sedimentary hiatus began (see section 3.3). Kristensen *et al.*, (1995) suggests that the Limfjord remained connected to the North Sea up until sometime between ~AD 1,100-1,200. Based on the much-debated Old Norse Kings' sagas in the Heimskringla of Snorri Sturluson, it has been suggested that the Norwegian King Harald Haardraade might have fled from the Danish fleet under King Sven Estridssøn in AD 1061 through the northern entrance of the Limfjord (Petersen, 1976). A notable increase in the regularity and abundance of *Chara* oospores (freshwater-low salinity tolerant charophytes) occurs above 454 cm (estimated date ~AD 1,140), which might mark the closure of the Limfjord. Increased abundance of *Chara* oospores was also evident in the Bjørnsholm Bay core between ~AD 1,200-1825 (Kristensen *et al.*, 1995). Additionally, marine diatoms remain present in the Kilen profile up until at least

depth 450 cm bpsl (currently dated at ~AD 1030; Figure 3.19). This might suggest relatively good dating throughout the Viking period and up until the end of the marine phase (~AD 1,100-1,200; Kristensen *et al.*, 1995).

In contrast, a diatom sample at depth 437 cm bpsl (currently dated at ~AD 1430), can almost certainly be placed within the low accumulation/low productivity or sedimentary hiatus phase. This sample is extremely low in diatom concentration $(6.2*10^6 \text{ g})$, mainly containing low salinity diatoms and substantial numbers of *Ellerbeckia arenaria*. This diatom grows on sandy sediments in low salinity alkaline or oligotrophic waters in North America (Spaulding and Edlund, 2009) and the Baltic Sea (Snoeijs *et al.*, 1993-1998). The combined presence of marine (e.g. *Paralia sulcata, Grammatophora oceanica*), oligotrophic (e.g. *Ellerbeckia arenaria*) and nutrient-favouring diatoms (e.g. *A. pediculus* and *C. choctawatcheeana*), suggests re-working and re-deposition of sediments. These conditions might be expected under slow accumulation and low productivity or during a sedimentary hiatus. The **F** index (0.35) is surprisingly high for either circumstance, which suggests relatively good valve preservation. There does, however, appear to be a high percentage of heavily silicified diatoms whose valves might be able to survive the re-working/re-deposition process.

KIL-E5 (**~AD 1856- present**) - *The limnic phase* - The building of the road and rail embankment in AD 1856 almost entirely cut the Kilen basin off from the Limfjord and began the limnic phase of Kilen history. Isolating the exact marker for this transition in the sedimentary record, however, has proved difficult due to lack of dating points between depths 420-460 cm bpsl (i.e. radiocarbon and ²¹⁰Pb/¹³⁷Cs stratigraphies), a slow accumulation rate (~1,000 years in 40 cm), very low concentration of diatoms, likely mixing/disturbance of sediments (or potential hiatus) and the complexity of the recent (last ~1000 years) history of the Limfjord (Kristensen *et al.*, 1995; Ellegaard, 2000; Holm and Bager, 2002; Christensen *et al.*, 2004; Poulsen *et al.*, 2007).



Figure 3.19 – Summary diagram of the upper section (~AD 300-present; top 110 cm) of the Kilen profile and associated dating problems. *Grey shading = total plant macrofossil concentration excluding ambiguous remains (i.e. deciduous leaf fragments, buds, wood, bark and twigs), black shading = total plant macrofossil concentration. 1 = approximate depth of AD 1856 using a straightforward extrapolation of the 210Pb/137Cs stratigraphy. 2 = approximate depth of AD 1825 using a straightforward extrapolation of the ²¹⁰Pb/¹³⁷Cs stratigraphy. AAR-12150 = uppermost radiocarbon date (depth 459-460 cm bpsl, AD 672–894 (95.4 %); Table 3.2, Figure 3.2).

The lowermost diatom sample in Figure 3.9 (at depth 417 cm bpsl in sequence; Figure 3.19) is dated to ~AD 1830, but the flora is almost identical to that in the samples immediately above it (i.e ~AD 1862-2007; depths 390-423 cm bpsl), characterised by diatoms favouring low salinity and relatively high nutrient status (e.g. *C. choctawatcheeana, O. mutabilis, A. pediculus*). It is therefore likely that this sample is post AD 1856 and that the dating in this section is erroneous. Prior to AD 1856-1857, Kilen was connected to the Limford and under marine conditions, due to an opening to the North Sea in the west. This western entrance was re-opened by a storm in AD 1825, and molluscan and foraminifera analyses at Bjørnsholm Bay suggest salinity was between 22-30 ‰ in the succeeding period (Kristensen *et al.*, 1995; Christensen *et al.*, 2004). Due to its close proximity to the western entrance, salinity at Kilen is likely to have been higher than Bjørnsholm Bay at this time, probably at or close to its present day value for the Struer area (~27 ‰).

A series of preliminary diatom samples (18 in total) between depths 427-482 cm bpsl (Figure 3.19) were scanned in an attempt to identify markers for ~AD 1825 and AD 1856-1857 via expected shifts in the diatom flora. Unfortunately, no obvious stratigraphic markers can be identified for these two transitions (via scanning alone), due to extremely low diatom concentrations and co-existence of both marine and brackish taxa (i.e. probable re-working of sediments). Therefore, at present the AD 1856 transition can only be broadly estimated at best, but no estimate is possible for ~AD 1825. Here it is suggested that ~AD 1856 falls somewhere between depths 442 cm bpsl (linear extrapolation of the ²¹⁰Pb/¹³⁷Cs stratigraphy, assuming constant accumulation which is, however, unlikely) and 429 cm bpsl, and most likely between 436-429 cm bpsl.

The latter depths are based on the likely, pre-AD 1856 position of the diatom sample at depth 437 cm bpsl (see above) and an increase in sedimentary pigment concentrations (and decrease in pheophytin-*a*: chlorophyll *a* degradation ratio) above 433 cm bpsl. An increase in productivity and/or improved pigment preservation might be expected, due to increased trapping of nutrients (and other material) within the Kilen basin and a faster accumulation rate. A diatom sample at depth 429 cm bpsl appears to be post-AD 1856, containing a similar diatom flora similar to the uppermost section of the profile (i.e. between 417-390 cm bpsl). This sample, however, again appears to be subject to some degree of sediment mixing, though numbers of marine and oligotrophic diatoms are much reduced. This might place it in close proximity to (i.e. soon after) the ~AD 1856 transition.

In the succeeding phase (i.e. above 429 cm bpsl and likely post-AD 1856), the diatom record (and associated metrics) clearly indicates increasing nutrient status and productivity at Kilen. An increase in plant macrofossil concentration (most notably wet ground, swamp and fen taxa) also occurs above 423 cm bpsl, which is again likely to represent either increased trapping of material within the basin, or improved preservation of material, perhaps due an increase in the accumulation rate. By ~AD 1930 (412 cm bpsl), planktonic diatoms account for over half the diatom assemblage, with C. choctawatcheeana being the most abundant taxon (accounting for ~23-38 % of the diatom assemblage between ~AD 1890-1960). This diatom is a strong indicator of enriched nutrient conditions, often found in modern coastal systems suffering from cultural eutrophication (Prasad et al., 1990; Håkansson, 1996; Clarke et al., 2003; Ryves et al., 2004). This drives the relatively high DI-TN values (>1100 µg L⁻¹), indicating eutrophic conditions within the basin. These conditions also appear to affect other organisms, such as Daphnia spp., which increase markedly after ~AD 1930. In contrast, there is a reduction in aquatic taxa (most notably *Chara* spp. and *Ranunculus sect. batrachium*) after ~AD 1920, and by AD 1945 they become quite sporadic. Eutrophic conditions are often detrimental to macrophyte communities, with increased nutrient/sediment loading and algal blooms often causing increasing water turbidity and reduced light penetration (Duarte, 1995; Brush and Hilgartner, 2000; Middleboe and Sand-Jensen, 2000; Nielsen et al., 2002).

In the 1960-90s there appears to be a slight a reduction in *Cyclotella choctawatcheeana* and an increase in the abundance of *Chaetoceros* resting cysts and *O. mutabilis*. Despite these taxa generally demanding higher-salinities than *C. choctawatcheeana*, measured salinity data (available for the period AD 1972-1987) suggest that higher salinity cannot explain this shift (Figure 3.20). These data show that salinity actually dropped between AD 1978-1984 due to decreased exchange of water between Kilen and Venø Bugt (Ringkjøbing Amtskommune, 1988). A preliminary comparison (low resolution, only four overlapping data points) of DIsalinity estimates (Figure 3.20) with salinity measurements (Ringkjøbing Amtskommune, 1988) suggest that the DI-salinity model performs poorly (i.e. incorrect pattern and overestimation of salinity; DI-salinity suggests a gradual increased up until ~AD 1988) in environments heavily impacted by humans and changing nutrient status. In Figure 3.20, the DI-TN model estimates are also compared against measured TN concentrations for the period 1976-1987. The DI-TN model tends to slightly overestimate the TN concentrations, but overall, this preliminary assessment suggests that the TN-model performs relatively well.



Figure 3.20 - Comparison of DI-salinity and DI-total nitrogen estimates with monitoring data for salinity, TN and TP, measured over the period AD 1972-1987.

Phosphorus is also likely to be important at Kilen over the recent past, with evidence for both increased TP concentrations in the 1960s following the building of four fish farms and rising TP levels over the monitored period AD 1976-1987 (Figure 3.20). The reduction of TP in the late 1970s (Figure 3.20) is believed to have been related to poor water exchange with the Limfjord and reduced TP loading from these fish farms. This resulted in an improved passage (or sluice) being installed in 1984, which successfully improved water exchange with the Limfjord and subsequently led to increased salinities and increased sediment release of phosphorus due to occasional salt water pockets forming above the sediments (Ringkjøbing Amtskommune, 1988). Low oxygen conditions can often prevail in the bottom waters after phosphorus release from the sediments. The importance of phosphorus in Danish limnic systems has previously been demonstrated in a number of studies (e.g. Anderson and Odgaard, 1994; Bradshaw *et al.*, 2002; Bradshaw *et al.*, 2005a; Bradshaw *et al.*, 2006) and certainly future potential exists for testing how well TP training sets (limnic and/or marine) perform in this environment (though higher resolution diatom data is required, over the period for which monitoring is available) and the potential long-term effects of TP at Kilen.

According the DI-TN reconstruction, the rate of TN loading drops between ~AD 1950-1980, but the nutrient status remains relatively high with C. choctawatcheeana still accounting for between ~10-20 % of the diatom assemblage (between ~AD 1970-1990), Chaetoceros cysts increasing in abundance and C. meneghiniana percentages remaining largely unchanged. As indicated above, some species of *Chaetoceros* can live in low salinity waters (Rines and Hargraves, 1988) and are often also affiliated with high nutrient status (Andrén et al., 2000; Leśniewska and Witak, 2008). In marine systems, cyst production often follows high productivity events (marked by high numbers of vegetative cells which rarely preserve in sediments), due to subsequent nutrient depletion (Leveneter et al., 1996). Despite the drop in C. choctwatcheeana percentages, this is unlikely to be the case here, due to likely regular input of nutrients, continued presence of these nutrient favouring diatoms (e.g. C. choctawatcheeana, C. meneghiniana), continued increase of DI-TN and high and increasing TP (e.g. annual mean phosphorus concentrations between AD 1989-1996 = 0.239 mg L^{-1} ; Jensen et al., 2006, Figure 3.20). Difficulties associated with taxonomy of resting cysts under light microscopy precludes identification down to species level for *Chaetoceros* cysts in this study, and therefore it is difficult to interpret, what environmental variables (e.g. salinity change, nutrients) they are reacting to (and when) throughout the Kilen profile.

Nevertheless, increased numbers of planktonic diatoms since the mid 1980s appear to have caused a reduction in both the abundance and diversity of the benthic community. With the exception of *O. mutabilis*, most other benthic species practically disappear from the record by ~AD 1985. The TN begins to increase rapidly after ~AD 1980 and Kilen becomes extremely eutrophic (measured TN > 2000 μ g L⁻¹, DI-TN ~1800-4300 μ g L⁻¹) with high phosphorus concentrations (Jensen *et al.*, 2006). This probably caused blooming of planktonic species, and subsequent decreasing water clarity and light penetration, often considered classic symptoms of eutrophication (Cederwell and Elmgren, 1990; Rosenberg *et al.*, 1990; Cooper and Brush, 1991; Nixon, 1995; Cornwell *et al.*, 1996; Clarke *et al.*, 2003). Planktonic blooms are further supported by high diatom concentrations (Figure 3.8), and both high measured summer chlorophyll *a* content in water column and shallow Secchi depths between AD 1989-1996 (Jensen *et al.*, 2006).

The high numbers of *O. mutabilis*, however, suggest that benthic conditions remained suitable for diatoms, possibly more confined to the shallower parts of the basin, and therefore light penetration alone cannot explain this reduction. There is a noticeable reduction in benthic (commonly epiphytic) diatoms (most notably *Ctenophora pulchella, Cocconeis placentula,*

Epithemia sorex, E. turgida, Diatoma moniliformis, Rhoicosphena abbreviata) at this time, which might have been due to the disappearance of macrophytes. No remains of aquatic macrophytes are recorded in the plant macrofossil record after ~AD 1970. It is possible that the remaining benthic taxa (e.g. A. *pediculus*) were outcompeted by *O. mutabilis* under higher nutrient and possibly salinity conditions. Sandier sediments might also have favoured *O. mutabilis* over other benthic taxa (e.g. *A. pediculus*), though this might be considered unlikely due to the high accumulation rates, likely depositing large volumes of fine organic content (Figure 3.19).

After ~AD 1999, *C. choctawatcheeana* becomes the most abundant species present in the profile again as eutrophication appears to be intensifying. There is an increase in *Betula* fruits between ~AD 1990-2005, which might reflect improved preservation conditions within the basin due to increased bottom water anoxia (e.g. reported after periods of phosphorus release from the sediments after inputs of salt water from Venø Bugt through the Kilerkanal). *Daphnia* spp. also disappear after ~AD 1996 which is possibly related to low oxygen conditions within the lake and the reduction in macrophyte abundances offering shelter from predation (e.g. Lauridsen *et al.*, 1998; Burks *et al.*, 2001).

3.12 Summary and conclusions

- The level of connection with the North Sea and Skagerrak (predominately governed by sea level, erosion and sedimentary accretion) is extremely important in driving changes in the marine environment (i.e. salinity, productivity, nutrient status and sedimentary regime) at Kilen.
- DCA results for the entire diatom dataset demonstrates a strong salinity gradient on axis 1, due to the presence of weakly brackish taxa at the top of the profile (in the limnic section), through to fully marine diatom taxa present in the pre-Roman period (~500-0 BC). For the limnic section only (i.e. since ~AD 1856), PCA axis 1 is predominately driven by nutrients, but with depth, sediment type and macrophyte cover probably also being important. For the marine section of the profile only (i.e. pre AD 1856), a salinity gradient is again apparent on DCA axis 1, with the diatom flora spanning from fully marine in parts of the core sequence (most notably pre-Roman period) through to brackish water taxa in the Roman period. For the marine dataset, DCA axis 2 appears to be driven by substrate type and habitat preference, with several of the epiphytic species being abundant during the Roman period (when there is

evidence for increased macrophyte abundances, particularly *Zannichellia palustris*) scoring negatively. Pelagic and other benthic diatoms (i.e. living on the sediments or attached to sand or stones) score more positively on this axis.

- The DI-salinity record suggests that between ~5,500-2,750 BC, salinity was lower (generally fluctuating between 18-25 ‰, but with occasional higher peaks) than at present for the Struer region (~27 ‰). It is therefore likely that the western Limfjord was either completely, or at least partially closed off at this time, probably due to lower sea levels in the North Sea, and the presence of the Jutland Bank (up until ~4,200 BC). Prior to the opening of the western Limfjord, the main connection to the North Sea (and source of high salinity water) was probably to the north, through the Skagerrak.
- Marine productivity is relatively high at the bottom of the profile, but subject to
 fluctuations up until ~4,400 BC, when a pronounced high marine productivity phase
 begins. Higher primary productivity appears to coincide with lower salinity
 conditions, possibly due to greater nutrient retention within the fjord, though it is
 likely that inputs of marine water are important for replenishing oxygen and nutrients
 levels and perhaps also, inputs of nutrients from the catchment. Productivity declines
 again after ~3,200 BC as salinity begins to increase within the fjord.
- An upward trend in salinity begins around ~3,300 BC, likely due to the opening of the western Limfjord, thereby establishing a direct connection with the North Sea. It is likely that the Northern Limfjord was also open at this time. Near fully marine conditions being established by ~1,500 BC (>30 ‰) and maximum salinities occur in the late pre-Roman period (~250-0 BC). Also associated with the intensification of marine conditions (i.e. higher salinity, stronger currents, likely coarser sediments and a more open system for transport of sediment, nutrients and remains out of the basin) are decreases in the sedimentary accumulation rate and marine productivity (both after ~3,200 BC).
- A large magnitude drop in salinity occurs after ~AD 0 (~10-15 ‰ decrease in ~50 years) suggesting that Limfjord became closed off from the North Sea/Skagerrak. Brackish-marine (~15-17 ‰) conditions persist up until ~AD 600, before a transition back to almost fully marine conditions throughout the late Germanic and Viking periods. Increased abundances of lower-salinity demanding molluscs and macrophytes, and slightly increased accumulation rates for several sedimentary pigments suggest that a more productive marine environment existed between ~AD 0-600 (i.e. Roman and early Germanic period).

- Poor temporal resolution or a sedimentary hiatus (and uncertain dating) prevents detailed assessment of the marine environment throughout medieval and historical times (i.e. ~AD 1110-1920, beginning of ²¹⁰Pb/¹³⁷Cs stratigraphy).
- Following the building of a road and rail embankment in AD 1856, Kilen became a brackish water limnic system that has been subject to increasing eutrophication for the last ~150 years. Symptoms include increased nitrogen and phosphorus (up until AD 1986) concentrations, increased summer chlorophyll-*a* and diatom concentrations, increased abundance of planktonic diatom taxa, a reduction in benthic diatom diversity (particularly after the late 1970s) and reduced water clarity/light penetration. Comparison with environmental monitoring data suggests that the diatom-based salinity model performs poorly in environments heavily affected by nutrients, but a preliminary assessment suggests that the DI-TN inferences are reasonably accurate.

Chapter 4. Results: Norsminde

4.1 Introduction and aims

In this section results are presented from palaeoenvironmental analyses performed at Norsminde Fjord. This includes a range of physical (lithology, water content, organic matter, carbonate content) and biological proxies (diatoms, molluscs, foraminifera, pollen, plant macrofossils) analysed from a sedimentary sequence spanning the time period ~6,900-1,700 BC. The chapter aims to address the following research questions:

- 1. What are the main lithological and physical changes that occur throughout the Norsminde sedimentary profile?
- 2. How has chronological control been attained for the Norsminde Fjord profile and how accurate is the age-depth model?
- 3. How have the biological communities (diatoms, foraminifera and molluscs) living within Norsminde Fjord changed over the study period?
- 4. How has the salinity, productivity and nutrient status of Norsminde Fjord changed between ~7,000-1,700 BC, as inferred by these biological communities?
- 5. To what extent can salinity change be accurately and quantitatively reconstructed using a diatom-based salinity transfer function and by what magnitude has salinity changed over time period spanned by the profile?
- 6. How has the vegetation of the Norsminde catchment area changed over the study period?
- 7. What are the main environmental factors driving the changes in the diatom community?
- 8. Are taphonomic signals evident in the record and to what extent do they obscure/override true signals of environmental change at this fjord site?
- 9. Is there any evidence for human impact upon the fjord and surrounding landscape from the physical and biological proxies analysed here?

4.2 Core Lithology

The Norsminde core lithology is summarised in Figure 4.1 and the original lithological descriptions (using a modified Troels-Smith system; Troels-Smith, 1955; Birks and Birks, 1980) are provided in Table 4.1. The upper 64 cm of the profile represents the water column, with this being the water depth at the coring location (Figure 2.3). The majority of the

underlying sedimentary profile (64 - 1,156 cm bpsl) is composed of a marine gyttja, containing varying amounts of sand, silt and clay. Scatterings of mollusc and plant material are generally found throughout, though in some sections these components can become more abundant or scarce. For example, increased plant material is present between 74-586 cm bpsl and 1,128-1,148 cm bpsl, and increased mollusc fragments between 861-866 cm bpsl and 1,096-1,114 cm bpsl. In contrast, plant material is particularly scarce between 509-861 cm bpsl and mollusc material between 590-804 cm bpsl and 900-1,096 cm bpsl. The uppermost sedimentary layer (64-74 cm bpsl) is dark brown in colour, loosely compacted with scattered findings of *Hydrobia* spp. and *Mya arenaria*. It is likely that this uppermost section has been regularly disturbed (churned up and re-deposited) since this shallow fjord reached its sedimentation limit (~3,700 years ago; see Section 4.11 for further discussion).



Figure 4.1 - Lithology of the Norsminde profile based on the original descriptions (Table 4.1) by Peter Rasmussen (GEUS), using a modified Troels-Smith system for characterisation of unconsolidated sediments (Troels-Smith, 1955)

Depth (cm bpsl)	Sediment description				
0–64	Water				
64–74	Dark-brown loose silty, sandy clay-gyttja with scattered molluscs (among others <i>Hydrobia</i> sp. and <i>Mya arenaria</i>). Ld^{1-2} 2–3, Clay and silt 1–2, Sand +++, Test. moll. +, Part. test. moll. +.				
74–594	Dark grey-green silty, sandy clay-gyttja with scattered molluscs and plant remains. Between 74 and 79 cm many <i>Hydrobia</i> sp. In 584 cm a larger piece of charcoal (0,5 cm). Ld^{0-1} 2, Clay and silt 2, Sand +++, Dg +, Test. moll. +, Part. test. moll. +. Limes superior 0.				
594–598	Sand with some clay and gyttja and very few mollusc fragments. Sand 2, Ld 1, Clay and silt 1, Part. test. moll. +. Limes superior $0-1$.				
598–875	Dark grey-green silty, sandy clay-gyttja with few mollusc and plant fragments. From 804 cm and downwards more molluscs and probably increasing content of minerogenic matter. Ld ⁰⁻¹ 2, Clay and silt 2, Sand +++, Dg +, Part. test. moll. (+). Limes superior 2–3.				
875-880	Dark grey-green silty, sandy clay-gyttja with relatively many molluscs (among others <i>Ostrea edulis</i>). A few small stones 3–4 mm in size. Ld 1–2, Clay and silt 2–3, Sand +++, Test. moll. +++ (1). Limes superior 0.				
880–914	Dark grey-green silty, sandy clay-gyttja with mollusc fragments (among others <i>Cerastoderma</i> sp.). Ld 1, Clay and silt 3, Sand +++, Part. test. moll. ++. Limes superior 0				
914–1,100	Dark grey silty, sandy clay-gyttja with very few molluscs. Ld 1, Clay and silt 3, Sand +++, Part. test. moll. (+). Limes superior 0 (primarily indicated by more molluscs in the layer above).				
1,100–1,118	Dark grey-green silty, sandy clay-gyttja with more coarse minerogenic material and more molluscs (among others <i>Cerastoderma</i> sp.) than in the layer above. Probably also a little more organic matter than in the layer above. Ld ⁰ 1, Clay and silt 3, Sand +++, Test. moll. +. Limes superior 0.				
1,118–1,132	Dark grey-green silty, sandy clay-gyttja with scattered molluscs (among others <i>Cerastoderma</i> sp.); the layer is darker than the layer above. Quite some coarse minerogenic material (sand/coarse sand). Ld ⁰ 1–(2), Clay and silt (2)–3, Sand/coarse sand +++, Test. moll. +, Part. test moll. +. Limes superior 0.				
1,132–1,152	Dark grey-green sandy, silty clay-gyttja with quite some coarse minerogenic matter and scattered plant remains (among others fruits of <i>Potamogeton pectinatus</i>). The layer gets darker downwards. Ld ⁰⁻¹ 1–(2), Clay and silt (2)–3, Sand/coarse sand +++, Dg +, Limes superior 0.				
1,152–1,156	Black-brown strongly sandy humic layer. Limes superior 0.				

Table 4.1- Original lithological descriptions of the sediments from the Norsminde profile, using a modified Troels-Smith system (Troels-Smith, 1955).

Below 74 cm bpsl, the sediment is predominately dark-greyish green in colour (see Table 4.1), changing only to a more greyish colour between 900-1,096 cm bpsl. Increasing amounts of coarse, minerogenic material (sand/coarse sand) are found below 1,096 cm bpsl, together

with elevated numbers of cockle (*Cerastoderma* spp.; 1,096-1,114 cm bpsl) fragments. The sediment gradually gets darker below 1,128 cm bpsl as more organic material becomes visible (particularly *Potamogeton pectinatus* fruits). At the very bottom of the profile (1,152-1,156 cm bpsl), a blackish brown, sandy humic layer underlies the marine gyttja, which is likely to represent a non-marine (limnic) environment, prior to the marine incursion into the Norsminde Fjord basin.

4.3 Age Model

Details of all AMS ¹⁴C dated samples (17 in total) are provided in Table 4.2. The final age model (Figure 4.2) was produced by Jesper Olsen (Aarhus University) using the age model program, B-peat v 1.0 (Blaauw and Christen, 2005), and consists of 14 AMS ¹⁴C dates derived from terrestrial plant material (see Table 4.2). Two dates were omitted from the final age model; AAR-3530 and AAR-3531. The ages generated for both samples were clearly too 'old' (i.e. outliers) when compared with the other AMS ¹⁴C dated samples in the profile. AAR-3531 was derived from a strongly corroded twig fragment, which may have been subject to substantial transportation, re-deposition or chemical alteration prior/during its burial. This may also be true for AAR-3530, which is derived from a wood fragment, though no visible signs of corrosion were evident.

In the uppermost section of the profile (post ~1,700 BC), it appears that a large hiatus exists, spanning the last ~3,700 years. This hiatus probably began when the shallow fjord (64 cm water depth at coring location; Figure 2.3) reached its sedimentation limit, and thereafter material has been constantly churned up, decomposed and possibly re-deposited elsewhere. The uppermost AMS ¹⁴C date (AAR-4266) of 1911-1675 cal. BC (2 SD) at 161 cm bpsl appears to fit well with the age-depth model, suggesting that this part of the sequence is reliably dated (Olsen *et al.*, submitted). Therefore, Norsminde Fjord must have been extremely shallow as early as ~3,700 years ago.

Laboratory Code	Depth (cm bpsl)	Material dated	¹⁴ C age (yr. BP)	Calibrated age BC (± 2 SD)	δ ¹³ C (‰ VPDB)
AAR-4266	156–166	15 leaf-fragments, 10 budscale-fragments	3475 ± 50	1932BC (95.4%) 1682BC	-25.1
AAR-3529	304-308	1 Quercus twig with bark and bud	3805 ± 50	2459BC (93.1%) 2132BC, 2082BC (2.3%) 2060BC	-25.8
AAR-4267	378–384	70 leaf-fragments	3990 ± 65	2850BC (3.2%) 2812BC, 2742BC (0.6%) 2729BC 2694BC (0.3%) 2687BC, 2680BC (91.3%) 2293BC	*-25
AAR-4268	428-434	224 leaf-fragments	3885 ± 50	2476BC (95.4%) 2204BC	-25.4
AAR-4024	484–494	132 leaf-fragments, 1 budscale-fragment, 1 Chenopodiaceae seed	4010 ± 60	2856BC (5.1%) 2812BC, 2747BC (1.5%) 2724BC 2698BC (85.4%) 2391BC 2385BC (3.5%) 2346BC	-25.6
AAR-4269	528–538	67 leaf-fragments, 13 budcale-fragments, 1 Chenopodiaceae seed-fragment	4215 ± 60	2920BC (95.4%) 2620BC	-25.5
AAR-3530	540-550	1 wood fragment	5635 ± 55	4592BC (95.4%) 4352BC	-25.3
AAR-4270	574–584	98 leaf-fragments, 6 budscale-fragments	4250 ± 50	3010BC (2.7%) 2978BC, 2958BC (0.5%) 2950BC 2942BC (51.4%) 2835BC, 2816BC (40.8%) 2668BC	-26.1
AAR-3531	608–618	1 twig-fragment (strongly corroded)	4720 ± 65	3637BC (56.2%) 3484BC, 3475BC (39.2%) 3370BC	-28.4
AAR-4271	652–662	90 leaf-fragments, 4 budscale-fragments, 1 <i>Phragmites communis</i> caryopsis, 2 <i>Betula</i> sp. fruits	4585 ± 50	3514BC (24.8%) 3424BC, 3404BC (0.3%) 3399BC 3384BC (35.3%) 3263BC, 3244BC (35.0%) 3100BC	-25.7
AAR-4025	712–716	305 leaf-fragments, 2 budscale-fragments, 1 <i>Epilobium</i> cf. <i>hirsutum</i> seed	4885 ± 60	3798BC (83.1%) 3625BC, 3600BC (12.3%) 3524BC	-26.7
AAR-4272	782–786	105 leaf-fragments, 2 budscale-fragments, 1 <i>Alnus</i> sp. Fruit	5255 ± 65	4254BC (95.4%) 3960BC	-26.1
AAR-4026	824–834	187 leaf-fragments, 5 budscale-fragments, 1 bud, 3 Chenopodiaceae seed-fragments, 1 Poaceae caryopsis	5555 ± 60	4520BC (93.7%) 4325BC, 4286BC (1.7%) 4269BC	-19.8
AAR-3532	876–878	2 twig-fragments	5980 ± 60	5007BC (95.4%) 4720BC	-30.8
AAR-3533	948–956	231 leaf-fragments	6770 ± 65	5788BC (95.4%) 5558BC	-27.7
AAR-3534	1150-1152	15 wood-fragments	7995 ± 55	7061BC (91.9%) 6734BC, 6726BC (3.5%) 6699BC	-27.5

Table 4.2 - Details of all AMS ¹⁴C dated samples for the Norsminde profile. Samples in grey were omitted from the final age model (see text). *= estimated value.



Figure 4.2 - Norsminde age model (produced by Jesper Olsen, Aarhus University, using B-peat (Blaauw and Christen, 2005).

Prior to ~1,700 BC, this ¹⁴C based age-depth model appears to be reliable and robust. The date of the *Ulmus* decline at Norsminde (~4,000 BC) appears to fit well with datings of this event at many Danish lake sites (~3,900-4,000 BC; e.g. Andersen and Rasmussen, 1993). The date for the marine transgression of the Norsminde basin around 6,900-6,800 BC at 1,152 cm bpsl, is in good accord with regional estimates (e.g. Århus Bay, ~6,800 BC, (Jensen and Bennike, 2008) discussed in more detail in section 4.11).

4.4 Physical parameters

Physical sedimentary analyses (including overall accumulation rate, water content, organic matter, carbonate and minerogenic content and flux rates) were performed on 530 samples

(average resolution 10 years, range 4-20 years) from the Norsminde profile and results are displayed in Figure 4.3. Results are described below in relation to four statistically significant zones (labelled NRS-1-NRS-4) determined using an optimal splitting technique (see section 2.10.1).

NRS-S1 (~7,000-6,300 BC) – At the bottom of the record, the overall accumulation rate is relatively high (~0.15-0.17 cm yr⁻¹), despite the flux of all three components (organic matter, carbonate and minerogenic matter) decreasing throughout. The only real exception is a brief increase in the minerogenic accumulation rate between ~6,760-6,580 BC. In terms of the sedimentary matrix, at the bottom of the profile the minerogenic component accounts for approximately 73-78 % of the total dry mass, but this rises to ~84-86 % after ~6,670 BC, causing both the organic and carbonate components to decrease in relative abundance.

NRS-S2 (~6,300-4,100 BC) - After remaining relatively constant in NRS-S1, the accumulation rate starts to decrease early in phase NRS-S2 (~6,300-4,100 BC). This decrease begins shortly after ~6,300 BC and continues steadily up until ~5,400 BC. Lowest accumulation rates for all parameters are recorded between 5,400-4,700 BC, with overall accumulation rate dropping as low as ~0.01 cm yr⁻¹. This changes after ~4,700 BC, as the flux of organic and minerogenic material steadily increases, causing the overall sedimentation rate to rise (which continues for the remainder of NRS-S2). The organic matter and minerogenic accumulation rate, however, is more variable, exhibiting a slight increase between ~5,400-4,250 BC and then becoming more erratic thereafter (e.g. two distinct peaks at ~4,930 BC and 4,410 BC).

NRS-S3 – (**4,100-2,700 BC**) - Early in this zone, the overall accumulation rate stops increasing (at ~4,000 BC), and remains constant (0.17 cm yr⁻¹) up until ~2,800 BC. This is largely true for each individual component, with the exception of a short lived peak in minerogenic flux between 3,070-2,900 BC and a steady decrease in carbonate accumulation after ~3,350 BC. This minerogenic peak features prominently in the relative percentage data for these components, with both the organic matter and carbonate content decreasing markedly, as increased minerogenic matter is input into the basin. At the very top of this zone (i.e. between ~2,800-2,700 BC), there is a sharp increase in the flux rate of all parameters, which subsequently causes the overall accumulation rate to rise.



Figure 4.3 - Sediment physical parameters, overall accumulation rate and flux rates for organic, carbonate and minerogenic matter for the Norsminde profile.

NRS-S4 (2,700-1,700 BC) – The major increase in flux rates at the top of NRS-S3 results in an accumulation rate of ~0.44 cm yr⁻¹ in this uppermost zone (NRS-S4; 2,700-1,700 BC), far exceeding the rate at any other time in the profile. Flux rates for each component also reach their highest levels for the entire profile in this zone but also become much more variable, exhibiting more regular and greater magnitude fluctuations than in any of the preceding zones. The only systematic change in the sedimentary matrix (i.e. percentage data) is an increase in the organic component and subsequent drop in minerogenic matter between ~2,700-2,400 BC.

4.5 Diatoms

A total of 69 diatom samples were counted, with a further 7 samples assessed for diatom presence (in which indicator species were recorded, where possible). The diatom record spans the time period ~6,900-2,020 BC, with elevated resolution over the marine transgression (6,900-6,700), the Mesolithic-Neolithic transition (~4,200-3,800 BC) and between ~3,000-2,300 BC which spans the Pitted Ware cultural period (~2,800-2,400 BC). A total of 89 different genera were recorded, with 263 taxa identified to species level. 38 species were present at >3 % abundance in one or more samples. The diatom assemblage is shown in Figure 4.4 and Figure 4.5 and associated metrics are presented in Figure 4.6. Results are described below in relation to four statistically significant biological zones (determined via optimal splitting; see section 2.10.1).

Two quantitative diatom-based salinity reconstructions are also provided in Figure 4.5. The first reconstruction includes all diatom taxa present in both the Norsminde profile (excluding *Chaetoceros* spp. and *Rhizosolenia* spp.) and the MOLTEN training set. The second reconstruction tests if DI-salinity inferences change/becomes clearer if *Fragilaria* sensu lato (s.l.) are removed. These species are often euryhaline and have great ability to colonise quickly, and survive in a broad range of unstable brackish/marine environments (e.g. Eronen, 1974; Stabell, 1985; Shennan *et al.*, 1993; Witkowski, 1994; Sabbe and Vyverman, 1995; Seppä *et al.*, 2000; Witkowski *et al.*, 2000; Miettinen *et al.*, 2007), which might in turn lead to some important shifts in salinity passing undetected. Furthermore, in the Norsminde profile *Fragilaria* s.l. percentages tend to exhibit great variability throughout (Figure 4.5, and see below), which probably drive some of the short term fluctuations in DI-salinity (Figure 4.5). These small salinity shifts, therefore, might just be the result of minor changes in community

structure, possibly related to other forcing factors (e.g. water depth, nutrients) operating within the fjord. Overall, there is general agreement (in terms of salinity change) between both reconstructions over the majority of the record. There are, however, two main conflicting points: 1. the absolute values for salinity differ throughout the record, with generally higher inferences when *Fragilaria* s.l. are removed, and 2. very different reconstructed values for salinity around ~5,400 BC (see NRS-D3). The salinity transfer function and the accuracy of the presented salinity reconstructions are discussed in more detail in section 6.2.



Figure 4.4 - Norsminde diatom record; ~7,000-6,870 BC. Data expressed as presence/absence.

Diatom metrics include the diatom concentration and accumulation rate, dissolution indices (**F** index and 4 stage dissolution assessment of *Paralia sulcata* valves), benthic: planktonic: tychoplanktonic ratio (B:P:T), centric:pennate ratio, DCA generated axes (discussed in section 4.10.1), Hills N2 measure of diversity and *P. sulcata* valve size. There is very little change in the predominant size of *P. sulcata* valves, with the smaller variety being more abundant throughout. In addition to diatoms, Chrysophytes and *Distephanus speculum* (marine silicoflagellate; not shown) were occasionally recorded.



Figure 4.5 – Norsminde percentage diatom record. (>3 % abundance in one or more slides). Values in brackets refer to salinity optima in the MOLTEN salinity training set (see section 2.10.2).



Figure 4.6 - Norsminde diatom associated metrics. Full model refers to diatom-inferred salinity including *Fragilaria* s.l. and for the reduced model *Fragilaria* s.l. were removed for reconstruction of salinity. Chrysophytes expressed as presence/absence data.

NRS-D1 (~7,000-6,600 BC) – No diatoms were present in the two samples analysed before 6,930 BC (Figure 4.4), and only a few valves (almost always fragmented) of heavily silicified robust diatoms (e.g. *Epithemia turgida*, *Tabularia fasciculata* agg. and *Diploneis* spp.) were present between ~6,930-6,870 BC, due to extremely poor preservation conditions. Extremely low diatom concentration prevented full assemblage counts (i.e. minimum 300 valves) up until ~6,800 BC. Low salinity, robust benthic diatoms, dominate the early assemblage with *E. turgida*, initially (6,860 BC) accounting for ~58 % of the assemblage, supplemented by smaller numbers of *Diploneis didyma* (7 %), *Ctenophora pulchella* (8 %) and *Rhiocosphenia abbreviata* (3 %). As the preservation conditions improve (rising **F** index; Figure 4.6) and the salinity increases, *E. turgida* and *C. pulchella* percentages fall and *Cocconeis scutellum*, *Cocconeis placentula* and *Grammatophora oceanica* all increase in abundance. Planktonic taxa are generally scarce (see B:P:T ratio and low centric:pennate ratio) in this zone.

NRS-D2 – (~6,600-5,490 BC) – There is a large increase in tychoplanktonic and planktonic diatoms in this zone, the latter of which accounts for ~40-50 % of the total assemblage between ~6,600-5,780 BC. An initial peak of the tychoplanktonic diatom, *P. sulcata* (31 %) at ~6,600 BC, marks the switch to higher salinity (~24 ‰), and probably deeper water, briefly replacing the benthic species, *C. scutellum* and *G. oceanica*, as the most abundant species. Above this, truly planktonic diatoms become more abundant with *Chaetoceros* cysts (17-30 %), *Thalassionema nitzschoides* (3-10 %) and *Rhizosolenia* spp. (2-7 %), as the main components of the phytoplankton. *P. sulcata* numbers drop drastically by ~6,360 BC (<5 %), but steadily begins to increase again after ~6,160 BC. The most abundant benthic raxa are *Bacillaria paxillifer* and *Staurosirella pinnata* (after ~6,360 BC), though a number of other species (*O. mutabilis, Opephora minuta, Navicula* spp., *Navicula perminuta* and *Dimmeregramma minor*) increase slightly after ~6,160 BC. After the initially high reconstructed salinity, associated with the *P. sulcata* peak, salinity appears to fall again up until ~6,160 BC, before starting a secondary rise in the upper section of this zone.

NRS-D3 – (~5,490-3,110 BC) – Planktonic taxa decrease again in NRS-D3, as large numbers of small *Fragilaria* s.l. (most notably *S. pinnata*, *S. elliptica* agg. *O. mutabilis*, *O. minuta* and *Fragilaria gedanensis*) become present in the record. These species all tend to exhibit irregular (non-systematic) fluctuations. A peak of *Fragilaria* s.l. (*S. pinnata* and *F. gedanensis*) at ~5,400 BC, may indicate that salinity drops as low as ~16 ‰ for a short period of time, but this is one of the major contrasting points between the two different DI-salinity reconstructions (see section 4.11). However, both reconstructions, suggest high salinity

between 5,320-5,160 BC, largely driven by the marine centric diatom, *P. sulcata* (representing 19-27 % of the assemblage). As *P. sulcata* begins to drop, *Chateoceros* cysts increase (after ~4,930 BC), reaching their highest abundances for this zone between ~4,360-3,800 BC. A number of benthic species also exhibits increases throughout this zone, most notably *G. oceanica* (after ~5,160 BC) and *D. minor* (after ~4,810), followed by *O. minuta* at ~4,690 BC and slightly later (after ~4,280 BC), by *C. scutellum* and *Rhopalodia acuminata*.

NRS-D4 (~3,110-1,800 BC) – A further reduction of the planktonic community occurs in NRS-D4, as *Thalassionema nitzschoides* numbers decrease (practically disappearing after ~2,460 BC) and *Chateoceros* cysts remain low throughout. For the tychoplanktonic taxa, *P. sulcata* numbers are relatively high throughout (5-20 %) and *Fragilaria* s.l. (most notably *S. pinnata, O. mutabilis, O. minuta* and *S. elliptica* agg.) continue to dominate the increasingly important benthic community. *R. acuminata, C. scutellum* and *D. minor* are the main non-*Fragilaria* s.l. benthic taxa, together with smaller numbers of *Cocconeis placentula*, *Tabularia fasciculata* agg., *Navicula ramossisma* and *Navicula perminuta* in an overall diverse benthic community. This is supplemented by occasional peaks of several other Naviculoid species (e.g. *Navicula* spp. 10A, at ~2,780 BC, *Navicula* spp. A, *Navicula* spp. B, both at ~2,400 BC, and several sporadic peaks of *F. cryptolyra*). Salinity continues to fluctuate throughout, with two peaks between ~3,100-2,500 BC.

4.5.1 Diatom Preservation

With the exception of NRS-D1 (see above), overall diatoms preservation is fairly good for a marine environment (**F** index = 0.3-0.5, after ~6,600 BC; Ryves *et al.*, 2004) and therefore dissolution is deemed unlikely to be a major problem for interpretation of natural environmental change at this site. The majority of *P. sulcata* valves also fall into the first two dissolution categories.

4.6 Foraminifera

The Norsminde foraminiferal assemblage spanning ~6,860-1,860 BC is presented in Figure 4.7. Due to low abundance and poor preservation of foraminiferal tests over large parts of the profile, foraminifera are expressed as concentration data (per 20 ml of wet sediment), rather than as relative abundances (i.e. percentage data). Zonation has also been avoided for these reasons and therefore the foraminiferal assemblage results are described below in relation to

the diatom-generated statistical zones (see section 0). 77 samples were analysed (using methodology outlined in section 2.7.3), containing a total of 10 species of calcareous foraminifera (from 6 genera), 6 agglutinated species (from 5 genera) and some undifferentiated planktonic species. Some re-deposited, pre-Quaternary foraminifera were recorded in the upper section of the profile (after ~3,500 BC) and are believed to represent changes in catchment depositional processes (Knudsen and Konradi, submitted).

NRS-DF1 (~7,000-6,600 BC) – Only very small concentrations of foraminifera are present in this zone with only three species recorded (*Ammonia beccarii*, *Haynesina germanica* and *Elphidium albiumbilicatum*), indicating shallow, brackish water conditions.

NRS-DF2 – (~6,600-5,490 BC) – There is a large increase in the overall concentration of calcareous foraminifera early in this zone (~6,380 BC), mainly driven by increased numbers of *H. germanica* and the arrival of *Elphidium excavatum* fo. *selseyensis*. These two species are however, supplemented by smaller concentrations of *A. becarrii* and *Elphidium williamsonii*. *Elphidium magellanicum* and *E. albiumbilicatum* are also present in a few samples and both exhibit a large peak at ~6,130 BC. *E. excavatum* decreases after ~6,130 BC. Variable numbers of calcareous forams throughout this zone causes the overall concentrations to fluctuate intensely.

NRS-DF3 – (~5,490-3,110 BC) – Foraminifera preservation is very poor in the lowermost section of this zone (pre ~5,160 BC), but conditions seem to improve markedly between ~5,160-4,340 BC. During this phase, calcareous foraminifera concentration (and diversity) reaches its highest levels for the entire profile. *H. germanica* and *E. excavatum* fo. *selseyensis* are again the most abundant species, but *A. beccarii, E. williamsonii* and *Elphidium incertum* also reach their maximum abundances in this period. The presence of *E. incertum* suggests relatively high salinity, particularly between ~5,160-4,910 BC, where highest concentrations for this species occur. This is further supported by small numbers of *Haynesina depressula* and *Elphidium voorthuysenii*, present only in this period. Agglutinated foraminifera briefly increase (most notably *Ammoscalaria runiana*; not shown in Figure 4.7) between ~4,630-4,340 BC. After ~4,340 BC, calcareous foraminifera concentration becomes highly variable with dissolution of tests, again, heavily affecting the assemblages. In samples where foraminifera are present, *H. germanica* and *E. excavatum* dominate the assemblage, with just sporadic recordings of *E. williamsonii, E. magellanicum* and *Aubignyna perlucida*. Both *A.beccarii* and *E. incertum* practically disappear from the record after ~4,340 BC.



Figure 4.7 - Norsminde foraminifera record (analyst Peter Konradi, (GEUS). Data expressed as concentrations per 20 ml of wet sediment.

NRS-DF4 (~3,110-1,800 BC) – With the exception of a small peak of *H. germanica* at ~2,620 BC, this zone is almost completely devoid of calcareous foraminifera. Agglutinated foraminifera, however, become far more abundant, reaching their highest concentrations for the entire profile. The most common agglutinated foramanifer is the brackish water species, *A. runiana*, but also of note is the presence of some common marsh species (*Miliammina fusca, Trochammina macrescens* and *Trochammina inflata*) after ~3,130 BC.

4.7 Molluscs

A total of 74 samples were analysed for molluscs from Norsminde Fjord spanning ~6,900-2,000 BC, and this record is provided in Figure 4.8. A total of 13 genera were recorded, containing 12 taxa identifiable to species level. Overall preservation of molluscs was poor, resulting in very low absolute abundances throughout. Data are therefore expressed as presence/absence in Figure 4.8 and divided according to the diatom-derived zones (see section 0). This also means that palaeoenvironmental interpretation is limited and inferences must be treated with caution, but can provide some insight, particularly when considered in conjuntion with other proxy data (e.g. diatoms, foraminifera; see section 4.11).

NRS-DM1 (~7,000-6,600 BC) – At ~6,870, there is an individual occurrence of the gastropod *Lymnaea peregra* which can live in both freshwater and brackish-water up to ~ 8 ‰. By ~6,850 BC, undifferentiated cockle (*Cerastoderma* spp.) fragments are present, and slightly later (at ~6660 BC), *Mytilus edulis, Macoma balthica* and *Cerastoderma edule* all appear, which probably reflects increasing salinity.

NRS-DM2 – (~6,600-5,490 BC) – In the lowermost section of this zone (~6,600-6,520 BC), there are two recordings each of *Mytilus edulis, Scrobicularia plana* and undifferentiated *Cerastoderma* fragments. The presence of *S. plana* and an individual specimen of the mud snail *Hydrobia ulvae* (at ~6,520 BC) suggests that salinity was still rising throughout this zone. This is also further supported by finds of higher-salinity demanding species (e.g. *Hinia reticulata* at ~6,020 BC and echinid remains at ~5,890 BC) later in NRS-DM2. Molluscs become particularly scarce after ~6,400 BC, with *M. edulis*, the only species recorded in more than one sample.



Figure 4.8 - Norsminde molluscan record and other marine animals (i.e. zooplankton, echinoderms and fish). The values in brackets refer to the lowest mean salinity inhabited by the taxa (according to Sorgenfrei, 1958). Analysts Kaj Strand Peteresen and Peter Rasmussen, GEUS). All mollusc species, zooplankton, echinoderm and fish bone data are expressed as presence/absence.

NRS-DM3 – (~5,490-3,110 BC) – Molluscs become more regular and diverse for much of this zone, despite the overall abundance remaining low. *S. plana* is the most regularly recorded taxon throughout, whereas *M. edulis, H reticulata* and *Corbula gibba* appear several times in the lowermost section of this zone (pre-4,320 BC). Higher salinity is indicated between 5,140-4,910 BC, by the presence of several higher-salinity demanding species (including *Ostrea edulis, Tapes decussates, Bittium reticulatum* and echinid remains). After ~3,800 BC, molluscs become scarce again and with the exception of *S. plana*, no species are recorded more than once for the remainder of NRS-D_M3.

NRS-DM4 (~3,110-1,800 BC) – No molluscs are recorded in the lowermost section (~3,110-2,630 BC) of this zone, but recordings of *H. reticulata* at ~2,490 BC, and *S. plana* and *H. ulvae* at ~2,170 BC, suggests salinity has not changed dramatically from NRS-D_M3. Above this, there is an individual occurrence of *M. edulis* together with an undifferentiated *Littorina* spp. at ~2,210 BC and some undifferentiated fragments of *Cerastoderma* spp. and *Hydrobia* spp. at ~2,040 BC and 2,020 BC respectively. No more molluscs are recorded after this date.

4.8 Pollen

The Norsminde pollen record is provided in Figure 4.9 (selected taxa) and aquatic plants are presented in Figure 4.10, along with associated pollen metrics (pollen sum and PCA axis 1 and 2; see section 4.10.1). A total of 62 pollen samples were counted (by Peter Rasmussen, GEUS) spanning the period ~7,000-1,700 BC. For each sample, 499-1,365 terrestrial pollen and spores were counted, containing a total of 88 terrestrial genera (31 taxa identifiable to species). A further six swamp and fenland taxa and four aquatic taxa were also recorded. Pollen types have been divided into ten groups (see Figure 4.10). Terrestrial taxa (i.e groups from "trees" to "wet ground"; Figure 4.10) are expressed as percentages of the terrestrial pollen sum (TPS). Swamp, fenland and aquatic taxa are excluded from the TPS, but their percentages are expressed relative to the TPS. Overall, 19 taxa are present at >1 % abundance in one or more samples. The stratigraphic profile has been divided up into 6 statistically significant zones (using terrestrial taxa with >5 % abundance) using an optimal splitting technique (by information content) and results are discussed in relation to these divisions.



Figure 4.9 - Norsminde percentage pollen record (selected taxa). WG = wet ground taxa, S/F = swamp and fen taxa, Cult. = Cultivated taxa. (Analyst Peter Rasmussen GEUS). Data are expressed as percentages of the terrestrial pollen sum (TPS) with the exception of cultivated taxa which are expressed as presence/absence data. NB. Swamp and fen taxa are not included in the TPS but are expressed as percentages relative to the TPS.



Figure 4.10 - Norsminde pollen groups, aquatic taxa and pollen associated metrics (i.e. PCA axis 1 and 2 scores and pollen sum. (Analyst Peter Rasmussen GEUS, PCA analysis Jonathan Lewis). Pollen groups and aquatic group taxa are expressed as percentages relative to the terrestrial pollen sum (i.e. trees to wet ground).
NRS-P1 (~7,000-6,850 BC) – This zone consists of just one pollen sample (at ~6,940 BC), in which *Pinus* (~26 %) and *Corylus avellana* (~27 %) pollen are the most abundant tree taxa, followed by *Quercus* (~10 %) and *Alnus* (~9 %) and supplemented by smaller numbers of Betula (3 %), *Ulmus* (~4 %) and *Tilia* (~3 %). Poaceae is the most abundant of the non-arboreal pollen (NAP) taxa, with minor occurrences of *Pteridium aquilinum*, and *Carex* type. Swamp and fen taxa are at their highest levels for the entire profile, being predominately represented by *Dryopteris* type spores (~5 %, relative to the TPS). *Salix*, which likely grew on damp ground has its highest abundance in the entire profile.

NRS-P2 (~6,850-5,300 BC) – *Pinus* pollen abundance drops at the beginning of this zone as *Alnus, C. avellana, Betula, Ulmus* and *Tilia* all increase in importance. *C. avellana* is the most abundant pollen taxon at ~6,760 BC (~40 %) and again after ~6,170 BC (for the upper section of NRS-P2; 27-37 %), but its abundance drops dramatically (by ~over 20 %) between ~6,560-6.360. *Alnus* and *Betula* increase up until ~6,560 BC, at which point *Betula* reaches it highest level in the entire profile. Both *Alnus* and *Betula* continue to fluctuate in the upper part of this zone. A short term increase in the frequency of *Pinus* also occurs at ~6,560 BC, before turning relatively stable for the remainder of the profile. *Quercus* generally increases throughout and *Tilia* (4-6 %) and *Ulmus* (6-8 %) are relatively stable. For the NAP taxa, Poaceae accounts for ~5-7% of the TPS, but its frequency drops after ~6,360 BC (thereafter ~2-4 % of the TPS). *P. aquilinum, Carex* type and *Dryopteris* type are ever-present, but always in very low abundances (<2 %), following decreases at the beginning of this zone.

NRS-P3 (~**5,400-4,570 BC**) – *C. avellana* is initially high (~39 % at ~5,260 BC), but decreases in the lowermost section of NRS-P3 (5,260-5,050 BC), as *Betula, Quercus, Alnus* and *Ulmus* all exhibit increases. *Quercus* continues to increase up until ~4,830 BC, but steadily drops off again above this. *Corylus* begins to rise steadily after ~5,050, whereas *Betula, Alnus* and *Ulmus* all stop increasing around this time. Little change occurs in the NAP taxa, with Poaceae remaining marginally the most abundant taxon, but in reduced numbers compared to the previous zone. There is a minor increase in Chenopodiaceae pollen at ~4,650 BC and an overall further reduction in *Dryopteris* type spores. Most other taxa appear sporadically.

NRS-P4 (~4,570-4,000 BC) – Another large drop in *Corylus* (by >10 %) occurs at the onset of this zone, and again *Quercus*, *Alnus* and to a lesser extent *Betula* synchronously exhibit

minor increases. After this initial drop, *C. avellana* gradually increases throughout (along with *F. excelsior*), whereas *Betula* and *Quercus* fluctuate and *Alnus* numbers fall again (up until ~4,180 BC). *Pinus*, *Tilia* and *Ulmus* are relatively stable in NRS-P4 with the exception of a drop in *Ulmus* numbers in the upper sample (~4,030 BC) of this zone. Again there is little change in the NAP pollen assemblage compared to the previous zone (NRS-P3), with the exception of very slightly elevated numbers of Chenopodiaceae pollen and *Drypoteris* type spores.

NRS-P5 (~4,000-3,000 BC) – The onset of NRS-P5 broadly coincides with the widespread elm decline (Andersen and Rasmussen, 1993; Peglar, 1993; Peglar and Birks, 1993; Rasmussen, 2005) where *Ulmus* pollen numbers drops to ~2 % of the terrestrial pollen sum (at Norsminde) and remains low (2-3 %) for the rest of NRS-P5. At this point several taxa increase, most notably *Betula, Corylus* and *Alnus*. Between ~3,850-3,320 BC, *Alnus* and *Corylus* continue to increase throughout, with the latter representing ~30 % of the terrestrial pollen sum by ~3,070 BC. *Quercus* fluctuates, but overall exhibits a decreasing trend, as does *Tilia* and *F. excelsior* but to a lesser extent. For the non-arboreal taxa, higher percentages of Poaceae and Chenopodiaceae pollen are present. An overall drop in arboreal pollen (~4-6 % decrease) occurs after ~3,710 BC, as grass pollen continues to increase and is joined by slightly elevated occurrence of *Pteridium aquilinum* pollen and slightly later (~3,580 BC) by *Plantago lanceolata. Tritiucm* type first appears at ~3,580 BC and *Hordeum* type is regularly recorded in the record after ~3,320 BC indicating arable farming within the Norsminde catchment area.

NRS-P6 (~3,000-1,700 BC) – *Corylus, Quercus* and *Alnus* are the most abundant taxa, but are all subject to minor fluctuations, particularly in the lower part of this zone (up until ~2,300 BC). Above this, minor rises in *Alnus*, (~5 % increase between ~2,500-2,400 BC), and *Quercus* (by 5 % between 2,330-2,250 BC) occur, whereas *C. avellana* rises by ~10 % in the uppermost section (2,370-1,770 BC; as *Quercus* numbers fall again). *Ulmus* and *F. excelsior* both increase at the onset of this zone, the latter of which reaches its highest levels for the entire profile. Poaceae is again the most abundant non-arboreal taxon, generally accounting for ~2-4 %), but peaking between ~2,850-2,530 BC (along with *P. lanceolata* and coinciding with a substantial drop in arboreal pollen). Chenopodiaceae, *P. aquilinum* and *Artemisia* are all ever-present, joined by regular appearances from several other non-arboreal taxa (including *Rumex acetellosa, Urtica, Carex* type and *Filipendula*). *Hordeum* type

(cereal) pollen is also regularly present throughout this zone, again indicating the importance of agriculture within the Norsminde catchment ares. The aquatics *Ruppia* and *Potamogeton* both reappear in the record.

4.9 Plant Macrofossils

The Norsminde plant macrofossil record is presented in Figure 4.11, generally as presence/absence profiles. A total of 74 samples (spanning 6,920-1,810 BC) were analysed (by Peter Rasmussen, GEUS) from 8-12 cm thick core slices, with the volume of sediment analysed generally ranging between 78-117 cm³. Larger volumes of sediment (392 cm³) were analysed in the uppermost 13 samples (after ~2,210 BC). Sufficient material was present for AMS ¹⁴C dating (see section 4.3), but results are limited in terms of palaeoenvironmental inferences, due to the overall low concentration (0-85 per 50 ml, average 2.9 per 50 ml) of plant macrofossil material identifiable to genus/species level. In the 59 samples which contained identifiable material, 18 different genera were recorded, but only 5 taxa were identifiable to species level. Charcoal was commonly recorded throughout. The plant macrofossil results are discussed below in relation to the vegetation zones, as determined from the pollen record.

NRS-PM1 (~7,000-6,850 BC) – Undifferentiated bud, wood and deciduous leaf fragments are the only remains recorded from arboreal taxa in NRS-PM1. Swamp, reed and aquatic taxa are the most abundant, with the lowermost sample containing high concentrations of *Typha* seeds, *Equisetum* stem fragments and *Nymphaea* seeds. In the upper sample, *Chara* oospores, *Ruppia* fruits, *Potamogeton* and *Zanichellia palustris* fruits appear, suggesting low salinity at this time. *Typha* seeds however disappear, whereas *Equisetum* and *Nymphaea* remain present, but both decrease in concentration. Charcoal fragments are present in both samples, but the concentration is higher in the uppermost (i.e. most recent; 6,860 BC) sample of NRS-PM1.



Figure 4.11 - Norsminde plant macrofossil record. Groups are organised according to pollen groupings. (Analyst: Peter Rasmussen GEUS). Data expressed as concentration per 50 ml wet sediment (bars) or as presence/absence (circles). *= Gray silhouette refers to plant macrofossil concentration excluding undifferentiated leaf, bud, wood and twig fragments.

NRS-PM2 (~6,850-5,300 BC) – Macrofossils from some aquatic taxa (*Ruppia*,

Potamogeton, Nymphaea and *Z. palustris*) remain present in the lowermost section of this zone, but all disappear by ~6,520 BC. There are also a few levels with Chenopodiaceae remains and one level containing preserved *Juncus* seeds. Above ~6,520 BC, only undifferentiated buds and deciduous leaf fragments are regularly recorded. Charcoal concentration is relatively high between 6,800-6.600 BC, before dropping down to it lowest levels for the Mesolithic section of the Norsminde profile (i.e. pre 3,900 BC). However, it does begin to rise again shortly after ~6,170 BC.

NRS-PM3 (~5,400-4,570 BC) – Undifferentiated buds and deciduous leaf fragments are the only regularly recorded plant macrofossils in this zone, as overall concentration is low. Individual finds of a Poaceae caryopsis, seed remains from *Eupatorium cannabinum, Juncus* and *Typha* seeds, stem fragments from *Euqisetum* and *Cenococcum geophilum* sclerotia are the only supplements to this record. Charcoal is present in all samples, increasing up until ~4,910 BC, before dropping off again in the upper part of this zone (after ~4,910 BC).

NRS-PM4 (~4,570-4,000 BC) – There is little change in the plant macrofossil record in NRS-PM4. Deciduous leaf and bud remains are again present in most levels, supplemented by several scattered findings of wood fragments, *Juncus* seeds and Chenopodiaceae seeds. A slight increase in charcoal concentration occurs at the beginning of this zone, after which it remains stable.

NRS-PM5 (~4,000-3,000 BC) – *Betula* fruits appear in the record at ~3,800 BC and thereafter are fairly regularly recorded up until ~3,220 BC. Some *Alnus* remains are also found (after ~3,410 BC). *Cenococcum geophilum* also makes a couple of appearances in the lower part of the zone (up until ~3,740 BC) and several other taxa are present in individual samples (e.g. *Equisetum* stems, *Urtica dioeca* seeds, *Phragmites* caryopsis and *Chara* oospores). Buds and deciduous leaf fragments are regularly recorded throughout. However, the charcoal concentration drops to its lowest levels for the entire record, almost disappearing between ~4,000-3,610 BC.

NRS-PM6 (~3,000-1,700 BC) – There is a substantial increase in both the concentration and the diversity of plant macrofossils in this zone. The only identifiable arboreal remains are *Betula* fruits, which become common between ~2,650-2,200 BC. There are, however,

remains from several different species of grass and herb taxa present, including multiple recordings of Chenopodiaceae, *U. dioeca* and Poaceae caryopses. Representing the reed swamp taxa, are several appearances of *Juncus* and *E. cannabium* seeds and for the aquatics, *Ruppia* fruits and *Chara* oospores reappear in the record, being regularly recorded throughout NRS-PM6. Deciduous leaf fragments and bud remains are present in most samples, and several samples also contain wood and twig fragments. There is a large increase in charcoal concentration at the onset of this zone, and with the exception of a couple of samples between (~2,820-2,760 BC), charcoal remains high up until ~2,390 BC, after which the concentration decreases again.

4.10 Statistical analysis

4.10.1 Trends in species datasets

The major trends in the diatom dataset were assessed using the unimodal technique of DCA (axis 1 gradient length >2 SD; Table 4.3). For the entire diatom dataset (69 slides with assemblage counts), 26 % of the variation was explained on the first DCA axis, but only an additional 6 % was explained by the second axis. Axis 1 appears to be driven by the change in salinity (r=0.87, p<0.05; DCA axis 1 sample scores vs. DI-salinity; full model) and probably to some degree, valve preservation (r=0.72, p<0.05; DCA axis 1 sample scores vs. **F** index) associated with the transition from brackish to marine conditions within the fjord between ~6,900-6,600 BC. The low/medium salinity, robust, benthic diatoms (*E. turgida, R. curvata, D. didyma, C. placentula C. scutellum;* >3 % abundance), that dominate the NRS-D1 assemblage (see Section 0) score highest on DCA axis 1 (Table 4.4). The majority of other species, present during the marine phase of the profile, generally have weakly-positive to weakly-negative scores (i.e. after ~6,600 BC), with substantial clustering around zero (~- 0.4 - 0.4). This explains why DCA axis 1 sample scores for the entire dataset are high in NRS-D1, but decrease almost as low zero by ~6,360 BC (early in NRS-D2; Figure 4.6), and remain low for the rest of the profile.

Dataset	No. of samples	No. of species	Total inertia	Axis	1	2	3	4	Technique
Diatoms	69	392	2.194	Eigenvalues	0.571	0.136	0.079	0.063	DCA
(%)				Length of gradient	3.389	2.268	2.055	1.38	
				% variance explained	26	6.3	3.6	2.8	
	63	385	1.387	Eigenvalues	0.169	0.085	0.048	0.037	DCA
				Length of gradient	2.349	1.602	1.232	1.017	
				% variance explained	12.2	6.2	3.4	2.7	
Pollen (%)	62	88	1	Eigenvalues % variance explained	0.406 40.6	0.148 14.7	0.075 7.5	0.053 5.3	PCA*

Table 4.3 - Summary of Norsminde Fjord diatom and pollen DCA/PCA results. Analyses performed on entire datasets for both pollen and diatoms and a reduced dataset (63 samples) for diatoms (see text for details; section 4.10.1.). PCA performed only on entire pollen dataset, applying Atchisons log ratio (*) for PCA linear data.

To examine more subtle gradients of change within the dataset, during the marine phase, a secondary DCA was performed on a reduced diatom dataset, excluding the lowermost 6 'brackish' samples (NRS-D1; ~7,000-6,600 BC; Section 0). Only ~12 % of variance was captured by the new DCA axis 1 (Table 4.3), with again ~6 %, explained on the second axis. The low percentage variance explained by both axis 1 and 2, is probably due to the great variability within the dataset, with a wide range of species regularly exhibiting short term, minor fluctuations in abundance, probably caused by a variety of factors (see Section 4.11). In the reduced dataset (63 samples), species scoring highest (at >3 % abundance) are the planktonic (and some benthic) species present during NRS-D2, most notably Rhizosolenia spp., Chaetoceros cysts, Thalassionema nitzschioides and Cyclotella cf choctawatcheeana (Table 4.4). Species scoring lowest (>3 % abundance) include the *Navicula* species, that appear sporadically in the uppermost diatom zone, NRS-D4 (i.e. Navicula sp. A, Navicula tripunctata and Naviucla sp B). Other benthic diatoms, commonly present throughout the rest of profile (e.g. C. scutellum, O. mutabilis, O minuta, C. placentula, O. horstiana and A. helenensis; Table 4.4) but in lower numbers in the NRS-D2 score close to zero on DCA axis 1.

Therefore the main compositional shifts captured by DCA axis 1 in the reduced dataset are not related to salinity (r=0.12, p>0.05 (insignificant); DCA axis 1 sample scores reduced dataset vs. DI-salinity, full model) but to changes in the relative abundances of

benthic/planktonic species and is therefore probably driven by water depth within the fjord. Despite only 12 % of the variance being explained by axis 1, changes in DCA axis 1 sample scores therefore, do appear to provide a good summary of the major compositional shifts during the marine phase of the Norsminde profile. This is characterised by high sample scores in NRS-D2 ~6,360-5,570 BC, when an almost evenly mixed benthic/planktonic community existed within the fjord (~6,600-5,780 BC). Sample scores drop in NRS-D3 as the number of planktonic valves decrease, prior to a further reduction in NRS-D4, as the fjord begins to fill up with sediment and conditions become somewhat restricted (e.g. shallower water, reduced habitat, increased competition for nutrients with benthic diatoms).

The Norsminde pollen data exhibit a linear gradient (DCA, axis 1 gradient = 0.93; Table 4.3), and therefore major trends in the terrestrial pollen sum (excluding swamp, fen and aquatic taxa) were examined using the linear technique of PCA. Over 40 % of the variation was explained by axis 1, with a further ~15 % captured by the second axis. This suggests that axis 1 and 2 sample scores (Table 4.5) provide a good summary of the major compositional changes in the pollen dataset. Axis 1 appears to be predominately driven by a light-shade gradient, with closed canopy (shade-giving) forest dominating up until ~3,900 BC, after which it becomes a culturally influenced landscape containing higher abundances of light-demanding plants. *Betula, Pinus, Corylus, Ulmus* and *Tilia* all exhibit negative PCA axis 1 scores, and therefore drive the low PCA sample scores in NRS-P1 and NRS-P4 (~7,000-4000 BC). The negative scores for *Pinus* and *Betula* (pioneer trees) are due to their exaggerated abundance at the bottom of the profile, probably resulting from catchment erosion and subsequent inwash.

Quercus is an is at its most abundant in the pollen record during the late Atlantic, but due to its highest abundance in the post 3,900 BC period it scores zero. The relatively high PCA axis 1 score for *Quercus* (compared to most other closed canopy trees) is misleading, because it causes the PCA axis 1 sample scores to rise in NRS-P3 and NRS-P4 (together with a synchronous decrease in *Corylus*, between ~5,380-4,380 BC; PCA axis score -0.3). In this instance, rising PCA scores do not indicate a more open landscape, as at this time the Norsminde catchment on the high ground was covered by a dense closed canopy forest, dominated by shade-trees (most notably *Tilia*, *Ulmus*, and *Corylus*). Absolute lowest (negative) scores are exhibited by *Ulmus* and *Tilia*, whose numbers drop dramatically at

~4,180-3,970 BC, causing a shift from negative to positive PCA sample scores at the beginning of NRS-P5 (~3,970 BC).

P. lanceolata, Alnus, F. excelsior and Urtica exhibit the highest (most positive) species scores (for taxa >1 % abundance), all of which are most abundant in the Neolithic phase of the record (post 3,900 BC). Other taxa scoring relatively high on PCA axis 1 are Chenopodiaceae and Artemisia, which are again more abundant in the post 3,900 BC period, together with Rumex acetellosa (important on pastoral lands; Rasmussen, 2005) and Urtica, both of which are practically absent from the Mesolithic section of the pollen record. The positive scores for Alnus and Fraxinus could be considered somewhat anomalous, as both are shade trees, but are both more abundant in the post \sim 4,000 BC period, suggesting that other factors might also be important in driving axis 1 (e.g. climate changes, local water table changes, edaphic properties). The early Neolithic increases in Alnus and Poaceae (the latter due to forest clearances for agricultural purposes) pollen largely explain the gradual increase in sample scores, between (~3,980-3,450 BC; NRS-P5). After ~3,450 BC, PCA sample 1 scores are consistently high (above 0.4), but with a phase of elevated scores occurring over the short term deforestation phase (between 2,580-2,530 BC; see section 4.11), predominately driven by increases in P. lanceolata, Poaceae and Alnus and decreases in Quercus, Corylus and Pinus.

Taxa	Axis 1	Taxa	Axis 1	Таха	Axis 1	Taxa	Axis 1
Epithemia turgida	3.70	Staurosira elliptica agg.	0.05	Cyclotella cf. choctawatcheeana	4.94	Rhoicosphenia curvata	0.33
Amphora copulata	3.69	Glyphodesmis cf distans	0.04	Rhizosolenia spp.	4.09	Opephora minuta	0.29
Ctenophora pulchella	3.60	Amphora helenensis	0.00	Navicula salinicola	3.36	Grammatophora oceanica	0.21
Diploneis stroemii	3.54	Navicula sp. B	-0.01	Bacillaria paxillifer	2.79	Amphora helenensis	0.09
Navicula rynchocephala	3.47	Bacillaria paxillifer	-0.04	Thalassionema nitzschioides	2.61	Cocconeis placentula	0.05
Diploneis didyma	3.27	Opephora horstiana	-0.05	Navicula sp. 10A MOLTEN	2.58	Opephora mutabilis	0.02
Rhoicosphenia curvata	3.12	Opephora minuta	-0.11	Chaetoceros cysts	2.54	Navicula perminuta	0.00
Cocconeis placentula	2.88	Chaetoceros cysts	-0.12	Amphora graeffeana	2.39	Cocconeis scutellum	-0.01
Cocconeis scutellum	2.60	Naviucla ramossisima	-0.14	Navicula ramossisima	1.55	Opephora horstiana	-0.01
Tabularia fasciculata agg.	2.15	Dimmeregramma minor	-0.15	Fallacia tenera	1.30	Cocconeis neothumensis	-0.31
Grammatophora oceanica	2.08	Staurosirella pinnata	-0.16	Tabularia fasciculata agg.	1.29	Navicula sp. B	-1.14
Amphoras gacialis	0.93	Fallacia tenera	-0.25	Staurosirella pinnata	1.26	Navicula tripunctata	-1.88
Navicula sp. A	0.81	Navicula palpebralis	-0.26	Fragilaria gedanensis	1.16	Navicula sp. A	-3.10
Paralia sulcata	0.31	Thalassionema nitzschioides	-0.28	Fallacia cryptolyra	1.10		
Navicula perminuta	0.29	Fragilaria gedanensis	-0.36	Glyphodesmis cf distans	1.10		
Rhizosolenia spp.	0.24	Amphora graeffeana	-0.38	Dimmeregramma minor	0.71		
Cocconeis neothumensis	0.19	Navicula sp. 10A MOLTEN	-0.38	Amphora gacialis	0.50		
Navicula tripunctata	0.06	Navicula salinicola	-0.44	Staurosira elliptica agg.	0.47		
Rhopalodia acuminata	0.06	Fallacia cryptolyra	-0.48	Navicula palpebralis	0.40		
Opephora mutabilis	0.06	Cyclotella cf. choctawatcheeana	-0.82	Paralia sulcata	0.39		
	Α				В		

Table 4.4- Norsminde Fjord diatoms DCA axis 1 ranked species scores (taxa >1 % abundance only). A = Full dataset (69 samples; 6,900-2,000 BC), B= Reduced dataset (63 samples; 6,600-2,000 BC).

Taxa	Axis 1
Plantago lanceolata	0.8
Alnus	0.7
Fraxinus excelsior	0.7
Urtica	0.7
Artemisia	0.7
Rumex acetellosa	0.6
Poaceae	0.5
Chenopodiaceae	0.4
Pteridium aquilinum	0.2
Salix	0.2
Carex type	0.0
Quercus	0.0
Calluna vulgaris	0.0
Betula	-0.2
Corylus avellana	-0.3
Pinus	-0.6
Ulmus	-0.9
Tilia	-0.9

Table 4.5 - Norsminde Fjord pollen PCA axis 1 ranked species scores (taxa >1 % abundance only) for the entire dataset (i.e. 62 samples).

4.10.2 Environmental-species relationships

A top-down approach (i.e. lowest trophic level = response dataset, e.g. Anderson *et al.*, 2008) was implemented to explore the relationships between environmental parameters and change in the Norsminde diatom assemblage. The environmental 'predictor' dataset included palynological analyses (as predictors for changes in vegetation, catchment and land-use), arbitrary charcoal units (see below; for catchment/land use changes), organic matter, carbonate and minerogenic matter (abundance and fluxes), water content, total accumulation rate (as proxies for productivity, sediment load/type and catchment erosion), regional temperature (Lake Trehörningen, Sweden; Antonsson and Seppä, 2007) and sea level data (Blekinge, Sweden; Berglund *et al.*, 2005). Only 39 samples contained both palynological data and diatom assemblage data, at either corresponding stratigraphical levels or within a maximum of 25 years of one another. Therefore, environmental-species relationships were only explored using this reduced subset (39 samples), spanning 6,760-2,020 BC. Sediment age was factored out (i.e. included as a co-variable) to eliminate purely time-related processes.

To reduce long gradient lengths and reduce the effect of high value outliers, log (base 10) transformations were applied to the following environmental variables: sea level, temperature, water content, organic matter, minerogenic matter, organic flux, carbonate flux and minerogenic flux. Charcoal was included in the predictor dataset as a dummy variable, being assigned a value 1 or 0 (for each predictor sample), with 1 representing high charcoal concentration (>15 per 50 ml fresh sediment) and zero for low concentration (<15 per 50 ml fresh sediment). In cases of 'missing' overlaps for diatom and charcoal samples, the assigned value is based on the closest stratigraphic representative. Species-environment relationships were explored for both the entire dataset, and over 20 'moving window' analyses (after Bradshaw et al., 2005b), to examine changing temporal trends. Each subset consisted of 20 samples, with the uppermost sample being removed and a lower sample added (i.e. samples 1-20, 2-21, 3-22...etc) for each progressive run. For each reduced subset, DCA was performed on the species data to establish the gradient length of the first axis, and appropriate linear/unimodal techniques were selected accordingly (i.e. RDA_{linear} = <1.5, CCA_{unimodal} >1.5). For CCA analyses, rare species were downweighted and for RDA analyses, species data was log transformed (after Aitchison, 1986) and centred by samples and species.

Using CCA/RDA analyses, variables (from the predictor dataset) were manually selected using a Monte Carlo (n=999 permutation) forward selection process. The significance of the resultant variables (ranked in order of importance, following CCA/RDA analysis) were tested prior to selection, and a model was progressively built up to include only those that significantly explained variation in the species dataset (p < 0.05, with Bonferroni corrections applied for selection of each additional variable). The selection process was stopped when no further variables could significantly improve the model. Minority variables, present in less than 4 samples per subset (or sporadically occurring throughout) were omitted from selection, independent of their significance level. After selection, the significance of each model (i.e. subset of selected variables) was independently tested (i.e. PCA/RDA re-run, containing only those variables, with sediment age, again, factored out as a co-variable). All models illustrated in Figure 4.12 were significant (p < 0.05).

For the whole dataset (i.e. subset 1, all 39 samples; 6,760-2,020 BC), five environmental variables (sea level, temperature, *Tilia*, organic matter (%) and water content) significantly explained 26.9 % of the variation in the diatoms dataset. Sea level is an important determinant of water depth within the fjord and the degree of connection with the Kattegat

(and therefore salinity, saline water input, energy). The water content/dry mass suggests that over the entire dataset, sedimentary processes (probably supplying both allocthonous and autochthonous material), water table changes and perhaps flooding of sediments (e.g. increasing water content, particularly towards the bottom of the profile during the marine transgression) are important in driving changes in the diatom community. In the individual subsets, these factors are broken down more, with a series of individual sedimentary predictors appearing (organic matter (%), minerogenic flux and minerogenic matter (%); Figure 4.12 and see below). It is important to note, that as the physical parameters are expressed as percentage data, relative to one another (and therefore are often co-linear), apparent changes in one of these parameters might also/actually be related to change in another sedimentary physical parameter. This inevitably complicates the extraction of true environmental signals. The other three significant predictors (*Tilia*, temperature and organic matter (%)) in the first subset, also appear in the moving window analyses and are discussed in turn below.

For the moving window analyses, overall, climate-related processes seem to be important at the bottom of the profile (i.e. subsets 18-20; 6,360-3,430), catchment processes predominate in the middle section (subsets 10-17; 5,400-2,570 BC) and a variety of catchment, climatic and human related process in the upper section (subsets 2-9; 4,360-2,020 BC). The selected variables for each model were significant in explaining between 7.5-32.8 % (generally, 7-20 %, and average 16 %) of the variation in the diatom dataset. Maximum variance (32.8 %) is explained in the lowermost subset (subset 21; 6,760-3,960 BC), by minerogenic flux, organic matter (%) and water content. High organic content might act as a predictor for extensive bedding of aquatic macrophytes as suggested by the high abundance of aquatic plants present in the plant macrofossil record between ~6,930-6,600 BC (e.g. Ruppia spp., Nymphaea spp.). Macrophyte bedding could also explain the high abundances of benthic taxa, commonly found living as epiphytes (e.g. Epithemia turgida, Cocconeis placentula, Rhoicosphenia abbreviata, Grammatophora oceanica, Ctenophora pulchella), present at the bottom of the record (~6,860-6,740 BC). The high organic matter content might also indicate large quantities of plant matter being washed in from the surrounding catchment (e.g. high frequencies of wetland grass and *Carex* type pollen and *Dryopteris* type spores, some coarse minerogenic matter apparent in the lithology) and/or changing productivity within the fjord, perhaps due to variable nutrient status. This might be further supported by the relatively high $(70-170 \text{ mg cm}^2 \text{ yr}^{-1})$ and more variable flux of minerogenic material at the bottom of the

profile, likely to be due to increasing sea levels and marine inundation of low lying lands, subsequently causing inwash of terrestrial material and erosion from the catchment. Shallow water, regular mixing up of sediment and high minerogenic loads might explain the high (and likely exaggerated) abundances of heavily silicified diatoms (particularly *Epithemia turgida* and *Diploneis didyma*) in the lower section of the profile and could explain the poor diatom preservation at the bottom of the profile (i.e. lower **F** index values at the bottom of the profile; generally \mathbf{F} = 0.1-0.2 in the period 6,900-6,600 BC).

Subset	Samples	Upper age	Lower Age	Length of axis 1 gradient	Gradient technique	Variance explained (%)	Plantago lanceolata	Sphagnum	Minerogenic matter	Ruppia	Temperature	Tilia	Charcoal	Corylus avellana	Organic flux	Poaceae	Minerogenic flux	Organic matter (%)	Water Content	Sealevel
1	1-39	2021	6762	3.083	CCA	26.9														
2	1-20	2021	3960	1.954	CCA	25.7													-	
3	2-21	2288	4020	2.102	CCA	24.4													\dashv	
4	3-22	2352	4110	2.086	CCA	17.7												\vdash	\dashv	
5	4-23	2402	4165	2.065	CCA	17.4												\vdash	\dashv	
6	5-24	2438	4193	1.061	CCA	14.0													-	
7	6-25	2456	4222	1.606	CCA	13.7												\vdash	\dashv	
8	7-26	2519	4281	1.589	CCA	15.5												\vdash	-	
9	8-27	2547	4359	1.592	CCA	15.9												\vdash	+	
10	9-28	2574	4425	1.328	RDA	9.0													+	
11	10-29	2606	4550	1.325	RDA	8.6	-											\vdash	+	
12	11-30	2651	4648	1.5/5	CCA	1/.8										_			+	
13	12-31	2696	4688	1.564	CCA	16.0													-	
14	13-32	2881	4811	1.543		15.5	-											\vdash	+	
15	14-33	2904	4931	1.536		15.8						_							+	
16	15-34	3070	5156	1.497		8.3												\vdash	+	
1/	10-35	2424	5401	1.48/		12.7			\vdash				\vdash					\vdash	+	
18	1/-30	3424	5908	1.829		12.7			\vdash				\vdash				\vdash		+	
20	10-3/	2060	6264	1.5		11.0	-									_		\vdash	+	
20	20.20	2060	6762	1.8/4		22.0			\vdash				\vdash							
21	20-39	3900	0/02	1.05/	LUA	32.8														

Figure 4.12 - Environmental predictor-species response relationships for Norsminde Fjord (6,760-2,020 BC). Length of axis 1 gradient refers to DCA analysis on reduced diatom dataset (if DCA axis 1 gradient length > 1.5 s.d. then unimodal CCA employed to explore trends. If DCA axis 1 <1.5. s.d. then linear RDA technique used). Subset 1 = entire dataset (39 samples). Subsets 2-21 = moving window analyses (20 samples per window). Shadings: Red = Anthropogenic indicator, Black = sedimentary regime/catchment erosion, Green: vegetation change predictors, Grey = climatic and Blue = fjord depth and level of connection with Kattegat.

Temperature is an important factor up in subsets 18-20 (i.e. between ~6,360-3,420 BC). The direct role of temperature on controlling diatom communities has previously been critiqued

(Anderson, 2000), with indirect temperature-related processes likely to have greater impact upon diatoms communities than temperature itself (e.g. Kilham et al., 1996; Rippey et al., 1997). Here, it is likely that temperature rise (and later decrease; see below) influenced the entire catchment (e.g. Anderson et al., 2008), causing changes in the overall vegetation composition and structure, influencing edaphic processes, affecting net evaporation and run off volumes from the catchment, and affecting the abundances of grazing organisms such as molluscs and foraminifera (e.g. Yonge, 1960). Lower temperatures at the bottom of the profile might also have affected winter snow cover on the catchment and subsequent thawing events, or even sea ice formation, though the latter is unlikely due to the absence of sea ice diatoms in the profile. Thermal stratification in the late spring/summer (assuming the fjord was deeper in Mesolithic and early Neolithic period than it is today, due to higher sea levels and minimal isostatic change) might also have influenced diatom communities, particularly as temperatures warmed during the Holocene thermal optimum. Temperature is again important between ~4,280-2,440 BC (subsets 6-8). In this period, pollen-inferred air temperatures for the Lake Trehörningen area (Antonsson and Seppä, 2007) suggest decreasing temperatures and, around this time, the thermophilous species Hedera helix disappears (~3,850 BC) from the record (P. Rasmussen, unpublished data and see below). Again cooler temperatures (following the thermal optimum) in this section of the record might have caused greater catchment snow/ice cover. Large inputs of cold, fresh water following thawing events (during more extreme winters) might explain some of the variation in the diatom data and perhaps some of the salinity fluctuations in the DI-reconstruction.

The presence of *Corylus* (in subsets 12 and 13) and perhaps *Tilia* (significant in subsets 6-7 and 13), as environmental predictors suggest that at different times, restructuring of the forest had propagative effects upon the Norsminde Fjord ecosystem. Such changes in the forestry structure might affect the pH of the fjord (though unlikely due to large amount of saline water input from the ocean acting as a buffer for pH), nutrient inputs to the fjord, the speed of sediment delivery, sediments loads and subsequently water turbidity and depth. *Corylus* is dynamic in the phase in which it becomes a significant variable for change in the diatom dataset (i.e. between ~4,690-2,650 BC, subsets 12 and 13). The fall in *Corylus* pollen between ~5,260-4,380 BC (Figure 4.9), might be related to climate change and/or edaphic factors (see section 4.11), whilst its increase again after ~4,380 BC, probably reflects a change in forest structure.

Tilia, on the other hand is a climax tree, belonging to high ground and favouring welldrained soils (Iversen, 1969; Andersen, 1978). *Tilia* appears to be the most common tree of the primeval forest, being underestimated in pollen records due to its lower pollen production (Andersen, 1970b). Both *Corylus* and *Tilia* were prominent trees in the Danish landscape during the mid-Holocene, and therefore changes in these two species are likely to result in largescale changes across the entire catchment. *Tilia* might also act as a proxy for humaninduced land-use changes and forest clearance for agricultural purposes. A reduction in *Tilia* frequencies (together with *Quercus* and *Ulmus*), with simultaneously increasing *Betula* and *Corylus* are characteristic features of Iversen's landnam (Landnam is an old Norse term for 'taking land') (Iversen, 1941; Iversen, 1973). Signs of a landnam phase are evident in the pollen diagram between ~3,900-3,000 BC (Figure 4.9, see also section 4.11). Again, this is likely to alter catchment run off rates, sediment loads and nutrient inputs into the fjord and might explain some of the variations in the diatom data (e.g. fluctuations in *Fragilaria* abundances due to increased instability of the marine environment, caused by short term removal events and/or other human impacts upon the catchment).

P. lanceolata, important in two subsets (2 and 9), is another anthropogenic proxy, considered to be a common indicator of pastoral farming, growing on pastures and meadows (Iversen, 1941; Behre, 1981; Gaillard, 2007). The lowermost of these two subsets (i.e. subset 9, 4,360-2,550 BC) incorporates the time when the species attains a continuous presence in the record, implying the beginning of pastoral farming in the landscape. This suggests that either the clearance of land for animal husbandry or that free-grazing of animals influenced the marine environment (and subsequently the diatom community). Therefore, a variety of factors might be involved, including a more open catchment, increased surface run off, greater nutrient input (from natural fertilisers) and possibly less direct human pressure placed upon the marine ecosystem following the introduction of agriculture (e.g. Andersen, 1989). Subset 2 (3,960-2,020 BC) incorporates the peak in *P. lanceolata* values between ~2,580-2,530 BC, which is synchronous with a large decrease in arboreal pollen, signifying a short-term deforestation phase. It is important to note, however, that the pollen source area is likely to be far greater than the scale of the Norsminde Fjord catchment area and therefore changes evident in the vegetation record, might to some extent have taken place outside the catchment. Care must therefore be taken with the interpretation of pollen data and the potential effects that vegetation might have on the Norsminde Fjord diatom communities as

coincidental changes cannot be fully ruled out, particularly with pollen taxa low in abundance for which their presence in the local catchment is more uncertain.

The minerogenic flux significantly explains a small amount of the variation in the diatom dataset between ~5,400-2,900 BC (subsets 15-17), and is the only significant explanatory variable in subsets 15 and 16. As the minerogenic flux and charcoal concentration curve are closely related throughout the profile, it is likely that the predominant source of minerogenic matter is from the surrounding catchment (see below). In fact, later, charcoal itself becomes significant in several subsets (subsets 8-12; 4,650-2,520 BC) and, therefore both are likely to act as predictors for changes in the sedimentary supply of material from the catchment. Specifically, charcoal probably reflects land clearance and subsequent catchment/edaphic processes (discussed in more detail in section 4.11).

The mineorgenic flux appears to be more important when sediment accumulation rate is lowest, whereas the overall percentage minerogenic content is more important in the later subsets (i.e. subsets 2-5 ~4,170-2,020 BC), that incorporate the large scale rise in sedimentation (after ~2,800 BC). The minerogenic flux might also affect the diatom assemblage, with minimum diatom concentrations (~5,160-4,360 BC), occurring in the same period as the low minerogenic fluxes. This suggests that the minerogenic flux might play an important role in optimising the habitat for the diatoms, perhaps reducing abundance of mollusc beds and subsequently filter-feeding, though this is unlikely to disrupt the benthic diatom community. This period also coincides with maximum foraminifera and mollusc abundance (Figure 4.7 and Figure 4.8), and archaeological shell midden accumulation in the fjord (Andersen, 1989) suggesting a favourable environment for grazing organisms. Alternatively, it might just be a coincidental link, with minerogenic matter being reduced due to the coring location being more distant from the shoreline, reducing terrestrial input of minerogenic matter. Sea level is also at its highest during this phase (i.e. ~5,650-4,250 BC; Berglund et al., 2005), and therefore, by this point, the sea might have already claimed all low-lying land, lowering catchment erosion rates and causing the input of minerogenic matter to slow down and perhaps with it, the input of key nutrients required for the diatom community to thrive (e.g. drop in Chaetoceros cysts between ~5,600-4,650 BC might indicate reduced spring and/or summer diatom blooms and cysts production when nutrients become limited). These results seem to highlight the important influence that the sedimentary regime operating within the fjord has upon ecosystem structuring. Changes in the

sedimentary matrix may influence the diatom community in several ways, including reduced light penetration with increased turbidity of the water column (with increased sediment loads), variations in the substrate, coarsening of sediments and changes in higher trophic communities (e.g. zooplankton, molluscs). Some species of molluscs (and foraminifera) might also find it difficult to establish and survive under conditions of rapid sedimentation (e.g. Rasmussen, 1958; Yonge, 1960). The large increase in minerogenic matter flux after ~2,800 BC, and subsequent infilling of the fjord appears to have reduced the abundance of the planktonic diatom assemblage (Figure 4.6). Increased run-off rates from the surrounding catchment are also likely to have resulted in coarser sediment loads, which might explain the slight increase in abundances of epipsammic diatoms (e.g. *Opephora* species; Figure 4.5) after ~3,400 BC (see section 4.11).

Towards the top of the profile, Ruppia spp. becomes significant (in subsets 3-5; ~4,170-2,290 BC). This is probably a predictor for shallower water and (potentially lower salinity) as living conditions become more restricted and difficult (increased competition for space and nutrients, potentially increased grazing pressure, greater sediment disturbance in shallower water) in the fjord, as it begins to reach its sedimentation limit (see section 4.11). Specifically, beds of *Ruppia* spp. might also become important as a new epiphytic habitat for benthic diatoms (e.g. relatively high abundances of S. pinnata, though no increase in epiphytic diatoms is apparent), as well as providing protective habitats for grazers (e.g. increase in *Daphnia* spp. in upper sediments; Figure 4.8) and reducing light penetration to the bed of the fjord. Sphagnum is also a significant predictor in the uppermost subset (i.e. subset 2; 3,960-2,020 BC), which might represent an increase in swamp/fenland vegetation surrounding the fjord, probably due to decreasing sea level. Increased sediment load (after ~2,800-2,700 BC) and lower water levels probably caused silting up the fjord sides, causing the fjord to contract, resulting in the emergence of wet swampy land. Therefore, Sphagnum is probably another predictor for increased sedimentation rate, shallower water and restricted living conditions.

4.11 Synthesis of Results and site discussion

In this section, the results from the individual proxy analyses (described in sections 4.2-4.9) are bought together to provide a synthesis and discussion of the Norsminde Fjord environment and surrounding catchment area over the early-mid Holocene (~7,000-1,700

BC). In Figure 4.13, a summary diagram is provided of the major changes exhibited by these proxies (see Figure caption for details). Temperature (from Lake Trehörningen; Antonsson and Seppä, 2007) and sea level curves (from Blekinge, southeastern Sweden; Berglund *et al.*, 2005) have also been included to examine changes in relation to regional climate and marine environmental change. Five environmental zones have been included to aid interpretation labelled NRS-E1, NRS-E2, NRS-E3a, NRS-E3b and NRS-E4. An environmental zone has been assigned where at least two of the original optimally-split proxy zones (see above) fall within close temporal proximity to one another. The boundaries for NRS-E1/NRS-E2, NRS-E2/NRS-E3a and NRS-E3b/NRS-E4, have been determined by close agreement between the diatom and pollen derived zones. The boundary split for NRS-3a and NRS-3b, marks close correspondence between changes in the pollen and physical parameters profiles. Changes within Norsminde Fjord and its surrounding catchment (spanning ~7,000-1,700 BC) are discussed below, in relation to these environmental zones.

NRS-E1 (6,900-6,730 BC) - The marine transgression phase - Preservation of both microand macro- fossils is very poor at the bottom of the record, which therefore limits palaeoenvironmental interpretation. The lowest recorded marine microfossils occur at ~6,930 BC and belong to three brackish-marine diatom species (E. turgida, G. oceanica and T. fasciculata agg.), suggesting that the fjord was either already under marine influence at this time, or at least had received an input of marine water (e.g. a short term pulse). This is further supported by finds of brackish-water foraminifera (A. beccarrii and H. germanica) and Ruppia spp. remains at ~6,860 BC, the latter of which only tolerates salinities down to 3 ‰ (Moeslund et al., 1990). The first recorded mollusc at ~6,870 BC, is a well-preserved specimen of Lymaeae peregra, which can live in both freshwater and brackish water with salinities up to 8 ‰ (Mandahl-Barth and Bondesen, 1949), suggesting that the fjord might still have been under freshwater conditions at this date. The first unambiguous brackish water molluscs appear at ~6,850 BC (Cerastoderma spp.) Establishing the exact timing of the limnic-brackish transition is difficult due to some probable signs of re-working and redeposition of sediments. Almost all diatom valves present between (6,930-6,870 BC; Figure 4.4) are heavily fragmented (and scarce) and the only identifiable valve fragments belong to robust diatoms, which to some degree, might be able to survive the re-deposition process.



Figure 4.13- Summary diagram of the major changes occurring at Norsminde Fjord (~7,000-1,700 BC). *= Gray silhouette refers to plant macrofossil concentration excluding undifferentiated leaf, bud, wood and twig fragments. Full model, refer to Di-salinity including all fossil taxa, reduced model refers to diatom-inferred salinity excluding *Fragilaria* s.l. from the fossil dataset. Regional sea level curve from Blekinge, SE Sweden (Berglund *et al.*, 2005) and pollen-based temperature record from Lake Trehörningen, Sweden (Antonsson and Seppä, 2007). †m a/b/ psl = metres above/below present sea level.

Nevertheless, the bulk of evidence suggests that some time between ~6,930-6,870 BC the first inundation of marine water into the Norsminde Fjord basin occurred, and that a full marine connection was definitely established before ~6.800 BC. This agrees well with other local estimates for the area such as Århus harbour (in-situ pine stump dated to ~7,200 BC; Dencker, 1999) and Århus Bay (oldest marine mollusc ¹⁴C dated to ~6,800 BC; Jensen and Bennike, 2008). Prior to this, it is likely that a lake or a stream existed at the coring point, but the exact date of the transition from limnic-brackish conditions cannot be further constrained, due to both poor preservation and/or reworking of sediments. This transition might have begun as intermittent pulses of saline water, but again this is difficult to establish. This would, however, provide an alternative explanation to re-deposition/dissolution, for findings of highsalinity tolerant diatoms, such as P. sulcata and G. oceanica, pre-dating (and after co-existing with) low-salinity fresh/weakly brackish water diatoms (e.g. N. scutelloides, N. rhyncocephala, A. copulata, E. turgida; Figure 4.4). It is important to note, that although this community almost certainly reflects low salinity conditions (e.g. 1 ‰ at ~6,860 BC), the precise DI-estimates of salinity in the lowermost samples must be treated with caution. The accuracy of these inferences is questionable due to the low concentration of diatoms and dominance of robust diatoms (e.g. E. turgida, 58 % at ~6,860 BC). It is highly likely that dissolution of the more brittle and less silicified taxa has resulted in exaggerated representation of some taxa in this zone skewing the inferred salinity.

Above this (i.e. after ~6,800 BC), the salinity appears to increase rapidly, as the Norsminde basin begins to develop as a marine system and sea level continues to rise. The diatom flora increases in diversity as the salinity rises and preservation improves, and brackish-marine diatoms become more abundant, replacing the low-salinity, robust diatoms characteristic of the bottom few samples (pre-6,600 BC). Molluscs are low in number and foraminifera absent in the upper section of this zone, probably due to poor preservation, low salinity and difficult living conditions (e.g. high sediment loads). The overall accumulation rate is relatively high (0.15 mg cm⁻² yr⁻¹) in NRS-E1, which is largely driven by the input from the surrounding catchment. This is suggested by the increase in minerogenic matter between 6,680-6,650 BC, and general agreement between the minerogenic flux and charcoal concentration curve (both within this zone and throughout the profile). Erosion of the catchment and drowning of low lying land during the marine transgression probably best explains this.

At the very bottom of the record, most major tree taxa are present (except possibly *F*. *excelsior*), but with *Pinus* and *C. avellana* being the most abundant pollen taxa. *C. avellana*

pollen accounts for 25-40 % of the TPS in this zone, suggesting that it featured prominently in the local landscape, together with *Pinus* and *Alnus* and to a lesser extent *Quercus, Ulmus* and *Tilia*. The arrival of *Tilia* in western and northwestern Jutland has been dated to ~7,000 BC, but it may have arrived earlier than this in the Norsminde area, as it already account for ~3 % of the TPS in the lowermost sample (~6,940 BC). Whether *F. exclesior* was present in the Norsminde area before ~5,380 BC, is uncertain due to low pollen frequencies. The particularly high numbers of *Pinus* (e.g. 26 % at ~6,940 BC), suggests that it was locally abundant, probably growing on wet soils or peats, but the high pollen percentages could be due to over-representation due to buoyancy of the saccate (vesiculate) pollen grain (P. Rasmussen, unpublished data). *Salix* is also at its highest numbers in this lowermost zone, probably also growing in these damper areas.

The high abundance of wet ground and marsh/fenland plants, present in both the pollen and plant macrofossil assemblages, suggest that the fjord was immediately surrounded by marsh/wetlands. The relative abundances of these taxa, however, may be exaggerated, due to catchment erosion and subsequent washing in of sediment from the sides of the fjord. *Carex* type pollen (~2 %) and *Dryopteris* type spores (~5 %) are the main indicators in the pollen record of damp/marshy conditions, but remains of *Typha* seeds and *Equisetum* stem fragments are also found in the plant macrofossil record. The majority of the grass pollen before ~5,750 BC, is most likely from wetland species such as *Phragmites australis* and *Phalaris arundinaceae*. Remains of aquatic plants are also relatively high in the earliest phase of the record, particularly *Nymphaea* spp., *Chara* spp., *Zanichellia palustris* and *Potamogeton* spp.. These taxa can tolerate low salinity and live in freshwater, and therefore, they cannot provide any further information regarding the timing of the marine transgression.

NRS-E2 (~6,730-5400 BC) – The early marine phase – Salinity continues to increase rapidly and by ~6,600 BC, DI-salinity (in both models) exceeds 20 ‰ (~15 ‰ rise between 6,760-6,600 BC). Water depth also appears to increase, resulting in a large increase in the proportion of planktonic diatoms present, which account for approximately half the diatom community between ~6,600-5,780 BC. According to the DI salinity reconstruction (with all taxa), a further increase in salinity occurs after ~5,880 BC, which broadly coincides with the second major sea level transgression documented in the Blekinge stratigraphy (i.e. L2 transgression, ~5,750-5,250 BC, Berglund *et al.*, 2005; Figure 4.13). Under the full model, this rise is interrupted by a peak of *Fragilaria* s.1. (most notably *S. pinnata* and *F. gedanensis*) at ~5,400 BC, which causes DI-salinity to fall to ~16 ‰ (discussed in more detail below;

NRS-E3). DI-salinity under the reduced model does not show this drop (Figure 4.13). Mollusc abundance remains very low in this zone, but as the water depth and salinity increases, preservation conditions for foraminifera appear to improve. Foraminifera become more common after ~6,500 BC, but abundances of individual species are somewhat variable (most notably *H. germanica* and *E. excavatum* fo. *selseyensis*), which might indicate fluctuating salinity. A peak of *E. magellanicum* at ~6,130 BC suggests a relatively open connection with the Kattegat at this time (Knudsen and Konradi, submitted).

The rise in minerogenic matter between ~6,680-6,650 BC coincides with rapidly increasing salinity (as evident in both DI-reconstructions; see Figure 4.5) within the fjord. As indicated above, this is probably due catchment erosion, as water level increases, further claiming low-lying land and washing material into the fjord. This is supported by the relatively high numbers of Poaceae pollen (most likely from wetland species) and wetland taxa (*Carex* type and *Dryopteris* type), which are more abundant in the record, up until ~6,170 BC. The accumulation rate begins to fall after ~6,180 BC. Interestingly, the entire phase of lowest accumulation throughout the Norsminde profile broadly coincides with highest sea levels (Mörner, 1976; Berglund *et al.*, 2005) and temperatures (Antonsson and Seppä, 2007) for the southern Baltic area (i.e. ~6,000-4,200 BC; Figure 4.13). When sea level is higher, a more open connection with the Kattegat might result in more material being lost to the open sea and increased energy within the system might lead to greater decomposition of material. This suggests that sea level (affecting accommodation space and level of connection with the Kattegat), climatic factors and later, human impact all influence the sedimentary regime of Norsminde Fjord (Rasmussen, submitted).

There is little change in the forest tree species present, with *C. avellana* still being the most abundant pollen type for the majority of this zone. The curves for *C. avellana, Betula* and *Pinus* all exhibit fluctuations in the lower section of this zone (e.g. *C. avellana* pollen drops by ~20 % (of the TPS) between 6,760-6,560 BC, with *Betula* and *Pinus* pollen simultaneously increasing), before stabilising after ~6,200 BC. This coincides with the reduction in both the flux of minerogenic matter and a drop in the overall accumulation rate. This probably means lower rates of catchment erosion, and therefore might indicate that the earlier fluctuations (between ~6,760-6,170 BC) in *Corylus, Betula* and *Pinus* abundance, were caused by erosion and re-deposition of sediments (P. Rasmussen, unpublished data), rather than large-scale changes in the forest composition. *Quercus* pollen frequencies rise throughout, suggesting

that it became more prominent in the landscape, accompanied by relatively extensive occurrence of *Alnus*, *Ulmus* and *Tilia*.

In terms of non-arboreal taxa, the low porportion of grass pollen (the majority of which after ~6,000 BC, most likely originate from dry ground taxa) and dry ground herbs (most abundant being P. aquilinum), suggests a closed-canopy forest with a sparse field-layer vegetation. Two halophyte groups (Chenpodiaceae and Artemisia) become more common in NRS-E2, probably growing on shore meadows along the edge of the fjord. The latter of these taxa is ever-present after 6,500 BC, which broadly coincides with the establishment of fully marine conditions in the fjord, and probably the promotion of shore-meadow plant communities (P. Rasmussen, unpublished data). Both summer and winter temperatures at this time were probably warmer than today, perhaps up to 2.5°C warmer in summer, and able to support the growth of the thermophilous species, Hedera helix and Viscum album within the catchment (Iversen, 1944; Antonsson and Seppä, 2007; Seppä et al., 2009; Brown et al., submitted; P. Rasmussen, unpublished data). Pollen and spores from wet habitats taxa (most notably Carex type and *Dyropteris* type) are less abundant than in NRS-E1, but remain present, probably still important in the nearshore area of the fjord. Due to lower rates of catchment erosion and greater distance away from the coring location under higher sea level conditions, they feature less prominently in the profile. This is again supported by a drop in the minerogenic flux in this zone. Aquatic species practically disappear from the plant macrosfossil record somewhere between 6,600-6,500 BC, consistent with the transitions to high salinities (>20 ‰ under both models), though minor quantities of *Ruppia* spp. and *Potamogeton* spp. are found up until ~6,360 BC in the pollen record (likely originating from fresher, upstream parts of the fjord).

NRS-E3a (5,400-4,050 BC) – The middle maximum phase – High salinity is indicated by all three marine proxies in the lowermost section of this zone. There is, however, some difference between the two diatom reconstructions, regarding the nature of salinity change in this part of the record. If *Fragilaria* s.l are removed, DI-salinity appears to rise steadily from 5,870-5,160 BC. However, a peak in *Fragilaria* species (*S. pinnata* and *F. gedanensis*) at ~5,400 BC results in a low-reconstructed salinity (~16 ‰) under the full model, but it is difficult to assess whether this is a true salinity decline, or just an opportunistic increase in *Fragilaria* s.l. (often abundant in unstable systems such as fjord/estuarine systems or recently isolated/transgressed basins (Eronen, 1974; Stabell, 1985; Seppä *et al.*, 2000; Miettinen, 2002). The period between ~5,550-5,360 BC is almost devoid of foraminifera, which might support a low salinity inference at this time (see below). However, it is clear that the

Norsminde Fjord environment is dynamic between ~6,000-5,000 BC, with the sedimentation rate decreasing and planktonic abundance (most notably, *T. nitzschoides* and *Chaetoceros* cysts, after ~5,570 BC) dropping, perhaps due to increased instability in the fjord (e.g. rise in Fragilaria s.l abundances), changing nutrient status and or increasing salinity, greater marine connection with increased sea level and increased inwash of robust marine diatoms of *P. sulcata* (e.g. increasing *P. sulcata* abundances between ~5,570-5,160 BC) . Therefore, changes in other parameters might also be influencing these communities.

While it remains unclear as to whether, salinity began rising at ~5,800 BC, or rose quickly after ~5,400 BC, all three proxies document high values between ~5,160-5,140 BC (~DIsalinity of 24 ‰ with full model; 29 ‰ under reduced model). High-salinity demanding molluscs appear (e.g. O. edulis, T. decussatus, B. reticulatum) at ~5,140 BC. E. incertum is the main foraminiferal indicator of high salinity (requiring at least 20-25 ‰; Knudsen and Konradi, submitted), particularly between (~5,160-4,900 BC) where it reaches its maximum abundance. It is further supported by small numbers of Elphidium voorthuysenii and Haynesina depressula (found only at salinities >24 ‰ in marginal areas of Kattegat and Skagerrak; Alve and Murray, 1999). This might also suggest that foraminifera preserve better under higher salinity conditions at Norsminde Fjord, which has also been observed at Horsens fjord, situated ~25 km south of Norsminde (Kisum, 2007). The maximum salinity phase appears to persist for longer in both the molluscan and foraminifera records (up until ~4,340 BC in the foraminiferal record than the diatom record, but both the DI-reconstructions suggest salinity has fallen again by 4,940 BC). In the succeeding phase (after ~5,160 BC), evidence from both diatoms (characterised by high and fluctuating percentages of Fragilaria s.l.), and later (after ~4,340 BC) the foraminifera (fluctuating H. germanica and E. excavatum concentrations) suggest variable salinity (and/or other environmental parameters).

A late peak in *Chaetoceros* cysts occurs between (~4,300-3,800 BC), possibly due to the increased sedimentation rate and increased input of nutrients from the catchment. This might result from both falling sea levels (see Figure 4.13; Berglund *et al.*, 2005) and later, the marked changes in forest structure associated with the *Ulmus* decline at Norsminde (beginning around ~4,100 BC) (see NRS-E3b) and/or human disturbance of the catchment. An alternative explanation might be that nutrient status is actually higher in Norsminde fjord under lower sea level, with shallower water perhaps preventing nutrient from the catchment from being washed out to sea. This might also be supported by the lowest diatom concentrations between ~5,400-4,300 BC, broadly coinciding with maximum sea levels and

lowest accumulation of sediments (Figure 4.13). The reduced input of sediment (particularly minerogenic matter) from the catchment during higher sea levels (i.e. ~6,200-3,900 BC) might also be important for providing stable habitats for molluscs and foraminifera, thereby placing added grazing pressures upon the diatom community.

Salinity might also play a role, with some evidence for lower DI-reconstructed salinities occurring simultaneously with high *Chaetoceros* cyst numbers, though there are uncertainties attached to the DI-reconstruction at this time (see below; NRS-E3b). Dissolution might also explain lower diatom concentrations, but neither the **F** index, or *P. sulcata* dissolution index suggest any shift towards poorer preservation. Diatom concentration increases again after ~4,430 BC, broadly coinciding with peak values for *Chaetoceros* cysts (and decreasing sea level; Figure 4.13) and thereafter remains relatively high.

Arboreal pollen is at its most abundant in phase NRS-3a and rising numbers of *Quercus*, together with high numbers of *Ulmus*, *Alnus* and a marked increase in *F. excelsior* (after ~5,380 BC) suggest that the Norsminde catchment is dominated by a dense forest over this period. *Corylus* pollen is still abundant but decreases throughout this zone, possibly being out-competed by *Quercus* and other climax trees. Alternatively climatic and/or edaphic factors may have contributed to its decrease. The very low percentages for grasses and dry ground herbs (1-4 %) suggest a continued densely forested landscape in this phase. There is very little change in the abundances of *Carex* type and *Dryopteris* type pollen, and minerogenic flux rates are at their lowest. Aquatics are practically absent in this phase of the record.

NRS-E3b (4,050-3,050 BC) – The late marine phase – The foraminifera indicate lower salinity in this phase, but this is not evident in the diatom flora or DI-salinity reconstruction (Figure 4.14). Calcareous foraminifera become much sparser (and less diverse), as the higher salinity species disappear from the record and those that remain (most notably *H. germanica* and *E. excavatum*), are subject to greater fluctuations in numbers. Molluscs also become fewer in number, but there is, however, little change in species composition compared to the uppermost section of NRE-3a. There is also very little change in the diatom flora, compared to the preceding phase. *Fragilaria* s.l. taxa continue to fluctuate, which inevitably cause fluctuations in the DI-salinities, making it difficult to isolate clear patterns of salinity change.



Figure 4.14 - DI- reconstructions with added loess smooth (0.25 span) for Norsminde Fjord between ~6,600-2,100 BC.

Therefore, on the basis of the multiproxy evidence presented here, lower salinity in the early Neolithic can only be considered equivocal at best. Furthermore, interpretation must be viewed in the light of likely large short-term variability (e.g. decadal-centennial scale fluctuations in DI-salinity possibly up to 5 ‰ in magnitude) and must consider the dynamic nature of the region at this time. Lowering of sea level, elevated sedimentation rate, forest restructuring (following the elm decline) and human impact upon the landscape all occur (see below) in this phase. Therefore, again salinity may not be the only factor influencing the diatom assemblage at this time, and therefore the apparent short term shifts in salinity, might actually be driven by small scale restructuring of the community, driven by other forcing factors (e.g. weak salinity relationship with DCA axis 1 for the reduced diatom dataset; 6,600-2,000 BC). In terms of the diatom community, perhaps one of the most important additional factors is water depth (likely to be indirectly exerting pressure through depth related process; e.g. Weckström and Juggins, 2006, and see section 6.2), as the shallowing of the fjord probably accelerated in this zone, driven by the higher accumulation rates (~0.17 mg cm⁻² yr⁻¹ after ~4,000 BC) and the drop in sea level between 4,200-3,800 BC (see Figure 4.13). Benthic diatoms gradually increase in number throughout this phase, as both the tychoplanktonic and planktonic community are reduced (*P. sulcata* decreases after ~4,080 BC and *Chaetoceros* cysts after ~3,800 BC). Variable nutrient status again, might also be influencing the diatom community (see NRE-3a).

In NRS-3b, large scale changes occur in the pollen record, with *Ulmus* values dropping down to as low as ~2 %, accompanied by increases in Betula, Alnus and Corylus. This suggests a restructuring of the forest, resulting from the reduction of elm in the catchment. Disease and/or human impact upon the landscape are the most commonly cited causes for the decline of elm in northern Europe (Troels-Smith, 1960; Ten Hove, 1968; Perry and Moore, 1987; Peglar and Birks, 1993; Parker et al., 2002; Rasmussen, 2005), but it is difficult to isolate the exact cause. At Norsminde, there is no direct evidence for human interference from the palaeoenviromental records, at the time of the elm decline. Poaeceae remains low and unambiguous agricultural indicators (e.g. cereals, *P. lanceolata*) do not appear until later in the record. Charcoal concentrations are also at their lowest for the entire profile between ~4,060-3,610 BC, which might suggest little activity in terms of fire-induced forest clearance by humans at this time. The archaeological evidence from surrounding sites (e.g. shell middens) provide evidence for changes in cultural practice (Andersen, 1989; Enghoff, 1989; Enghoff et al., 2007) at the time of the Mesolithic/Neolithic transition, but again nothing that directly suggests forest clearances between ~4,180-3,710 BC. This probably means disease is most likely to be responsible for the elm decline at Norsminde.

A gradual increase in Poaceae pollen frequency occurs throughout, which is probably related to both a natural reduction in forest density associated with the elm decline and the introduction of agriculture. The first hard evidence for agriculture appears shortly after ~3,600 BC, when a continuous curve for *P. lanceolata* begins, signifying pastoral farming (Iversen, 1941; Behre, 1981; Gaillard, 2007). Pollen from cultivated crops also appears around this time; *Triticum* type at ~3,580 BC and slightly later, but much more commonly thereafter, *Hordeum* type pollen appears at ~3,320 BC.

Signs of Iversen's classic landnam (Iversen, 1941; Iversen, 1973) are identifiable between the elm decline and ~3,000 BC, with decreasing *Ulmus, Tilia, Quercus* and *Fraxinus* values paralleled by rising *Betula* and *C. avellana* values accompanied by increases in Poaceae and *P. lanceolata* and the presence of cereal pollen. These events are inferred as the cumulative effects of many short-term clearance episodes followed by arable and pastoral farming. Whether or not the catchment was frequently disturbed by fire after ~3,500 BC is inconclusive. *Betula* remains high throughout, but changes in *Populus* and *Pteridium aquilinum* are minimal. Charcoal exhibits a minor increase at ~3,610 BC, but overall remains relatively low, probably mainly being washed in from the catchment, rather than air-borne (see below).

NRS-E4 (3,050-1,700 BC) – The infilling phase – The fjord almost certainly becomes shallower in this phase as it approaches its sedimentation limit and living conditions become somewhat restricted (reduced habitat, increased competition for nutrients amongst benthic diatoms, increased grazing pressure). This is suggested by both the reduction in planktonic diatom ratio (which begins decreasing at ~3,150 BC, and again after ~2,400 BC) and the rise in agglutinated marsh foraminifera (*T. inflata*, *M. fusca* and *T. macrescens*). Despite sea level records suggesting that sea level is slowly decreasing throughout this zone (Figure 4.13), rapid infilling with sediment is the main cause of this shallowing, particularly after ~2,800-2,700 BC, when the average accumulation rate almost trebles within the basin (from 0.17 to 0.44 cm yr⁻¹). This probably explains why salinity (as inferred by both diatom reconstructions) does not really decrease in this upper section (i.e. after ~3,000 BC) of the profile, which would be expected under rapidly falling sea levels. In fact, DI-salinity (especially under the reduced model) suggests two phases of increased salinities between ~3,100-2,500 BC, which might also explain the re-accumulation of oyster dominated middens at Norsminde over the Pitted Ware cultural period (~2,800-2,400 BC; Andersen, 1983, discussed in section 6.3.2.1). Likely fluctuating salinities and sporadic peaks of benthic diatoms suggest living conditions are highly variable within the fjord after ~3,000 BC.

What exactly caused these large increases in the flux of organic matter, carbonate and minerogenic material, in just over 100 years (from 2,850-2700 BC), remains equivocal. Rasmussen, (unpublished) suggests that anthropogenic processes and subsequent catchment soil erosion best explains the contemporaneous increase between the minerogenic and organic components. Catchment erosion is further supported by the more regular appearance of pre-Quaternary foraminifera after ~3,000 BC (Knudsen and Konradi, submitted), which must

have been washed in from the surrounding catchment following erosion from pre-Quaternary sediments. In addition, a more open catchment (and subsequent higher run-off rates), a shallower fjord, narrower Kattegat entrance (perhaps due to silting up of the entrance and development of beach ridges; Andersen, 1989), sea level change, high productivity (e.g. high diatoms concentration, high organic content), extensive bedding of macrophytes trapping sediments and climate change might all, also have contributed either directly or indirectly to the infilling of the Norsminde basin.

Calcareous foraminifera are almost absent, and the few remains that are found do not shed light on the salinity of the fjord. Mollusc remains also become even sparser in this phase, which probably reflects both poor preservation and poor living conditions, most notably, shallower water, higher sediment loads into the fjord and perhaps coarser sediments. An increased number of epipsammic diatoms, such as *O. minuta* and *O. mutabilis* (Sabbe and Vyverman, 1995) might also suggest a coarsening of sediments, which is consistent with shallower water and likely more physical turbulence and wave action. Despite this, there is little change in the molluscan faunal composition, compared to the previous phase. The presence of *H. reticulata* at ~2,490 BC and *S. plana* at ~2,170 BC, however, both suggest that medium-high salinity existed within the fjord at these dates, which supports the diatom inferred salinity (under both models).

Quercus, Alnus and *Corylus* are again the most abundant pollen taxa, but the latter of these experiences a drop at ~3,070 BC, as another overall shift in the forest structure occurs in this phase. *Betula* decreases, while *Ulmus, F. excelsior* and *Tilia* abundance all increase, indicating a change in land-use pattern and forest density. The persistent occurrence of relatively high percentage values for Poaceae and *P. lanceolata*, being joined by *R. acetellosa* after ~2,930 BC, indicates a continued presence of grassland. Arable farming is documented by sporadic occurrences of *Hordeum* type pollen. Between ~2,580-2,530 BC there is a substantial increase in Poaceae (~5-7 %) and *P. lanceolata*. At the same time, there is a distinct drop in the percentage of arboreal pollen (~10 % decrease), which signifies a marked and short term episode of deforestation. High concentrations of charcoal are present in the plant macrofossil record in this phase (particularly between 3,000-2,530 BC), which might reflect forest clearance (via burning) for agricultural purposes.

However, increased charcoal concentrations correspond with increased minerogenic fluxes into the fjord, which rather suggests that the likely cause is inwash from the catchment, as

(see Figure 4.13). Therefore, care must be taken with the interpretation of the charcoal record, as this is not a direct proxy for fire, but could rather represent catchment erosion, changes in input loads or source area. There is no direct evidence for air-borne charcoal input (e.g. no correspondence exists between the non-arboreal pollen curve and the charcoal record at Norsminde), which acts as a more direct proxy for fire, and therefore it can only be speculated that the contemporaneous high charcoal concentrations and vegetation signals of human-induced forest clearance and land-use change are related. Aquatic plants increase again in this phase, providing more evidence for infilling and shallowing of the fjord, enabling species such as *Chara* spp. (Figure 4.11) and *Ruppia* spp. (Figure 4.10, Figure 4.11) to spread across large areas of the fjord. After ~1,700 BC, Norsminde Fjord became too shallow to accommodate any regular (stratigraphic) accumulation pattern and since then sediments have been continuously disturbed, thereby preventing palaeoenvironmental analysis of the late Holocene at this site.

4.12 Summary and conclusions

- Norsminde was first inundated around ~6,900 BC, with a lake or stream existing at the coring point prior to this date. Salinity rises rapidly in the succeeding phase, reaching DI-salinities values of above 20 ‰ by ~6,600 BC. Exaggerated abundances of wet ground and swamp and fen taxa in the pollen and plant macrofossil records probably indicated catchment erosion and inwash from surrounding wetlands immediately surrounding the fjord as sea level rise claims low lying lands.
- A further increase in salinity is apparent in the reduced DI-salinity model between ~5,800-5,100 BC, which broadly coincides with rapid sea level rise in other parts of the Baltic basin and possibly global sea level rise. Highest salinities for the Mesolithic period are inferred in the period between ~5,400-4,850 BC (i.e. in the DI-reduced salinity model, foraminifera and molluscan records).
- Salinity appears to drop again by ~4,800 BC and thereafter, both DI-salinity models suggest salinity fluctuates (generally between ~17-22 ‰ in the full model, with the exception of two pronounced peaks between ~3,000-2,900 BC and 2,700-2,570 BC) throughout the remainder of the Norsminde Fjord sequence. Fluctuating salinity is particularly apparent in the full model, due to the high and variable abundances of *Fragilara* s.l. taxa, which are opportunistic species, able to survive in unstable habitats (i.e. subject to constant changes in important environmental parameters such as

salinity and nutrient status). This is likely to have implications for the accuracy of DIsalinity reconstructions in some sections of the sequence.

- Decreased rates of sedimentary accumulation appear to coincide with maximum sea levels (i.e. lowest rates in the period ~6,000-3,900 BC), suggesting that sea level exerts an important control on the sedimentary regime at Norsminde, prior to human disturbance of the landscape (after ~3,900 BC).
- Multivariate 'moving window analyses' (using CCA and RDA techniques) suggest that climate, sea level rise and associated processes/responses (e.g. minerogenic input from both marine resources and from the catchment as land is claimed) significantly explain some of the variation in the diatom dataset in the bottom section of the profile (i.e. 6,760-2,900 BC).
- The first indicators of large-scale agricultural impact upon the landscape appear in the pollen record at ~3,700-3,600 BC (increase in Poaceae pollen, beginning of a continuous curve for *Plantago lanceolata* and appearance of cereal crops). Multivariate analysis (CCA and RDA) suggests that human-related catchment changes (e.g. *Plantago lanceolata*, *Corylus avellana*, *Tilia*, Poaceae and charcoal) are important in explaining a significant proportion of the variation in the diatom dataset in the 'moving window' analyses covering the time period ~4,930-2,020 BC.
- DCA results for the full diatom dataset suggests that DCA axis 1 is driven by salinity, but this changes to depth and when the low-salinity samples at the bottom of the profile are removed (i.e. samples between ~7,000-6,600 BC). For the pollen dataset, PCA (linear data) axis 1 appears to be predominately driven by a light-shade gradient, with shade-giving trees dominating in the lowermost section of the record (i.e. ~6,900 -3,900 BC), while higher abundances of light-demanding plants are present in the record as the catchment become culturally disturbed.
- After ~3000 BC, benthic diatom taxa increase in abundance and submerged macrophyte taxa (most notably *Ruppia* spp and *Chara* spp.) reappear in the pollen and plant macrofossil records as the fjord begins to shallow due to sedimentary infilling and declining sea level. This infilling accelerates after ~2,800 BC, following a large increase in the accumulation rate, most likely due to land-use change and subsequent catchment soil erosion (supported by increase in charcoal and pre-Quaternary foraminifera from the catchment). Predictors of shallower water conditions (e.g. *Ruppia* spp.) also appear to explain some of the variation in the diatom dataset in the uppermost section (i.e. moving window analyses covering the period 4,160-2,020 BC) of the Norsminde Fjord record.

Chapter 5. Results: Korup Sø

5.1 Introduction and aims

This section presents the results from a multiproxy palaeoenvironmental investigation at Korup Sø, aimed at reconstructing changes in both the terrestrial and marine environment of the Djursland area over the early-mid Holocene (~7,600-2,200 BC) years. The sedimentary analyses described in this chapter include core lithology and sedimentation history and palaeoecological analyses include diatoms (analysed as part of this thesis), pollen (H. Krog, unpublished data) and molluscs (Petersen, 1993). Salinity has been reconstructed quantitatively using both a diatom-based salinity transfer function (see section 2.10.2) and a weighted-average minimum salinity molluscan transfer function. This chapter specifically aims to address the following questions:

- 1. How was chronological control achieved at Korup Sø in order to be able to consider changes in the environmental record in their proper context?
- 2. How does the sedimentary environment at Korup Sø change over the mid-Holocene as reflected by the sedimentation rates?
- 3. What are the main changes in the structure and composition of the marine biological communities (diatoms and molluscs) present at Korup Sø over the mid-Holocene?
- 4. How does the salinity at Korup Sø change over the mid-Holocene as inferred by diatoms and molluscs?
- 5. How does vegetation of the surrounding area change over the mid-Holocene?
- 6. To what extent do changes in the terrestrial (e.g. vegetation, land-use) and marine (e.g. sea level change, accumulation rates, productivity) environments affect the diatom community?
- 7. What are the most significant variables driving the changes in the marine biological communities and do these variables change throughout different sections of the record?

5.2 Core Lithology

Descriptions of the lithology were made by Harald Krog and Bent Odgaard (Geological Survey of Denmark, now Geological Survey of Denmark and Greenland). These original descriptions (for Well no. 1) are presented in Table 5.1 and a summarised lithology column is provided in Figure 5.1. Due to drainage and modern agricultural use of the Korup Sø area, the last ~4,200 years are not present in the sediment record. The top 50 cm of the record are disturbed by drying up of the former basin and agricultural activities. Below this, the majority of the profile is marine clay gyttja, with varying amount of shells (occasional layers) and other macrofossils. Between 727-753 cm (5,510-5,940 BC), the sediment was particularly rich in shells (notably *Ostrea edulis, Mytilus edulis* and Barnacle spp.) exhibiting characteristics similar to a natural shell bank. Slight colour changes occur towards the bottom of the profile and between 753-849 cm (5,940-7,080 BC), small dark lumps of peat material were present. There is eventually a shift to a limnic white-grey calcareous gyttja at 865 cm (~7350 BC), which was deposited before the Korup Sø was inundated by the sea (see also Iversen, 1937).

Depth (cm)	Lithological description
0-50 cm	Disturbed
50-727 cm	Olive-grey-green, clay-gyttja with shells of <i>Cerastoderma edule</i> and <i>Mytilus edulis</i> . Scattered <i>Ruppia</i> spp. fruit-stalks until 150 cm depth.
727-753 cm	Olive-grey, clay-gyttja with more clay and sand than in the layer above. Rich in shells and has the character of a natural shell bank with shells of among others <i>Ostrea edulis</i> and <i>Mytilus edulis</i> plus barnacles.
753-846 cm	Predominantly dark olive-brown, clay-gyttja with a relatively high content of peat with woody material. <i>Cerastoderma</i> shells until 765 cm depth. Shell layer between 842 and 846 cm depth. Lower boundary slanting and irregular.
846-849 cm	Dark olive-grey, clay-gyttja with dark lumps of peat material like in the layer above. Shells of <i>Cerastoderma edule</i> and <i>Mytilus edulis</i> . Lower boundary irregular and gradual (not sharp).
849-853 cm	Dark olive-grey, clay-gyttja with few shells. Lower boundary gradual.
853-856 cm	Dark olive-green, clay-gyttja, with scattered small snails.
856-865 cm	Light olive-green, clay-gyttja with scattered small snails. Lower boundary sharp.
865-881 cm	Light white-grey calcareous gyttja.

 Table 5.1 - Original lithological desciptions of the sediments from well no. 1 Korup So, (H. Krog and B. Odgaard, Geological Survey of Denmark, now Geological Survey of Denmark and Greenland).



Figure 5.1 - Summary of Korup Sø lithological profile (based on original description by Harald Krog and Bent Odgaard (Geological Survey of Denmark, now Geological Survey of Denmark and Greenland). For colour changes, see Table 5.1.

5.3 Age Model

Details of all radiocarbon dates obtained at Korup Sø are provided in Table 5.2. Twelve shell samples (5 from well no. 1 and 7 from well no. 2) were dated at Copenhagen Carbon-14 Dating Laboratory and results are reported in (Petersen, 1993). Four additional samples were dated more recently from Well no. 1 at the AMS ¹⁴C Dating Centre at Aarhus University (Table 5.2) using methods described in 2.5. The final Well no. 1 Age model (produced by J. Olsen, Aarhus University; Figure 5.2) includes 8 (out of the 9) ¹⁴C dates. Date AAR-4569 (645 cm depth, *Hinia reticulata*) was excluded from the model because of a low agreement index with the applied model (J. Olsen pers. comm.). The age-depth models for core 1982 and core 1983 are constructed using depositional models in OxCal 4.1 (Ramsey, 2008, 2009). The ¹⁴C samples are calibrated to calendar years using the IntCal04 calibration curve (Reimer *et al.*, 2004) with a ¹⁴C reservoir age correction of 400 ± 50 yrs. The availability of pollen records for both Korup Sø wells (H. Krog unpub.) enabled eight unambiguous (i.e. clearly visible in both profiles) pollen makers Table 5.3 to be identified. These pollen markers have been built

into the OxCal depositional model as additional chronological constrains on the age model. The depositional model has been constructed using a k value of 10 yielding model agreement indices A_{model} of 58.2% (J. Olsen pers. comm.).

All lithological and biological (diatom, molluscan and pollen) analyses described below are based on sedimentary material from Well no. 1, but the Well no. 2 age model is also presented (Figure 5.2, Table 5.2) here as it provides spatial information about the sedimentary environment at Korup Sø over the study period (~7,500 -2,200 BC). The Well no. 2 age model is based on 7 AMS ¹⁴C dates.



Figure 5.2 - Korup Sø age models. Correlated dates refer to the stratigraphic pollen links (Table 5.3) identifiable in both wells and incorporated into the depositonal models (J. Oslen, P. Rasmussen, unpub.). Analyses performed as part of this thesis are all from core-1983 sediments (well no. 1).
Laboratory Code	Depth cm	Material dated (marine molluscs)	Age* ¹⁴ C BP	Age (cal. BC)	δ13C ‰ VPDB	δ18O ‰ VPDB
Well No. 1	Korup Sø 1983					
K-4233	134	Cardium edule	4369 ± 65	2,644-2,395 (95.4 %)	-3.3	
K-4234	168	Cardium edule	4290 ± 65	2,736-2,492 (95.4 %)	-2.7	
AAR-4567	295	Cerastoderma sp.	4275 ± 50	3,082-2,867 (95.4 %)	-1.3	-2.7
AAR-4568	495	Cerastoderma edule	4840 ± 50	3,630-3,405 (95.4 %)	-0.3	-2.3
AAR-4569	645	Hinia reticulata	5325 ± 50	Excluded	-1.0	-1.2
AAR-4570	702.5	Mytilus edulis	6140 ± 55	5,226-4,921 (95.4 %)	-1.0	-1.0
K-4235	736	Mytilus edulis	7161 ± 100	5,845-5,480 (95.4 %)	-1.3	
K-4236	749	Ostrea edulis	7402 ± 100	6,057-5,686 (95.4 %)	-0.4	
K-4237	849.5	Mytilus edulis, Cardium edule	8401 ± 110	7,355-6,809 (95.4 %)	-1.6	
Well No. 2	Korup Sø 1982					
K-3681	212.5	Cardium edule	4271 ± 85	2,708-2,478 (95.4 %)	-1.0	
K-3984	215	Cardium edule	4259 ± 85	2,712-2,582 (95.4 %)	-3.6	
K-3986	712.5	Mytilus edulis, Cardium edule	5012 ± 90	3,690-3,459 (95.4 %)	-0.6	
K-3987	875	Mytilus edulis, Cardium edule, Ostrea edulis, Macoma baltica	5321 ± 90	4,010-3,702 (95.4 %)	-1.0	
K-3988	1155	Corbula gibba	7652 ± 110	6,254-5,900 (95.4 %)	-0.3	
K-3989	1187.5	Cardium edule	7779 ± 110	6,426-6,125 (95.4 %)	-3.2	
K-3990	1187.5	Corbula gibba	7790 ± 110	6,418-6,120 (95.4 %)	-2.5	

Table 5.2 - Details of ¹⁴C dated samples from the two Korup Sø boreholes (well no. 1 & well no.2). *The dating results are reported according to international convention (Stuiver and Polach, 1977) as conventional ¹⁴C dates in ¹⁴C yr BP (before AD 1950) based on the measured ¹⁴C/¹³C ratio corrected for the natural isotopic fractionation by normalizing the result to the standard δ^{13} C value of -25‰ VPDB Andersen *et al.*, 1989).

Pollen	Core 1982	Core 1983	Pollen event	
link	Depth (cm)	Depth (cm)		
8	175	145	Late Hystrix/Dinoflagellate peak	
7	305	199	Upper Corylus peak/Quercus dip	
6	405	301	Lower Corylus peak/Quercus dip	
5	515	420	Second Plantago lanceolata peak	
4	575	449	Major Hystrix/Dinoflagellate peak	
3	735	549	First Plantago lanceolata peak	
2	905	660	Ulmus peak (before decline)	
1	1155	775	Pinus peak (before decline)	

 Table 5.3 – Unambiguous pollen stratigraphic marker levels incorporated into the depositional age model

 (J. Olsen, P. Rasmussen, unpublished data).

5.4 Sediment accumulation rate

The Korup Sø sedimentation rates for both wells are shown in Figure 5.3. The predominant pattern for both wells is low accumulation of sediments up until ~3,900 BC, at which point the rate increases sharply and remains high for the upper section of each profile. Super-imposed upon this general pattern are a number of smaller shifts, which are described below for Well no. 1 only (analysed as part of this thesis). Between ~9,000-3,900 BC (Mesolithic or Atlantic section of the record), the Well no.1 accumulation rate ranges between ~0.04-0.09 cm yr⁻¹ (average 0.07 cm yr⁻¹), with highest rates (i.e. ~0.09 cm yr⁻¹) occurring between ~7250-6,300 BC. Lowest values for the entire Well no. 1 profile occur in the period between ~5,000 - 3,900 BC.

The accumulation rate in the Neolithic (i.e. post ~3,900 BC) period is far more variable, though this may be explained to some extent by the higher number of dating/tie points for this period. Highest accumulation rates (~0.5 cm yr⁻¹) for the whole Well no. 1 profile occurs between ~3,600-3,500 BC, after which there is relatively sharp decrease down to ~0.25-0.3 cm yr⁻¹ (up until ~3,200 BC). Between ~3,200 BC-2,800 BC the accumulation rate is ~0.40-0.45 cm yr⁻¹, prior to a short term drop (~300 years) down to its lowest (~0.23 cm yr⁻¹) Neolithic/Subboreal level (i.e. post ~3,900 BC). Another minor rise occurs in the uppermost section of the profile (0.37 cm yr⁻¹; 2,500-2,300 BC). The overall pattern of sedimentation rate is very similar for both wells situated ~300 m apart, but absolute sedimentation rates are higher in Well no. 2.



Figure 5.3 – Sediment accumulation rates at Korup Sø (for both the1982 and 1983 wells). Lithology column included for the 1983 well (see Table 5.1 and Figure 5.1 for key).

5.5 **Biological Analyses**

5.5.1 Diatoms

A total of 52 diatom slides were counted spanning ~2200-7,500 BC at an average temporal resolution of ~100 years with increased resolution over the Mesolithic-Neolithic transition and the period of the Pitted Ware and Single Grave Culture see Table 2.2). A total of 201 taxa were identifiable to species level (from 71 genera), with 31 species having an abundance of >3% in one or more slides. The Korup Sø diatom record (>3 % abundance) for is presented in Figure 5.4 together with a diatom-inferred quantitative salinity reconstruction. Figure 5.5 includes a range of associated metrics, including diatom concentration, diatom dissolution (**F** index), benthic: planktonic: tychoplanktonic and centric: pennate ratios, percentage of taxa incorporated into the salinity model and species richness (Hills N₂ diversity). Three statistically significant diatom assemblage zones have been identified (using taxa >3 %;

Figure 5.4) using optimal splitting, together with one further sub-zone (added for ease of interpretation).

KRP-D1 (~7,330-6,930 BC) – *Fragilaria* sensu latu (s.l.) dominate the zone, with *Pseudostaurosira brevistriata* (~60 %) the most abundant species at ~7,330 BC, but accompanied by smaller numbers of *Staurosira construens* fo *venter* and *Staurosira elliptica* agg. The presence of *Chaetoceros* resting cysts and the occasional valves of *Paralia sulcata* and *Dimeregramma minor* indicate some degree of saline water input into the system at this time. *P. brevistriata* quickly disappears and the upper sample of KRP-D1 (~7,180 BC) is dominated by *S. construens* (*venter* and *subsalina* varieties), *Staurosirella pinnata* and *Rhopalodia constricta*, yielding a DI-inferred salinity of ~9 ‰. Very few planktonic species (e.g. low centric:pennate ratio and high B:P:T ratio) are present and diatom concentration and diversity are low.

KRP-D2 (~6930-5830 BC) – Diatoms indicating higher salinity (e.g. *Paralia sulcata, D. minor*) become more abundant, and overall diversity and concentration increase. Benthic diatoms still dominate with high percentages of *Fragilaria* s.l. (*S. pinnata & S. elliptica* agg.) and *Opephora (mutabilis* and *minuta* agg.) species present. Planktonic diatoms increase in abundance, however, around ~6,300 BC, with an increase in *Chaetoceros* cysts and a peak of *Cyclotella striata* at ~6,270 BC. Small amounts of the high nutrient favouring diatom, *Cyclotella choctwatcheeana* are also present between ~5,970-5,930 BC. A drop in DI-salinity occurs at ~6,000 BC, driven mainly by the decrease in *P. sulcata* and *D. minor* and the increase of *Fragilaria* S.L. (*S. pinnata, S. elliptica* agg., *O. mutabilis*, and *.O minuta* agg.).

KRP-D3a (~**5830-5490 BC**) - Salinity reaches it highest levels (between 19-27 ‰) in this zone, driven by the highest percentages of the marine diatom *P. sulcata* (salinity optima 19.6 ‰ in training set), for the entire profile. *S. pinnata* (~13-26 %) and *S. elliptica* agg. (~5-10 %) are also abundant, but both *O. minuta* and *O. mutabilis* decline. Small amounts of D. minor are also present in the benthos. *Chaetoceros* cysts are the most abundant of the true planktonics (~3-8 %), as *C. choctawatcheeana* disappears. Overall diatom concentration and diversity decreases, whereas the centric: pennate ratio and tychplanktonic abundances reaches their highest levels (due to the high percentage of *P. sulcata* valves).



Figure 5.4 - Korup Sø percentage diatom record (>3% abundance). Taxa sorted by salinity optima in diatom training set (values quoted in brackets). Error bars for DI-salinity reconstruction plotted in Figure 5.5.



Figure 5.5 - Diatom associated metrics.

KRP-D3b (5,490-3,330 BC) – *P. sulcata* drop markedly at the beginning of KRP-D3b and with it the DI-salinity also falls, thereafter fluctuating between 18-20 ‰ for the majority of the zone. One exception to this is a short phase between ~3,840-3,770 BC, where it drops to as low as ~16-17 ‰. *S. pinnata* is the generally most abundant species, but percentages do tend to fluctuate. *Chaetoceros* cysts increase at the beginning of this zone (>15 % between ~5,470-4,730 BC) and remain relatively high until ~3,530 BC, after which the decline sharply (down to <2%). *O. mutabilis* and *S. elliptica* agg. both increase after ~4,060 BC. Several species also exhibit minor increases between (4,000-3,800 BC), including *Fallacia fenestrella* agg, *Fallacia cryptolyra, C. choctwatcheeana, S. pinnata, P. sulcata* and *Opephora horstiana* agg. A further increase in *O. mutabilis* and *O. minuta* occurs at ~3,430 BC.

KRP-D4 (3,330-2,290 BC) – There is a further reduction in the planktonic community in this zone as *Chaetoceros* cysts decrease. Benthic diatoms account for ~83-97 % of the assemblage, with high numbers of *Fragilaria* s.l. present throughout (most notably *S. elliptica* agg. and *S. pinnata*, which co-dominate). Marine benthic species such as, *F. fenestrella* agg., *D. minor*, and *A. helenensis* are present in low abundances and begin to disappear (successively) after ~2,780 BC. DI-salinity fluctuates throughout, but overall exhibits a downward trend in this zone, dropping from ~20 ‰ at ~3,280 BC, to ~11 ‰ by the uppermost sample (~2,290 BC). At this point a series of more brackish diatoms appear in the record, most notably *Psammothidium kuelbsii* (~6 %) and *Achnanthes submarina* (~4 %). Superimposed upon this general trend are two DI higher-salinity peaks at ~2,840 BC (19 ‰) and ~2,380 BC (18 ‰), both characterised by a decrease in *Fragilaria* s.l and minor increases of higher salinity-tolerant diatoms (e.g. *Achnanthes brevipes* and *D. minor* at ~2,840 BC, *O. minuta* and *F. cryptolyra* at ~2,380 BC).

Diatom Preservation

Valves preservation is generally good (**F** index > ~0.3 for the majority of the profile and range 0.22-0.86), but **F** index values decrease (i.e. greater dissolution) towards the top of the profile as the fjord gets shallower (0.25-.0.54 after ~3,400 BC, discussed in section 5.7). In all samples, the majority of *Paralia sulcata* valves fall into dissolution categories 1 and 2.

5.5.2 Molluscs

Molluscan assemblages from 65 samples (volume 166-3475 cm³, 3-30 cm slices) were analysed by Kaj Strand Petersen (Petersen, 1993) spanning the time period ~2,300-7,500 BC with an average temporal resolution per sample of ~81 years (26 – 398 yrs). A total of 25 species were recorded (from 20 different genera), 11 of which were bi-valves and 14 gastropods. The full molluscan record (and supplemented statistics) is shown in Figure 5.6. Supplemented statistics include mollusc concentration, bivalve-gastropod ratio, DCA axis 1 sample scores and a weighted-averaged minimum salinity reconstruction (MI-salinity). Molluscan abundances are expressed as concentrations per L⁻¹, with square root transformation for improved presentation of data (in Figure 5.6). The following section describes the Holocene molluscan assemblages sequentially based on the implementation of statistical zoning, application of a new age model and application of a weighted averagesalinity transfer function (see section 2.10). Five statistically significant zones have been determined (via optimal splitting by information content, on all species with >5 % abundance; see section 2.10.1) with three additional zones, added to aid interpretation (additional zones do not affect the significant optimal splits).

KRP-M1 (7,500-7,300 BC) - This zone consists of just 2 samples, both of which are low in concentration and yield a combined total of 2 species; *Cerastoderma edule* (edible cockle), present in both samples and *Omalogyra atomus* (atom snail) at ~7,350 BC. Reconstructed MI-salinity in this zone is low (~6 ‰).

KRP-M2a (7,300-6820 BC) – MI-minimum salinity increases in this zone (~10-14 ‰) as *O. atomus* is replaced by brackish-marine taxa, most notably *Hydrobia ventrosa* and *Hydrobia ulvae*. These *Hydrobia* species reach particularly high concentrations between ~7,060-7,010 BC (*H. ventrosa* ~2200 per L⁻¹ at ~7,060 BC and *H. ulvae* ~420 per L⁻¹ at ~7,010 BC). Diversity also increases at this time as smaller numbers of *Mytilus edulis, Onoba vitrea, Hinia reticulata, Retusa trunculata* and *Rissoa membrancea* are present in the record. *H. ulvae* remains the most abundant mollusc species in the upper part of the zone, but overall molluscan concentration drops dramatically after ~7,010 BC.



Figure 5.6 Korup Sø molluscan record (from data originally published in Petersen, 1993). Taxa sorted by minimum salinity based on values stated in Sorgenfrei, 1958. Mollusc abundances and total concentration expressed as concentration per L^{-1} (with square root transformation applied to the data for presentation purposes). Dotted lines indicate optimally split sub-zones. It is important to note that sub-zones have added to aid interpretation, and are therefore not necessarily statistically significant splits (see section 2.10.1).

KRP-M2b (6,820-5,930 BC) – Throughout the majority of this zone (~6,800-6,150 BC), molluscan concentration and diversity are low. *H. ulvae* is the most abundant throughout, supplemented by regular appearances from *C. edule, M. edulis, O. vitrea* and *R. trunculata*. A large increase in molluscan concentration occurs at ~6,150 BC, predominantly driven by a ten-fold increase in *H. ulvae* concentration. A number of higher salinity species also appear at this time including *R. trunculata, Ostrea edulis* and *Corbula gibba*. The MI-minimum salinity estimates range between ~13-15 ‰.

KRP-M3a (**5,930-5,240 BC**) – *O. edulis* and *R. albella* continue to increase, together with the first appearances of a number of other high-salinity demanding species; *Bittium reticulatum, Brachystomia eulimoides* and *Tapes decussates. H. ulvae*, remains the most abundant species throughout, and *C. gibba* and *R. trunculata* reach their maximum abundances (for the entire profile) in this zone. MI-minimum salinity is high throughout (~19-26 ‰) reaching its maximum level for the whole record between ~5,430-5,250 BC. Molluscan concentration is variable, rising up until ~5,600 BC, prior to dropping off again in the uppermost section of KRP-M3a.

KRP-M3b (5,240-3,910 BC) – *H. ulvae* concentrations fluctuate intensely, causing both the MI-minimum salinity (16-23 ‰) and molluscan concentration to fluctuate accordingly (i.e. low salinity and high concentration between ~5,160-5,070 BC and 4,670-4,530 BC). *B. reticulatum* is abundant throughout, suggesting relatively high salinity and is supplemented by several sporadic appearances of *H. reticulata* and *R. albella*.

KRP-M4 (3,910-3,100 BC) - *H. ulvae* is again the most abundant species throughout this zone, with *M. edulis*, the only other ever-present, and the rest only making brief or sporadic appearances (e.g. *C. edule, H. ventrosa*). The higher-salinity tolerant species, *B. reticulatum* and *T. decussates* both disappear by ~3,530 BC, but *O. vitrea* remains present throughout. MI- minimum salinities are lower in this zone (fluctuating between ~12-17 ‰) and generally exhibit a decreasing trend. Molluscan diversity and concentration are also relatively low throughout.

KRP-M5a (**3,100-2,560 BC**) - *H. ventrosa* and *H. ulvae* are the generally most abundant species, with the exception on the uppermost sample (*C. glaucum* most abundant at~2,590 BC). MI-minimum salinity rises between ~2,930-2,750 BC (reaching ~19 ‰ at ~2,750 BC),

predominately driven by the reappearance of *O. vitrea* in the record. *R. membranacea* is everpresent and *M. edulis* and *C. glaucum* are present in most samples. *C. edule* also re-appears at ~2,850 BC and *Littorina tenebrosa* (lagoon periwinkle) makes is first appearance at ~2,750 BC. After ~2,600 BC, the MI-minimum salinity drops again, falling as low as ~7 ‰ by ~2,590 BC.

KRP-M5b (2,560-2,330 BC) – Molluscan concentration steadily increases, due mainly to the increasing numbers of *H. ventrosa* (most abundant species in KRP-M5b). *C. glaucum* steadily decreases throughout, and small numbers of *O. vitrea* and *R. membranacea* are present in the lowermost section, but disappear by ~2,480 BC and ~2,430 BC respectively. *L. tenebrosa* and *H. ulvae* disappear slightly later at ~2380 BC. Both salinity and diversity are low throughout, with the MI-minimum salinity initially rising, up to ~10 ‰, where it stabilises for the rest of the zone.

5.5.3 Pollen

A total of 110 pollen samples were counted from Well No.1 at an average resolution of 50 years. Between 737 and 1257 terrestrial pollen grains were counted per sample, containing a total of 89 terrestrial taxa (9 identifiable to family level only, 72 genera with 32 taxa identifiable to species level), 5 swamp and fen taxa (from 5 genera, 2 of which were identifiable to species level a further 7 aquatic taxa (7 genera, 2 identifiable to species level). Besides the pollen and spores *Pediastrum* spp. (freshwater green algae), dinoflagellates and hystrichosphaerids were recorded in the pollen samples. Percentage abundances for these organisms are calculated relative to the pollen sum (see Figure 5.8 caption). Data for these are however expressed independently, relative to the terrestrial pollen sum. A pollen diagram including selected terrestrial taxa is provided in Figure 5.7 and Figure 5.8 includes percentage curves for major plant groupings (as for Norsminde; see section 4.8), PCA sample scores for axis 1 and a summary of the aquatic pollen, and other aquatic organisms (e.g. *Pediastrum*, Hystrix, Dinoflagellates) present in the palynological slides.



Figure 5.7 - Korup Sø percentage pollen record (selected taxa). Solid zone lines indicate significant zones (determined using the broken stick method; Bennett, 1996). Analyst: Harald Krog, GEUS. (Abbreviations: WG = wet ground). Data are expressed as percentages of the terrestrial pollen sum (TPS) with the exception of cultivated taxa which are expressed as presence/absence data. NB. Swamp and fen taxa are not included in the TPS but are expressed as percentages relative to the TPS.



Figure 5.8- Major plant groupings percentage pollen diagram for Korup Sø and aquatic species (including aquatic plants, Pediastrum, dinoflagellates and hystrix; all recorded during palynological analyses and expressed relative to the terrestrial pollen sum. NB. The percentage terrestrial pollen sum is only made up of (all) taxa from the following groups: Trees. Shrubs and herbs, halophytes, grasses, dry ground herbs, cultivated, other herbs and wet ground. Percentage abundances for all other taxa (i.e. swamp and fen, aquatics, Pediastrum and dinoflagellates) are calculated relative to the terrestrial pollen sum.

KRP-P1 (7,600-6,980 BC) – *Corylus* (10-28 %) and *Alnus* (14-25 %) are the most abundant pollen taxa at the bottom of the record, with the latter decreasing throughout and *Corylus* fluctuating. *Pinus* (9-25 %) and *Betula* (13-21 %) also fluctuate and *Tilia* pollen (5-10 %) increases up until ~7,060 BC. *Quercus* and *Ulmus* both independently represent ~5-7 % of the terrestrial pollen sum (TPS) and very small amounts of *Populus* pollen (up to 3 %) are also present. The shrubs, dwarf and terrestrial herbs are represented by small and generally sporadic appearances of *Salix, Hedera helix, Calluna vulgaris* and *Pteridium aquilinum*. A number of species indicative of wet conditions are found in the profile, most notably *Poaceace, Cyperaceae, Dryopteris* and *Sparganium*. *Poaceae* pollen (probably derived from wetland grasses at this time) increases throughout the zone, representing between 4-9% of the TPS. *Pedisastrum* numbers are also initially high, but decline after ~7,470 BC and small amounts of *Ruppia* and *Coleogeton* appear towards the top of the zone (after ~7,330 BC). There is also a *Lingulodinium* spp. (dinoflagellate) peak (~20 %) at ~7,060 BC.

KRP-P2 (6,980-5,850 BC) - *Corylus* is the single most abundant species in zone KRP-P2 increasing substantially at the very start (up to >24%) and then fluctuating throughout. *Alnus* (10-17%) and *Pinus* (9-22%) also fluctuate, though the latter of these begin to decreases after ~6,270 BC. There is generally little change in the pollen frequencies of *Betula, Ulmus* and *Tilia,* whereas *Quercus* remains relatively constant up until ~6,120 BC, after which it begins to increase. For the non arboreal taxa, increased percentages of *Chenopodicieae* (up tp ~3%) and *Pteridium aquilinum* (up to ~2%) pollen are present in this zone and *Calluna vulgaris* is regularly recorded. High numbers of pollen and spores from local wetlands remain present, particularly *Dryopteris* (11-35%), Poaceae (2-7%) and Cyperaceae (1-3%). In fact, Drypoteris spores increase dramatically between ~7,180 (in KRP-P1) - 6,650 BC (~30% increase), before declining again in the upper section of KRP-P2. The aquatics species (*Coleogeton* and *Ruppia*) disappear at the start of this zone and *Pediastrums* never exceeds 1%. There are two distinct dinoflagellate peaks during this zone; *Hemicystodinium* spp. at ~7,035 BC and *Lingulodinium* species at ~6650-6700 BC.

KRP-P3 (5,850-3,900 BC) – *Quercus* (19-20 %) and *Ulmus* (5-9 %) both increase in this zone, the latter of which reaches its highest Holocene frequencies in the period between 5,200-4,500 BC. *Corylus* is initially high (~28 %) but declines after ~5,000 BC, at which *Quercus* takes over as the most abundant pollen type. *Betula* (7-16 %) fluctuates throughout and small numbers of *Fraxinus excelsior* (ash) pollen are also present in the record. There is

very little change in the frequencies of *Pinus* (5-10 %), *Alnus* (12-18 %) and *Tilia* (3-7 %) in this zone. For the non-arboreal taxa, *P. aqulinum* and *Chenopodiciae* are ever present and *C. vulgaris* and *Hedera helix* are present in most samples, but always at very low numbers. *Poaceae, Cyperaceae* and *Dryopteris* percentages all decrease in abundance and *Sphagnum* disappears (by ~6,600 BC) from the pollen record. The aquatics also disappear, and both hystrix and dinoflagellate frequencies increase.

KRP-P4 (**3,900-3,720 BC**) - There is substantial change in the arboreal populations in KRP-P4, with increases in *Quercus, Betula, Corylus* and *Alnus*, synchronous with a very pronounced and rapid *Ulmus* decline (~5 % decrease between ~3,940-3.880 BC. *Pinus* continues its steady decrease and *Tilia* also experiences a small drop in pollen frequencies. The first agricultural signs appear in KRP-P4, as pollen from cultivated cereals such as *Triticum* (wheat) and *Hordeum* (barley) appear in the record for the first time (at ~3,850 BC). Swamp, fen and aquatic representation on the pollen record remains very low and dinoflagellate and hystrix numbers both decrease.

KRP-P5 (3,720-3,510 BC) – There is a large decrease in *Quercus* pollen (~15 % decrease between ~3,740-3,690 BC) at the beginning of this zone, with Betula and to a lesser extent, *Corylus* simultaneously increasing in abundance. This increase is relatively short lived for *Betula*, which declines after ~3,660 BC, and falls by ~13% in the upper section of this zone (3,680-3,550 BC). After its initial drop *Quercus* gradually increases again, along with *Corylus* between ~3,700-3,550 BC. There is little change in the pollen frequencies of the other major tree types (*Alnus* ~18%, *Pinus* ~5%, *Fraxinus* <1%, *Tilia* 2%. and *Ulmus* <1%) or for shrub and dwarf herb species. *Poaceace* (probably now largely from dry grassland species) pollen increases slightly and continuous curves for both *P. lanceolata* and *Rumex* begin in this zone, reflecting the first signs or pastoral farming (Iversen, 1941; Behre, 1981; Gaillard, 2007) in the Djrursland area. Cultivated crops (*Triticum* and *Hordeum*) also remain regularly present in the record. There are no aquatic taxa present in the pollen record and both hystrix and dinoflagellate numbers are very low, despite a slight increase in *Lingulodinium* spp. at the onset of KRP-P5.

KRP-P6a (**3**,**520-2**,**570 BC**) – *Quercus* (13-31 %) and *Corylus* (21-36 %) are the most abundant pollen types and their frequencies fluctuate synchronously in a cyclical fashion (i.e. as one taxa increases, the other simultaneously decreases). *Alnus* percentages remain high

(~17-25 %) and *Tilia, Ulmus* and *Fraxinus excelsior* all increase in abundance, the latter of which reaches its highest percentages for the entire profile in this zone. *Pinus* and *Betula* are relatively stable throughout. *Poaceae* (1-6 %) continues to increase along with *Pteridium aquilinum*, and *P. lanceolata* and *Rumex* remain ever-present. Slight increases of *Chenopodiciae* and *Artemisia* also occur after ~3,400 BC, and a number of terrestrial herbs (and cultivated crops) make sporadic appearances throughout this zone, the most regular of these being *Filipendula*. A short lived, but distinct decrease in arboreal pollen, together with a rise in halophyte (*Artemisia, Chenopodiaceae*), grass and herb pollen (*Poaceae, P. lanceolata*) also occurs between ~3,320-3,230 BC. *Ruppia* reappears in the record in this zone (at ~3,460 BC) and *Pediastrum* is occasionally recorded. Overall, dinoflagellate percentages remain low, with the exception of a peak in *Lingulodinium* spp. between ~3,390-3,250 BC. Increased numbers of hystrix are present between ~3,460-3,120 BC.

KRP-P6b (2,570-2,290 BC) – *Corylus, Quercus* and *Alnus* are again the most abundant pollen types, with *Quercus* and *Alnus* fluctuating throughout and *Corylus* increasing after an initial drop. *Pinus* (~5%) and *Betula* (~8%) remain unchanged from the previous zone and small percentages of Ulmus (1-4%), Tilia (1-5%) and Fraxinus (1-6%) are also present. The amount of open grassland appears to increases as overall tree pollen falls to as low ~82% of the TPS, with both *Poaceae* (4-7%) and *P. lanceolata* (1-2%) pollen frequencies increasing. Additionally, the continued presence of cultivated cereals (*Triticum* and *Hordeum*) indicates the importance of arable farming on the catchment. Towards the top of the profile *Cyperaceae* and *Dryopteris* increase slightly (after ~2,500 BC) and for the aquatics, *Ruppia* continues to rise, reaching its highest level for the whole profile in the uppermost sample (8% at ~2,300 BC). Hystrix numbers experience a slight increase at the very start of the zone, before decreasing again towards the top (after ~2,550 BC) of KRP-P7 and very few dinoflagellates are present.

5.6 Statistical analysis

5.6.1 Trends in species datasets; DCA and PCA analyses

Major trends in the diatom dataset were explored using detrended correspondance analysis (DCA) and the results are summarised in Table 5.4. The variation in the entire diatom dataset captured by the first two DCA axes is relatively even explaining 20.1 % and 15.4 %

respectively. Axis 1 is driven by salinity with the early low-salinity benthic species (e.g. *Epithemia sorex, Staurosira construens* varieties, *Rhoicosphenia curvata, Pseudostaurosira brevistriata*) characteristic of the lowermost (oldest) samples yielding the highest species scores. The lower/negative end of the gradient is broadly characterised by higher salinity species. However, due to the large magnitude salinity increase associated with the marine transgression (compared to more the subtle shifts in salinity later in the record) and high diatom diversity, but often low abundance (i.e. 201 species in dataset, 37 species at >3% abundance) the individual species scores are noisier (with respect to salinity) and more compressed at the low end of the gradient. As a result there is little change in DCA axis 1 gradient sample scores after KIL-D1 (Figure 5.5).

DCA was also performed on a slightly reduced dataset, excluding the lowermost two samples due to their very different nature to the rest of the profile (i.e. very low salinity). In summary, in this reduced dataset, DCA axis 1 sample scores fluctuate up until ~3,400 BC, after which they fall and remain low for the upper section of the profile. Salinity still appears to be important in explaining part of the variation in the diatom dataset (e.g. r=0.52, p<0.05(significant); DCA axis 1 sample scores vs. DI-salinity), with the more brackish taxa present in the uppermost section of the profile (e.g. Psamothidium kuelbsii, Achnanthes submarina, Staurosira elliptica agg. and Fragilaria gedanensis) scoring lowest (most negatively) on DCA axis 1 and the majority of other higher-salinity demanding taxa scoring positively (Table 5.5). This axis, however, cannot be explained by salinity alone as some 'high' salinity demanding species such as *Cymatosira belgica* and *Opehora minuta* agg. also score low on DCA axis 1. Both C. belgica and O. minuta are benthic species, which reach their highest abundances in the upper part of the profile (after ~3,400 BC; see section 5.5.1) when the fjord is beginning to shallow, DI-salinity is decreasing and brackish-marine benthic diatoms generally dominate the profile. As C. belgica and O. minuta agg. diatoms have higher salinity demands, then other processes are also likely to be important on DCA axis 1, most likely sedimentary regime, water depth and indirect depth related processes (e.g. light penetration, sediment accumulation rate, competition for space, sediment disturbance). At the other end of the gradient, highest scoring species (>3 % abundant in one or more slides) are all planktonic (i.e. Chaetoceros cysts, Skeletonema costatum, and Cyclotella striata), rather than being the highest salinity demanding species, further supporting the case that water depth and indirect depth related processes (in addition to salinity) are also important on DCA axis 1.

Dataset	No. of samples	No. of species	Total inertia	Axis	1	2	3	4	Technique
Diatoms	52	242	1.653	Eigenvalues	0.333	0.253	0.12	0.09	DCA
(%)				Length of gradient	1.979	3.042	1.754	1.902	
				% variance explained	20.1	15.4	7.2	5.5	
	50	236	1.104	Eigenvalues	0.167	0.099	0.056	0.032	DCA
				Length of gradient	1.702	1.587	1.124	1.037	
				% variance explained	15.1	9	5	2.9	
Pollen	110	89	1	Eigenvalues	0.381	0.188	0.095	0.050	PCA*
(%)				% variance explained	38.1	18.9	9.5	5	
Molluscs	65	25	1.104	Eigenvalues	0.586	0.178	0.095	0.072	DCA
(conc. per L with SQRT tarnsformation)				Length of gradient	4.426	2.246	1.779	1.684	
)			% variance explained	29.9	9.1	4.9	3.6	
Molluscs	63	24	1.789	Eigenvalues	0.586	1.175	0.088	0.072	DCA
(conc. per L with SQRT tarnsformation)				Length of gradient	4.42	2.042	1.586	1.272	
	1)			% variance explained	32.7	9.8	4.9	4	

Table 5.4 - Summary of DCA/PCA results for the full pollen, mollusc and diatoms datasets.

It is difficult to determine precisely what is driving DCA axis 2, but it appears to be at least partly influenced by specific living conditions and dissolution. For example species scoring highest are *P. kuebsii*, *A. submarina*, and *P. sulcata*, all of which are most abundant when **F** index scores are low (0.2-0.4). Furthermore, both *A. submarina* and *P. kuelbsii* are present in the upper sample of the profile, when the fjord is very shallow and sediment input is high, which combined with lower salinity, likely increased turbidity and increased presence of macrophytes (e.g. *Ruppia* spp.) probably produced quite unique habitats and therefore favoured diatoms well adapted to living in these specific conditions. Absolute lowest sample scores occur in the period when *P. sulcata* is most abundant (~5,730-5,510 BC) and DI-salinity is highest, suggesting maximum marine conditions and probably poorer preservation, increased energy and deeper water. Species scoring highest on DCA axis 2 are *Bacillaria paxillifer*, *Nitzschia frustulum* agg. and *T. fasciculata* which are all most abundant at ~6,680 BC, when conditions within the fjord were still rapidly changing (i.e. increasing water level and salinity, low but increasing nutrient status). The ability of these taxa to survive in such a

wide range of conditions (e.g. widely euryhaline; Hendey, 1964; Snoeijs *et al.*, 1993-1998), probably enabled them to reach their highest percentages at this time. A relatively high score for *S. costatum* (an indicator of good preservation) on DCA axis 2, also further supports a role for dissolution in explaining the variation on this axis.

DCA was also applied to the square-root transformed mollusc dataset (Table 5.4). The variation explained by DCA axis 1 was relatively high (29.9 %) and therefore axis 1 sample scores (Figure 5.6, Figure 5.7) provide good summaries of the major compositional changes in the mollusc dataset. Species scores for almost all molluscan species plotted along DCA axis 1, which appears to be a salinity driven gradient. Species tolerant of high salinities generally scored higher (e.g. *O. edulis, B eulimoides, B. reticulatum*) with lower/negative scores represented by species indicative of more brackish (e.g. *H. ventrosa, L. tenebrosa, C. glaucum*) conditions (Table 5.6). DCA was also performed on a reduced dataset, excluding the lowermost 2 samples. In this case, 32.7 % of the variation was explained by axis 1, but there was very little change in the axis species or sample scores (see Table 5.6) and again a strong salinity gradient was apparent.

PCA axis 1 of the pollen dataset tends to exhibit a strong temporal pattern and is therefore probably defined by the interaction of a number of environmental variables influencing the vegetation (e.g. long-term temperature, edaphic conditions and human impact) rather than representing a single gradient. There appears to be a Mesolithic-Neolithic divide, with species only found in the Mesolithic exhibiting the highest positive PCA axis 1 species scores and vice versa (i.e. negative for species only present in the Neolithic). Of the major tree species, Pinus, Tilia, Ulmus and Poplulus (more abundant pre ~3,900 BC) exhibit positive axis 1 species scores, whereas Fraxinus increases after ~3,500 BC and therefore scores negatively on axis 1. Alnus (positive), Corylus and Quercus (negative) have scores close to zero being similarly abundant both pre- and post- 3,900 BC, whereas Betula remains positive despite its sharp increase following the elm decline due to its particularly high abundance in the earliest part of the record. Poaceae also scores weakly negative, being abundant between ~7,600-6,100 and again after ~3,500 BC (in the Neolithic). It is likely that before 6,100 BC, that the high numbers of Poaceae pollen was probably derived from wetland grasses, being washed in following catchment erosion, whereas the second increase is most likely from dryland grasses associated with agricultural land-use changes. Other plants commonly associated with

agricultural activities (e.g. *Rumex*-type, *P. lanceolata, Ononis*-type, *Ranunculus acris, Triticum* type and *Hordeum* type) also exhibit negative species scores.

Таха	Axis 1	Таха	Axis 2
Cyclotella striata	5.0554	Bacillaria paxillifer	4.5473
Skeletonema costatum	2.6895	Nitzschia frustulum	4.1442
Chaetoceros cysts agg.	2.5807	Tabularia fasciculata agg.	2.5029
Grammatophora oceanica	2.5413	Skeletonema costatum	2.418
Rhopalodia acuminata	2.4459	Rhopalodia acuminate	2.4033
Plagiogramma staurophorum	2.0523	Cyclotella choctawatcheeana	2.1537
Dimmeregrammer minor	1.9333	Cocconeis scutellem [parva + scutellum]	1.9935
Paralia sulcata	1.5128	Grammatophora oceanica	1.9022
Bacillaria paxillifer	1.4949	Chaetoceros cysts agg.	1.768
Fallacia fenestrella agg.	1.3374	Opephora mutabilis	1.6531
Cyclotella choctawatcheeana	1.0012	Plagiogramma staurophorum	1.5343
Staurosirella pinnata	0.9024	Fragilaria gedanensis	1.4882
Planothidium delicatulum agg.	0.8008	Amphora helenensis	1.3704
Cocconeis scutellem [parva + scutellum]	0.5668	Dimmeregrammer minor	1.3598
Nitzschia frustulum agg.	0.5445	Opephora burchardtiae/horstiana agg.	1.3254
Tabularia fasciculata agg.	0.493	Opephora minuta agg.	1.292
Opephora burchardtiae/horstiana agg.	0.4251	Planothidium delicatulum agg	1.1064
Fallacia sp. undiff.	0.3862	Cyclotella striata	1.0311
Amphora helenensis	0.3824	Cymatosira belgica	0.8415
Opephora mutabilis	0.3337	Staurosira elliptica agg.	0.7296
Fallacia cryptolyra	0.251	Fallacia cryptolyra	0.3516
Opephora minuta agg.	-0.0218	Staurosirella pinnata	0.1054
Fragilaria gedanensis	-0.6719	Fallacia fenestrella agg.	-0.1046
Staurosira elliptica agg	-0.8279	Paralia sulcata	-0.8848
Cymatosira belgica	-1.3495	Fallacia sp. undiff.	-0.9515
Psammothidium kuelbsii	-3.0609	Psammothidium kuelbsii	-1.8202
Achnanthes submarina	-3.0638	Achnanthes submarina	-1.8525

Table 5.5- Summary of DCA results for the reduced diatom dataset (50 samples)

A further 18.9 % of the variation in the pollen dataset is explained by PCA axis 2 (Table 5.4). This axis is likely to be driven, initially, by inwash and catchment erosion as suggested by exaggerated percentages of Poaceae, Cyperaceae and *Pinus* pollen probably from local wetlands (see section 5.5.3), but later by forest density as catchment erosion is reduced and shade trees such as *Quercus*, *Corylus* and *Ulmus* become more abundant in the catchment. High PCA axis 2 sample scores at the bottom of the profile (~7,600-6,100 BC; Figure 5.8) are caused by the high, and exaggerated percentages of Cyperaceae and Poaceae pollen (both of which score positively on PCA axis 2; Table 5.7). These taxa give the impression that the Korup Sø catchment was relatively open, but these wetland sedges and grasses probably mainly grew in low-lying areas in the immediate vicinity of the fjord together with some

Betula, Pinus and *Alnus*. The rest of the catchment was dominated by a dense forest consisting predominately of *Tilia, Corylus* and *Quercus*. The lowest PCA axis 2 scores occur between ~6,000-3,500 BC, when a dense closed-canopy forest dominated by *Tilia, Corylus* and *Quercus* (up until ~3,900 BC) covered the Korup Sø catchment. The lowest (most negatively) scoring species (abundant at >1 % in one or more slides) on axis 2 are *Quercus* and *Ulmus*, the latter of which reaches its highest percentages between ~5,560-3,900 BC. *Corylus* is relatively abundant throughout the whole record and therefore scores close to zero.

Таха	DCA axis 1	DCA axis 1
	(all samples)	(63 samples)
Macoma baltica	6.87	6.86
Abra alba	6.51	6.51
Brachystomia eulimoides	4.96	4.95
Bittium retulatum	4.40	4.39
Ostrea edulis	3.94	3.93
Tapes descussatus	3.82	3.81
Rissoa albella	3.62	3.61
Littorina littorea	3.56	3.55
Parvicardium exiguum	3.50	3.50
Mysella bidentata	3.50	3.49
Corbula gibba	3.24	3.24
Srobicularia plana	2.99	2.98
Hinia reticulata	2.87	2.86
Retusa truncata	2.69	2.68
Omalogyra atomus	2.12	Absent
Hydrobia ulvae	2.06	2.05
Mytilus edulis	1.93	1.92
Cerastoderma edule	1.78	1.78
Onobo vitrea	1.22	1.23
Skeneopsis planorbis	0.86	0.87
Onobo semicostata	0.61	0.61
Rissoa membranacea	0.44	0.44
Cerastoderma glaucum	0.28	0.28
Hydrobia ventrosa	-0.02	-0.02
Littorina tenebrosa	-0.77	-0.77

Table 5.6 - DCA species scores for the molluscan dataset

PCA axis 2 sample scores rise again in the Neolithic, as overall arboreal pollen decreases and elevated percentages of Poaceae (derived from dryland grasses) and other open ground taxa (most notably *P. lanceolata* and *Rumex* type) become present in the record. These taxa all score positively on PCA axis 2, with Poaceae exhibiting the highest score (Table 5.7). Axis 2

sample scores generally increase throughout the Neolithic period, primarily driven by the continually increasing percentages of Poaceae pollen and the increase in wet ground taxa (e.g. Cyperaceae) after ~2,800 BC, the latter probably caused by the expansion of wetlands in the Korup Sø vicinity as the fjord begins to dry up. Superimposed upon this general trend are occasional high scores (e.g. ~3,250 BC) which might represent short-term episodes of forest clearances for agricultural purposes. Fluctuations in arboreal pollen also result (e.g. increase in *Quercus* pollen at ~3,160 BC; ~31 %) in occasional lower sample scores in the post-3,900 BC period.

Taxon	Axis 1	Taxon	Axis 2
Pinus	0.84	Poaceae	0.90
Tilia	0.81	Cyperaceae	0.73
Ulmus	0.76	Plantago lanceolata	0.46
Betula	0.55	Rumex type	0.39
Populus	0.39	Pinus	0.39
Cyperaceae	0.34	Alnus	0.26
Salix	0.02	Artemisia	0.23
Poaceae	-0.04	Chenopodiaceae	0.13
Pteridium aquilinum	-0.28	Corylus avellana	-0.02
Chenopodiaceae	-0.31	Tilia	-0.13
Alnus	-0.43	Salix	-0.13
Quercus	-0.50	Fraxinus excelsior	-0.14
Corylus avellana	-0.53	Betula	-0.16
Rumex type	-0.56	Pteridium aquilinum	-0.19
Artemisia	-0.69	Ulmus	-0.24
Plantago lanceolata	-0.73	Populus	-0.40
Fraxinus excelsior	-0.82	Quercus	-0.78

Table 5.7 - Summary of PCA axis 1 and 2 species scores for the Korup So pollen record (taxa > 1 % abundance in one or more slides)

5.6.2 Environmental-Species relationships

Environmental-species response relationships were examined using a top-down approach with diatoms (the lowest trophic level) used as the response dataset. Palynological analyses were used as predictors for vegetation, catchment and land-use changes, and molluscs (and dinoflagellates) for changes in the marine environment and grazing pressure. Independent sea level data (from Blekinge, Sweden; Berglund *et al.*, 2005) and temperature data (pollenbased quantitative temperature reconstruction from Lake Trehörningen, Bohuslän, southwest Sweden; Antonsson and Seppä, 2007) were also included in the environmental dataset. Both

of these reconstructions were considered to be good regional estimates for these parameters and applicable to the Danish area (P. Rasmussen, pers, comm.; Brown *et al.*, submitted). Lastly, sedimentation rate data (from Well No. 1, see section 5.3.) was included as another predictor of biotic change in the marine environment. For the predictor dataset, the accumulation rate data was expressed as cm yr⁻¹ and sea level as metres above present sea level (as in Berglund *et al.*, 2005). The pollen and mollusc data were expressed as percentages, but rare or sporadically occurring species (i.e. less than four occurrences and/or only sporadically present through the profile) were removed prior to analysis. In addition, temperature was log (base 10) transformed to reduce the relatively long gradient length exhibited by this variable.

Differences in the sampling strategy between proxies meant that a reduced subset of 46 samples was used for predictor-response analyses. Each sample contained pollen, mollusc and diatom data either from the same stratigraphic level or within a maximum of ~25 years of one another (as determined by the age model). Coarser sampling resolution for molluscs (analysed for the entire profile) meant that several pollen and diatom samples often corresponded to one mollusc sample. In these circumstances the time-averaged mollusc samples were duplicated. Rare pollen and mollusc species were eliminated if they occurred less than 4 times in the 46-sample environmental dataset and only at sporadic intervals and low abundances. Sediment age was included as a co-variable to partial out variance explainable by time-related processes (e.g. ontogeny, succession). Moving window analyses were again performed to examine the possible influence of different environmental parameters at different times throughout the profile (after Bradshaw et al., 2005a and see Chapter 4). Subsets of 20 samples were analysed starting at the top of the record and working backwards, eliminating the upper sample and adding a lower sample each successive run (e.g. 1-20, 2-21, 3-22...etc). Each time the datasets were reduced, DCA was re-performed on the species subset for selection of the appropriate ordination technique. Axis 1 gradient lengths ranged between 1.07-2.63 s.d. units and therefore both canonical correspondence analysis (CCA) and redundancy analysis (RDA) techniques were employed (Figure 5.9). For each CCA/RDA analysis, significant variables (p < 0.05; with Bonferroni corrections applied for each subsequent selection) were chosen using a forward selection process (999 unrestricted permutations). After this process, CCA/RDA analysis was performed using just the selected variables to test the significance (p < 0.05) of the reduced model and time factored out as a covariable.

CCA and RDA results are presented in Figure 5.9. For each subset only significant variables are included, together with the amount of variance explained. The length of the DCA axis 1 gradient together with the type of analysis performed is also included. CCA on the full response-predictor dataset produced a statistically significant model (p <0.01) with five independent predictor variables (Pediastrum, Dryopteris, Pinus, sea level and temperature; discussed in detail below) explaining 36.7 % of the variation in the diatom dataset. In the moving window analyses ~7.8-49 % of variation is explained by the selected variables with all models proving statistically significant (p < 0.05). The greatest variation is explained in the lower subsets (subsets 27-35; Figure 5.9), where the system is at it most dynamic and the diatom dataset appears to be responding strongly to changes in the marine environment, particularly salinity, water depth and nutrients, all probably related to rising sea level (discussed in more detail below). Above this (i.e. subsets 5-26; ~5,730-2,380 BC), individually selected environmental predictors generally only explain relatively small amounts (<20%) of the variation with the diatom dataset. In subsets 21-35 (~7,330-3,400 BC) predictors for marine environmental change (e.g. sea level, sedimentation rate, changing mollusc faunas) are generally more common, whereas catchment processes/vegetation changes (probably heavily driven by human disturbance after ~3,900 BC) are more important in the upper section of the record (subsets 5-20; 4,870-2,380 BC).

Sea level is important in explaining some of the variation in the diatom dataset at two separate times throughout the profile (i.e. subsets 28-32; ~6,680-3,840 BC, subsets 12-14; ~3,930-2,980 BC) as well as being a significant proxy for the entire dataset (subsets 1 and 2). The lowermost block of subsets that sea level is important in (i.e. subsets 28-32) covers the most dynamic phase of the *Littorina* Sea (i.e. L1-L3 transgressions; Berglund *et al.*, 2005). Sea level is likely to influence salinity, marine exposure (i.e. currents and waves) catchment erosion and inwash (see below), water depth (and light penetration) and macrophytic vegetation. For example, aquatic taxa (*Ruppia* spp. and *Coleogeton*) practically disappear from the pollen record after ~7,000 BC as sea level (water depth) and salinity increase, altering the benthic habitat and enabling planktonic diatom taxa to increase in abundance (Figure 5.5). Furthermore, the DI-salinity reconstruction appears to be closely connected with sea level rise, in the lowermost section of the sequence (~7,300-5,500 BC). In this period, several floral changes are identifiable in the diatom record, as salinity rises and marine conditions intensify. This also includes a peak of planktonic diatoms at ~6,270 BC possibly related to increasing water depth and maybe higher nutrient status, perhaps being washed in

from the catchment as low lying lands were sub-merged. In the uppermost block of subsets (~3,930-2,980 BC), declining sea levels (together with high sedimentation rates) causes the basin to get shallower and salinity to decline (after ~3,400 BC, according to the DI-salinity record). This directly affects the diatom community as benthic diatom taxa increase in abundance and macrophyte vegetation (e.g. *Ruppia* spp.) appear again in the pollen record.

1 1-46 2379 7332 1.99 CCA 36.7 <td< th=""><th>Scrobicularia plana</th></td<>	Scrobicularia plana
2 1-44 2379 6680 1.07 CCA 33.1 3 1-27 2379 3884 1.61 CCA 19.8 4 28-46 3913 7332 2.51 CCA 61.3 5 1-20 2379 3769 1.59 CCA 18.9 6 2-21 2480 3782 1.48 RDA 161	Ť
3 1-27 2379 3884 1.61 CCA 19.8 4 28-46 3913 7332 2.51 CCA 61.3 5 1-20 2379 3769 1.59 CCA 18.9 6 2-21 2480 3782 1.48 RDA 161	
4 28-46 3913 7332 2.51 CCA 61.3 5 1-20 2379 3769 1.59 CCA 18.9 6 2-21 2480 3782 1.48 RDA 16.1	\Box
5 1-20 2379 3769 1.59 CCA 18.9 6 2-21 2480 3782 1.48 RDA 16.1	
6 2-21 2480 3782 1.48 RDA 16.1	
7 3-22 2530 3801 1.47 RDA 8.0	
8 4-23 2588 3808 1.34 RDA 16.5	
9 5-24 2782 3841 1.46 RDA 16.2	
10 6-25 2842 3861 1.50 RDA 7.8	
11 7-26 2937 3874 1.52 CCA 16.1	
12 8-27 2978 3884 1.57 CCA 16.9	\bot
13 9-28 3028 3913 1.57 CCA 17.5	\bot
14 10-29 3084 3939 1.59 CCA 16.4	\perp
15 11-30 3117 4063 1.62 CCA 17.4	\perp
16 12-31 3162 4125 1.58 CCA 18.8	\perp
17 13-32 3201 4323 1.55 CCA 20.2	\perp
18 14-33 3228 4460 1.55 CCA 19.8	\vdash
19 15-34 3284 4734 1.37 RDA 19.5	\vdash
20 16-35 3387 4868 1.41 RDA 18.8	⊢
21 17-36 3425 5207 1.40 RDA 10.1	⊢
22 18-37 3538 5382 1.39 RDA 19.1	⊢
23 19-38 3703 5470 1.43 RDA 19.0	+
24 20-39 3769 5558 1.37 RDA 12.1	⊢
25 21-40 3782 5645 1.52 CCA 10.0	⊢
26 22-41 3801 5727 1.26 RDA 18.4	⊢
27 23-42 3808 5966 1.47 RDA 28.5	⊢
28 24-43 3841 6270 1.77 CCA 47.6	⊢
29 25-44 5861 6680 1.68 CCA 44.2	╋
30 20-44 3874 6680 1.67 CCA 45.1	╋
31 27-44 3884 0080 1.09 CCA 43.9	+
52 28-44 5915 0680 1.72 CCA 47.2	⊢
33 29-44 3939 0080 1.73 CCA 40.9	╋
34 20-43 3074 7184 2.03 CCA 39.9	╇

Figure 5.9 - Summary of 'moving window' results from CCA/RDA analyses assessing predictor-response (diatoms) relationships through time. Length of axis 1 gradient refers to DCA analysis on reduced diatom dataset (if DCA axis 1 gradient length > 1.5 s.d. then unimodal CCA employed to explore trends. If DCA axis 1 <1.5. s.d. then linear RDA technique used). Subset 1 = entire dataset (46 samples). Subsets 2-35 = moving window analyses. Shadings: Red = Anthropogenic indicator, Black = sedimentary regime/catchment erosion, Green: vegetation change predictors, Grey = climatic and Blue = fjord depth and level of connection with Kattegat, Brown = molluscs (grazers).

Average temperature has changed by $\sim 3^{\circ}$ C (Figure 5.10) over the course of the Korup Sø record, rising in the early Holocene and reaching a 'thermal maximum' between $\sim 5,550$ -3,550 BC, but beginning to decline again after $\sim 4,000$ BC. Changes in temperature can both directly and indirectly affect the diatom community (see section 4.10.2). For example, cooler temperatures (particularly before and/or after the thermal maximum) might have affected the severity of winters and the volume of snow and ice on the catchment. This might have induced large thawing events inputting significant volumes of cold-fresh water into the fjord and subsequently causing short-term fluctuations in both temperature and salinity (e.g. numerous salinity fluctuations are evident in the DI-salinity reconstruction; Figure 5.4). Sea ice might also have developed in extremely cold winters, though there is no evidence for any major sea ice event in the diatom record (e.g. complete absence of sea ice diatoms).

Interestingly, temperature is absent from the reduced 'moving window' subsets at Korup Sø but appears to explain some of the variation when all samples are included (i.e. present in subsets 1 and 2; 2,380-7,330 BC). This is likely to result from temperature more strongly exerting an indirect influence upon the diatom community, by driving changes in other important variables (and at different times) such as hydrology, vegetation structure, edaphic processes and grazing communities (see also Chapter 4; Norsminde). For example, reproduction in grazing organisms such as molluscs and foraminifera is often temperaturedependent (e.g. oysters require 15° C temperatures for spawning, whilst cockles (~9° C) and mussels (9-12° C) can tolerate slightly lower temperatures; Yonge, 1960; Petersen, 2004) and if temperatures fall below the minimum required, the length of the spawning season can be reduced or even prevented entirely (Yonge, 1960; Johannesson et al., 1989). In prolonged cold periods, molluscan and foraminifera populations might become severely reduced, thereby potentially reducing grazing pressure upon diatoms. Alternatively, a change in species composition might occur (i.e. colonisation by a colder water mollusc or foraminiferal fauna) which might alter the habitat for benthic diatoms and/or alter ecosystem structure (e.g. feeding patterns, predators).

It is also possible that the influence of temperature on the diatom communities is purely coincidental, as for example maximum Holocene temperatures (i.e. during the Holocene climatic optimum, ~5,550-3,550 BC; Snowball *et al.*, 2004; Antonsson and Seppä, 2007) broadly coincide with the highest abundances of dense closed canopy tree taxa (e.g. *Quercus*, *Ulmus*) in the pollen record (~6,000-3,900 BC; Figure 5.7). Denser woodland is likely to

influence catchment run off rates and inputs of sediment (e.g. lowest accumulation rates broadly coincide with highest temperatures), nutrients and minerals. This is particularly identifiable later in the profile when human impact upon the catchment (i.e. forestry clearances, crop-growing and expansion of agricultural grasslands) becomes more important in driving changes in the diatom record (see below). Summer thermal stratification might also have occurred under higher temperatures (e.g. during the thermal optimum), though this is probably unlikely due to the depth of the fjord (probably relatively shallow even in the late Mesolithic when sea level was higher). The lack of evidence for a halocline (e.g. no freshwater planktonic diatoms present in the profile) suggests that the fjord was probably well mixed at all/most times, though short-term stratification events might have occurred in particularly warm years. Such events might offer an alternative explanation (to water depth or productivity) for the high abundances of *Chaetoceros* cysts between ~5,380-3,540 BC, with cyst production events following spring/summer planktonic blooms when nutrients in the surface layers became depleted.

Pediastrum (present in the lowermost subset and significant for the entire dataset; i.e. subset 1, 46 samples) is relatively abundant (>3 %) in the record between ~7,600-7,360 BC and is a likely predictor for fresh to weakly brackish conditions, prior to the marine transgression (after ~7,400 BC). At this time the diatom record is also dominated by fresh/brackish water diatoms (e.g. Pseudostaurosira brevistriata, Staurosira construens and some freshwater Cyclotella species). Pediastrum percentages drop as salinity in the fjord rises, and higher salinity diatoms appear in the record (i.e. <1 % after ~7,000 BC). It reappears again (sporadically and at low abundances) towards the top of the record (after 3,100 BC) as the salinity in the fjord decreases. Cerastoderma edule is also important in the lowermost subsets (and in subsets 29-33), probably acting as another predictor for increasing salinity at the very bottom of the profile. Ostrea edulis is important in subsets 29-33 (~6,680-3,860 BC), covering the time period in which this taxon reaches its highest abundances (~6,000-5,570 BC). Maximum O. edulis abundances coincide with maximum DI-salinities (~24-27 ‰, between ~5,730-5,510 BC) and high MI-salinities (19-23 ‰ between ~5,570-5,510 BC) suggesting that O. edulis is a predictor for high salinity conditions (and greater marine influence on the fjord). This is probably also true of *Brachystomia eulimoides* (high-salinity demanding; Sorgenfrei, 1958; Petersen, 2004), which also achieves its highest abundance in this period (~5,870-5,440 BC) and is important in subset 26-28; (~6,270-3,800 BC). Varying molluscan abundances might also directly exert grazing pressures upon the diatom

communities, though this is difficult to isolate in a system subject to such rapidly changing environmental conditions.

The vegetation predictors in the lower subsets (i.e. Pinus, subset 35, Poaceae, subset 34, Dryopteris, subsets 27-34) probably represent catchment erosion and inwash at this time, associated with rising sea levels and submergence of low lying lands. The relatively high abundances of Poaceae pollen (before ~6,000 BC) is likely to be predominately derived from wetland grass species, whilst extremely high percentages of *Dryopteris* between ~7,060-5,890 BC suggest that the Korup Sø area either was surrounded by, or situated in close proximity to, extensive swamp and fenlands (also Sparganium present between ~7,470-6,600 and Sphagnum between ~6,700-5,650 BC). Along with inputs of plant material, catchment erosion and submergence of terrestrial swamps and wetlands is likely to have resulted in greater inputs of nutrient and minerals into the fjord. Several shifts in the diatom flora are identifiable between ~7,330-5,930 BC (Figure 5.4) and, whilst the diatoms appear to be primarily responding to sea level and salinity change at this time, it is possible that additional factors might also be important. For example, C. choctawhatcheeana (high-nutrient favouring diatom) appears in the record between ~5,970-5,930 BC, Cyclotella striata and Skeletonema costatum exhibit a peak at 6,270 BC and Chaetoceros cysts fluctuate throughout, which might suggest that nutrient inputs either from the catchment following changes in vegetation structure or from inputs of nutrient rich water from the North Sea (through the Kattegat) are also important in driving changes in the diatom dataset.

It is important to stress that the pollen catchment area is likely to be far greater than the hydrological catchment area of the fjord, and therefore care must be taken with the interpretation of predictors of vegetation change. Nevertheless, it remains likely that local vegetation changes within the catchment will exert pressure on fjord diatom communities. Despite changes in the marine environment dominating in the lower section of the record, some proxies for vegetation change are also significant in the Mesolithic (e.g. *Tilia* is important in subset 34; 7,180-3,870 BC). The decrease in *Tilia* and *Pinus* pollen at ~7,060 BC (Figure 5.7) might reflect a change in the size of the pollen catchment due to increasing sea level around this time (i.e. a larger pollen catchment meaning a change from a more local to regional signal). Whilst a change in the size of the pollen catchment is unlikely to directly affect diatom communities, in the moving window analyses, *Tilia* (in subset 34) and *Pinus* (in subset 35) might be acting as predictors for a change in the size (and/or configuration) of the

hydrological catchment area as this probably also increased under rising sea levels. A larger hydrological catchment might have affected factors such as surface run-off, sediment and nutrient supply to the fjord and perhaps increased water turbidity. Later, the establishment of a closed canopy forest is likely to have reduced inputs of material (including nutrients) from the catchment, perhaps partly explaining the low sediment accumulation rates in the Mesolithic section of the record (i.e. pre 3,900 BC).

Ulmus is important in subset 28 (~6,270-3,840 BC), covering the period of maximum elm abundance in the catchment (according to the Korup Sø pollen record) and its rapid decline at ~3,900 BC. The elm decline triggered a large restructuring of the woodlands in the early Neolithic (~3,900-3,550 BC), which in turn is likely to have contributed to the large increase in the sedimentary accumulation rate (important in subsets 23-27: 5,970-3,700 BC) after ~3,900 BC (Figure 5.3). During this woodland transition, Betula (important in subset 19, 4,730-3,280 BC), Corylus and Quercus (important in subset 14; 3,900-3,080 BC) increase rapidly (Figure 5.7). The reduction of elm trees and shift in forest structure might also have released nutrients into the fjord (e.g. Bormann et al., 1974; Bayley et al., 1992), possibly explaining the peak in diatom concentration around ~3,800 BC and increased abundances of Cyclotella choctawhatecheeana at ~3,780 BC. There is also a general increase in Fragilaria s.l. around this time which might further reflect changing (or variable) nutrient status and/or instability (perhaps associated with increased sediments loads) within the fjord. Lower sea levels and human impact (e.g. cereal pollen first appears at ~3,850 BC; Figure 5.7) might also have influenced the Korup Sø sedimentary regime at this time, whilst climate change and edaphic factors might have exerted some pressure upon the vegetation.

Mytilus edulis (subsets 20, 23-23 and 28; 6,270-3,380 BC) and *Hydrobia ulvae* (subsets 21-21; 5,380-3,430 BC) appear as significant predictors for environmental change in subsets covering the late Mesolithic/early Neolithic period. *Hydrobia ulvae* is abundant for much of the profile, but exhibits a brief drop in the late Mesolithic. After a substantial increase in the early Neolithic, it becomes the dominant species as overall mollusc concentrations drop. Whilst the decrease in *Bittium reticulatum* and general decline in high-salinity demanding taxa might be interpreted as a salinity decline, the DI-reconstruction and the continued presence of *Onoba vitrea* and *Tapes decussates* (albeit at low concentrations) suggest that salinity does not decline at this point. Petersen, (1993) suggested that the decline in molluscan concentration is caused by an increase in the sedimentary accumulation rate (i.e. at

~3,900 BC; Figure 5.3). Therefore, it is likely that these two taxa (i.e. *M. edulis* and *H. ulvae*) are proxies for a change in the sedimentary regime and substrate (likely driven by forest restructuring and later human impact), a decrease in mollusc concentration and a shift in grazing pressure. These taxa have the ability to survive living in muddy conditions, with higher sediment loads and finer substrates, which other molluscs might be unable to survive (e.g. oysters; Rasmussen, 1958; Yonge, 1960).

Plantago lanceolata (important in subsets 12-13, ~3,900-2,980 BC and 15-19, ~4,730-3,100 BC) is a likely predictor for expanding grasslands (Iversen, 1941; Behre, 1981; Gaillard, 2007) in the Korup Sø catchment area. This taxon is consistently present in the pollen record (Figure 5.7) after ~3,730 BC, marking the replacement of areas of woodland with agricultural land for both pastoral and arable farming (Iversen, 1941; Troels-Smith, 1960). Again, this is likely to have affected rates of surface run off and inputs of sediment and nutrient loads (e.g. animal waste and other natural fertilisers) from the catchment. Variable sediment and nutrient loads probably explain some of the fluctuations in Fragilaria s.l. and Chaetoceros cyst abundances after ~3,700 BC. In the uppermost section of the profile (i.e. subsets 5-11; 2,380-3,870 BC), changes in the major tree taxa (i.e. Pinus, Betula, Corylus and Alnus) significantly explain some of the variation (8-19%) in the diatom record. Whilst climatic (e.g. decreasing temperatures; Antonsson and Seppä, 2007) and edaphic factors might also be important, fluctuations in these tree taxa are likely to be heavily associated with anthropogenic disturbance, particularly rapid forest clearance events (e.g. decrease in arboreal pollen at ~3,250 BC). Features such as short term increases in *Betula* followed by expansions of Corylus are described in Iversen's landnam model (Iversen, 1941; Iversen, 1973; Edwards, 1993) further supporting the case here, that fluctuations in the major tree taxa are likely predictors of anthropogenic disturbance. Corylus nuts might also have been an important resource, and therefore this tree might have been directly coppiced by Neolithic (and late Mesolithic) cultures.

5.7 Synthesis of results and discussion of site

This section attempts to tie together the results from the independent proxy records to build up a more complete picture of environmental change in and around the Korup Sø area during the mid-Holocene. Figure 5.10 provides a summary of the major environmental changes described in the above sections, together with sea level and temperature curves, included to examine changes in a more regional context. For ease of interpretation, five environmental zones (labelled KRP-E1 to E5) have been identified, based on relatively good agreement between the independently assigned (optimal split) proxy zones. An environmental zone has been assigned at points where independent zone splits for all 3 proxies fall in close temporal proximity to one another. There is one exception to this rule; the split between KRP-E3 and KRP-E4 is based on agreement of only 2 proxies (molluscs and pollen). This has been added due to the almost identical timing of the split in both proxies and the substantial changes exhibited between zones. The multiproxy changes are discussed in relation to these environmental zones.

KRP-E1 – Early brackish phase (~7,600-6,900 BC) – The high abundance of *Pediastrum* (freshwater algae recorded in the pollen samples) at the bottom of the profile (i.e. ~7,600 BC), together with a variety of freshwater diatoms (preliminary assessment only) suggests that Korup Sø was probably limnic at this time. By the start of the diatom (~7,330 BC) and molluscan (*C. edule* recorded at ~7,490 BC) records presented here, there was certainly a marine influence. Both DI- and MI-inferred salinities (~4-6 ‰ between ~7,500-7,300 BC) suggest brackish conditions, but at this point it is difficult to determine the degree of connection with the sea. Korup Sø may have been isolated from the sea, but breached by intermittent pulses of saline water, perhaps during storm conditions. Alternatively, it may have been only partially connected at this time (i.e. ~7,500-7,300 BC). The molluscan fauna at this point is very poor, consisting of just two species (*O. atomus* and *C. edule*) which could potentially live under both scenarios. If completely cut off, regular inputs of saline water would be required to support these species (and maintain this salinity) and therefore at the very least, we must assume that the Korup Sø was in very close proximity to the Randers/Kolinsund Fjord systems at this time.



Figure 5.10 - Summary diagram for all proxies analysed at Korup Sø, together with regional temperature (Antonsson and Seppä, 2007) and sea level (Berglund *et al.*, 2005) reconstructions. Well no. 1 sedimentation rate is based on latest age model calculations (J. Olsen, P. Rasmussen, unpublished data). *m a/b psl = metres above/below present sea level.

In the succeeding phase, salinity increases quite rapidly in response to global sea level rise (e.g. Mörner, 1976; Berglund *et al.*, 2005; Yu *et al.*, 2007), and by ~7,200 BC a permanent marine connection is definitely established. The molluscan fauna increases both in diversity and concentration, characterised by high numbers of the brackish water gastropods *Hydrobia ventrosa* and *Hydrobia ulvae* but supplemented by several other brackish/marine species. The diatom flora at this point is still dominated by opportunistic *Fragilaria* s.l. that are highly competitive in unstable environments such as transitional zones (Eronen, 1974; Stabell, 1985; Seppä *et al.*, 2000; Miettinen, 2002), but the overall shift in the flora indicates strengthening of brackish conditions and the DI-salinity of ~9 ‰ (at ~7,200 BC) would probably be unachievable by intermittent intrusions alone. Dinoflagellates also appear in the record around this time and *Pediastrum* numbers fall (after ~7,370 BC). Small amounts of *Ruppia* and *Coleogeton* (both living in weakly brackish water systems) are also present probably occupying the more littoral areas, but both disappear around ~7,000 BC, likely due to rising salinity and an increase in water depth.

The pollen record suggests that the Korup Sø basin (and probably large areas of low-lying Kolindsund coastal land) were surrounded by extensive wetlands (including swamp and fenlands) at this time. The high and exaggerated percentages of Poaceae (probably derived from wetland grass species at this time), Cyperaceae pollen and *Dryopteris* spores in the lower part of the pollen record are most likely to be the result of catchment erosion and inwash, associated with rising sea levels and gradual submergence of the lower lying wetlands. At this time the catchment was covered by a mixed deciduous forest, with *Tilia*, *Corylus*, *Ulmus* and *Quercus* as the main taxa on the high ground and *Betula*, *Pinus* and *Alnus* likely to be prevailing on low-lying moist ground (Iversen, 1969; Andersen, 1978; Aaby, 1983). The relatively high abundances of *Pinus*, *Betula* and *Alnus* at this time might suggest that their representations in the pollen record are also affected by catchment erosion and inwash under rising sea levels.

KRP-E2 - The early Korup Sø marine phase (~**6**,**900-5**,**900 BC**) – The diatom sampling resolution is low in this phase (4 samples), but the diatom flora changes several times in this period due to rapidly changing conditions. The lowermost sample is completely dominated by benthic species (particularly *Bacillaria paxillifer* and *Fragilaria* s.l.), but by ~6,270 BC, higher-salinity planktonic/tychoplanktonic diatoms (e.g. *Paralia sulcata, Chaetoceros* cysts, *Cyclotella striata*) become more abundant in the profile due to increasing salinity, water depth and nutrient status, driven by rising sea levels. A drop in salinity and a shift back to a

Fragilaria s.l. dominated system occurs at ~5,970 BC, possibly driven by a sea level regression phase (identifiable in the Blekinge stratigraphy between 6,250-6,050 BC; Berglund *et al.*, 2005), which has been tentatively linked to the 8.2 k cooling event (Alley *et al.*, 1997; Rohling and Pälike, 2005).

In contrast to the diatoms, there is little change in the molluscan faunal composition in this phase, which remains dominated by *Hydrobia ulvae* throughout, with a minimum MI-salinity of ~13-15 ‰. However, despite the huge number of *H. ventrosa* snails in the upper section of KRP-E2 (after ~6150 BC), salinity appears to be increasing as a number of higher-salinity demanding species (*Corbula gibba, Retusa truncatula, Ostrea edulis, Onoba vitrea*), appear in the record. There is no evidence for decreased salinity associated with a marine regression (6,250-6,050 BC) in the molluscan record, though this is more likely to go undetected due to the coarser sampling resolution (5-10 cm thick slices in KRP-M2; 50-120 year resolution), absolute dominance of *H. ulvae* and generally lower sensitivity of molluscs (when compared to diatoms) to more minor salinity shifts.

Pinus are still frequent in the pollen record, perhaps related to catchment erosion and inwash. *Tilia, Ulmus, Corylus, Quercus, Betula* and *Alnus* all remain relatively abundant in the catchment, with *Quercus* increasing slightly after ~6,100 BC. High percentages of wet ground, swamp and fen taxa (Figure 5.8) are present in the pollen record, most notably *Dryopteris* spores (14-35 %), but also Cyperaceae, *Sphagnum* and Poaceae (from wetland grass species), again indicating large scale catchment erosion and drowning of extensive low lying swamp and marsh areas. The aquatic taxa disappear from the record after ~6,700 BC as the salinity rises (and water depth increases) and only small percentages of *Pediastrum* remain present in this phase.

KRP-E3 – **The Korup Sø marine maximum (5,900-3,900 BC)** - The beginning of KRP-E3 broadly coincides with the second phase of the Littorina marine transgression (L2; Berglund *et al.*, 2005) and marks the beginning of the maximum marine phase at Korup Sø. Rapid sea level rise appears to seriously disrupt the early Holocene system at Korup Sø, causing DI-salinity to rise sharply, reaching its highest inferred levels in the whole record between ~5,700 -5,200 BC. This salinity rise causes a substantial change in the mollusc fauna with the appearance of a number of high salinity tolerant species and a reduction in *H. ulvae* numbers (though it remains the dominant species). Several of the high-salinity demanding molluscs practically disappear from the record again shortly after ~5,200 BC, but the continued

presence of *B. reticulatum* suggests salinity remained high. The high DI-salinity values (between ~5,700-5,200 BC) result from a large increase in *Paralia sulcata* valves, perhaps present in exaggerated numbers due to poorer preservation (e.g. **F** index drops below 0.4 in several samples between ~5,700-5,500 BC and decreased diatom concentration). Its abundance drops off again by ~5,470 BC and *Fragilaria* and *Opephora* species dominate the record thereafter (*Fragilaria* s.l.). The decrease in DI-salinity after ~ 5,510 BC might be caused by a minor sea level regression (~1 m drop in sea level between the L2 and L3 transgression in the Blekinge stratigraphy; Berglund *et al.*, 2005) and a subsequent drop in water level, enabling *Fragilaria* s.l and other benthic taxa to increase in abundance.

There is no renewed increase in DI-salinity associated with the L3 transgression (~4,850-4,250 BC; Berglund *et al.*, 2005), probably due to the high abundances of *euryhaline Fragilaria* s.l. present in the record at this time. Variable abundances of *Fragilaria* s.l. suggests that this phase is characterised by general instability (e.g. Eronen, 1974; Stabell, 1985), likely being subject to fluctuations in salinities, nutrient status, abundances of grazing populations (e.g. molluscs and foraminifera) and variable currents, perhaps being driven by rising sea level and greater exposure to the Kolindsund. In such circumstances, several environmental variables might be influencing the diatom assemblage and perhaps skewing the DI-salinity estimates. Alternatively, salinity rise associated with higher sea levels might be counteracted by freshwater inputs. Cooler temperatures at the end of the Holocene thermal optimum (e.g. Antonsson and Seppä, 2007; Seppä *et al.*, 2009; Brown *et al.*, submitted) might have increased winter ice cover on the catchment, leading to larger inputs of freshwater (and nutrients) in the melt season.

Chaetoceros cysts are also abundant through KRP-E3, probably indicating high nutrient status and high productivity, at least in the spring and summer months. Higher productivity probably results from increased input of nutrient rich water from the Kolindsund during the Mesolithic high sea level stand (Berglund *et al.*, 2005) and perhaps nutrient inputs from the catchment, particularly during the late winter/early spring. The dominance of the larger sized *P. sulcata* valves (> 20 μ m, more abundant in nutrient poor conditions) might be explained by nutrient depletion in winter (when it reaches its highest abundances; Hobson and McQuoid, 1997), following likely spring/summer *Chaetoceros* blooms. At the top of this phase, the largest Littorina regression occurs at ~4,200 BC (~3 m decline in sea level in the Blekinge stratigraphy; Berglund *et al.*, 2005), but there appears to be little change in the biological records until ~3,900 BC (see below; KRP-E4).

In this phase the woodland is dominated by *Tilia*, *Ulmus*, *Corylus*, *Quercus*, *Alnus* and *Betula* while *Pinus* and especially *Fraxinus* are more sparsely represented. The catchment probably closely resembles the Atlantic woodland structure suggested by pollen analyses from small hollows and soils in Denmark with *Tilia* prevailing on dry well-drained soils, together with *Corylus* and *Quercus*, whilst *Ulmus*, *Alnus* and *Fraxinus* could be found on damp or wet areas, presumably mixed with some *Corylus* and *Quercus* (e.g. Iversen, 1969; Andersen, 1978; Aaby, 1983). The presence of *Hedera helix* in the pollen record is an indicator of warmer conditions than today (Iversen, 1944) associated with the Holocene thermal maximum (5,550-3,550 BC; Snowball *et al.*, 2004; Antonsson and Seppä, 2007; Seppä *et al.*, 2009; Brown *et al.*, submitted). Poaceae pollen decreases in abundance in this phase, as does *Dryopteris* (less than 2 % after 5,560 BC) and the other wet ground, swamp and fen taxa, suggesting that rising sea levels (Berglund *et al.*, 2005; Yu *et al.*, 2007) had probably drowned most of the low lying swamp and marsh areas surrounding the fjord by the late Kongemose/early Ertebølle period.

KRP-E4 – The late Korup Sø marine phase (3,900-2,600 BC) – The MI-reconstruction suggests salinity decreases at the onset of KRP-E4 (~3 ‰ decrease at ~3,900 BC, largely driven by the decrease in B. reticulatum and increase in H. ulvae) and continues to decline steadily throughout this phase. There is, however, little change in DI-salinity up until ~3,400 BC, with the exception of a relatively short-lived decline between ~3,850-3,750 BC. This short drop coincides with a peak in *C. choctawatcheeana* (~12 % at ~3,780 BC) suggesting higher nutrient status within the fjord, probably resulting from increased inwash of nutrients (and sediment) from the catchment, following the Ulmus decline and subsequent restructuring of the woodland (Figure 5.7 and see below). Unfortunately, changing nutrient status and increased sediment loads might weaken the DI-salinity estimates in this section of the record (e.g. DI-salinity model performs poorly under high nutrient conditions in the upper section of the Kilen record, see section 3.11). The bulk of evidence presented here suggests that there was no major decline in salinity at the Mesolithic-Neolithic transition and/or lower salinity throughout the early Neolithic period (as proposed by Rowley-Conwy, 1984). Despite the apparent decrease in the MI-salinity reconstruction, salinity remains high enough to support small (but sporadic) concentrations of B. reticulatum R. albella and T. decussatus (up until ~3,400 BC, all high salinity-demanding molluscs; Figure 5.6) and occasional finds of O. vitrea up until ~2,750 BC (which might be considered to be in better agreement with the diatom record). Salinity fluctuations might have been important potentially disrupting or preventing spawning of oysters and other mollusc populations on yearly to decadal scales (but
likely to be at too fine a resolution to be picked up by the diatom and molluscan records presented here). The DI-salinity reconstruction provides some evidence for fluctuating salinities (on decadal to centennial scales) between ~4,100-3,700 BC; Figure 5.10), but as indicated above, it is difficult to be certain whether or not these are true salinity variations, or just changes in the diatom flora associated with other environmental parameters (most notably changing nutrient and sedimentary regime).

Both molluscan diversity and concentration decrease in this phase, probably in response to the rapid increase in the sedimentary accumulation rate (~ 0.4 cm yr^{-1} increase) at ~3,900 BC. Petersen, (1993) previously attributed this sedimentary shift to the lowering of the tidal range in the early Neolithic period (e.g. Nielsen, 1938), but this is rejected on the basis of the lack of evidence supporting this hypothesis. It is more likely to be associated with increased sediment loads from the catchment following the *Ulmus* decline and subsequent restructuring of the woodland (discussed in more detail below). Thereafter, the accumulation rate continues to fluctuate throughout the remainder of the profile, probably in response to a mixture of land use changes (affecting sediment loads) and variations in the marine environment (e.g. sea level variation, productivity, sediment inwash from the Kolindsund perhaps under storm conditions).

By ~3,400 BC, both proxies indicate decreasing salinity, which is further supported by the reappearance of *Ruppia* in the pollen record. In fact *Ruppia* continues to gradually increase in abundance throughout KRP-E4 as the fjord begins to get shallower, due to decreasing sea levels and the high sediment accumulation rates within the basin (see below). Both the molluscs and the diatoms indicate a short term rise in salinity in the upper part of KRP-E4 (~2,800-2,700 BC), with *O. vitrea* reappearing in the mollusc record, albeit at low numbers, and small increases in marine benthic diatoms such as *Cymatosira belgica* (likely washed in from the outer exposed parts of the fjord) and *D. minor*. This coincides with the beginning of the Pitted Ware Cultural period (2,800-2,400 BC), and the reappearance of oyster-dominated coastal shell middens in some parts of Denmark (Andersen, 1983; Rasmussen and Richter, 1991; Rasmussen, 1993a; Andersen, 2007).

In KRP-E4 there is also a substantial shift in the vegetation, driven by a restructuring of the forest following the rapid *Ulmus* (elm) decline and the introduction of agriculture (Figure 5.7). *Betula, Quercus* and *Corylus* (all prolific pollen producers) quickly increase at the onset of KRP-E4 causing a brief rise in the total arboreal pollen (2-3 % higher between ~3,940-

3,800 BC), before plants indicative of open land, related to agricultural impacts, become more abundant in the profile. The elm decline in the Korup Sø catchment is likely to be predominately driven by an elm pathogen, though the early signs of agriculture in the pollen record (i.e. *Triticum* and *Hordeum* appear around ~3,850 BC) might suggest that humans also played some role (e.g. Peglar, 1993; Peglar and Birks, 1993). The appearance of cereal pollen is followed shortly after by increased frequencies of Poaceae pollen (predominately from terrestrial grassland taxa at this time) and plants characteristic of pastoral and arable farming such as *Plantago lanceolata* and *Rumex*-type (Iversen, 1941; Behre, 1981; Gaillard, 2007) both appear around 3,750 BC.

The gradual reduction in *Tilia* pollen between ~3,900-3,550 BC further reflects woodland clearances, supported by the increased abundances of secondary woodland taxa (i.e. Betula and Corylus), apophytes (most notably Poaceae, P. lanceolata and Artemisia) and anthropochores (Triticum-type and Hordeum-type; Iversen, 1941; Iversen, 1973; Rasmussen, 2005). After ~3,550 BC, Tilia re-establishes itself, accompanied by increases in Quercus, Ulmus and Fraxinus, perhaps reflecting reduced cultural pressure on the landscape at this time. This might be supported by more sporadic appearances of cereal and *Rumex*-type pollen between ~3,500-2,580BC and low percentages of *P. lanceolata* (except for occasional peaks, see below). The slight increase in grass pollen at this time might be exaggerated by the reduction in prolific pollen producing trees (e.g. *Betula* and *Corylus*). This might also suggest that the pollen record underestimates the area of open land in the earliest part of the Neolithic (i.e. 3,900-3,500 BC) due to the abundance of prolific pollen producers (e.g. Betula abundant between ~3,900-3,700 BC and Corylus increase up until ~3,500 BC). It is also possible that in the later part of the record larger proportions of pollen from wetland grass taxa are being washed into the fjord, as low lying wetlands begin to emerge, as high sediment accumulation rates cause the basin to shallow and sea levels decline (e.g. Mörner, 1976; Berglund et al., 2005).

The four-six fold increase in the sediment accumulation rate (identifiable in both wells) around ~3,900 BC (Figure 5.3) is probably the most characteristic change of this phase and therefore it is important to expand upon it here. The timing of the accumulation rate increase at Korup Sø suggests that it is driven by forest restructuring and anthropogenic disturbance associated with the introduction of agriculture, leading to increased rates of soil erosion and surface run off into the Korup Sø basin. However, as the chronology of the Korup Sø profile is based on radiocarbon dates from marine shells (Table 5.2) then it is possible that errors

associated with the marine reservoir effect are inherent in the age model, though it is argued here that these errors are likely to be relatively small in magnitude for several reasons. Firstly the sequence of radiocarbon dates appears to fit well with known palynological events particularly the Ulmus decline (~3,900 BC; Andersen and Rasmussen, 1993). This is particularly rapid in the Korup Sø pollen record (decrease of 5-6 % in ~50 years; Figure 5.7) and can therefore be considered a precise dating marker in the Korup Sø stratigraphy. Whilst there is no specific regional date for the introduction of *Plantago lanceolata*, *Rumex* and the first appearance of cereal cultivation (Triticum and Hordeum), these 'events' are in good accord with other Danish coastal, limnic and mire pollen records (e.g. Troels-Smith, 1960; Aaby, 1983; Odgaard, 1994; Odgaard and Rasmussen, 2000; Rasmussen, 2005; Rasmussen and Olsen, 2009, P. Rasmussen, unpublished data). Secondly, dates generated from the two boreholes are largely in agreement and can be tied via events in the pollen stratigraphy (Figure 5.2; H. Krog, unpublished data). Thirdly, all dates appear in stratigraphic order. The latter two points might be considered unlikely if the marine reservoir effect was as problematic at Korup Sø as in other Danish fjord sites such as Tempelkrog and Horsens Fjord (Olsen et al., 2009). Fourthly, an increase in sediment accumulation rate is often recorded in northwestern Europe after the Ulmus decline, subsequent forest restructuring and introduction of agriculture in the early Neolithic period (Odgaard, 1989; Edwards and Whittington, 2001; Rasmussen and Bradshaw, 2005; Odgaard, 2006; Rasmussen and Olsen, 2009) and it is therefore consistent with, rather than remarkable that it is recorded at this site.

It is, however, acknowledged that even minor errors (i.e. $\pm 0-200$ years) in the age model could affect the rate and overall pattern of sedimentary accumulation in the Korup Sø profile. As calculations of sediment accumulation rates are determined by chronological fix points (either ¹⁴C dates or other tie points such as palynological markers) sedimentary dynamics between dates/tie points remains unknown and therefore assumed constant (i.e. linearly interpolated). Whilst it is argued here that there is a true increase (of substantial magnitude) in the sedimentary accumulation rate following the *Ulmus* decline and the introduction of agriculture, there is potential scope for variable sedimentation, particularly in the Mesolithic section of the record. For example, as sea level begins to decline (at ~4,200 BC; Berglund *et al.*, 2005), a slight increase in the sedimentary accumulation rate might be expected (e.g. as seen at Norsminde Fjord; see section 4.11) due to a factors such as decreased size and depth of the basin, lower velocity currents, reduced marine connection and less output of sediment from the fjord/estuary into the open sea. The nearest date to the *Ulmus* decline (precise dating marker) in the Mesolithic section of the Korup Sø Well no. 2 profile is located ~50 cm below

this (at depth 702.5 cm; Table 5.2) and dated to \sim 5,070 BC (\sim 1,100 years earlier). Therefore, with the potential error in the dating model it must be considered plausible that accumulation rates might have started rising earlier than \sim 3,900 BC, perhaps in response to sea level change in the late Mesolithic period (\sim 4,200 BC; Berglund *et al.*, 2005, Figure 5.10).

Consideration must also be given as to whether a four to six fold increase in the sediment accumulation rate is possible around ~3,900 BC (under the current age model). This large magnitude increase occurs at a time when no reduction in overall arboreal pollen (despite the elm decline and subsequent forest restructuring) and very little expansion of open land (i.e. Poaceae pollen only begins to increase after ~3.800 BC and *Plantago lanceolata* and *Rumex* spp. appear after ~3,740 BC; Figure 5.7). However, the lack of change in overall arboreal pollen percentages might be explained by a higher pollen capacity of the existing forest vegetation at this time (Aaby, 1988) and the expanse of light demanding (and some prolific pollen producing) trees such as Betula, Corylus and Quercus (Figure 5.7). Aaby, 1988 suggested that openings in the forest created by Neolithic people (by tree felling) enabled a larger part of the tree's crown to flower, explaining the positive relationship between arboreal pollen influx and cultural activity in the early to mid Neolithic. Conversely, in the late Mesolithic period, when the forest was dense, arboreal pollen influx was lower (cf Iversen, 1973). The early appearance of cereal crops in the Korup Sø pollen record (*Triticum* appears at ~3,850 BC and *Hordeum* at ~3,840 BC) and lack of pastoral indicators might suggest that arable farming was the dominant mode of farming in the earliest part of the Neolithic. Arable land is believed to be more prone to soil erosion than permanently grass-covered pastures (Veihe et al., 2003), which might partly explain why there is such a substantial (initial) increase in the accumulation rate at ~3,900 BC. This might also explain why later, larger magnitude agricultural changes such as the expansion of pastoral grasslands after ~3,800 BC (i.e. increase in Poaceae, *Plantago lanceolata* and *Rumex* spp. in the pollen) and additional rapid but short term deforestation phases (e.g. ~3,350-3,200 BC; Figure 5.7) appear to have less impact on the rate of sediment accumulation in the Korup Sø basin. Unfortunately there is little data available from other Danish coastal or fjord systems for comparison, largely due to poor dating and the lack of complete palaeo records from these environments. Much more data exists for lake or mire systems and some of these records also document an early increase in soil erosion and sediment accumulation (e.g. Odgaard, 1989; Rasmussen and Bradshaw, 2005). Whilst the accumulation rates in these systems do not increase by such a large magnitude as in Korup Sø, it is likely that factors such as a reduction in the size and depth of the basin with declining sea level, reduced currents and reduced connection with the

Kolindsund will also have contributed to the large increase in the sedimentation rate at Korup Sø.

It is also important to highlight that increased soil erosion and sediment accumulation is often heavily affected by localised factors (e.g. Edwards and Whittington, 2001) and subsequently no repeatable, synchronous regional signal is identifiable across Denmark or Scandinavia. For example, in several Danish lakes and mires there is a delay evident, with accumulation rates increasing slightly later (e.g. Aaby, 1988; Odgaard, 2006; Rasmussen and Olsen, 2009), often coinciding with the appearance of the unambiguous indicators for arable and pastoral farming (i.e. cereal pollen and *Plantago lanceolata*). The few well dated coastal records available (e.g. Norsminde Fjord (see Chapter 4), Horsens Fjord and Tempelkrog; P. Rasmussen et al. unpublished data) all appear to exhibit unique sedimentation records, likely being affected by factors such as response to sea level change, fjord configuration and local vegetation and catchment changes. Furthermore, Colman et al., 2002 demonstrate that accumulation rates in shallow water sites in the Chesapeake Bay region reflect more localised factors, whereas the deeper sites appear to respond more consistently to larger-scale regional changes. On a global scale Dearing and Jones, 2003 identify that small basins are the most responsive to human impacts and show the greatest changes in sediment flux, typically up to 5-10 fold higher than under undisturbed conditions, though this figure can rise depending on the intensity of the impact or local environmental conditions. At the Mesolithic-Neolithic transition, Korup Sø can almost certainly be classified as a small shallow water fjord and probably extremely sensitive to catchment changes. It might, therefore, be considered unsurprising that a large increase in sedimentation rate occurs following a substantial forestry change (i.e. rapid decrease in *Ulmus* and structural changes of the forest) and with respect to clear early evidence for arable farming taking place in the immediate catchment (Figure 5.7).

Outside Denmark, other systems have shown substantial increases in the sedimentary accumulation rate associated with the introduction of (primitive) agriculture and even reduced sensitivity (i.e. little or no increase in sedimentary accumulation rates) to later, larger scale agricultural changes. For example, in a closed lake-catchment system in northern Guatemala, Anselmetti *et al.*, 2007 found that the most rapid soil loss (~1000 t/km² yr⁻¹) occurred during the initial land clearance when Mayan population densities were relatively low. A dramatic rise of 30-80 times the undisturbed sediment yield (used as a proxy for soil erosion) is also evident in sediments from Frains Lake, Michigan (USA) following the initial phase of deforestation by pioneer settlers (Davis, 1976). Sediment yield stabilises again after the initial

disturbance peak, but post settlement levels remain higher (~10 fold) than pre-clearance levels. Both these studies demonstrate that land clearance and primitive agriculture, likely carried out by small numbers of people can have quite large effects on erosion rates and subsequent sediment delivery to aquatic systems. Additionally, lake records from a steepland catchment in New Zealand document an increase in sedimentation/erosion rates following fire-induced changes from indigenous forest to fern/scrub following Polynesian settlement (Page and Trustrum, 1997). Later, European settlement in this catchment and a change to pastoral land cause the lake sedimentation rate to further increase, reaching a rate of 8-17 times than under indigenous forest. Therefore, viewed in the light of such drastic changes in other systems, a four to six fold increase in the sedimentation rate at Korup Sø can be considered both feasible and likely with respect to the environmental and cultural change occurring within the catchment at this time.

KRP-E5 – **Late brackish Zone** (~2,600-2,300 BC) - In the uppermost zone, sea level continues to decrease gradually and signs of this weakening marine influence become apparent in the proxy records. With the exception of *Rissoa membranacea* (present up until ~2,430 BC), there is an absence of high-salinity demanding mollusc species. Instead, the molluscan fauna is dominated by the brackish water snail *H. ventrosa* and lower concentrations of *Littorina tenebrosa* and the lagoon cockle *Cerstoderma glaucum*. This is supported by relatively high abundances of *Ruppia* (2-8 %) in the pollen record, a decrease in the abundance of dinoflagellates (Figure 5.8) and the disappearance of some of the marine diatom species such as *Dimmeregramma minor*, *Fallacia fenestrella* agg. and *Chaetoceros* resting cysts. Diatom concentration continues to increase as *Fragilaria* s.l. further increase in abundance, though this probably indicates the ability of these species to colonise unstable environments (e.g. subject to sediment disturbance, reducing habitat availability, fluctuating nutrients, salinity and sediment loads) and/or perhaps shallower water at the coring location, rather than increasing productivity and/or nutrient status at this time.

Low numbers of planktonic species throughout this zone suggest shallower water and the appearance of taxa such as *Psammothidium kuelbsii*, *Pseudostaurosira brevistriata* and *Achnanthes submarina* in the uppermost diatom sample suggest that by ~2,300 BC the Korup Sø basin was brackish, probably maintaining only a very narrow connection with the Kolindsund (e.g. DI-salinity of ~11 ‰ at this time). The appearance of scattered *Ruppia* stalks in the sediments after ~2,500 BC, together with a 1-2 % *Ruppia* rise in the pollen record likely indicates extensive emergent bedding occupying the shallow littoral areas. Some high-

salinity tolerant species remain present in the profile such as *P. sulcata* and *Fallacia cryptolyra*, though the latter of these is never particularly abundant and disappears by ~2,380 BC. Despite *P. sulcata* favouring higher salinities, its euryhaline nature, ability to cope with low nutrient conditions and tychoplanktonic lifestyle might have ensured its survival into the later stages of the fjords development. Alternatively, the relatively high percentages of *Paralia sulcata* (and continued presence other higher salinity diatoms) might be explained by greater disturbance and reworking of sediments under shallower conditions, or by inwash from the Kolindsund and/or the more exposed parts of the fjord. The sedimentation rate of the profile remains high, but continues to fluctuate, probably being affected by variable sediment loads from the catchment due to human impact (e.g. forest clearance episodes), but climate changes (or events such as storms) and minor variations in sea level might also be important.

Alnus becomes more abundant in this zone, perhaps due to expanding wetland areas, whilst *Corylus* and *Quercus* continue to fluctuate, again probably being influenced by human activity on the catchment. There is a slight decrease in *Fraxinus* pollen, whilst the other major tree taxa (*Tilia*, *Ulmus*, *Betula* and *Pinus*) remain relatively unchanged in this phase. There is a further increase in Poaceae pollen, together with a minor rise in *Plantago lanceolata* indicating increasing stretches of open catchment used for pastoral and arable farming (*Rumex* and cereal pollen also still present). A very slight increase in Cyperaceae pollen at the top of the profile probably reflects increasing wetland areas surrounding the fjord, as sea level continues to decrease and the basin begins to fill with sediment. The sediment record ends at ~2,300 BC due to continual disturbance of sediments after this date, and therefore the late Holocene history (e.g. the transition from brackish to limnic and later terrestrial land) of Korup Sø remains uncertain.

5.8 Summary

- The diatom and molluscan marine records presented here suggest that the Korup Sø basin had already been subjected to marine water input by ~7,500 BC (in accord with previous results), but a permanent connection to the Kolinsund might not have been established until after ~7,400 BC.
- In the lower section of the record (i.e. ~7,300-5,500 BC) there is a close connection between salinity and sea level change. The early Littorina transgression phases (i.e. L1-L2 in Blekinge stratigraphy; 6,550-5,350 BC) cause DI-salinity to increase rapidly between ~7,300-5,640 BC, with the exception of a brief drop (~16-17 ‰) around

~6,000 BC (likely caused by a minor sea level regression). According to the MIsalinity reconstruction, a maximum marine phase occurs between ~5,870-3,940 BC, after which salinity begins to decline. However, the DI-salinity record suggests that salinity does not begin to decline until after ~3,400 BC. Absolute maximum DIsalinities (~27 ‰) occur between ~5,730-5,510 BC, and highest MI-salinity occurs (~25-26 ‰) between ~5.340-5,250 BC.

- After ~5,500 BC, DI-salinity becomes less sensitive to changes in sea level (e.g. no response to late Mesolithic transgression), likely due the high percentages of euryhaline *Fragilaria* s.l. present in the diatom record at this time. This genus is known for its ability to tolerate a wide range of (constantly fluctuating) environmental conditions and therefore high abundances might reflect generally instability within the fjord (e.g. fluctuating salinity, nutrient status, water depth, current strength, marine exposure, substrate) and possibly even skew salinity estimates. Alternatively, other factors might also be influencing salinity (perhaps offsetting the increase expected under rising sea level, e.g. greater freshwater input) in this section of the record.
- The multivariate 'moving window' analyses suggest that in the lower section (between ~7,330-3,400 BC) predictors for marine environmental change (e.g. sea level rise, catchment erosion and inwash, sedimentation rate, changing mollusc faunas) tend to explain relatively large amounts of the variation in the diatom dataset (up to 49 %). There is, however, a shift in the upper section of the profile (i.e. ~4,870-2,380 BC) as the diatom distributions within the fjord becomes more heavily affected by catchment changes associated with human impact (i.e. deforestation and expanding agricultural land likely affecting sediment loads and nutrient inputs into the fjord).
- A large increase in the accumulation rate occurs at ~3,900 BC, probably associated with increased sediment loads from the catchment following the elm decline (probably predominately caused by an elm pathogen) and subsequent restructuring of the catchment woodland. The first clear indicators of agriculture (i.e.cereal pollen) appear in the Korup Sø pollen record at ~3,850 BC, followed shortly afterwards (~3,800 BC) by increases in Poaceae pollen and the appearance of *Plantago lanceolata* and *Rumex*. The relatively early sign of agriculture in the Korup Sø catchment might suggest that humans were partly responsible for the elm decline.
- For the full diatom dataset, DCA axis 1 is heavily driven by salinity due to the high abundance of brackish water taxa at the very bottom of the profile prior to the rapid sea level rise associated with the Littorina transgression (beginning at ~6,550 BC). When the lowermost samples are removed, salinity and water depth appear to be the

most important variables, but neither can explain all the variation on DCA axis 1. A strong salinity gradient is apparent on DCA axis 1 for the mollusc dataset. For the pollen dataset, PCA axis 1 appears to be a temporal trend, probably driven by a variety of factors including climate (particularly long-term temperature), edaphic conditions and human impact. PCA axis 2 is likely to be driven by a light-shade gradient with dense closed canopy tree taxa most abundant in the Mesolithic scoring highly and taxa indicative of more open habitats and more abundant in the Neolithic period scoring low/negative. At the very bottom of the profile (before ~7,000 BC), it is likely that erosion and inwash from the catchment is also important for explaining some of the variation in the pollen dataset captured on PCA axis 2. This is supported by the exaggerated abundances of Drypoteris (fern) spores, Poaceae (most likely from wetland grass species) and Cyperaceae (sedges) in the lowermost section of the pollen record (~7,600-5,800 BC), which suggests that extensive wetlands (including swamp and fenlands) either surrounded the basin or situated in close proximity (e.g. in lowlying areas of the Kolindsund Fjord catchment) to Korup Sø at this time. Erosion and later sub-mergence of these wetlands (resulting in greater inwash of material) was likely to be caused by rapidly rising sea levels characteristic of the early Holocene (e.g. particularly between ~6,550-5,800 BC associated with stages L1 and L2 in the Belkinge sea level startigraphy).

After ~3,400 BC, salinity begins to decline and the fjord continues to get shallower in response to declining sea levels and high sediment loads (as suggested by the high accumulation rate after ~3,900 BC). This is supported by the reappearance of *Ruppia* beds, an increase in benthic diatom taxa (accounting for >83 % of the diatom after ~3,400 BC) and the appearance of low-salinity tolerant mollusc species. Continuous disturbance of the sedimentary record after ~2,300 BC prevents assessment of the late-Holocene history at Korup Sø.

Chapter 6. Discussion

6.1 Introduction

In Chapters 3 - 5, the environmental history of each study site (Kilen, Norsminde Fjord and Korup Sø) has been independently discussed, with particular focus on localised changes within the fjord or immediate catchment area. Whilst some attention has been given to regional influences (e.g. sea level change, temperature) these have largely been discussed in the context of their potential impact upon a particular site. In this chapter, the accuracy of the presented palaeoenvironmental reconstructions (e.g. DI-salinity inferences) is assessed, and results from the individual site profiles are considered in a more regional context (i.e. comparison of results between sites and with other regional records of environmental change). The environmental changes identified in this thesis are also discussed with respect to their likely impact upon cultures and societies living in Denmark over the Holocene and existing environmental hypotheses for cultural change are reviewed (e.g. Rowley-Conwy, 1984). The specific aims of this chapter are:

- 1. To assess critically the accuracy and quality of the palaeoenvironmental reconstructions presented in Chapters 3 5.
- To compare results with other palaeoenvironmental records and establish if there are any regional patterns of environmental change (in the sedimentary regime, salinity, nutrients and/or productivity) in coastal Denmark over the Holocene.
- 3. To attempt to explain what caused the major regional environmental changes (i.e. forcing factors) identifiable in the sedimentary records.
- 4. To assess how the coastal environment of Denmark changed over key periods of cultural change and assess how influential such changes might have been in driving cultural and societal change (i.e. compare regional interactions between land, sea and society over the study period).
- 5. To evaluate how this study has improved our understanding of coastal environmental change in Denmark over the Holocene and identify key questions that require further research in future.

It is important to note that changes in the vegetation records (i.e. pollen and plant macrofossil records) are not discussed any further than the site summaries provided in Chapters 3 - 5. Only interactions between the terrestrial and marine environment (as detailed by these

records) and any associations with cultural/societal changes are considered in this chapter. Good regional accounts of the vegetation (and climate) history for Denmark and southern Scandinavia can be found in the following references: Iversen, 1973; Odgaard, 1994; Andersen, 1995b; Snowball *et al.*, 2004; Rasmussen, 2005; Antonsson and Seppä, 2007.

6.2 Interactions between environmental variables and their impact upon biological communities

Isolating the impact of a single environmental variable such as salinity in systems where several variables are changing at once is extremely difficult (e.g. Underwood *et al.*, 1998; Zong and Horton, 1999; Anderson, 2000; Engstrom *et al.*, 2000; Muylaert *et al.*, 2000; Anderson *et al.*, 2008; Velle *et al.*, 2010). Variables often co-vary or exert indirect influences upon biological communities by driving changes in other important variables and/or processes. For example, Anderson, (2000) criticized diatom-based temperature reconstructions from freshwater lakes on the basis that indirect temperature related changes or processes (e.g. soil and vegetation changes, pH, nutrients dissolved organic carbon) can exert greater influence upon the diatom assemblage than temperature itself. A number of complex statistical techniques (e.g. multivariate techniques such as RDA and CCA, variance partitioning) are now often used to explore the major trends in past and present datasets and for isolation of the most important for assessing the strength and accuracy of transfer function inferences and understanding when and where they perform well and cases where (and why) they do not.

Using transfer functions for environmental reconstructions is a popular technique, but beset with difficulties and potential sources of error, relating to both the statistical techniques and assumptions implicit (Juggins, 1992; Birks, 1998; Racca and Prairie, 2004; Telford *et al.*, 2004; Telford and Birks, 2005, 2009), and the complexity of natural biological systems (Underwood *et al.*, 1998; Anderson, 2000; Underwood and Provot, 2000; Velle *et al.*, 2010). Ryves *et al.*, (2004) highlight that often as much as 60-80 % of the variation in the dataset remains unexplained (e.g. Zong and Horton, 1999) and that a substantial amount of this information might be lost from the environmental dataset due to sampling strategies. Environmental datasets, such as the one used in this study (http://craticula.ncl.ac.uk/Molten/jsp/; Andrén *et al.*, 2007a) are often limited to a set number

of carefully selected variables (e.g. ten environmental variables measured for MOLTEN sites; see section 2.10.2), which likely results in the omission of some factors that are important in explaining part of the variation in the biological dataset (e.g. iron, dissolved organic carbon, potential metal pollutants such as mercury or lead not measured in this coastal training set). Complex temporal (e.g. diurnal, seasonal) processes are often summarised by using either spot measurement taken at the time of sampling or averaged data sampled over a period of time (e.g. Juggins, 1992; Parsons *et al.*, 1999; Clarke *et al.*, 2003; Ryves *et al.*, 2004; Clarke *et al.*, 2006).

A significant proportion of the variation in biological communities might also be explained by spatial or temporal characteristics (e.g. Lapointe, 2000; Muylaert et al., 2000). In systems which exhibit high spatial variability, large training sets are needed to fully capture true relationships between biological communities and key environmental parameters (e.g. Lapointe, 2000). Large datasets, however, are likely to introduce additional causes of variation (e.g. differences in site configuration, catchment lithology and vegetation cover, human impact). For a successful salinity transfer function, ideally a wide salinity gradient will be sampled, over which all other environmental factors are 'frozen' (i.e. exhibit minimal change throughout the dataset). It is particularly difficult to produce such ideal training sets in open coastal and estuarine systems, due the operation of a large number of factors important in driving these systems, and common co-variation of variables (e.g. salinity and nutrients; Underwood et al., 1998; Ryves et al., 2004). For large datasets spanning a wide geographical range, 'freezing' of environmental variables also becomes increasingly difficult. There is also the additional problem that modern species-environmental relationships may not hold true in the past. For example, in modern training sets there might be stressors from several environmental variables (e.g. nutrient enrichment, metal and chemical pollutants, acidification) that did not exist in the past.

There are a number of taxonomic issues which might lead to further sources of error within environmental-species training sets. Producing large training sets which capture all the necessary ecological information is labour intensive, often requiring several taxonomists (e.g. MOLTEN, DEFINE and DETECT projects) and subject to potential human error. Additionally, taxa abundant within a fossil dataset might be absent or rare in the contemporary training set for a variety of reasons (e.g. previously unidentified taxa, or taxonomic revisions in later floras, or taxa which have become locally or regionally extinct, or misidentified). In coastal system, often poor preservation and increased dissolution of diatom

valves can cause further problems (e.g. difficulties with taxonomic identification, underrepresentation or overrepresentation of taxa). Taxonomic and preservation issues are discussed in more detail below, with specific reference to the environmental and fossil datasets involved in this study.

Despite all these problems, transfer functions can perform well in coastal and estuarine systems and high quality reconstructions of parameters such as salinity and nutrient status have been produced in a number of Danish and Baltic coastal systems (Clarke *et al.*, 2003; Ryves *et al.*, 2004; Weckström *et al.*, 2004; Clarke *et al.*, 2006; Ellegaard *et al.*, 2006; Weckström, 2006; Andrén *et al.*, 2007a). In some cases diatom inferences have been compared to short-term monitoring data (e.g. Ellegaard *et al.*, 2006; Andrén *et al.*, 2007a) or longer-term documentary records (Ryves *et al.*, 2004), which have verified that transfer function inference can produce accurate reconstruction in some circumstances.

Diatom-salinity model

Salinity, depth, total nitrogen (TN) and total phosphorus (TP) explain 16 % of the variation in the modern day training set presented here (based on 210 MOLTEN and DEFINE sites, salinity gradient 0-31 ‰, 318 species; A.L. Clarke, *et al.* unpub.). Salinity is the single most important variable uniquely explaining 6.58 % of the variance, when TN, TP and depth are partialled out. Depth was also important, uniquely explaining 3.26 %, whilst TN and TP accounted for 1.96 and 1.85 % respectively. These four variables have also been identified in other coastal Danish training sets (and sedimentary profiles) as important drivers of the change within diatom communities (Jiang, 1996; Jiang *et al.*, 1998; Clarke *et al.*, 2003; Ryves *et al.*, 2004; Ellegaard *et al.*, 2006).

A strong association between salinity and the distribution of biological organisms (such as diatoms, molluscs, foraminifera) has been demonstrated for the inner Danish coastal waters and the transitional waters between the Baltic Sea and North Sea (e.g. Sorgenfrei, 1958; Jiang, 1996; Petersen, 2004; Ryves *et al.*, 2004). Diatoms have been shown to be sensitive indicators of salinity, exhibiting change to osmotic stress over short and long salinity gradients in both freshwater and brackish-marine systems (e.g. Fritz, 1990; Juggins, 1992; Gasse *et al.*, 1995; Jiang, 1996; Underwood *et al.*, 1998; Parsons *et al.*, 1999; Ryves *et al.*, 2002; Ryves *et al.*, 2004). The relationship between diatom distributions and salinity is not however necessarily linear, with often small changes in freshwater or weakly brackish systems causing a similar magnitude of species turn over as large changes might in higher salinity marine systems (van

der Werff and Huls, 1958-1974; Ryves *et al.*, 2004). It is also important to note that whilst salinity might be the most important variable, in this study higher salinity conditions tend to be associated with intensified marine conditions and/or large-scale environmental changes. For example, intensified marine conditions in Kilen occur between ~3,100-0 BC, driven by increased exposure to the North Sea and/or Skagerrak (see Chapter 3). Greater exposure to the North Sea is also likely to affect current systems, energy, productivity (Figure 6.3), accumulation rates, sediment type, and in some cases diatom preservation (see below). Salinity can also affect macrophyte abundances (important for epiphytic diatoms) and grazing populations such as zooplankton (e.g. Amsinck *et al.*, 2003), mollusc populations (Petersen, 2004; this study) and fish (Enghoff, 1999; Holm and Bager, 2002), all of which can exert further pressure upon diatom communities and influence the accuracy of single-parameter diatom-based salinity reconstructions. This is apparent in this study at Kilen between ~AD 0-600, with increased abundances of epiphytic diatoms coinciding with a regular occurence of *Zannichellia palustris* in the plant macrofossil record in response to lower salinity, decreased marine exposure and potentially shallower water conditions (see section 3.11).

In coastal systems, water depth might be considered as an indirect environmental influence upon the diatom communities with associated factors such as light and UV penetration, habitat availability, macrophyte cover, energy and wave (and perhaps tidal) action exerting more direct influences (Campeaua *et al.*, 1999; Wulff *et al.*, 2005; Weckström and Juggins, 2006). Increased water depth will reduce light penetration to benthic areas limiting macrophyte and benthic organisms (e.g. Carter and Rybicki, 1990; Kurtz *et al.*, 2003). In coastal and estuarine systems, changes in sea level will also likely affect the salinity, energy, nutrients and oxygen conditions within the fjord or estuary. In the modern training set, there is very little co-variation between depth and salinity (<0.5 % under variance partitioning), but depth appears to be important in the sedimentary records and is likely to be associated with salinity, particularly at the lower end of the salinity gradient (e.g. before the marine transgression and in the upper section of the Korup Sø record; see Chapters 4 and 5).

Only a small amount of variation in the diatom training set is explained by TN or TP, with the latter showing a small amount of co-variation (0.87 % using variance partitioning, Borcard *et al.*, 1992) with salinity. This is likely to result from the source of higher salinity water (i.e. from the North Sea) being also richer in phosphorus. The small covariation between salinity and phosphorus is deemed unlikely to cause major changes in DI-salinity inferences using this model. Furthermore, the modern training set spans a large salinity gradient, with salinity

uniquely explaining far more of the variation in the diatom dataset and therefore probably overriding the effects of TP on the diatom community. In terms of the role played by nutrients in determining planktonic growth, TN is generally considered far more important than TP in coastal and estuarine systems (Nixon et al., 1996), though TP can become a limiting factor for phytoplankton growth, particularly in the spring (Conley, 2000 and references therein). Seasonal or year round TP limitation appears to occur more often in the lower salinity areas of the Baltic Sea such as the Bothinian Bay or in close proximity to large freshwater inputs such as the Himmefjärden estuary (Andersson et al., 1996; Conley, 2000). Recent coastal and estuarine systems from Denmark often contain high TN concentrations due to anthropogenic loading, but the long-term history of TN in these systems (prior to recent major anthropogenic disturbance) is unknown. Whilst the small amount of variation explained by TN (<2 %) suggests it is not particularly important in explaining the variation in the modern diatom dataset, this does not mean that TN has not been important sometime in the past (e.g. Bianchi et al., 2000). Care must therefore be taken when interpreting DI-salinities in the sedimentary fossil records, particularly when potential key indicators of changing nutrient status are present (e.g. increased planktonic taxa, high abundances of Cyclotella choctwatcheeana) in the record. At Kilen, sedimentary pigment analysis has also been performed to examine changes independently in productivity and nutrient status.

Taxonomic problems and representation of fossil taxa in the modern training set are also key issues, which might weaken transfer function inferences. Over the course of the MOLTEN, DEFINE and DETECT projects a series of taxonomic workshops were held in order to maintain taxonomic consistency and harmonize any differences in taxonomic interpretation. In several cases, aggregate taxa were produced by clumping two or more species together (see Andrén et al., 2007a), which might have resulted in the loss of some ecological information. Diatoms from the genera *Chaetoceros* and *Rhizosolenia* are also problematic, often being extremely important components of seasonal planktonic cycles in coastal and estuarine systems, yet their vegetative cells often preserve poorly in surface sediments (e.g. MOLTEN training sites) and the sedimentary record (rare at all three sites analysed here). These taxa are generally represented by resting cysts (Chaetoceros) or remnants of spines (Rhizosolenia) which are largely unidentifiable to species level, but can account for significant proportions of the diatom community. Both genera contain a large number of species, which inhabit a wide range of environments (Rines and Hargraves, 1988; Hasle and Syvertsen, 1996), making them ambiguous salinity indicators (unless they can be successfully identified to species level). For both the modern and fossil training sets, all Chaetoceros cysts were clumped and therefore

must also be omitted from the model. In this study, there are also a large number of other taxa present in the fossil sediments that are absent from the training set. In addition, high abundances of ubiquitous euryhaline taxa (most notably *Fragilaria* s.l. in this study) make it increasingly difficult to isolate a true environmental signal (see Chapters 4 and 5).

Classically, the predictive power of a transfer function model is evaluated by its regression coefficient (r^2 for observed versus inferred values) and its root mean square error of prediction (RMSEP) following cross-validation (usually via bootstrapping or jackknifing). This criterion was used for model selection in this study with the WA-PLS model outperforming other methods tested including classic and inverse weighted average models, modern analogue technique (MAT), locally weighted-weighted averaging (LW-WA) and maximum likelihood models. An additional problem with WA-PLS models (particularly over WA techniques) is that greater care is needed in model selection and model evaluation (Birks, 1998) in order to choose the correct number of components (additional components improve the 'fit' of data by improving the r^2 and lowering the RMSEP, but can introduce 'apparent' statistics that have little or no predictive value; Birks, 1995; Birks, 1998). In statistical modelling the aim is to choose the minimum adequate model (Crawley, 1993; Birks, 1998) and therefore a common cut-off for the addition of components is if a component reduces the prediction error by 5 % or more under internal validation (ter Braak and Juggins, 1993). Whilst the WA-PLS component-3 model employed here only improved RMSEP by ~ 2 %, it was chosen because it was shown to 'significantly' lower the RMSEP from the WA-PLS model component 1 and 2 models under a randomised t-test (p<0.05) incorporated into the C2 package (Juggins, 1991-2009) and produced more realistic species estimates/coefficients.

Problems associated with model selection can be exaggerated in spatially structured environments as close similarities between neighbouring sites (e.g. similar environmental conditions and species present) can lead to spatial autocorrelation, which violates assumptions underlying many statistical tests (Telford and Birks, 2005, 2009). Spatial autocorrelation can result in over-optimistic (and therefore unrealistic) estimates of model predictive power (Telford and Birks, 2005, 2009) and subsequently lead to inappropriate model selection and unreliable inferences from fossil data. Over optimistic predictive power was demonstrated by Telford and Birks, (2009) for several different transfer function models based on calibration sets exhibiting differing degrees of spatial autocorrelation, with the RMSEP of WA-PLS (and MAT) models being substantially higher (and r^2 much lower) when 'neighbouring' sites were removed from the training sets (i.e. to reduce spatial structuring). No independent test for

spatial autocorrelation was employed in this study (e.g. h-block: Telford and Birks, 2009) but it is argued that the selected WA-PLS component 3 model produces reasonable salinity estimates that are more reliable than the other tested models. Under preliminary assessment (via comparison with other proxy records, subjective assessment of past salinity and based on ecological knowledge) all other tested models appear to underestimate salinity, with the modern analogue techniques and maximum likelihood models performing particularly poorly in almost all circumstances. It is, however, acknowledged that a degree of spatial autocorrelation is likely to be inherent in the MOTEN-DEFINE-DETECT dataset and that in future this should be properly tested (e.g. Telford and Birks, 2009) and subsequently the RMSEP and r^2 values for different transfer function models should be re-evaluated correcting for spatial autocorrelation. The final applied DI-salinity model should then be selected accordingly based on these new estimates of predictive power. Future salinity inferences using this training set might also benefit from using a consensus reconstruction approach (e.g. Bartlein and Whitlock, 1993) based on all (reasonable) model reconstructions, as it is clear that in certain circumstances the WA-PLS-component 3 model employed here also performs poorly (e.g. overestimation of some DI-salinity values (>35 ‰) at Kilen in the late pre-Roman period, ~250-0 BC).

Nevertheless, using the 'classical' approach the WA-PLS component-3 model used in this study appears to have a high predictive power (i.e. performs well under internal validation by bootstrapping with 1000 cycles; $r_{boot}^2 = 0.97$ and RMSEP = 0.36 square root salinity units), comparing well with other published salinity-based transfer functions (e.g. Ryves et al., 2004; r²_{jackknife}=0.887, RMSEP=0.246 log salinity, Wachnicka *et al.*, 2010; r²_{jackknife}=0.95, RMSEP=0.39 square root salinity units). The application of the diatom based salinity model (see above and section 2.10.2) to the three fossil records analysed in this study is therefore deemed to have produced reasonably accurate reconstructions of salinity change (qualitatively confirmed by ecological knowledge) over the study period. It is important to distinguish between patterns of salinity change and precise quantitative estimates for salinity, as the latter are extremely difficult to verify, particularly in physical coastal systems, influenced by many variables (see above) and host to high abundances of euryhaline taxa (e.g. Fragilaria s.l.). Existing palaeosalinity records for the Limfjord are only available for the last ~2,500 years, over which period the Kilen DI-salinity record is in relatively good agreement with other published records of salinity change (Kristensen et al., 1995; Christensen et al., 2004; discussed in more detail below). Salinity estimates based on foraminifera and molluscan analyses (Kristensen et al., 1995; Christensen et al., 2004) are also in good accord with the

DI-reconstructed values for Kilen, suggesting this model performs well, at least in the upper section of the Kilen profile. For Norsminde, reconstructed DI-salinity values in the reduced model (excluding *Fragilaria* s.l.) are also generally in accord with 'expected' salinities during the mid-Holocene, placing the salinity (~19-22 ‰) in the Mesolithic at slightly higher than present day (~18-20 ‰; Muus, 1967). For Korup Sø, DI-salinity inferences might be slightly lower than previously thought (e.g. Petersen, 1993) and in disagreement with the previous suggestion that lower salinity existed in the early Neolithic period (discussed in more detail below). It is, however, highly likely that the diatom record is producing more realistic estimates, based on the location of Korup Sø (i.e. right at the back of a southward limb of the Kolindsund Fjord; Figure 2.5) and the lack of major floral shifts in the Neolithic period (at both Korup Sø and Norsminde).

The diatom-salinity model appears to perform poorly at the high end of the salinity gradient, where salinity appears to be overestimated (e.g. DI-salinity inferences at Kilen >35 ‰ ~250-0 BC; see Chapter 3). Higher salinity phases are also often marked by high (and likely exaggerated) abundances of robust *Paralia sulcata* valves which might be skewing the DI-salinity estimates (see below). Across the salinity gradient, high abundances of euryhaline *Fragilaria* s.1 at both Norsminde and Korup Sø are also likely to be affecting the accuracy of the DI-reconstruction at these sites. However at Norsminde, both the full model (i.e. including eurhaline *Fragilaria* s.1.) and reduced model produce largely similar curves (though inferred values differ), which suggests that the high (and fluctuating) abundances of *Fragilaria* s.1. are not completely muting/overriding the salinity signal at this site.

6.2.1 Diatom preservation

The differential susceptibility of diatom valves to dissolution can be problematic for the creation of high quality modern day training sets that accurately detail species ecological distributions (Fritz, 1990; Reed, 1998; Ryves *et al.*, 2004; Ryves *et al.*, 2006). For example, salinity training sets from both coastal and limnic systems have demonstrated close correlation between salinity and diatom dissolution (e.g. Ryves *et al.*, 2004; Ryves *et al.*, 2006), with a general consensus that diatom preservation is poorer under higher salinity conditions. In addition to salinity, physical processes (erosion, wave and tidal action, sediment re-working, sediment transportation), water chemistry, pH and bacterial breakdown of silica and stratification pattern (e.g. meromixis) can also affect the rate of diatom dissolution in aquatic systems (Ryves *et al.*, 2009). Such processes can often be exaggerated

in coastal and estuarine systems (i.e. naturally complex and high energy environments), making fossil organisms particularly susceptible to dissolution in these environments. Understanding the impact of dissolution upon a fossil diatom assemblage is paramount to assessing the quality of the dataset, and ultimately, the confidence in palaeoenvironmental reconstructions (e.g. Ryves *et al.*, 2004).

Ryves *et al.*, 2004 found a strongly positive relationship ($r^2 = 0.33-0.74$) between diatom dissolution (as measured by the **F** index) and salinity in the Vejlerne (Limfjord) sedimentary profiles. There was also a tendency towards lower **F** index values (i.e. poorer preservation) as salinity increased in the lake samples from the training set, but valve preservation was not assessed in the higher salinity fjord sites. Unfortunately, diatom dissolution was not measured for the MOLTEN, DETECT or DEFINE training sites and therefore its influence upon the diatom distribution (i.e. the amount of variance explainable by dissolution) in the modern environmental dataset cannot be assessed. Sites heavily affected by dissolution would probably be identifiable as outliers in the training set (e.g. likely contain high abundances of large heavily silicified diatoms and overall low concentration). Nevertheless, it is likely that dissolution explains at least some of the variation in the MOLTEN/DETECT derived modern training set, which might add a further minor source of error to DI-salinity inferences in some sections of the record. This is most likely to occur at the high end of the salinity gradient, where near-fully marine conditions exist (e.g. at Kilen during the pre-Roman period, ~500-0 BC).

For the fossil dataset, **F** index sample scores can be directly plotted against the DI-salinity values (assuming reasonably accurate DI-salinity reconstruction; see above) in order to assess the relationship between these variables. The relationship between DI-salinity and diatom **F** index score was very weak (and not significant) for both Kilen (r=0.14, p>0.05) and Korup Sø (r= 0.05, p>0.05, with an inverse relationship (i.e. negative correlation) between salinity and dissolution), but for Norsminde Fjord, it was only weak (r=0.04, p>0.05; not significant) after the low salinity (fresh-brackish) samples at the bottom of the profile (i.e. 6,900-6,600 BC) were removed. For the full Norsminde dataset, there is relatively strong (significant) inverse gradient (r=0.52, p<0.05), with greater dissolution (i.e. lowest **F** index) occurring under lower salinity. This is driven by poor preservation in the lowermost salinity samples (i.e. prior to the rapid salinity rise associated with the Littorina transgression (6,900-6,600 BC; see section 4.11). In this period, Norsminde Fjord is characterised by shallow water (but subjected to rising sea level and salinity), catchment erosion, sedimentary mixing, transport and re-

deposition (see section 4.11), all of which are likely to increase dissolution. After ~6,360 BC, the preservation at Norsminde Fjord does not change systematically and is generally good (above $\mathbf{F} = 0.4$), but subject to occasional fluctuations. Canonical correspondence analysis (CCA) suggests that dissolution significantly explains 10 % of the variation in the Norsminde Fjord diatom dataset (from 38 samples for which an \mathbf{F} index assessment is available). When the lowermost 6 samples are removed (i.e. ~6,900-6,600 BC) the variance explained by dissolution is insignificant (p > 0.05; Monte Carlo test; 999 permutations).

Periods of poorer diatom preservation also exist in sections of both the Kilen and Korup Sø records, which might subsequently affect the accuracy of the DI-reconstructions (Figure 6.2). CCA analysis of the Kilen diatom record suggests dissolution is significant (p < 0.05) in explaining ~ 5 % of the variation in the diatom dataset. At Kilen, poorer preservation tends to exist throughout the more marine phase (i.e. lower **F** index values, generally **F** = 0.2-0.5) between ~3,500-0 BC, probably associated with higher salinity and/or greater exposure to the North Sea and subsequently greater energy and coarser sediments. Absolute lowest values coincide (**F** = 0.2-0.3) with the late pre-Roman salinity maxima, when common North Sea diatoms (e.g. *Delphineis minutissima* and *Cymatosira belgica*) become abundant in the diatom record. Therefore, salinity is probably also important, but relatively low **F** index values in the succeeding brackish period (~AD 0-600), probably weakens the apparent dissolution-salinity relationship (i.e. r=0.34, p<0.05 (significant) when brackish samples from the Roman/Germanic period are removed). Poorer preservation in the Roman/Germanic period might be related to lower accumulation rates (i.e. slower burial rates) or increased predation (e.g. higher molluscan abundances)..

CCA suggests that dissolution significantly also explains ~5 % of the variation in the Korup Sø diatom dataset. This is likely to be driven by the poorer preservation in the upper sediments (**F** index = ~0.25-0.45 after ~3,400 BC) of the Korup Sø profile, particularly as salinity begins to decrease, but perhaps more importantly as the basin begins to shallow, due to decreasing sea levels and sediment infilling. Poorer preservation probably results from greater volumes of sediment in the basin and sediment disturbance under shallower conditions (e.g. movement, mixing and reworking of valves, particularly during storm events, strong currents or sea surges). This period is characterised by extremely high percentages of benthic diatoms, particularly *Fragilaria* s.1. which are generally insensitive salinity indicators and likely reflect the dynamic, unstable nature of the Korup Sø environment at this time. There are also several samples with reduced **F** index values (**F** = < 0.4) between ~5,930-5,470 BC,

which broadly ties in with increasing sea level, salinity and high abundances of *Paralia sulcata* (see below).

In this study, higher salinity phases are often associated with higher abundance of *Paralia sulcata*, which probably skews the accuracy of salinity inferences at the high end of the salinity gradient. *P. sulcata* is present at a wide range of salinities, but generally reaches it highest abundances at higher salinities (Hendey, 1964; Cooper, 1995b; Jiang, 1996; Ryves *et al.*, 2004; this study), often in winter due to being able to cope with low light conditions (Hobson and McQuoid, 1997; McQuoid and Nordberg, 2003). Therefore, in cases where *P. sulcata* is abundant, higher salinity is probably a true signal, but the salinity estimates generated by the model value might be biased by potential overrepresentation due to the robust, heavily silicified nature of *P. sulcata* valves (Ryves *et al.*, 2004; see also Chapters 3-5). This is typified by the general lack of change in *P. sulcata* dissolution indices throughout any of the sedimentary profiles, even in cases where the **F** index decreases substantially (i.e. preservation conditions deteriorate). Alternatively, confidence in DI-salinity reconstructions might be increased when weakly silicifed diatoms such as *Skeletonema costatum* or *Chaetoceros* vegetative cells (very seldom found in the present study) are preserved in the sediments.

6.3 Environmental change in coastal Denmark and interactions between land, sea and society; ~7,600-1,700 BC

Obtaining high quality palaeoenvironmental records from dynamic coastal and estuarine systems is extremely challenging as a wide range of problems must be overcome (e.g. coring logistics, chronological issues (see section 2.2.1), sediment mixing and disturbance, taphonomic problems). This study has demonstrated that well preserved, high resolution sedimentary records, securely datable via ¹⁴C dating of terrestrial material, are obtainable from Danish fjord and estuarine systems. These records also offer potential for multiproxy analyses (i.e. preservation of the remains of a wide range of organisms) including proxies detailing changes in both the marine (e.g. diatoms, foraminifera) and terrestrial (pollen and plant macrofossil) environment. Such archives are therefore invaluable for testing environmental push hypotheses for cultural change (Rowley-Conwy, 1984) and for understanding how large scale changes in oceanic (and terrestrial) systems drive changes in commercially important coastal areas which continue to be foci for human settlement and

activity (both at present and in the past; Holm and Bager, 2002; MacKenzie *et al.*, 2002; Andersen, 2007; MacKenzie *et al.*, 2007; Poulsen *et al.*, 2007; Andersen, 2008a).

In Figure 6.1-Figure 6.3, selected independent proxy records (analysed in this study) are plotted on common temporal scales (i.e. calendar years, BC; based on age-depth models) to compare changes in the sedimentary regime (Figure 6.1), salinity (Figure 6.2) and productivity/nutrients (Figure 6.3) at each site. The similarities (and differences) between these sedimentary records are discussed below (in chronological order), with comparisons also being made to other palaeoenvironmental records from the region (e.g. Petersen, 1981; Berglund et al., 2005; Emeis, 2005; Antonsson and Seppä, 2007; Rasmussen et al., 2007) and to the archaeological records documenting cultural change (e.g. Andersen, 1989; Andersen and Rasmussen, 1991; Andersen, 1995a; Enghoff, 1995; Andersen, 2005; Andersen, 2007). The broad aims of this section are two fold; firstly to assess the extent to which the results from this study improve/contradict (i.e. by comparison with other regional studies) our previous understanding of environmental change in coastal areas of Denmark over the Holocene. Secondly, to consider how changes in the marine environment might have influenced past cultures and societies living in Denmark throughout the Holocene and to review critically environmental push hypotheses for cultural change (e.g. Rowley-Conwy, 1984) on the basis of the evidence presented here.



Figure 6.1- Sedimentary physical parameters for all three sites (Korup Sø, Norsminde Fjord and Kilen) plotted on a common temporal scale (~6,900-1,700 BC). Cultural division are based on those given in Hvass and Storgaard, 1993.



Figure 6.2- Selected parameters providing information about salinity at all three sites (Kilen, Norsminde Fjord, Korup Sø; ~7,400-1,700 BC). Cultural division are based on those given in Hvass and Storgaard, 1993. Regional sea level curve from Blekinge, SE Sweden (Berglund *et al.*, 2005, m a/b psl = metres above/below present sea level.). *Includes the high salinity demanding foraminifer *Elphidium incertum* (shaded black). † = Includes *Ruppia* spp. (shaded grey). DI-salinity = diatoms inferred salinity, MI-salinity = mollusc-inferred salinity.



Figure 6.3 - Selected parameters providing information about productivity and/or nutrient status at all three sites (Kilen, Norsminde Fjord, Korup Sø; ~6,900-1,700 BC). Cultural division are based on those given in Hvass and Storgaard, 1993. *= Mollusc concentration data square root transformed for Korup Sø.

6.3.1 The marine transgression and the early Mesolithic Maglemose (9,800-6,800 BC) and Kongemose (6,800-5,400 BC) cultures

The ingression of marine water into Norsminde Fjord and Korup Sø appear to differ by ~200-300 years. Korup Sø was inundated first (between ~8,000-7,500 BC; Iversen, 1937; Petersen, 1993; this study), with a limnic phase identifiable, prior to its full submersion during the Littorina transgression (Iversen, 1937; Petersen, 1993, H. Krog unpublished data). Sediments containing lacustrine molluscs (e.g. Anodonta sp.) were originally dated to the Boreal period (8,000-7,000 BC) based on pollen analyses (H. Krog, unpub.). This is further supported in this study by the occurrence of a number of freshwater diatoms present in the lowermost sediments (identified via a preliminary scanning of slides). The first ingression of seawater into Århus Bay (Dencker, 1999; Jensen and Bennike, 2008) and Norsminde Fjord (Rasmussen, submitted; and this study) occurred approximately 6,900-6,800 BC, with again a likely freshwater (limnic or fluvial) phase prior to submersion. The Kilen record (presented here) does not extend back as far as the marine transgression and so the date of inundation can only be estimated based on regional records. Petersen, (1981) suggests a marine transgression of the Limfjord occurred between ~7,000-6,000 BC and that by ~5,000 BC, sea level in the Limfjord was ~3 m above its present day level. More recently a preliminary date from deeper sediments (~2,900 cm bpsl; not analysed as part of this thesis) collected from the Kilen basin indicate marine conditions (i.e. contain marine molluscs and diatoms) at ~6,930 BC. This suggests that a marine transgression occurred earlier than ~7,000 BC, at least in the Struer area of the Limfjord.

Prior to the Littorina transgression, the Maglemose culture inhabited Denmark and Scandinavia, but the extent to which this culture exploited coastal resources remains uncertain. The majority of Maglemose sites are based inland, often near rivers and lakes (Grøn, 1995), portraying them as an inland hunter gatherer culture, exploiting freshwater fish species (particularly pike and perch) and hunting for elk, aurochs, red deer, roe deer, wild boar and small fur animals. The Maglemose coastline was very different to that today, and most of the coastal sites belonging to the Maglemose culture were probably drowned during the Littorina transgression. During this transgression (beginning somewhere between ~6,700-6,400 BC), sea level rose by approximately ~28 m in the Kattegat and Storebælt, and ~15 m in the southern Baltic Sea in just six centuries (e.g. Mörner, 1976; Petersen, 1981; Christensen, 1995; Schmölcke *et al.*, 2006). This completely transformed the Baltic and

Kattegat coastlines, drowned the Øresund and Storebælt, creating a permanent marine connection to the Baltic Sea, and resulting in brackish-marine salinities within the Baltic basin (Winn *et al.*, 1998; Andrén *et al.*, 2000; Bennike *et al.*, 2004; Berglund *et al.*, 2005). Based on the DI-salinity (and mollusc inferred-salinity, MI-salinity) records presented here, by ~6,600 BC, the salinity was relatively high at both Norsminde Fjord (~25 ‰) and Korup Sø (~15 ‰), suggesting they were quickly inundated, very early in this rapid phase of sea level rise.

The succeeding Kongemose culture certainly exploited marine resources as detailed by the numerous coastal sites (including the earliest kitchen middens) belonging to this cultural group (e.g. Andersen, 1970a; Larsson, 1990). Therefore, in archaeological terms, the Littorina transgression really marks the transition between the Maglemose and Kongemose cultures (e.g. Christensen, 1995). The Littorina transgression completely changed the landscape for humans, reducing the total land area, cutting off the land-bridge between Denmark and Sweden and likely submerged or re-configured many formerly important freshwater bodies (lakes and fluvial systems). As the marine waters poured into low lying areas such as former glacially-scoured troughs and basins, shallow accessible fjords and estuaries were created. Increasing temperatures (Iversen, 1944; Snowball et al., 2004; Antonsson and Seppä, 2007; Brown et al., submitted) and salinity (Mikkelsen, 1949; Petersen, 1993; this study) enabled edible marine fish and molluscs (e.g. mussels and cockles; Figure 5.6) to colonise these waters providing valuable resources for the Kongemose. It appears that the Kongemose quickly recognised the prosperity of the marine environment, more regularly inhabiting coastal areas and increasingly gearing their economy towards its resources (e.g. Andersen, 1970a, 2007).

6.3.2 From Ertebølle (~5,400-3,900 BC) to Funnel Beaker Culture (TBK; ~3,900-2,800 BC) and the Mesolithic-Neolithic transition (~3,900 BC)

The beginning of the Ertebølle period (~5,400 BC) broadly coincides with an acceleration in sea level rise in the Baltic Sea, Kattegat (Mörner, 1976; Christensen, 1995; Berglund *et al.*, 2005; Brenner, 2005; Yu *et al.*, 2007) and other globally distributed sea level records (Blanchon and Shaw, 1995; Blanchon *et al.*, 2002; Siddall *et al.*, 2003; Sloss *et al.*, 2005; Behre, 2007; and see section 1.3). It is likely that this rapid rise in sea level (between ~5,600-5,000 BP) triggered the large increase in shell midden accumulation in the late

Kongemose/Ertebølle period. Several factors associated with this sea level rise might have been responsible for increased cultural exploitation, including increased salinity (initially), changing sedimentary conditions (e.g. lower accumulation rates between ~6,000-3,900 BC at Norsminde Fjord and 6,200-3,900 BC at Korup Sø; Figure 6.1) and an overall increase in marine productivity. Increasing temperature is also likely to have been important as Scandinavia entered the Holocene thermal optimum (Snowball *et al.*, 2004; Antonsson and Seppä, 2007; Brown *et al.*, submitted).

The poorly constrained sea level history of the Limfjord makes it difficult to understand how this period of rapid sea level rise affected the Limfjord region. Petersen, (1981) suggests that by ~5,000 BC sea level was ~3 m above the present day sea level, but the dating of this study has been has been criticized by Christensen, (2001) and no other detailed sea level studies for the Limfjord cover this period. Based on the timings of marked changes at Korup Sø (i.e. salinity rise at ~5,600 BC; this study) and in the Baltic Sea (e.g. Brenner, 2005; Yu et al., 2007), it is likely that rapid sea level rise would have impacted the Limfjord before ~5,400 BC, and most likely through its northern opening at this time (see section 3.11). This also ties in with the submergence of the oldest known Kongemose site from the Limfjord (~5,600 BC; e.g. Brovst, layer 11, Andersen, 1970a, 2007) and slightly later, the appearance of the kitchen middens from the Ertebølle culture (~5,400 BC; Andersen, 1995a). This almost entirely predates the Kilen record presented here and so changes associated with this sea level rise at Kilen cannot be investigated. In the lowermost section of the Kilen record (i.e. between ~5,500-5,000 BC), there is very little change in DI salinity, suggesting that any potential impact upon salinity had either ceased by ~5,500 BC or by this point too small in magnitude to influence the Kilen profile. Cores collected more recently from Kilen (May 2008) offer future potential to investigate environmental changes at this site between ~6,930 BC and 5,500 BC according to the new AMS ¹⁴C dates of this deeper part of the record, but this lies beyond the scope of this study.

Both the Korup Sø DI- and MI-salinity reconstructions suggest a relatively close connection between sea level rise in the Baltic Sea (Berglund *et al.*, 2005) and salinity change between ~6,000-5,200 BC (Figure 6.2). Increased salinity between ~5,880-5,160 BC is also apparent in the reduced Norsminde DI-salinity model, but not in the full model due to DI-salinity decline at ~5,400 BC (see section 4.11). The sea level-salinity connection, however, appears to weaken in the mid-Ertebølle period (after ~5,500 BC for Korup Sø and ~4,900 BC BC for

Norsminde Fjord under the reduced model), particularly highlighted by inferred maximum salinities occurring in the late Kongemose/early Ertebølle period (somewhere between \sim 5,600-4,900 BC; see below and Figure 6.2), several hundred years before the maximum sea levels of the late Atlantic period (4,800-4,200 BC; Berglund et al., 2005). Whilst all proxies analysed in this study place the salinity maximum in this period (i.e. somewhere between ~5,600-4,900 BC), there are considerable differences between these estimates (Figure 6.2). At Korup Sø, both the DI and MI-salinity reconstruction suggest salinity begins to increase almost simultaneously (after ~6,000 BC), but highest DI-salinities (23-27 ‰; ~5,730-5,520 BC) occur before maximum MI-salinities (~5,340-5,250 BC). The earlier maximum in the DI-reconstruction is largely driven by high abundances of Paralia sulcata and therefore estimates might be exaggerated due to the robust nature of this diatom and often over representation in marine sediments (Ryves et al., 2004). A drop in the F index is apparent in several samples between \sim 5,930-5,470 BC might further support this, but dissolution of P. sulcata valves remains unchanged (see section 5.7). After ~5,520 BC, P. sulcata percentages drop as Fragilaria s.l become more abundant, suggesting that preservation might have improved and that the system is subject to instability, perhaps caused by variable salinity, nutrients and productivity (see section 5.7).

At Norsminde, highest DI-salinities under the reduced model (~5,400-5,160 BC) also occur before the high-salinity demanding species of foraminifera (particularly Elphidium incertum and Elphidium voorthuysenii) reach their maximum abundances (~5,160-4,900 BC). It is difficult to explain the drop in salinity after ~4,900 BC, perhaps being related to the minor drop in sea level between transgressions (i.e. minor regression between L2 and L3 transgressions in Berglund et al., 2005) or the drowning of low lying shallow marsh areas which might form productive habitats for diatoms, giving a fresher water signal, than the 'higher-salinity' conditions in the main part of the fjord itself. If the latter argument was true, however, then higher abundances of wet ground/marsh plants/pollen might be expected, but there is very little evidence of this in the pollen or plant macrofossil records. Alternatively, the lower salinities after ~4,900 BC might suggest that other factors exert greater influence upon salinity than sea level in the mid-late Ertebølle period. Isolating a clear salinity signal in this section of the records is extremely difficult due to the dynamic nature of the marine environment at this time, with apparent variations in salinity (Figure 6.2), nutrient status, productivity (Figure 6.3) and sedimentation rates (Figure 6.1) all occurring. All these factors might drive changes in the diatom communities and have implications for producing accurate

estimates of salinity over this period. This is exemplified by the regular salinity fluctuations (particularly in the Norsminde Fjord record; Figure 6.2) and the high numbers of *Fragilaria* s.l. present in both the Norsminde Fjord and Korup Sø records, which are highly adapted to living in unstable environments (Eronen, 1974; Stabell, 1985). It is important to note that despite decreasing in the mid-late Ertebølle period, all salinity proxies suggest that salinity remained higher (i.e. more marine) than in the earlier Magelmose/early Kongemose periods (i.e. ~7,200-5,900 BC). This probably explains the continued presence of higher-salinity demanding organisms (absent in the earlier periods), higher marine productivity and continued intense human exploitation of marine resources throughout the entire Ertebølle phase (e.g. Andersen, 1989; Andersen, 1995a, 2007; Andersen, 2008a).

In the Kilen record, maximum DI-salinity for the Ertebølle period occurs between ~5,100-4,800 BC, which likely reflects an opening or widening of the western entrance (e.g. North Sea drift diatoms such as *C. belgica* appear in the record). Whilst salinity in the preceding (i.e. before ~5,500 BC) period remains uncertain, the Kilen record suggests that DI-salinity was generally lower throughout the Ertebølle period than in the Neolithic (see Chapter 3, Figure 6.2). Reduced marine connection (and lower salinity), however, appears to have driven higher marine primary productivity (Figure 3.16) in the Limfjord throughout the Ertebølle and early Neolithic period. This probably provided sufficient food for higher organisms (e.g. molluscs and fish), enabling them to reach high abundances, subsequently providing an environment rich in marine resources for cultures and societies inhabiting the Limfjord catchment (Andersen and Johansen, 1986; Andersen and Rasmussen, 1991; Enghoff, 1991; Andersen, 1995a; Andersen, 2005).

6.3.2.1 The oyster decline and the Mesolithic-Neolithic transition

The large scale shift from oyster dominated middens in the Ertebølle period to cockle dominated middens in the Neolithic (Funnel Beaker Culture) has generated substantial interest regarding the role of environmental change, particularly salinity change (e.g. Rowley-Conwy, 1984), in potentially forcing the introduction of agriculture into Denmark (see section 1.6.2). Contrary to the "salinity hypothesis", there appears to be no synchronous, regional decrease in salinity, at or near the Mesolithic-Neolithic transition, or any convincing evidence to suggest that salinity was markedly lower in the early Neolithic period. In fact, salinity actually increases in the Neolithic period at Kilen (see section 3.11) whilst at

Norsminde Fjord, it remains relatively high until the fjord reaches its sedimentation limit (at ~1,700 BC). At Korup Sø, a shift towards lower DI-salinities occurs later in the record (after \sim 3,300 BC), when the fjord basin begins to shallow. There is, however, some disagreement between the Korup Sø DI- and MI-inferences (Figure 6.2), with the latter supporting a decrease in salinity at ~3,900 BC and subsequent lower inferred salinities in the Neolithic period. This transition broadly coincides with a large increase in the accumulation rate (Figure 6.1), which might have driven the changes in mollusc record, rather than salinity (Petersen, 1993; see section 5.7 for more detailed discussion). Elsewhere, preliminary diatom analyses from two other Danish coastal sites (Tempelkrog in Isefjord and Horsens Fjord; Figure 2.1) also fail to document any marked change in salinity over the Mesolithic-Neolithic transitions (A. Clarke, unpublished data, K. Jensen unpublished data). Furthermore, in the NW Baltic Sea salinity continued to increase up until~3,000 BC (Emeis et al., 2003), despite the drop in sea levels (Berglund et al., 2005). Bailey and Milner, (unpub.) also dismiss salinity as causal factor for the quantitiative compositional changes in the Norsminde shell midden due to the continued presence of both Ostrea edulis and Tape decussatus (highsalinity demanding species; Jensen and Spärck, 1934; Yonge, 1960) throughout the Neolithic shell layers.

The lack of any long-term salinity decline does not necessarily mean that salinity change had no impact whatsoever upon mollusc abundances and/or distributions. Despite some uncertainties with the DI-salinity signal (see above) it remains highly likely that salinity was subject to fluctuations (magnitude 0-5 ‰ in the three DI-salinity reconstructions; Figure 6.2) at, or around the Mesolithic-Neolithic transition. It is, however, extremely difficult to assess the impact that short term salinity 'events' might have on mollusc abundances due to the low numbers of molluscs generally present in sedimentary archives and the coarse resolution of archaeological shell midden deposits. Whilst it is difficult to establish any firm linkages between short term salinity events and molluscan change, it certainly remains plausible that non-breeding populations (e.g. when salinity drops below 23-25 ‰ for O. edulis; Jensen and Spärck, 1934; Yonge, 1960) might become stressed in relatively short time periods (e.g. 1-20 years), particularly if regular and/or intensive human exploitation continues. Following deterioration, molluscan populations can also exhibit slow re-generation times (e.g. after the Limfjord re-opened to the North Sea in AD 1825, oysters did not appear until 24 years later (AD 1849; Bailey and Milner, submitted), which might further affect human resource scheduling and encourage the introduction of alternative resources (e.g. Rowley-Conwy,

1984). The most striking DI-salinity decrease occurs at Kilen between ~4,100-4,000 BC, with average salinity appearing to remain low up until ~3,900 BC (Figure 6.2). This might be related to a narrowing of the northern entrance of the Limfjord (see section 3.11) and it is likely that lower salinity exerted at least some pressure upon oyster abundances in the nearshore areas of the Limfjord in the late Mesolithic (e.g. the documented oyster decline in the shell midden at Bjørnsholm Bay; Andersen and Rasmussen, 1991).

Based on the data presented in this study, it is unlikely that any single environmental factor alone, can explain the widespread oyster decline apparent in various Danish shell middens (e.g. Andersen, 2007). There is, however, plenty of evidence suggesting that both the marine and terrestrial environment changed substantially over the late-Mesolithic/early Neolithic period (e.g. Mikkelsen, 1949; Petersen, 1993; this study). This period is also marked by deteriorating climate (e.g. decreasing temperatures; Antonsson and Seppä, 2007) associated with the end of the Holocene climatic optimum (Seppä and Birks, 2001; Snowball et al., 2004; Antonsson and Seppä, 2007; Brown et al., submitted). It is argued here, that environmental change did play an important role in the oyster decline and the changing prosperity of the marine environment over the Mesolithic-Neolithic transition, with both sea level and climate change (particularly temperature) being the predominant drivers. The precise changes (i.e. response to sea level and/or climate change) often differ between sites and therefore the manner in which changes manifest within individual fjords and estuarine systems appear to be particularly important. For example, sea level is an important driver of changes in salinity (see section 1.4 and above), sedimentary regime (including substrate and accumulation rate), water depth, productivity and/or nutrient status, all of which are likely to have implications for community dynamics and societies exploiting marine resources. Climate and temperature change can influence catchment processes (e.g. precipitation, run off, winter ice cover and subsequent thawing events), vegetation change, coastal sea ice extent, marine productivity (e.g. Andrén et al., 2000; Emeis et al., 2003) and sea level (Berglund et al., 2005), as well as directly affecting the reproduction and metabolism of marine organisms (e.g. oysters; Jensen and Spärck, 1934; Yonge, 1960; see below).

The sedimentary changes at both Norsminde and Korup Sø, suggest that the substrate and sedimentary regime might be more important than previously thought for explaining/forcing the shell midden oyster decline (e.g. Andersen, 1995a, 2007; Andersen, 2008a; Bailey and Milner, 2008). The period of lowest accumulation at Norsminde and to a lesser extent Korup

Sø broadly coincides with the maximum sea levels of the late Atlantic period (~5,600-4,200 BC), as evident in Baltic Sea and Kattegat sea level curves (Mörner, 1976; Christensen, 1995; Berglund *et al.*, 2005) and with maximum oyster abundances in Danish shell middens (Andersen, 1989; Andersen, 1995a, 2007; Andersen, 2008a). At Korup Sø, this relationship with sea level is less obvious as the accumulation rate is very low (0.03-0.09 cm yr⁻¹) throughout the entire Mesolithic section of the profile (Figure 6.1). Whilst absolute lowest rates do coincide with the maximum sea levels of the late Atlantic, the accumulation rate does not really increase until after ~3,960 BC, thereby post-dating the late Atlantic regression (~4,200 BC as dated by Berglund *et al.*, 2005). Whilst sea level fall might be a contributory factor to the rise in accumulation rate after ~3,960 BC (by reducing the strength of currents, thereby lowering the energy of the system and encouraging deposition of finer sediments within the fjord), catchment changes (e.g. forestry restructuring after the *Ulmus* decline) and increased marine productivity are also likely to have been important factors (see section and 5.7). Nevertheless, the site clearly demonstrates a large shift in the sedimentary regime, near the time of the oyster decline (Andersen, 2007; Bailey and Milner, 2008).

Ostrea edulis struggles with higher volumes of fine sediment, being less efficient at ejecting continuously accumulating sediment from its mantle cavity than species such as Cerastoderma edule, which are better adapted to living in finer sediments (Yonge, 1960; Bailey and Milner, 2008; Bailey and Milner, submitted). The gradual increase in accumulation rate at Norsminde (after ~4,700 BC) might have eventually been sufficient to stress O. edulis populations, particularly within the inner more human accessible parts of the fjord, where often less turbulent (or even still) waters might have led to greater deposition of fine material. Lower salinity might be also expected in conjunction with lower sea level, due to a likely reduction in the connection with the Kattegat (and Skagerrak/North Sea). The lack of any systematic salinity decline (see above) might be explained by other factors, such as increased terrestrial input from the land being more important for the change in the sedimentary regime (e.g. particularly for Korup Sø; see below), or that the sedimentary regime is more sensitive to changes in fjord (e.g. sea level) configuration than salinity. Alternatively, records from the Skagen and Skagerrak region document an increase in the strength of the Jutland current and a reduction in the outflow of Baltic water at the surface, and subsequent decrease in the strength of the Baltic current beginning around 4,300 BC (~6,300 BP in Conradsen and Heier-Nielsen, 1995; Jiang et al., 1997; Gyllencreutz et al., 2006). These current systems might have helped maintain the salinity within the Kattegat and

inner Danish coastal waters, despite the drop in sea level and subsequent reduction in the size of the fjord entrances (plus likely siltation).

Unfortunately, there is a lack of suitable records available for the inner Danish waters or Kattegat sites with which to compare changes in the sedimentary regime over the early-mid Holocene. This is partly due to poor dating, as a robust age model is essential for determining changes in the accumulation rate at a site, but only two other sites from eastern Denmark (Tempelkrog and Horsens Fjord; P. Rasmussen and D.B. Ryves unpub., Figure 2.1) have long-term records with chronologies based on AMS ¹⁴C dates on terrestrial material. At Tempelkrog, a sharp peak in accumulation rate occurs somewhere between ~4,400-3,900 BC, which broadly coincides with the late Mesolithic regression in the Blekinge sea level stratigraphy (~4,200-3,800 BC; Berglund et al., 2005). This peak also succeeds a relatively low accumulation phase throughout the majority of the late Mesolithic period at Tempelkrog. The Horsens Fjord record, however, does not exhibit any increase in the accumulation rate in the late Mesolithic or early Neolithic period. This might be due to the sheer size of Horsens Fjord, being much larger than Tempelkrog or any of the other sites analysed in this study. It is likely to have maintained a large, open connection with the Kattegat, even during the late Mesolithic sea level regression (Berglund et al., 2005), and therefore relatively minor changes in sea level might have had little impact upon the sedimentary regime.

For the other systems, however, a more open connection with the Kattegat and intensified marine conditions under high sea level conditions (between ~5,600-4,200 BC) is likely to result in increased energy within the system (stronger currents), greater transport of material out of the fjords and possibly greater breakdown and decomposition of material. It is also important to note here, that whilst sea level appears to influence the sedimentary regimes at Norsminde, Korup Sø and Tempelkrog, particularly during the late Mesolithic period, it is not a straightforward relationship, as all three sites exhibit quite different accumulation patterns. This might be related to physical differences between sites (e.g. site configuration, exposure to currents, depth, topography) and/or additional factors (e.g. productivity, catchment vegetation, human impact), potentially influencing the independent sedimentary regimes.

The idea that substrate was perhaps more important than salinity in driving the distribution of oysters in the Mesolithic has previously been suggested by Rasmussen, (1958). In the Mesolithic *Tapes pullastra* and *O. edulis* exhibit practically identical distributions

(Rasmussen, 1958), but their salinity tolerances today appear to be quite different. T. *pullastra* occurs at 18 ‰ or maybe even at lower salinities (Rasmussen, 1958), whereas salinities of at least 23-25 ‰ are needed to support breeding populations of *O. edulis* (Jensen and Spärck, 1934; Yonge, 1960). Rasmussen, (1958) suggests that their affinity to a similar substrate better explains their identical distributions, with both species preferring hard bottom substrates (i.e. sand and coarse sediments) with sparse vegetation cover. Such conditions were more likely in the late Mesolithic when the sea was more open (e.g. Berglund *et al.*, 2005) and deeper water limited vegetation cover (e.g. eel-grass beds).

In addition to sea level, a number of other factors might also influence the sedimentary regime of a particular site. For example, increased productivity at Norsminde Fjord between 4,550-3,510 BC is likely to have contributed to the higher accumulation rates in the late Mesolithic. Higher productivity is indicated by marked increases in both the diatom concentration (particularly planktonic diatoms from the genus *Chaetoceros*) and organic matter flux (Figure 6.3). A period of increased marine productivity is also evident at Kilen between ~4,400-3,100 BC (see Chapter 3). In contrast, the short term drop in *Chaetoceros* cysts at Korup Sø between ~3,900-3,770 BC (Figure 6.3) might represent a decrease in productivity. However, the diatom concentration does not decrease, due to higher abundances of benthic taxa, probably as a response to lower sea levels and subsequently shallower water between ~4,200-3,800 BC (Berglund *et al.*, 2005; Figure 6.2). It is therefore possible that reduced habitat availability and increased competition for resources is driving the *Chaetoceros* decline rather than marine productivity. Alternatively the *Chaetoceros* decline might be a data artefact, with the *Chaetoceros* abundances actually remaining the same, but being diluted by an increase in benthic diatoms due to shallower water.

The large increase in accumulation rate after ~3,960 BC at Korup Sø appears to be unexplainable by sea level alone, occurring towards the end of the late Mesolithic/early Neolithic regression phase. Whilst sea level is probably important (e.g. rise in accumulation rate occurs during likely shallower water associated with sea level regression phase; ~4,200-3,800 BC; Berglund *et al.*, 2005), the timing of this sedimentary change (i.e. rise) is very close to the sharp *Ulmus* decline at Korup Sø (Figure 5.7), possibly suggesting a role for increased input of terrestrial material and nutrients following vegetation change. The pronounced increase in *Quercus, Betula* and *Corylus* pollen immediately after the *Ulmus* decline suggest large scale restructuring of the catchment woodlands (see section 5.7) which

certainly might have influenced the volume and rate of sediment input. High accumulation rates were likely maintained by human impact upon the landscape (particularly forest clearances for arable and pastoral farming) with the first signs of agriculture appearing relatively early in the Korup Sø pollen record (e.g. first cereal pollen present at ~3,850 BC). A possible increase in marine productivity occurs shortly after ~3,700 BC (e.g. increase in *Chaetoceros* cysts and *C. choctawhatcheeana*), probably in response to both rising sea level (i.e. L4 transgression, Berglund *et al.*, 2005; Figure 6.2) and increased nutrient input from the catchment associated with forest restructuring and land use change.

The very different sedimentary regime at Kilen suggests that this site is either heavily driven by more localised changes, or that mechanisms important in governing accumulation and fluxes in the Limfjord and/or northwestern Jutland region differ from those operating along the eastern coast of Jutland (see section 3.11). The accumulation rate at Kilen is very high $(0.33-0.49 \text{ cm yr}^{-1})$ between ~5,500-3,150 BC, far exceeding the rates at both Korup Sø (before $\sim 3,900 \text{ BC}$; 0.03-0.09 cm yr⁻¹) and Norsminde Fjord (0.09-0.17 cm yr⁻¹). Such high accumulation rates might explain why shell midden sites are absent, to date, from the western reaches of the Limfjord (e.g. Figure 1.3 - Location map for Danish archaeological sites referenced in the text.). Fast accumulation of fine sediments in the far southern and western reaches of the Limfjord might have limited the oyster's ability to survive there (see above) there, particularly in the inner more accessible areas. The accumulation rate at Kilen drops after ~3,200 BC, which coincides with increasing salinity, likely coarsening of sediments due to stronger currents meaning less settling out of finer material (both suggesting intensified marine conditions) and a decrease in marine productivity (Figure 6.1). This might suggest that conditions actually became more suitable for oysters in the western Limfjord after ~3,200 BC as a result of higher salinity and reduced accumulation of finer substrates. This might also help explain the high numbers of oyster present in some of the Neolithic layers of the Krabbesholm shell midden (Andersen, 2005; Nielsen, 2008), which contrasts with many of the other shell middens because it does not exhibit a 'conventional' oyster decline (e.g. Andersen, 1989; Andersen and Rasmussen, 1991).

As suggested by Petersen, (1993), the case for sea level as a predominant driver of changes in the accumulation rate and substrate type might be further strengthened by the lack of change in the sedimentary regime in other archives such as lakes (e.g. Odgaard, 1989; Odgaard, 1992; Rasmussen and Bradshaw, 2005). For example, if large scale catchment changes (and
associated processes) were responsible, then increased accumulation rates might be expected in these other archives. Additionally, the reduction in the strength of the Baltic current after ~4,300 BC (Conradsen and Heier-Nielsen, 1995; Gyllencreutz *et al.*, 2006; see above) might also have further reduced energy within the Kattegat system and connected coastal waters, and subsequently led to greater deposition of fine sediments.

It is also likely that temperature exerted an additional stress on oyster populations in the late Mesolithic. Spawning of *O. edulis* occurs in the summer months when temperature reaches above ~15°C (Yonge, 1960). Scandinavian temperature records suggests that temperature was declining following the Holocene climatic optimum (~5,500-3,500 BC) and that by the Mesolithic-Neolithic transition, average annual temperatures had fallen by almost ~1°C (Seppä and Birks, 2001; Snowball et al., 2004; Antonsson and Seppä, 2007; Seppä et al., 2009). There is evidence suggesting that temperature might have decreased more rapidly during the last 500 years of this thermal optimum (~4,000-3,500 BC; Snowball et al., 2004), possibly coinciding with one of the North Atlantic ice rafting events (Bond et al., 1997). At this timer there was increased activity in Swedish and Norwegian glaciers (Karlén, 1988; Karlén and Matthews, 1992; Matthews and Karlén, 1992; Dahl and Nesje, 1994; Dahl and Nesje, 1996), a shift to colder and drier winters in western Norway (Nesje et al., 2001) and a distinct fall in mean July temperatures in Finland (Korhola et al., 2002). More recently, quantitative temperature inferences have been extended into Denmark (Brown et al., submitted), exhibiting similar patterns to other Fennoscandian records (e.g. Seppä and Birks, 2001; Antonsson and Seppä, 2007). A reduction in temperature has been shown to have affected the distribution of other organisms such as *Emys orbicularis* (European pond turtle), which became extinct from Denmark and southern Sweden in the early Subboreal period (3,750-1,750 BC; Sommer et al., 2007). There is also an southward expansion of Picea abies in southern Sweden (Giesecke and Bennett, 2004) and a general decline in the abundance of thermophilous vegetation (e.g. Hedera helix, Viscum album) as the climate cooled (Antonsson and Seppä, 2007; Brown et al., submitted).

Oysters are likely to have been more sensitive to short term decreases in temperature (and/or salinity) than cockles in this period, being close to both their temperature and salinity tolerance limits (Jensen and Spärck, 1934; Yonge, 1960). The shift to cooler summers probably suppressed the length of the oyster spawning season, whilst colder drier winters and likely increased irregularity and frequency of extreme cold events might have led to elevated

oyster mortality. Temperature is actually considered to be more important than salinity for oyster reproduction (Korrigna, 1952; Yonge, 1960) and in some areas of Norway and Sweden, colder temperatures prevent oyster from spawning at all in some years (Johannesson *et al.*, 1989). Carbotte *et al.*, 2004 have also previously linked the demise of the eastern oyster (*Crassostrea virginica*) from the Hudson River estuary to cooler climates and increased sediment accumulation rates. Prolonged cold weather (temperatures around 0°C) throughout the winter months of 1939-1940 is also believed to have destroyed more than half the native population and almost the entire imported population of *O. edulis* in the Thames estuary (Yonge, 1960). Rapid thawing events in Denmark might also have temporarily lowered salinity within the coastal waters, which might have further increased mortality. As with the salinity events (see above), it is extremely difficult to pick up short-term temperature changes in sedimentary records and directly assess their impact upon mollusc populations. Rapid thawing events might also explain some of the fluctuations in DI-salinity throughout this period (Figure 6.2).

Temperature decrease certainly offers a plausible explanation for the oyster decline yet continued presence and exploitation of oysters into the Neolithic, as evident in the shell middens (Andersen, 2007). *Ostrea edulis* can survive in substantially lower temperatures and salinities than it requires for successful spawning (Yonge, 1960). Oysters can also live up to 20 years in oyster beds and therefore, even during prolonged periods of colder weather (and/or lower salinity), oysters probably survived in the inner waters. Despite this, shorter spawning periods and possibly years without any reproduction would have inevitably impacted oyster populations, leading to greater stress, particularly within the nearshore accessible populations, if exploitation continued at the same rate in the early Neolithic as in the Mesolithic.

The suggestion that a decline in oysters alone forced the neolithisation of Denmark is certainly oversimplifying a complex transition. However, the decline in oysters appears to be a symptom of general change in the marine environment and it is increasingly likely that a change played at least some role in the transition from a Mesolithic to a Neolithic economy. Carbon (δ^{13} C) and nitrogen (δ^{15} N) isotopic evidence from human bones and teeth suggests greater consumption of marine resources in the Mesolithic with a switch to a more terrestrial diet in the Neolithic (Tauber, 1981; Milner *et al.*, 2004; Andersen, 2007; Andersen, 2008a), though this evidence remains equivocal to date (e.g. Milner *et al.*, 2004, 2006; Richards and

Schulting, 2006a). The isotopic shift, however, fits well with a change in the exploitation of marine resources (e.g. reduced oyster collection and change in seasonality of shell gathering; Bailey and Milner, 2008; Bailey and Milner, submitted) and the appearance of agricultural indicators in the Danish pollen records (Iversen, 1941; Behre, 1981; Odgaard, 1994; Andersen, 1995b; Rasmussen, 2005). Whereas the changes in both the isotope (δ^{13} C and δ^{15} N) and the pollen records, could potentially be explained by cultural factors alone (e.g. environmentally-unforced human resource shift), substantial evidence exists to suggest that the oyster decline and the shift in marine resources is synchronous (within the constraints of dating shell midden deposits) across Denmark and strongly related to environmental change (see above).

In this study it is suggested that changes in the substrate and sedimentary regime (Figure 6.1) probably made it more difficult for oysters to survive, particularly in the innermost, more accessible areas of the fjord where shallower, still water might have increased fine mud deposition. In addition, short term stresses associated with salinity, productivity and temperature events, probably driven by sea level, vegetation and climatic changes (see above) are likely to have prevented spawning in some years and even caused widespread mortality, suppressing oyster populations. More variable abundance, combined with continued human exploitation would have placed greater pressure on oyster beds, probably making them less of a dependable resource, causing their decline in the middens as humans subsequently shifted towards a more terrestrial-based diet (e.g. Tauber, 1981; Andersen, 2007; Andersen, 2008a).

There is still substantial work to be done before a clear and certain link can be made between changes in the marine environment forcing changes in the shell middens. Thorough investigation of the accumulation rates and changes in the substrate from more localities across the inner coastal waters of Denmark and even eastern Sweden is essential to understand fully changing sedimentary conditions, particularly in response to sea level changes over the late Mesolithic and early Neolithic period. Such inferences can be strengthened by use of proper sedimentological techniques (e.g. grain size analyses, magnetic susceptibility), in conjunction with the more standard sedimentary parameters and descriptive lithological techniques employed in this study.

More long term records of salinity change are also required, particularly from the Limfjord region, where there is evidence to suggest lower salinities might have existed in the late

Mesolithic period. Future diatom counts should involve larger census counts (i.e. 300-500 valves excluding Fragilaria s.l, Chaetoceros and Rhizosolenia species), so that salinity models can be tested excluding euryhaline indicators, or taxa that cannot be differentiated below the genus level. This is particularly important for east Jutland and Kattegat coastal sites which contain particularly high abundances of Fragilaria s.l., likely due to regular disturbances, instability and variable salinity, nutrient concentrations and productivity. Exclusion of Fragilaria s.l. is likely to overestimate salinity inferences, but comparison of full and reduced models might help to establish periods where transfer functions are performing well from sections where other factors might be driving changes in the diatom distributions (i.e. disagreement in patterns of salinity change between models). Future diatom inferences would be strengthened by high resolution foraminifera, molluscan (where possible) and ostracod records for comparison of salinity inferences and to begin to understand better under what circumstances these independent proxy records agree or disagree with one another. Other proxies not considered here, can also be used, for example, dinoflagellates (abundant at Korup Sø; Figure 5.8). Cyst morphology (particularly process length of dinoflagellates) has been shown to be closely associated with salinity change, making it another potential palaeosalinity indicator (Ellegaard et al., 1993; Ellegaard, 2000; Ellegaard et al., 2002; Mertens et al., 2009).

High resolution records of water temperature (e.g. alkenone records or Mg-Ca records) from fjord sites might help to improve our understanding of the role of temperature in driving the changes in oyster (and cockle) abundances. Such records would also provide a useful comparison with the quantitative temperature records (particularly regional pollen- or chrionomid based studies; Seppä and Birks, 2001; Antonsson and Seppä, 2007; Seppä *et al.*, 2009), which might be subjected to temperature lags due to the slower response of vegetation to climatic changes (e.g. Pennington, 1986; Williams *et al.*, 2002). Pollen-based temperature records are also less likely to be able to pick-up any short-term extreme temperature events and might also be affected by human impact too (particularly after the introduction of agriculture in Denmark; e.g. Hannon *et al.*, 2000; Odgaard and Rasmussen, 2000; Brown *et al.*, submitted).

Furthermore, as suggested by Bailey and Milner, (2008), it still remains likely that a mixture of environmental and cultural factors explain the changes in the shell middens (and the introduction of agriculture). In addition to changes in the substrate and temperature, they

speculate that population dynamics (i.e. a larger Neolithic population), changes in economic scheduling and/or a reduction in conservation measures (i.e. more careful measures taken to protect marine food source in the Mesolithic, when it was more important to their subsistence) might have placed more pressure on marine resources (such as cockles and oysters) in the Neolithic. In terms of the introduction of agriculture, the evidence presented in this study supports the idea that environmental change was important in forcing the oyster decline and subsequent shift in shell midden composition, but more work is needed before the agricultural question can be confidently addressed and non-environmental hypotheses can be dismissed (e.g. Rowley-Conwy, 1985; Klassen, 1996-97; Petersson, 1996-97; Johansen, 2006). For example, more archaeological work is also needed from the middens to understand better the rapidity of midden changes (both ecological and cultural) and regional differences in speed and timing of these changes. Whilst timing and rapidity is beset with difficulties, particularly when using chronologies based on ¹⁴C dating, further complications might be commonly encountered in shells middens associated with discontinuous accumulation and sedimentary gaps. A better understanding of the physical environment that the shell midden formed in is also required and greater focus placed on understanding exactly why Mesolithic and Neolithic artefacts are sometimes found in mixed layers (e.g. reworking or cultural dualism; Petersson, 1996-97). Isotopic, age and seasonality analyses (e.g. Milner, 2001a, 2001b; Bailey and Milner, 2008) of the shells present from the actual shell middens and parallel palaeoenvironmental investigations in the immediate vicinity of other major shell middens might help answer some of these questions.

Oyster abundance generally remains low throughout the entire TBK section of Danish shell middens (i.e. up until ~2,800 BC), as marine resources appear to become less important in the TBK diet (Tauber, 1981; Andersen, 2007; Andersen, 2008a; see section 1.6.2). A period of lower and more variable abundance of oysters (and other marine resources) probably ensued, as sea level and (probably) salinity (see above) continued to fluctuate and air temperature further decreased (Snowball *et al.*, 2004; Antonsson and Seppä, 2007; Seppä *et al.*, 2009; Brown *et al.*, submitted). Large scale human impacts upon the catchment also become more visible in the sedimentary records in the TBK period, as seen for example at Norsminde and Korup Sø (see section 4.11 and 5.7). Episodes of forest clearances and land-use change are probably the main causes of the fluctuations in the accumulation rate at these sites, though sea level, climate and natural vegetation change probably all contributed. The impact of humans upon the Kilen catchment area, and the level of interaction between the terrestrial

and marine environment, is more ambiguous, due to lack of information regarding the local terrestrial environment (i.e. no real pollen record or poor plant macrofossil record; Figure 3.3 and Figure 3.13). There is a minor increase in the accumulation rate between 3,700-3,400 BC, which broadly coincides with an appearance by *Plantago lanceolata* and increasing Poaceae pollen. This is possibly related to increased input of nutrients and sediment from the catchment, causing the organic and minerogenic matter to increase and helping to maintain high pigment concentrations/accumulation at this time. It is difficult, however, to be certain that terrestrial input is the source with such little information available regarding changes in the surrounding catchment. After ~3,200 BC, accumulation rate drops substantially, and changes in the marine environment (i.e. greater exposure to the North Sea, higher energy and higher salinity) appear to control the sedimentary regime at this site.

6.3.3 The Pitted Ware and Single Grave cultural period (~2,800-2,400 BC)

The increase in the number of shell middens and shell banks in the Pitted Ware/Single Grave cultural period (hereafter PW-SG), and the switch back to oyster-dominated middens, might suggest either a return to higher marine productivity, intensified marine conditions (e.g. increased salinity, coarser sediments) or a change in cultural resource exploitation. Temperature reconstructions (e.g. Antonsson and Seppä, 2007) suggest a minimal change (gradually decreasing) throughout the PW-SG cultural period and is therefore unlikely to be able to explain shell midden patterns, although occasional warm temperature events might have been important (see above). In the Limfjord region, salinity increases over the PW/SG periods, almost reaching its present day value for the Struer region (24-27 ‰ according to the Kilen DI-reconstruction). This is probably due to opening and/or widening of the western entrance (see section 3.11) in response to rising sea levels (Behre, 2007). Higher salinities and a change in the sedimentary regime (i.e. drop in the accumulation rate at Kilen and likely coarsening of sediments) after ~3,100 BC might have resulted in a more suitable environment for oysters and enabled them to penetrate into the inner reaches of the fjords/estuaries. It is important to note, that despite decreasing, the accumulation rate at Kilen remained relatively high (0.15-0.18 cm yr⁻¹) and therefore oysters probably might still have been confined to the more exposed, outer reaches of high-accumulating fjords such as Kilen.

There is also some evidence suggesting that higher salinities existed at Norsminde Fjord and Korup Sø, either prior to, or during the PW-SG cultural period. At Norsminde, the DI-salinity

reconstruction suggests that higher salinity might have existed between ~3,070-2,900 BC and ~2,700-2,570 BC. Under the full model, average DI-salinities reach 23-25 ‰ in these two phases, suggesting that in some areas of the fjord spawning populations of oysters might have existed (i.e. requiring salinities above 23-25 ‰). Again, oysters were probably still confined to the more exposed parts of Norsminde Fjord due the persistence of relatively high accumulation rates throughout the TBK period and the large increase in accumulation rate at ~2,800 BC probably driven by local catchment changes (Figure 6.1 and see section 4.11). For example, the Kalvø site (a shell midden dated to the Single Grave culture, where oysters make up 68 % of shellfish remains; Andersen, 1983) is located along the north eastern coast of the former island within Norsminde Fjord (Figure 2.4), placing it relatively close to the Kattegat opening and potentially subject to greater currents and coarser sediments than in the inner reaches of the fjord. Alternatively, sediment focussing might have resulted in higher accumulation rates at the coring location, but perhaps not in the shallower reaches of the fjord, oysters might have been able to survive in these locations. Their confinement to the Kalvø site (at Norsminde Fjord) might therefore be related to salinity (i.e. greater exposure to higher salinity Kattegat) rather than sediment accumulation.

At Korup Sø, the main indicators of higher salinity over the PW-SG period are increased abundances of *Hinia reticulata* and *Onoba vitrea* between ~3,000-2400 BC, which causes the MI-salinity to peak (~14-18 ‰) between ~2,800-2,660 BC. There is little change in the DI-salinity reconstruction over the PWC/SGC cultural period with the exception of a possible brief rise at ~2,840 BC. The accumulation rates for both Korup Sø core profiles (i.e. Well no. 1 and Well no.2) remains relatively high, despite a drop in Well no. 1 shortly after ~2,800 BC, which together with increasing *Ruppia* beds and salinities below the required 23-25 ‰ for oysters to spawn (Jensen and Spärck, 1934; Yonge, 1960) might have prevented oysters from living in the immediate vicinity of Korup Sø. Oysters are, however present in the Djursland area at this time, being found at the Pitted Ware culture sites Kirial Bro and Kainsbakke (Rasmussen and Richter, 1991; Rasmussen, 1993a). Both these sites are located north east of Korup Sø (Figure 2.5) and closer to the Kattegat entrance, where higher salinity and lower sediment accumulation rates probably existed.

Higher sea level also might be able to explain some of the changes evident in the DI- and MIsalinity records at Norsminde Fjord and Korup Sø (Figure 6.2), but as discussed above, there is no simple relationship between rising sea level and salinity in this period. Furthermore, the

sea level history is generally poorly understood throughout the Funnel Beaker and PWC/SGC periods, but it remains likely that minor fluctuations in sea level continued throughout this period (e.g. Mörner, 1976; Miller, 1982; Berglund *et al.*, 2005; Behre, 2007). To date, no sea level transgression has been specifically linked to the PW-SG period but it has been suggested that sea level might have been at a similar level to the Ertebølle period (e.g. Becker, 1951).

This study demonstrates that environmental change might have provided better conditions for oysters during the PW-SG period in coastal Denmark, but more work needs to be done before strong links between environmental change, oyster abundance and human exploitation can be made. Coastal sites from the PWC/SGC are still quite rare compared to the Mesolithic and Iron Age (e.g. Andersen, 2007) and despite demonstrating signs that marine resources were important to these two cultures (Becker, 1950; Andersen, 1983; Rasmussen and Richter, 1991; Rasmussen, 1993a; Andersen, 2007), the full extent of their marine exploitation remains uncertain. Better dated sea level records and more information (i.e. more spatially distributed records and higher resolution analysis of several proxies), particularly concerning salinity change and the sedimentary regimes of coastal sites, are required to improve our understanding of the marine environment over this period of marine exploitation. There is also a clear need to undertake palaeoenvironmental analyses in the immediate vicinity of PWC/SGC settlements containing marine resource deposits (e.g. shell layers or banks). It is likely that abundant, breeding oyster populations were confined to smaller, specific areas during the PWC/SGC period, than in the late Atlantic (i.e. 5,600-3,900 BC; see above), only existing (and exploited) where very specific environmental conditions were met (e.g. accessible areas of high salinity, more exposed to the open sea and with lower accumulation rates and perhaps only in warmer years).

It is also again likely that no single factor will explain the shell midden patterns observed during the PWC/SGC, and that both environmental and cultural factors are important. For example, increased abundance of oysters in the Kalvø shell midden broadly coincides with a large increase in accumulation rate and charcoal concentrations in Norsminde Fjord (Figure 4.13). As this likely reflects land-use change and subsequently greater input of material from the catchment area, then it is possible that greater pressure was placed on both terrestrial and marine resources at this time, perhaps to support a population increase or a diminishment in terrestrial resource for some reason (e.g. poor growing seasons perhaps due to decreasing

temperatures; Antonsson and Seppä, 2007, death of livestock due to disease). Increased exploitation of marine resources might therefore have been more essential to supplement shortfalls in the terrestrial diet. Marine resources might always have been present, but more heavily exploited when alternative resources were subject to greater pressure. Alternatively, marine resources might have been both more abundant and accessible during the PWC/SGC period due to more favourable environmental conditions.

6.4 Environmental change in the Limfjord region; ~1,700 BC – present

6.4.1 The Bronze Age (1,700-500 BC)

The DI-record from Kilen suggests that high salinity persisted throughout the Bronze Age and that an open connection existed between the Limfjord and the North Sea and Skagerrak throughout the period. Salinity generally increases up until the late Bronze Age (~1,000 BC), at which point there is a brief drop, probably driven by a sea level regression (e.g. regression 2 in southern North Sea beginning at ~1,500 BC and followed by persistently lower sea levels up until ~500 BC; Behre, 2007; see Figure 3.16). Additionally, the sedimentary pigment record suggests that marine primary productivity is relatively low throughout this period. Whilst the archaeological record demonstrates that coastal resources were still exploited in the Bronze Age (e.g. occasional shell layers and fishing tools and artefacts; see section 1.6.4), marine resources appear to be much less important to the subsistence of Bronze Age cultures and societies. It is possible that the higher salinity conditions, greater currents, coarser sediments and generally lower marine productivity confined natural shell beds to specific areas of the Limfjord. The absence of coastal shell middens indicate that shell fish were not collected in such large numbers, unless there was a change in the preparation method during the Bronze Age (e.g. transported elsewhere prior to processing) or greater erosion of coastal sites.

6.4.2 The Iron Age shell midden accumulation period (~500 BC-AD 1050)

The relationship between sea and society in the Limfjord region appears to be anomalous with much of Denmark over the Iron Age period. Numerous shell and kitchen middens are found along the coastlines of eastern Denmark and north eastern Germany (predominately dating to the Roman period and largely consisting of *Mytilus edulis* (blue mussel),

Cerastoderma edulis (cockle) and *Littorina littorea* (periwinkle); Rasmussen *et al.*, 2007) but are absent from the shorelines of the Limfjord (between ~500 BC-AD 1050; see section 1.6.5). The results presented in Chapter 3 (and previous palaeoenvironmental analyses; Kristensen *et al.*, 1995; Christensen *et al.*, 2004), suggest that the Iron Age was a dynamic phase in Limfjord (and north-western Danish) history. Large scale shifts in salinity occur as a result of varying degrees of connection of the Limfjord to the North Sea, predominately driven by changes in sea level (Christiansen *et al.*, 1990; Tanner, 1993; Behre, 2007; Pedersen *et al.*, 2009) and sedimentary processes (i.e. sedimentary accretion and erosion) along the northern and western coasts of the Limfjord (see section 3.11).

6.4.2.1 Pre-Roman (500 BC-AD 0) and Roman period (AD 0-400)

The most important changes over the Iron Age section of the Kilen profile are presented in Figure 6.4, together with regional curves for sea level change (Behre, 2007), and temperature (Antonsson and Seppä, 2007). In summary (for detailed discussion, see section 3.11) Kilen undergoes a major shift from high salinity (30-35 ‰) in the pre-Roman period, to brackish salinities (14-20 ‰) in the Roman period (~AD 50-600). This salinity decline is sharp (occurring over ~50 years) and is accompanied by changes in both primary (e.g. increase in the accumulation rate of several sedimentary pigments) and secondary productivity (e.g. increase in mollusc and ostracod concentrations). The molluscan record also exhibits a shift in faunal composition, with low-medium salinity demanding species (Cerastoderma sp., Mytilus edulis, Hydrobia ventrosa, Hydrobia ulvae, Corbula gibba, Littorina saxatilis and Parvicardium ovale) becoming more abundant. In the plant macrofossil record, the brackish horned pondweed, Zannichellia palustris, becomes regularly present between ~AD 0-600. The sedimentary regime also changes in the late pre-Roman/early Roman period (after ~AD 40) driven largely by a decrease in the minerogenic flux (decreasing between ~AD 0-500), which causes the overall accumulation rate to fall (see section 3.11 for more detailed discussion).

In Figure 6.5, the results from this study are compared with three previous late-Holocene palaeoesalinity studies from the Limfjord (Kristensen *et al.*, 1995; Ellegaard, 2000; Christensen *et al.*, 2004). There is generally good agreement between salinity estimates between these four studies (i.e. including Kilen), but dating problems (e.g. marine reservoir effect) inherent in all previous studies have prevented a precise dating of the various late

Holocene changes (~500 BC - present). For example, previous studies document a large decrease in salinity (i.e. a shift from marine to brackish conditions) towards the end of the pre-Roman period, but the date of this closure has previously only been broadly estimated to sometime between ~500-0 BC (Petersen, 1976, 1980; Christensen *et al.*, 2004). This study suggests that the Limfjord was closed off from the North Sea and Skagerrak at ~0 AD/BC, probably following the late pre-Roman drop in sea level (~150 BC-AD 50; Behre, 2007) and likely increased sedimentary accretion and dune activity under lower energy conditions (Christiansen *et al.*, 1990; Tanner, 1993). In the succeeding Roman period, the DI-salinities (~14-17 ‰; ~AD 50-600) at Kilen are in good agreement with the microfossil estimates (~15 ‰) of Kristensen *et al.*, 1995, and the upper macrofossil estimates of Christensen *et al.*, (2004).

Increased productivity of the marine environment (particularly of marine molluscs) during the Roman period occurs throughout other parts of Denmark, the Baltic Sea and Skagerrak (Andrén *et al.*, 2000; Hebbeln *et al.*, 2006; Rasmussen *et al.*, 2007; Zillén *et al.*, 2008; D.B. Ryves *et al.* unpublished data). The Limfjord, however, appears to differ from other localities, in the sense that lower salinity prevails throughout this period (Figure 6.4). For example, molluscan analyses from Horsens Fjord (east Jutland) and Tempelkrog (southern Isefjord, see Figure X; sites map) document up to two periods of increased molluscan concentration throughout the Roman period, during which higher-salinity demanding species such as *Ostrea edulis* and echinoid remains are present in the record (Rasmussen *et al.*, 2007). In fact, *O. edulis* is found in shell middens as far south as Flensburg Fjord and Eckernförder Bucht (in northern Germany) during the Iron Age (Anger, 1973). Quantitative salinity reconstructions (using diatoms) from two sedimentary cores from the Skagerrak also suggest higher surface salinity in the Roman period, particularly between ~AD 0-280 (Hebbeln *et al.*, 2006), and higher DI-salinity is apparent over this period at both Tempelkrog and Horsens Fjord (D.B. Ryves *et al.* unpublished data).



Figure 6.4 - Summary of key changes occurring over the Iron Age period (~500 BC-AD 1050) at the Limfjord site Kilen. Regional temperature (from Lake Trehörningen; Antonsson and Seppä, 2007) and sea level curves (southern North Sea; Behre, 2007) have also been included. DCA axis 1 reduced dataset excludes the uppermost 28 diatom samples (see section 3.10 for more details). *m NN asl = metres in relation to German ordnance datum above/below present sea level.

	Kristensen <i>et al.</i> , 1995 Christensen <i>et al.</i> , 2004 J. Lewis <i>et al.</i> , this study.											J. Lewis <i>et al.</i> , this study	/
Depth cn	Microsofte Balling	Areco 10851		nacrologistis	Di Di Di Di Di Di Di Di Di Di Di Di Di D	Depth cri	5	Phil Cin	Con	stien salinity	Perseccition of the second	Salini milianos	Cultural Derriod
	¹ 26 ‰	^I ∼26 ‰		8 22-30 ‰		20		2000-		K5 Brackish lake phase	AD 1856/57-	Documented building of road and rail embankment. Switch to brackish water diatom taxa favouring high nutrient status, high measured phosphorus and nitogen concentrations.	Modern
40	5-10 ‰	Ш		⁷ 8-10 ‰ Intermediate fluc-	_	-40	40 -	1800 1600		K4 Low salinity	~AD 1825 ····	Largely based on documentary evidence and Bjørsnholm Bay data. Evidence is poor at Kilen to date, but regular <i>Chara</i> oospores and	
- 60	2	7-10 ‰		6 tuating salinity 5	ity	60	- 60 -	1400 1200 1000		Decreasing salinity ???	AD 1,170 - 940 AD	brackish water diatoms at depth 47 cm (437 cm bpsl) support low salinity. Problems with sediment mixing, dating and low temporal resolution in this section of the profile (see Chapter 3).	Medieval
- 80	5	III		6-7 ‰		80				K3 High salinity	~AD 600	DI-salinity ~33 ‰ at ~AD 800 and high salinity diatoms present in scanned slides. Re-appearance of <i>Bittium reticulatum</i> .	vikilig
-100	30 ‰	inflow of North Sea water - near marine rich		4 22-30 ‰		-100		400		K2		Decrease in DI-salinity down to >20 ‰. Regular Zannichellia palustris in plant macrofossil records and substantial numbers of	Germanic
-120	^{4a} >20 ‰	IVa		³ Gradually increasing salinity		-120	120 -	200		~15-20 /00		lower salinity molluscs are also present in the profile.	Roman
- 140		25 ‰		2	-	- 140	140 -	0-	*******	K1	~0 BC/AD*****		
160	4b	IVb Diminished inflow of North		7-15 ‰	=	160	160					DI salinities >28 ‰ indicating marine connection to both the	Pre-Roman
180	>15 ‰	Sea water - more brackish conditions				-180	180 -	-		~30-35 ‰		north and west of the Limfjord.	
-200	Marine conditions	V Marine conditions	[````````	¹ 30 ‰		-200	200 -	400-					

Figure 6.5 – Comparison of Limfjord palaeosalinity records for the last ~2,500 years (i.e. Kristensen *et al.*, 1995; Ellegaard, 2000; Christensen *et al.*, 2004; this study). An updated chronology is proposed based on the AMS ¹⁴C dates from terrestrial material in this study. The box in the 'Kilen salinity' column covers the period of dating uncertainty in the age-depth model for the Kilen profile (see Chapter 3).

In the central Baltic Sea, salinity estimates (based on $\delta^{13}C_{TOC}$ analyses) suggest a shift towards increasing salinity after ~1,100 BC, which culminates in a severe reduction in the number of freshwater diatom taxa (being replaced by brackish taxa) in the early Roman period (Emeis et al., 2003). In this Gotland basin core, the shift towards increasing salinities is also accompanied by the formation of laminated sediments (beginning at ~2,600 BP; Emeis et al., 2003). Periods of increased productivity, salinity, warmer temperatures (leading to thermal and haline stratification) and higher total organic carbon (Andrén et al., 2000; Zillén et al., 2008) appear to amplify hypoxia, and laminations form due to the absence of large benthic fauna, which cannot live without oxygen. These organisms are responsible for the bioturbation of sediments and in their absence, alternating layers of diatomaceous (formed mainly during the spring bloom) and clay-rich authigenic mud (Zillén et al., 2008) or calcium rich rhodochrosite form (common in the Gotland sediments; Huckreide and Meischener, 1996; Sohlenius et al., 1996). In the period between ~AD 0-1200 laminations appear to form across the Baltic basin (Andrén et al., 2000; Emeis et al., 2003; Zillén et al., 2008), which coincides with periods of increased marine productivity (e.g. high TOC; Figure 1.2 Andrén et al., 2000) and possibly warmer climate associated with the Roman warm period (~AD 100-400; Lamb, 1982; Seppä et al., 2009) and, later, the Medieval warm period (Briffa et al., 1990; Briffa et al., 1992; Broecker, 2001).

The importance of salinity in driving changes in the Baltic Sea in the early Roman period, however, remains uncertain. Whilst it is likely that higher salinity existed in the Roman period (as suggested by Emeis *et al.*, 2003), due to greater exchange of saline water through the Danish straits, not all salinity studies document an increase at this time. For example, Gustafsson and Westman, (2002) employed an oceanographic model to investigate the exchange of water and salt through one of the Danish straits, and their results suggest that salinity briefly dropped ~2,000 years ago and has remained relatively constant since. It is difficult, however, to perceive how oysters were able to extend in the northern reaches of the Baltic Sea (e.g. Flensburg Fjord and Eckernförder Bucht, where surface salinity is today ~15 ‰; see above, Figure 2.1) without greater exchange of saline water through the Danish straits and Øresund, and hence higher salinity. Furthermore, evidence from south-eastern Sweden suggests an early Roman sea level transgression (Miller, 1982; Miller and Hedin, 1988), increasing water exchange with the North Sea, and subsequently higher salinity within the Baltic Sea over the late Holocene.

The spatial extent of evidence for high marine productivity across Denmark and the Baltic Sea clearly demonstrates that the Iron Age (and particularly the early Roman period) was a productive and diverse period. It appears, therefore, that the large number of shell middens dated to the Iron Age (but concentrated in the Roman period; Poulsen, 1978) is no coincidence and that humans responded to the richer marine biotope by heavily exploiting its resources. Based on the size and cubic content of some of the Iron Age middens located in Denmark and Germany, Andersen, (2007) has suggested marine resources (particularly mussels) might have been exploited almost industrially, perhaps for widespread distribution.

The absence of Iron Age shell middens right on the shoreline of the Limfjord is difficult to explain. Molluscan records from the Limfjord (e.g. Figure 3.12) all document that *Mytilus edulis* (generally the most common molluscan species present in Iron Age middens (Poulsen, 1978; Andersen, 2007) was abundant during the Roman period. The high abundance and widespread spatial distribution of mussels during the Iron Age, might have meant that site selection was more heavily determined by additional resources available to these cultures. If mussels could be farmed at any location then, primary importance might have been placed on alternative food sources such as oysters or particular fish species, potentially absent from the Limfjord due to lower salinity waters (and/or unsuitable substrate perhaps due to the regular presence of macrophytes such as *Zannichellia palustris*) in the Roman period. Alternatively mussels might have been collected more sporadically (perhaps as a dietary supplement), from different localities, preventing the temporal accumulation of shells have been transported to more inland settlement sites prior to shelling (Poulsen, 1978; Ringtved, 1992).

6.4.2.2 Germanic (AD 400-750) and Viking (AD 750-1050) periods

The low salinity and high productivity conditions at Kilen extend into the Germanic period, with higher salinity organisms re-appearing around ~AD 600 (e.g. *Bittium reticulatum*). This date broadly agrees with Kristensen *et al.*, 1995, after which they suggest conditions in the Limfjord follow a gradual shift back towards higher salinity, associated with steadily rising sea level (e.g. Christiansen *et al.*, 1990; Tanner, 1993). The rapidity of the shift is unknown for the Kilen profile as no diatom samples have been analysed between ~AD 600 (still low salinity; ~15 ‰) and ~AD 800 (~33 ‰). Unlike, the anomalous Roman period, high salinity in the Limfjord during the Viking period is consistent with high surface salinity stands in the Skagerrak (Hebbeln *et al.*, 2006) and possibly in the Baltic Sea (Emeis *et al.*, 2003). Evidence for a high sea level stand during the Viking period is also apparent in a number of spatially

distributed sea level records from the North Sea and, into the Baltic region (Mörner, 1976; Miller, 1982; Miller and Hedin, 1988; Christiansen *et al.*, 1990; Tanner, 1993; Behre, 2007). At least for part of the Viking period, both the north and west entrances were probably open, enabling salinity to reach near fully marine values (Kristensen *et al.*, 1995; Christensen *et al.*, 2004; this study, Figure 6.4). The exact dates for the various openings and closing of these entrances throughout the Germanic, Viking and Medieval period, however, remain open to debate.

The fall in sedimentary pigment accumulation rates (and concentrations) in the Kilen profile at ~AD 800 might mark the beginning of a maximum salinity phase in the Viking period (e.g. DI-salinity of 33 ‰). The intensified marine conditions (e.g. greater energy and coarser sediments) possibly caused pigment preservation conditions to deteriorate, as indicated by the increase in the pheophytin-a/chlorophyll *a* (degradation) ratio. At ~AD 900, productivity (and surface salinity) within the Skagerrak also appears to decrease, which has been attributed to increased Baltic outflow of low salinity water (Hebbeln *et al.*, 2006). This decrease in productivity, however, appears to be confined to the Limfjord-Skagerrak region, as the Baltic Sea records indicate higher productivity (Andrén *et al.*, 2000; Zillén *et al.*, 2008), likely associated with increased sea level (Miller, 1982), greater water exchange and warmer temperatures associated with the Medieval warm period (~AD 750-1200; Briffa *et al.*, 1990; Briffa *et al.*, 1992; Seppä *et al.*, 2009).

In terms of commercial value, the archaeofaunal evidence suggests that the Limfjord remained an important fishery throughout the Viking Age with flounder, garfish and whitefish the most abundant catches (Enghoff, 1999). Plaice and eel were also important, probably during the maximum salinity phase (Enghoff, 1999; Holm and Bager, 2002). High resolution diatom analyses are required to fully examine changes in salinity and the Limfjord environment over the last ~2,000 years. This remains possible at Kilen, providing the late-Holocene dating issues can be overcome, but additional analyses from other Limfjord sites would also offer information regarding spatial variability and regional differences. Further research in this area is essential for making more confident linkages between sea and society, and understanding why shell middens are absent from the Limfjord throughout the Iron Age.

6.4.3 The Limfjord during the Historical Period (AD 1050-present)

Despite the absence of shell middens in the post Iron Age period, documentary evidence suggests that humans continued to rely on marine resources from the Limfjord throughout

historical times (Enghoff, 1999; Holm and Bager, 2002; Poulsen et al., 2007). For example, the important plaice and eel fisheries of the Viking period (Enghoff, 1999) were replaced by a rich herring fishery after ~AD 1200 (Holm and Bager, 2002; Poulsen et al., 2007), following the closure of the Limfjord (see section 3.11). These lower salinity waters provided ideal feeding grounds for herring, but the prosperity of this fishery remained very closely connected with environmental conditions, most notably salinity and exposure to the North Sea. On several occasions this fishery was disrupted, particularly in the 1600s when storms breached through the western sand deposits (e.g. AD 1624), enabling highly saline marine water from the North Sea to enter and salinity to rise (Holm and Bager, 2002; Ryves et al., 2004). The fishery was able to recover from these temporary openings several times, prior to its final collapse shortly after ~AD 1825, when marine water broke through the Agger Tange sand dunes and created a permanent marine connection with the North Sea, kept open in modern times by active dredging of the channel (Holm and Bager, 2002). Unfortunately, the poor dating in this section of the Kilen profile (i.e. ~AD 1200-1920), prevents detailed examination of interactions between sea and society throughout this brackish phase of the Limfjord's history. Furthermore, it is unlikely that short term storm events and shifts in salinity would be discernible from the record due to the poor resolution and mixing of sediments in this section (though until full diatom counts are made this remains uncertain).

6.4.4 The limnic phase of Kilen (AD 1856-present)

The uppermost section of the Kilen profile represents a contemporary problem in many Danish lake/coastal sites (Conley *et al.*, 2000; Conley *et al.*, 2002b; Clarke *et al.*, 2003; Clarke *et al.*, 2006; Ellegaard *et al.*, 2006), where anthropogenic activities (e.g. agriculture, sewage treatment, industrial effluents) are causing nutrient enrichment of aquatic environments, which subsequently affects the ecological status of these systems (see Chapter 3). Unfortunately, Kilen does not offer the potential to explore baseline nutrient conditions (before the large-scale anthropogenic disturbances associated with the industrial revolution) due to its history as a marine environment, prior to the building of the road and rail embankment in AD 1856. The establishment of baseline conditions has been important issue in many nutrient-based transfer function studies in order to be able to understand the natural conditions of a lake/fjord, and to assess fully the extent of human impacts (Andrén *et al.*, 2007a). Furthermore, baseline conditions must be treated with caution, particularly in Denmark, where the landscape has been culturally impacted since the Neolithic period. Kilen does, however, provide a good example of how rapidly humans can influence a system after

isolation/reclamation from the marine environment. There is also potential for future comparison of the diatom stratigraphy with contemporary monitoring data available for Kilen between AD 1972-1997, sampled as part of the Danish Nationwide monitoring project (Ringkjøbing Amtskommune, 1988; Kronvang *et al.*, 1993). This will be extremely useful for testing the transfer function inferences (e.g. Figure 3.20) and for a better understanding the interactions between recent diatom communities, key environmental variables and potentially help manage heavily human impacted systems such as Kilen.

6.5 Conclusions

- The diatom-salinity model used in this study (WAPLS component-3 model (r²_{boot} = 0.923 and RMSEP = 0.36 square root salinity units) is deemed to have produced reasonably accurate reconstructions of palaeosalinity (qualitatively confirmed by ecological knowledge) at each site over the study period. However, there are some circumstances when the diatom-salinity model appears to perform poorly, such as at the high end of the salinity gradient, where salinity appears to be overestimated (i.e. DI-salinity inferences at Kilen >35 ‰ ~250-0 BC). The salinity-model also performs poorly in circumstances where other environmental variables (particularly high nutrient conditions) appear to exert pressure upon the diatom assemblage (e.g. poor agreement between DI-salinity estimates and measured salinities in the uppermost section of the Kilen profile under eutrophic conditions).
- Diatom preservation was generally good at each site (generally $\mathbf{F} = > 0.4$ for the majority of the three core sequences), though some exceptions to this were apparent (e.g. lowermost section of the Norsminde Fjord diatom sequence, F= < 0.3 between ~6,900-6,600 BC). Whilst high salinity probably affected valve preservation at certain times in the record (e.g. lower $\mathbf{F} = 0.2$ -0.4 at Kilen during the pre-Roman period, under high salinity condition, DI-salinity >28 ‰), overall, the salinity-dissolution relationship was weak, with increased dissolution appearing to be more strongly affected by factors such as depth, accumulation rate, sediment re-working, transport and deposition.
- Korup Sø was likely first inundated by the sea sometime between 8,000-7,500 BC, but probably remained a limnic system (subject to the occasional breaches) up until the

second half of the Boreal chronozone (~7,500-7,200 BC). After a permanent connection to the Kolindsund was eventually established (i.e. between ~7,500-7,200 BC) salinity increased rapidly in the fjord due to rising sea levels, reaching >15 ‰ by ~6,700 BC. Norsminde Fjord was inundated slightly later than Korup Sø, with a marine connection being established between ~6,900-6,800 BC, also following a limnic or fluvial phase, and by ~6,600 BC was again under high salinity conditions (DI-salinity >20 ‰). The date at which Kilen was inundated by the sea remains uncertain but a recent AMS ¹⁴C date (at which time marine diatoms and molluscan are present) suggests that the Limfjord was already under marine conditions by ~6,900 BC, likely predating earlier estimates (i.e. 7,000-6,000 BC).

- The earliest Danish coastal shell middens are dated to the late Kongemose period (i.e. ~5,600 BC), thereby coinciding with an abrupt (and potentially global) rise in sea level. At this time there is also evidence for increased salinity (initially), decreased sedimentary accumulation rates (at Norsminde Fjord), and an increase in molluscan and foraminiferal concentrations. It is likely that this rapid rise in sea level (between ~5,600-5,000 BP) was the main driver behind these (environmental and ecological) changes and potentially triggered the accumulation of shell middens (i.e. increased exploitation of marine resources) in the late Kongemose/Ertebølle period. Increasing temperatures was most likely an important factor (e.g. longer spawning seasons, reducing winter freezing and thawing) and at this time Scandinavia entered the Holocene thermal optimum (5,550-3,550 BC).
- Maximum salinity at both Korup Sø and Norsminde appears to have occurred somewhere between ~5,700-4,900 BC (rising after ~6,000 BC), but estimates differ slightly between proxies. At Korup Sø, highest DI-salinities (23-27 ‰) occur between ~5,730-5,520 BC, thereby predating the maximum MI-salinities (~25 ‰ between ~5,340-5,250 BC). At Norsminde Fjord, highest DI-salinities occur before (~5,400-5,160 BC under the reduced model) the highest-salinity demanding foraminiferal species reach their maximum abundances (~5,160-4,900 BC). This suggests that salinity and sea level change are not linearly related, and that other factors might be important in driving salinity change at different times in the record, (e.g. freshwater inputs from the catchment, reduced marine connection/exposure due to sediment accretion). Furthermore, biological communities (in which the salinity inferences are

based on, e.g. diatoms, molluscs and foraminifera) generally respond to a wide range environmental variables (e.g. changes in nutrients, sediment, depth), some of which might exert greater (or additional) pressure on these organisms at different times, thereby potentially skewing salinity estimates (as demonstrated at Kilen). In this study, the high abundances of *Fragilaria* s.l. at both Norsminde Fjord and Korup Sø (probably indicating general instability within these fjord systems) often made it more difficult to isolate a true salinity signal.

- Contrary to the salinity hypothesis for the oyster decline, there is no convincing evidence from the sites analysed in this study, for a regional shift to lower salinity in the Neolithic period (including a widespread decline at the Mesolithic-Neolithic transition). There is, however, plenty of evidence suggesting that both the marine and terrestrial environmental changed over the late Mesolithic/early Neolithic period and that several environmental factors, probably contributed to the oyster decline. Here it is argued that both regional sea level and climate changes were important drivers, but that the response to these forcing mechanism differed between sites, due to being heavily affected by more localised characteristics (e.g. vegetation cover and change (possibly both climatic and human forced, site configuration, exposure etc).
- The oyster decline in the shell middens was likely to be driven by a complex series of events including changes in the sedimentary regime (i.e. increased sedimentary accumulation rates evident at Norsminde Fjord and Korup Sø) and short term stresses associated with salinity (i.e. evidence for a short term decline at Kilen and fluctuating DI-salinities in both the Norsminde Fjord and Korup Sø record), productivity and temperature events (i.e. decreasing temperatures and less stable climate after the Holocene thermal maximum). Higher sedimentary accumulation rates (associated with lower sea level and catchment vegetation changes) might have pushed oysters out of the inner more human accessible parts of the fjord. Lower temperatures and fluctuating salinities might have reduced/ prevented spawning in some years and even caused widespread mortality in extreme winters, suppressing oyster populations. Subsequent human exploitation would have placed greater pressure on oyster populations, reducing their dependability as a resource and therefore culminating in their decline in the middens, as humans shifted towards a more terrestrial-based diet.

In terms of the introduction of agriculture it is likely that the oyster decline formed part of a complex transition driven by a mixture of environmental and cultural factors.

- There is some evidence for higher salinity over the Pitted Ware and Single Grave cultural period (~2,800-2,400 BC) from the records presented here. At Kilen, DI-salinity increases throughout the course of the Pitted Ware/Single Grave cultural period (24-27 ‰). At Norsminde Fjord, higher DI-salinity exists between ~3,070-2,900 BC and ~2,700-2,570 BC (23-25 ‰ under the full model) and at Korup Sø, increased abundances of *Hinia reticulata* and *Onoba vitrea* are present between ~3,000-2400 BC. Higher salinity (and perhaps changing sedimentary conditions) might have increased oyster abundances and/or enabled them to re-colonise the inner more accessible parts of the fjord at this time. However, before a positive connection between environmental change, oyster abundance and human exploitation can be made more work needs to be done (including better dated sea level records covering this period and more information concerning salinity change and the sedimentary regime at a variety of coastal sites).
- The Iron Age (~500 BC-AD 1050) is a dynamic phase in Limfjord history characterised by large scale shifts in salinity and productivity, resulting from varying degrees of connection with the North Sea (to the west) and the Skagerrak (to the north), and heavily affected by sea level and sedimentary processes (i.e. sedimentary accretion and erosion). The large number and size of shell middens (generally dominated by Mytilus edulis) dating to the early Iron Age (particularly ~300 BC-AD 200) clearly demonstrates that marine resources were important to cultures and societies living in Denmark at this time. Near fully marine salinities in the late pre-Roman period (i.e. >30 ‰) suggest that the Limfjord was heavily exposed to the North Sea and Skagerrak, likely being connected in both the west and north in this period. This is succeeded by a rapid decline in salinity in the early Roman period, driven by a sea level regression in the North Sea (~150 BC-AD 50) and subsequent accretion of sediments, blocking off the Limfjord entrances and isolating it from the North Sea and Skagerrak. Consistent with other Danish and Baltic studies, increased marine productivity is apparent (particularly macrophytes and molluscs, including *Mytilus edulis*) at Kilen over the Roman period, but the contrasting point is that this occurs under lower salinity conditions in the Limfjord region (rather than high salinity

in these other regions). The reasons for the lack of shell middens being present along the shoreline of the Limfjord remains uncertain. Several possible reasons have been suggested here, (e.g. primary importance placed on alternative food sources absent from the Limfjord (e.g. oysters), sporadic collection of mussels from different localities, preventing the temporal accumulation of shell middens) but these remain speculative.

Modelling species-environment relationships (using multivariate techniques) has demonstrated that a wide range of terrestrial and marine changes can influence the diatom distributions present in Danish coastal systems. The most important variables driving changes in the diatom datasets also appear to change over time as these complex coastal systems develop both naturally, and in response to humans altering of the landscape. At Norsminde Fjord, climate and climate-related processes (e.g. indicated by temperature), together with predictors for marine environmental change (e.g. macrophyte bedding, catchment inwash and erosion probably associated with sea level rise) tend to be most important at the bottom of the profile ($\sim 6.360-3.430$ BC). Catchment processes (as indicated by charcoal, minerogenic fluxes and predictors of vegetation change) are the most significant variables in the middle part of the profile (5,400-2,570 BC), and a series of catchment (e.g. minerogenic input), climatic (decreasing temperatures) and human related processes (forest clearances and expansion of agricultural lands) dominate in the upper section (4,360-2,020 BC). Similarly, at Korup Sø, in the lower section (between ~7,330-3,400 BC) predictors for marine environmental change (e.g. sea level, sedimentation rate, changing mollusc faunas) are generally more common (and tend to explain large amounts of the variation, up to 49 %), whereas catchment processes/vegetation changes (probably heavily driven by human disturbance after ~3,900 BC) are more important in the upper section of the record (~4,870-2,380 BC, generally explaining 8-20 % of the variation in the diatom dataset).

Chapter 7. References

- Aaby, B. (1983) Forest development, soil genesis and human activity illustrated by pollen and hypha analysis of two neighbouring podzols in Draved forest, Denmark. *Danmarks Geologiske Undersøgelse*, II, **114**, 1-114.
- Aaby, B. (1986). Trees as anthropogenic indicators in regional pollen diagrams from eastern Denmark. In Anthropogenic indicators in pollen diagrams (ed K.-E. Behre), pp. 73-93. Balkema, Rotterdam.
- Aaby, B. (1988). The cultural landscape as reflected in percentage and influx pollen diagrams from two Danish ombrotrophic mires. In *The cultural landscape - past, present and future* (eds H.H. Birks, H.J.B. Birks, P.E. Kaland & D. Moe), pp. 209-228. Cambridge University Press, Cambridge.
- Aitchison, J. (1986) *The statistical analysis of compositional data*. Chapman and Hall, London, pp. 416.
- Albrethsen, S.-E. & Brinch Petersen, E. (1977) Excavation of a Mesolithic Cemetery at Vedbæk, Denmark. *Acta Archaeologica*, **47**, 1-28.
- Alley, R.B., Mayewski, P.A., Sowers, T., Stuiver, M., Taylor, K.C., & Clark, P.U. (1997) Holocene climatic instability: A prominent, widespread event 8200 yr ago. *Geology*, 25, 483-486.
- Alve, E. & Murray, J.W. (1995) Benthic foraminiferal distribution and abundance changes in Skagerrak surface sediments: 1937 (Hoglund) and 1992/1993 data compared. *Marine Micropaleontology*, 25, 269-288.
- Alve, E. & Murray, J.W. (1999) Marginal marine environments of the Skagerrak and Kattegat: a baseline study of living (stained) benthic foraminiferal ecology. *Palaeogeography Palaeoclimatology Palaeoecology*, **146**, 171-193.
- Amsinck, S.L., Jeppesen, E., & Ryves, D. (2003) Cladoceran stratigraphy in two shallow brackish lakes with special reference to changes in salinity, macrophyte abundance and fish predation. *Journal of Paleolimnology*, 29, 495-507.
- Andersen, B. (1992) En undersøgelse af holocæne foraminiferfaunaer fra Vilsund i det vestlige Limfjordsomraade. Dansk Geologisk Forening, Arsskrift for 1990-1991, 47-52.
- Andersen, B.G., Mangerud, J., Sørensen, R., Reite, A., Sveian, H., Thoresen, M., & Bergström, B. (1995) Younger Dryas ice-marginal deposits in Norway. *Quaternary International*, 28, 147-169.
- Andersen, G.J., Heinemeier, J., Nielsen, H.L., Rud, N., Thomsen, M.S., Johnsen, S., Sveinbjörnsdóttir, A.E., & Hjartarson, A. (1989) AMS C-14 dating on the Fossvogur Sediments, Iceland. *Radiocarbon*, **31**, 592-600.
- Andersen, J.H. & Conley, D.J. (2009) Eutrophication in coastal marine ecosystems: towards better understanding and management strategies. *Hydrobiologia*, **629**, 1-4.
- Andersen, K. (1985) Frihavnen den første Kongemoseboplads. *Nationalmuseets Arbejdsmark*, 42-47.
- Andersen, S.H. (1970a) Brovst. En kystboplads fra ældre stenalder. Kuml, 1969, 67-90.
- Andersen, S.H. (1980) Tybrind Vig. Foreløbig meddelelse om en undersøisk stenalderboplads ved Lillebælt. *Antikvariske Studier*, **4**, 7-22.
- Andersen, S.H. (1983) Kalvø. A coastal site from the single grave culture. *Journal of Danish Archaeology*, **2**, 71-80.
- Andersen, S.H. (1987). Tybrind Vig: A sub-merged Ertebølle settlement in Denmark. In European Wetlands in Prehistory (eds J.M. Coles & J.L. Lawson), pp. 253-280. Clarendon Press, Oxford.

Andersen, S.H. (1989) Norsminde. A "Køkkenmødding" with Late Mesolithic and Early Neolithic Occupation. *Journal of Danish Archaeology*, **8**, 13-40.

Andersen, S.H. (1995a). Coastal adaption and marine exploitation in Late Mesolithic
 Denmark - with special emphasis on the Limfjord Region. In *Man and Sea in the Mesolithic. Coastal settlement above and below present sea-level* (ed A. Fischer), pp. 41-66. Oxbow Books, Oxford.

Andersen, S.H. (2000) 'Køkkenmøddinger' (shell middens) in Denmark: a survey. *Proceedings of the Prehistoric Society*, **66**, 361-384.

Andersen, S.H. (2005) Køkkenmøddinderne ved Krabbesholm. *Nationalmuseets Arbejdmark* 2005, 151-171.

Andersen, S.H. (2007). Shell middens ("Køkkenmøddinger") in Danish Prehistory as a reflection of the marine environment. In *Shell Middens in Atlantic Europe* (eds N. Milner, O.E. Craig & G.N. Bailey), pp. 31-45. Oxbow Books, Oxford.

Andersen, S.H. (2008a) The Mesolithic-Neolithic transition in Western Denmark seen from a kitchen midden perspective. A survey. *Analecta Praehistorica Leidensia*, **40**, 67-75.

- Andersen, S.H. (2008b) A report on recent excavations at the shell middens of Havnø in Denmark. *Mesolithic Miscellany*, **19**, 2-5.
- Andersen, S.H. & Johansen, E. (1986) Ertebølle revisited. *Journal of Danish Archaeology*, **5**, 31-61.
- Andersen, S.H. & Malmros, C. (1966) Norslund. En kystboplads frældre stenalder. *Kuml*, **1965**, 35-114.
- Andersen, S.H. & Rasmussen, K.L. (1991) Bjørnsholm. A stratified Køkkenmødding on the Central Limfjord, North Jutland. *Journal of Danish Archaeology*, **10**, 59-65.
- Andersen, S.T. (1970b) The relative pollen productivity and pollen representation of North European trees, and correction factors for tree pollen spectra. *Geological Survey of Denmark. II*, 96, 99.
- Andersen, S.T. (1978) Local and regional vegetational development in eastern Denmark in the Holocene. *Danmarks Geologiske Undersøgelse Årbog*, **1975**, 5-27.
- Andersen, S.T. (1979) Identification of wild grass and cereal pollen. *Danmarks Geologiske Undersøgelse Årbog*, **1978**, 69-92.
- Andersen, S.T. (1995b) History of vegetation and agriculture at Huse Mose, Thy, Northwest Denmark, since the Ice Age. *Journal of Danish Archaeology*, **118**, 57-79.
- Andersen, S.T. & Rasmussen, K.L. (1993) Radiocarbon wiggle-dating of elm declines in northwest Denmark and their significance. *Vegetation History and Archaeobotany*, 2, 125-135.
- Anderson, N.J. (2000) Miniview: Diatoms, temperature and climatic change. *European Journal of Phycology*, **35**, 307-314.
- Anderson, N.J., Brodersen, K.P., Ryves, D.B., McGowan, S., Johansson, L.S., Jeppesen, E., & Leng, M.J. (2008) Climate versus in-lake processes as controls on the development of community structure in a low-arctic lake (South-West greenland). *Ecosystems*, **11**, 307-324.
- Anderson, N.J. & Odgaard, B.V. (1994) Recent palaeolimnology of three shallow Danish lakes. *Hydrobiologia*, 256/257, 411-422.
- Andersson, A., Hajdu, P., Haecky, J., Kuparinen, J., & Wikner, J. (1996) Succession and growth limitation of phytoplankton in the Gulf of Bothnian (Baltic Sea). *Marine Biology*, **126**, 791-801.
- Andersson, H.C. (2002) Influence of long-term regional and large-scale atmospheric circulation on the Baltic sea level. *Tellus*, **54A**, 78-88.
- Andrén, E., Andrén, T., & Kunzendorf, H. (2000) Holocene history of the Baltic Sea as a background for assessing records of human impact in the sediments of the Gotland Basin. *The Holocene*, **10**, 687-702.

- Andrén, E., Clarke, A.L., Telford, R.J., Wesckström, K., Vilbaste, S., Aigars, J., Conley, D., Johnsen, T., Juggins, S., & Korhola, A. (2007a) Defining reference conditions for coastal areas in the Baltic Sea. *TemaNord*, **583**, 81.
- Andrén, T., Andrén, E., Berglund, B.E., & Yu, S.Y. (2007b) New insights on the Yoldia Sea low stand in the Blekinge archipelago, southern Baltic Sea. *Gff*, **129**, 277-285.
- Anger, K. (1973) Untersuchungen an eisenzeitlichen Muschelhaufen an der Flensburger Förde. *Offa*, **30**, 55-59.
- Anselmetti, F.S., Hoddell, D.A., Ariztegui, D., Brenner, M., & Rosenmeier, M.F. (2007) Quantification of soil erosion rates related to ancient Maya deforestation. *Geology*, 35, 915-918.
- Antonsson, K. & Seppä, H. (2007) Holocene temperatures in Bohuslän, southwest Sweden: a quantitative reconstruction from fossil pollen data. *Boreas*, **36**, 400-410.
- Appleby, P.G. (2001). Chronostratigraphic techniques in recent sediments. In *Tracking environmental change using lake sediments* (eds W.M. Last & J.P. Smol), Vol. 1: Basin analysis, coring and chronological techniques, pp. 171-203. Kluwer Academic Publishers.
- Ascough, P., Cook, G., & Dugmore, A. (2005) Methodological approaches to determining the marine radiocarbon reservoir effect. *Progress in Physical Geography*, **29**, 532-541.
- Attrillf, M.J. & Rundle, S.D. (2002) Ecotone or Ecocline: Ecological Boundaries in Estuaries. *Estuarine and Coastal Shelf Science*, **55**, 929-936
- BACC (2006). Assessment of Climate Change for the Baltic Sea Basin The BACC Project, Rep. No. 1681-6471, pp. 26 pp.
- Bailey, G. & Milner, N. (2008). Molluscan archives from European prehistory. In *Early Human Impacts on Megamolluscs* (eds A. Antczack & R. Cipriani), Vol. 165, pp. 111-134. BAR International Series.
- Bailey, G.N. (1975) The Role of Shell Middens in Prehistoric Economies. Ph.D Thesis, University of Cambridge, Cambridge.
- Bailey, G.N. & Milner, N. (submitted). The marine molluscs from the Mesolithic and Neolithic deposits of the Norsminde shell midden. In *Stone Age settlement in the coastal fjord of Norsminde, Jutland, Denmark* (ed S.H. Andersen).
- Bard, E., Arnold, M., & Duplessy, J.-C. (1991) Reconciling the sea-level record of the last deglaciation with δ18O spectra from deep sea cores. *Quaternary Proceedings*, 1, 67-73.
- Bard, E., Arnold, M., Duprat, J., & Duplessy, J.-C. (1987) Reconstruction of the last deglaciation: deconvolved records of δ18O spectra from deep sea cores, micropalaeontological variations and accelerator mass spectrometric 14C dating. *Climate Dynamics*, **1**, 102-112.
- Bard, E., Hamelin, B., Arnold, M., Montaggioni, L., Cabioch, G., Faure, G., & Rougerie, F. (1996) Deglacial sea level record from Tahiti corals and the timing of global meltwater discharge. *Nature*, **382**, 242-244.
- Barnes, R.S.K. (1974) Estuarine Biology. Edward Arnold, London
- Bartlein, P.J. & Whitlock, C. (1993). Paleoclimatic interpretation of the Elk Lake pollen record. In Elk Lake, Minnesota: Evidence for rapid climate change in the northcentral United States. Geological Society of America Special Paper (eds J.P. Bradbury & W.E. Dean), Vol. 276, pp. 275-293.
- Bateman, C. (1981) An alternative classification and profile type map applied to the Holocene deposits of the Belgian coastal plain. *Bulletin Société Belge de Géologie*, **90**, 257-180.
- Battarbee, R.W. (1986). Diatom analysis. In Handbook of Holocene Palaeoecology and palaeohydrology (ed B.E. Berglund), pp. 527-570. John Wiley & Sons, Chichester, New York.
- Battarbee, R.W., Carvalho, L., Jones, V.J., Flower, R.J., Cameron, N.G., Bennion, H., & Juggins, S. (2001). Diatoms. In *Tracking Environmental Change Using Lake*

Sediments (eds J.P. Smol, H.J.B. Birks & W.M. Last), Vol. Volume 3: Terrestrial, Algal, and Siliceous Indicators, pp. 155-202. Kluwer Academic Publishers, Dordrecht.

- Battarbee, R.W. & Kneen, M.J. (1982) The Use of Electronically Counted Microspheres in Absolute Diatom Analysis. *Limnology and Oceanography*, **27**, 184-188.
- Bayley, S.E., Schindler, D.W., Beaty, K.G., Parker, B.R., & Stainton, M.P. (1992) Effects of multiple fires on nutrient yields from streams draining boreal forest and fen watersheds: nitrogen and phosphorus. *Canadian Journal of Fisheries and Aquatic Sciences*, 49, 548-596.
- Becker, C.J. (1950). Den grubekeramiske kultur i Danmark (The Pitted-Ware Culture in Denmark). In *Aarbøger for nordisk Oldkyndighed og Historie*, pp. 153-274, Copenhagen.
- Becker, C.J. (1951) Den grubekeramiske kultur i Danmark (The Pitted-Ware Culture in Denmark). *Aarbøger for nordisk Oldkyndighed og Historie*, **1950**, 153-274.
- Behre, K.-E. (1981) The interpretation of anthropogenic indicators in pollen diagrams. *Pollen et spores*, **23**, 225-245.
- Behre, K.-E. (2007) A new Holocene sea-level curve for the southern North Sea. *Boreas*, **36**, 82-102.
- Bengtsson, L. & Enell, M. (1986). Chemical Anlaysis. In Handbook of Holocene Palaeoecology and Palaeohydrology (ed B.E. Berglund), pp. 423-451. John Wiley & Sons, Chichester, New York.
- Bennett, K.D. (1996) Determination of the number of zones in a biostratigraphical sequence. *New Phytologist*, **132**, 155-170.
- Bennett, K.D. (2003-2009) psimpoll http://www.chrono.qub.ac.uk/psimpoll/psimpoll.html.
- Bennike, O., Jensen, J.B., Lemke, W., Kuijpers, A., & Lomholt, S. (2004) Late- and postglacial history of the Great Belt, Denmark. *Boreas*, **33**, 18-33.
- Berglund, B.E., Sandgren, P., Barnekow, L., Hannon, G., Jiang, H., Skog, G., & Yu, S.Y. (2005) Early Holocene history of the Baltic Sea, as reflected in coastal sediments in Blekinge, southeastern Sweden. *Quaternary International*, **130**, 111-139.
- Bianchi, T.S., Engelhaupt, E., Westman, P., Andrén, T., Rolff, C., & Elmgren, R. (2000) Cyanobacterial blooms in the Baltic Sea: Natural or human-induced? *Limnology and Oceanography*, 45, 716-726.
- Birks, C.J.A. & Koç, N. (2002) A high-resolution diatom record of late-Quaternary seasurface temperatures and oceanographic conditions from the eastern Norwegian Sea. *Boreas*, **31**, 323-344.
- Birks, H.H. & Birks, H.J.B. (2006) Multi-proxy studies in palaeolimnology. *Vegetation History and Archaeobotany*, **15**, 235–251.
- Birks, H.J.B. (1995). Quantitative palaeoenvironmental reconstructions. In *Statistical modelling of Quaternary science data*. *Technical Guide 5* (eds D. Maddy & J.S. Brew), pp. 161-254. Quaternary Research Association, Cambridge.
- Birks, H.J.B. (1998) D.G. Frey & E.S. Deevey Review #1: Numerical tools in palaeolimnology - progress, potentialities, and problems. *Journal of Paleolimnology*, 20, 307-332.
- Birks, H.J.B. & Birks, H.H. (1980) *Quaternary Palaeoecology*. Edward Arnold, , London, UK, pp. 289.
- Björck, S. (1995) A Review of the History of the Baltic Sea. 13.0-8.0 ka BP. *Quaternary International*, **27**, 19-40.
- Björck, S. (2008) The late Quaternary development of the Baltic Sea. <u>www.geol.lu.se/personal/seb/The%20late%20Quaternary%20development%20of%20t</u> <u>he%20Baltic%20SeaKompendium.pdf</u>.
- Björklund, K.R., Bjørnstad, H., Dale, B., Erlenkeuser, H., Henningsmoen, K.E., Høeg, H.I., Johnsen, K., Manum, S.B., Mikkelsen, N., Nagy, J., Pederstad, K., Qvale, G., Rosenqvist, I.T., Salbu, B., Schoenharting, G., Stabell, B., Thiede, J., Throndsen, I.,

Wassman, P., & Werner, F. (1985) Evolution of the upper Quaternary depositional environment in the Skagerrak: a synthesis. *Norsk Geologisk Tidsskrift*, **65**, 139-149.

- Blaauw, M. & Christen, J.A. (2005) Radiocarbon peat chronologies and environmental change. *Applied Statistics*, **54**, 805-816.
- Blaber, S.J.M., Albaret, J.-J., Chong, C.V., Cyrus, D.P., Day, J.W., Elliot, M., Fonseca, D., Hoss, J., Orensanz, J., Potter, I.C., & Silvert, W. (2000) Effects on fishing on the structure of estuarine and nearshore ecosystem. *ICES Journal of Marine Science*, 57, 590-602.
- Blanchon, P., Jones, B., & Ford, D.C. (2002) Discovery of a submerged relic reef and shoreline off Grand Cayman: further support for an early Holocene jump in sea level. *Sediment Geology*, **147**, 253-270.
- Blanchon, P. & Shaw, J. (1995) Reef drowning during the last deglaciation: Evidence for catastrophic sea-level rise and ice-sheet collapse. *Geology*, **23**, 4-8.
- Bond, G., Showers, W., Cheseby, M., Lotti, R., Almasi, P., deMenocal, P., Priore, P., Cullen, H., Hajdas, I., & Bonani, G. (1997) A pervasive millennial-scale cycle in North Atlantic Holocene and glacial climates. *Science*, **278**, 1257-1266.
- Bonsdorff, E., Aarnio, K., & Sandberg, E. (1991) Temporal and spatial variability of zoobenthic communities in archipelago waters of the northern Baltic Sea -consequenes of eutrophication. *Internationale Revue der gesamten Hydrobiologie und Hydrographie*, **76**, 433-449.
- Borcard, D., Legendre, P., & Drapeau, P. (1992) Partialling out the spatial component of ecological variation. *Ecology*, **73**, 1045-1055.
- Borgendahl, J. & Westman, P. (2007) Cyanobacteria as a trigger for increased primary productivity during sapropel formation in the Baltic Sea A study of the Ancylus/Litorina transition. *Journal of Paleolimnology*, **38**, 1-12.
- Bormann, F.H., Likens, G.E., Siccama, T.G., Pierce, R.S., & Eaton, J.S. (1974) The export of nutrients and recovery of stable conditions following deforestation at Hubbard Brook. *Ecological Monographs*, 44, 255-277.
- Bradshaw, E.G., Anderson, N.J., Jensen, J.P., & Jeppesen, E. (2002) Phosphorus dynamics in Danish lakes and the implications for diatom ecology and palaeoecology. *Freshwater Biology*, 47, 1963-1975.
- Bradshaw, E.G., Nielsen, A.B., & Anderson, N.J. (2006) Using diatoms to assess the impacts of prehistoric, pre-industrial and modern land-use on Danish lakes. *Regional Environmental Change*, 6, 17-24.
- Bradshaw, E.G., Rasmussen, P., Nielsen, H., & Anderson, N.J. (2005a) Mid- to late-Holocene land-use change and lake development at Dallund So, Denmark: trends in lake primary production as reflected by algal and macrophyte remains. *Holocene*, **15**, 1130-1142.
- Bradshaw, E.G., Rasmussen, P., & Odgaard, B.V. (2005b) Mid- to late-Holocene land-use change and lake development at Dallund So, Denmark: synthesis of multiproxy data, linking land and lake. *Holocene*, **15**, 1152-1162.
- Bratlund, B. (1991) The bone remains of mammals and birds from the Bjørnsholm shell mound; a preliminary report. *Journal of Danish Archaeology*, **10**, 97-104.
- Brenner, W.W. (2005) Holocene environmental history of the Gotland Basin (Baltic Sea) a micropalaeontological model. *Palaeogeography Palaeoclimatology Palaeoecology*, 220, 227-241.
- Briffa, K.R., Bartholin, T., Eckstien, D., Jones, P.D., Schweingruber, F.H., & Zetterberg, P. (1990) A 1400-year tree-ring record of summer temperatures in Fennoscandia. *Nature*, **306**, 434–439.
- Briffa, K.R., Jones, P.D., Bartholin, T.S., Eckstein, D., Schweingruber, F.H., Karlén, W., Zetterberg, P., & Eronen, M. (1992) Fennoscandian summers from AD 500: temperature changes on short and long timescales. *Climate Dynamics*, 7, 111-119.

- Brinch Petersen, E. (1973). A Survey of the Late Palaeolithic and the Mesolithic of Denmark. In *The Mesolithic in Europe* (ed S.K. Kozlowski), pp. 77-127. Warsaw University Press, Warsaw.
- Brinch Petersen, E. (1993). The Late Palaeolithic and the Mesolithic in Denmark. In *Digging into the Past. 25 Years of Archaeology* (eds S. Hvass & B. Storgaard), pp. 46-49, Århus.
- Broecker, W.S. (2001) Was the Medieval Warm Period global? Science, 291, 1497-1499.
- Brown, K.J., Seppä, H., Schoups, G., Fausto, R., Rasmussen, P., & Birks, H.J.B. (submitted) A spatio-temporal reconstruction of Holocene temperature change in southern Scandinavia. *Holocene*.
- Brush, G.S. & Hilgartner, W.B. (2000) Paleoecology of submerged macrophytes in the upper Chesapeake Bay. *Ecological Monographs*, **70**, 645-667.
- Burks, R.L., Jeppesen, E., & Lodge, D.M. (2001) Littoral zone structures as Daphnia refugia against fish predators. *Limnology & Oceanography*, **46**, 230-237.
- Burman, J. & Schmitz, B. (2005) Periwinkle (*Littorina littorea*) intrashell delta δ^{18} O and delta δ^{13} C records from the mid-Holocene Limfjord region, Denmark: a new high-resolution palaeoenvironmental proxy approach. *Holocene*, **15**, 567-575.
- Campeaua, S., Pienitza, R., & Héquette, A. (1999) Diatoms as quantitative paleodepth indicators in coastal areas of the southeastern Beaufort Sea, Arctic Ocean *Palaeogeography Palaeoclimatology Palaeoecology.*, **146**, 67-97.
- Carbotte, S.M., Bell, R.E., Ryan, W.B.F., McHugh, C., Slagle, A., Nitsche, F., & Rubenstone, J. (2004) Environmental change and oyster colonisation within the Hudson River estuary linked to Holocene climate. *Geo-Marine Letters*, **24**, 212-224.
- Carlson, A.E., Clark, P.U., Raisbeck, G.M., & Brook, E.J. (2007) Rapid Holocene deglaciation of the Labrador sector of the Laurentide Ice Sheet. *Journal of Climate*, 20, 5126-5133.
- Carter, V. & Rybicki, N.B. (1990) Light attenuation and submersed macrophyte distribution in the tidal Potomac River and estuary. *Estuaries*, **13**, 441-452.
- Cederwell, H. & Elmgren, R. (1990) Biological effects of eutrophication in the Baltic Sea. *Ambio*, **19**, 109-112.
- Chen, N., Bianchi, T.S., McKee, B.A., & Bland, J.M. (2001) Historical trends of hypoxia on Louisiana shelf: application of pigments as biomarkers. *Organic Geochemistry*, **32**, 543-561.
- Christensen, C. (1995). The Littorina transgressions in Denmark. In *Man and sea in the Mesolithic. Coastal settlement above and below present sea-level* (ed A. Fischer), pp. 15-22. Oxbow Books, Oxford.
- Christensen, C. (2001a). Kystbosættelse og havniveauændringer i stenalderen. In *Danmarks jægerstenalder status og perspektiver* (eds O.L. Jensen, S.A. Sørensen & K.M. Hansen), pp. 183-193. Hørsholm Egns Museum.
- Christensen, C. (2001b). Kystbosættelser og havniveauændringer i stenalderen. In *Danmarks Jægerstenalder—Status og Perspektiver* (eds A.L. Jensen, S.A. Sørensen & K.M. Hansen), pp. 183-193. Hørsholm Egns Museum.
- Christensen, J.T., Cedhagen, T., & Hylleberg, J. (2004) Late-Holocene salinity changes in Limfjorden, Denmark. *Sarsia*, **89**, 379-387.
- Christiansen, C., Dalsgaard, K., Møller, J.T., & Bowman, D. (1990) Coastal Dunes in Denmark. Chronology in relation to sea level. *Catena Supplement*, **18**, 61-70.
- Claridge, P.N., Potter, I.C., & Hardisty, M.W. (1986) Seasonal changes in movements, abundance, size composition and diversity of the fish fauna of the Severn Estuary. *Journal of the Marine Biological Association*, **66**, 229-258.
- Clark, J.G.D. (1975) *The earlier Stone Age settlement of Scandinavia*. Cambridge University Press, Cambridge, pp. 252.

- Clarke, A., Juggins, S., & Conley, D. (2003) A 150-yr reconstruction of the History of coastal eutrophication in Roskilde Fjord, Denmark. *Marine Pollution Bulletin*, **46**, 1615-1629.
- Clarke, A.L., Weckström, K., Conley, D.J., Anderson, N.J., Adser, F., Andrén, E., de Jonge, V.N., Ellegaard, M., Juggins, S., Kauppila, P., Korhola, A., Reuss, N., Telford, R.J., & Vaalgamaa, S. (2006) Long-term trends in eutrophication and nutrients in the coastal zone. *Limnology & Oceanography*, **51**, 385-397.
- Clemmensen, L.B., Richardt, N., & Andersen, C. (2001) Holocene sea-level variation and spit development: data from Skagen Odde, Denmark. *Holocene*, **11**, 323-331.
- Colman, S.M., Baucom, P.C., Bratton, J.F., Cronin, T.M., McGeehin, J.P., Willard, D., Zimmerman, A.R., & Vogt, P.R. (2002) Radiocarbon dating, chronologic framework, and changes in accumulation rates of Holocene estuarine sediments from Chesapeake Bay. *Quaternary Research*, 57, 58-70.
- Conley, D.J. (2000) Biogeochemical nutrient cycles and nutrient management strategies. *Hydrobiologia*, **410**, 87-96.
- Conley, D.J., Humborg, C., Rahm, L., Savchuk, O.P., & Wulff, F. (2002a) Hypoxia in the Baltic Sea and basin-scale changes in phosphorus biogeochemistry. *Environmental Science & Technology*, **36**, 5315-5320.
- Conley, D.J., Kaas, H., Mohlenberg, F., Rasmussen, B., & Windolf, J. (2000) Characteristics of Danish estuaries. *Estuaries*, **23**, 820-837.
- Conley, D.J., Markager, S., Andersen, J., Ellermann, T., & Svendsen, L.M. (2002b) Coastal eutrophication and the Danish National Aquatic Monitoring and Assessment Program. *Estuaries*, 25, 848-861.
- Conradsen, K. (1995) Late Younger Dryas to Holocene palaeoenvironments of the southern Kattegat, Scandinavia. *The Holocene*, **5**, 447-456.
- Conradsen, K. & Heier-Nielsen, S. (1995) Holocene paleoceanography and paleoenvironments of the Skagerrak-Kattegat, Scandinavia. *Paleoceanography*, **10**, 801-813.
- Cooper, S. (1995a) Chesapeake Bay watershed historical alnd use: impact on water quality and diatom communities. *Ecological Applications*, **5**, 702-723.
- Cooper, S.R. (1995b) Diatoms in sediment cores from the mesohaline Chesapeake Bay, U.S.A. *Diatom Research*, **10**, 39-89.
- Cooper, S.R. & Brush, G.S. (1991) Long-Term History of Chesapeake Bay Anoxia. *Science*, **254**, 992-996.
- Cornwell, J.C., Conley, D.J., Owens, M., & Stevenson, J.C. (1996) A sediment chronology of the eutrophication of Chesapeake Bay. *Estuaries*, **19**, 488-499.
- Crawley, M.J. (1993) *GLIM for Ecologists*. Blackwell Scientific Publications, Oxford, pp. 379.
- Dahl, K., Lundsteen, S., & Helmig, S.A. (2003) *Stenrev havets oaser*. Gads Forlag, Copenhagen, pp. 104.
- Dahl, S.O. & Nesje, A. (1994) Holocene glacier fluctuations at Hardangerjøkulen, central southern Norway; a high-resolution composite chronology from lacustrine and terrestrial deposits. *Holocene*, 4, 269-277.
- Dahl, S.O. & Nesje, A. (1996) A new approach to calculating Holocene winter precipitation by combining glacier equilibrium-line altitudes and pine-tree limits: A case study from Hardangerjokulen, central southern Norway. *Holocene*, 6, 381-398.
- Daniel, J.S. & Chen, Z.Y. (1996) Neolithic settlement distribution as a function of sea level controlled topography in the Yangtze delta, China. *Geology*, **12**, 1083–1086.
- Davies, J.L. (1964) *Geographical variation in coastal development*. Oliver & Boyd, Edinburgh, pp. 212.
- Davis, M.B. (1976) Erosion rates and land use history in southern Michigan. *Environmental conservation*, **3**, 139-148.

- Day, J.W., Hall, C.A.S., Kemp, W.M., & Yáñez-Arancibia, A. (1989) *Estuarine ecology*. John Wiley & Sons, Toronto, Ontario, pp. 532.
- Dean, W.E. (1974) Determination of Carbonate and Organic-Matter in Calcareous Sediments and Sedimentary-Rocks by Loss on Ignition - Comparison with Other Methods. *Journal of Sedimentary Petrology*, **44**, 242-248.
- Dearing, J.A. (1986). Core Correlation and total sediment influx. In *Handbook of Holocene Palaeoecology and Palaeohydrology* (ed B.E. Berglund), pp. 247-270. John Wiley & Sons, Chichester & New York.
- Dearing, J.A. & Jones, R.T. (2003) Coupling temporal and spatial dimensions of global sediment flux through lake and marine sediment records. *Global and Planetary Change*, **39**, 147-168.
- Dellwig, O., Watermann, F., Brumsack, H.-J., & Gerdes, G. (1999) High-resolution Reconstruction of a Holocene Coastal Sequence (NW Germany) using inorganic geochemical data and diatom inventories. *Estuarine, Coastal and Shelf Science*, 48, 617-633.
- Dencker, J. (1999) Nyt og Noter. En stenalderskov. *Marinarkæologisk Nyhedsbrev fra Roskilde*, **12**, 35-36.
- Duarte, C.M. (1995) Submerged aquatic vegetation in relation to different nutrient regimes. *Opephelia*, **41**, 87-112.
- Dubois, G. (1924) Recherches sur les terrains quaternaires du nord de la France Mémoires de la Société Géologique du Nord, 8, 355.
- Edwards, K.J. (1993). Models of mid-Holocene forest farming for north-west Europe. In *Climate Change and Human Impact on the Landscape. Studies in palaeoecology and environment archaeology* (ed F.M. Chambers), pp. 133-145. Chapman and Hall, London.
- Edwards, K.J. & Whittington, G. (2001) Lake sediments, erosion and landscape change during the Holocene in Britain and Ireland. *Catena*, **42**, 143-173.
- Eisma, D. & Kalf, J. (1987) Dispersal, concentration and depsoition of suspended matter in the North Sea. *Journal of teh Geological Society, London*, **144**, 161-178.
- Ekman, M. & Mäkinen, J. (1996) Recent postglacial rebound, gravity change and mantle flow in Fennoscandia. *Geophysical Journal International*, **126**, 229-234.
- Ellegaard, C. & Skovgaard, H. (2007). Application of a 3D bio-chemical model for Norsminde Fjord. Part of task 1.1 of the EU-LIFE project AGWAPLAN. Environmental Centre Århus, pp. 7.
- Ellegaard, M. (2000) Variations in dinoflagellate cyst morphology under conditions of changing salinity during the last 2000 years in the Limfjord, Denmark. *Review of Palaeobotany and Palynology*, **109**, 65-81.
- Ellegaard, M., Christensen, N.F., & Moestrup, O. (1993) Temperature and salinity effects on growth of a non-chain-forming strain of *Gymnodinium catenatum* (Dinophyceae) established from a cyst from recent sediments in the sound (Øresund), Denmark. *Journal of Phycology*, **29**, 418-426.
- Ellegaard, M., Clarke, A.L., Reuss, N., Drew, S., Weckstrom, K., Juggins, S., Anderson, N.J., & Conley, D.J. (2006) Multi-proxy evidence of long-term changes in ecosystem structure in a Danish marine estuary, linked to increased nutrient loading. *Estuarine, Coastal and Shelf Science*, **68**, 567-578.
- Ellegaard, M., Lewis, J., & Harding, I. (2002) Cyst-theca relationship, life cycle, and effects of temperature and salinity on the cyst morphology of Gonyaulax baltica sp nov (Dinophyceae from the Baltic Sea area]. *Journal of Phycology*, **38**, 775-789.
- Ellis, B.F. & Messina, A. (1949) *Catalogue of Foraminifera (Supplements, including 2007)*. American Museum of Natural History and Micropaleontology Press, New York
- Emeis, K.C. (2005) Salinity changes in the central Baltic Sea (NW Europe) over the last 10 000 years: a reply to Wastegard and Andren. *Holocene*, **15**, 474-475.

- Emeis, K.C., Struck, U., Blanz, T., Kohly, A., & Voss, M. (2003) Salinity changes in the central Baltic Sea (NW Europe) over the last 10 000 years. *Holocene*, **13**, 411-421.
- Enghoff, I.B. (1989) Fishing from the Stone Age Settlement Norsminde. *Journal of Danish Archaeology*, **8**, 41-50.
- Enghoff, I.B. (1991) Mesolithic Eel-Fishing at Bjørnsholm, Denmark, Spiced with Exotic Species. *Journal of Danish Archaeology*, **10**, 105-118.
- Enghoff, I.B. (1994) Fishing in Denmark during the Ertebølle period. *International Journal of Osteoarchaeology*, **4**, 65-96.
- Enghoff, I.B. (1995). Fishing in Denmark during the Mesolithic Period. In *Man and Sea in the Mesolithic* (ed A. Fischer), pp. 67-74. Oxbow Books, Oxford.
- Enghoff, I.B. (1999) Fishing in the Baltic Region from the 5th century BC to the 16th century AD: evidence from fish bones. *Archaeofauna*, **8**, 41-85.
- Enghoff, I.B., MacKenzie, B.R., & Nielsen, E.E. (2007) The Danish fish fauna during the warm Atlantic period (ca. 7000-3900 BC): Forerunner of future changes? *Fisheries Research*, **87**, 167-180.
- Enghoff, I.E. (1986) Freshwater Fishing from a Sea-Coast Settlement the Ertebølle *locus* classicus Revisited. Journal of Danish Archaeology, **5**, 62-76.
- Engstrom, D.R., Fritz, S.C., Almendinger, J.E., & Juggins, S. (2000) Chemical and biological trends during lake evolution in recently deglaciated terrain. *Nature*, **408**, 161-166.
- Eronen, M. (1974) The history of the Litorina Sea and associated Holocene events. *Societas Scientiarum Fennicae, Commentationes Physico-Mathematicae*, **44**, 79-195.
- Ervynck, A., Bateman, C., Demiddele, H., Hollevoet, Y., Pieters, M., Schelvis, J., Tys, D., Strydonck, M., & van Verhaeghe, F. (1999) Human occupation because of a regression, or the cause of a transgression? a critical review of the interaction between geological events and human occupation in the Belgian coastal plain during the first millenium AD. *Probleme der Küstenforschung im südlichen Nordseegebiet*, 26, 97-121.
- Fægri, K. & Iversen, J. (1975) Textbook of pollen analysis. Munksgård, Copenhagen, pp. 340.
- Fairbanks, R.G. (1989) A 17,000 year glacio-eustatic sea-level record: Influence of glacial melting rates on the Younger Dryas event and deep-ocean circulation. *Nature*, 342, 637–642.
- Falt, L.-M. (1982) Late Quaternary sea-floor deposits off the Swedish west coast. *Department* of Geology, University of Goteborg and Chalmers University of Technology, *Publication A37*, 259 pp.
- Feistel, R., Nausch, G., & Hagen, E. (2006) Unusual Baltic inflow activity in 2002-2003 and varying deep-water properties. *Oceanologia*, **48**(**S**), 21-35.
- Feistel, R., Nausch, G., Matthäus, W., & Hagen, E. (2003) Temporal and spatial evolution of the Baltic deep water renewal in spring 2003. *Oceanologia*, **45**, 623-642.
- Feyling-Hanssen, R.W., Jørgensen, J.A., Knudsen, K.L., & Lykke-Andersen, A.-L. (1971) Late Quaternary Foraminifera from Vendsyssel, Denmark and Sandnes, Norway. Bulletin of the Geological Society of Denmark, 21, 1-317.
- Field, C.B., Behrenfeld, M.J., Randerson, J.T., & Falkowski, P. (1998) Primary production of the biosphere: Integrating terrestrial and oceanic components. *Science*, **281**, 237-240.
- Fischer, A. (1993). Mesolitisk inlandsbosættelse. In *Da klinger i muld. 25 års arkæologi i Danmark* (eds S. Hvass & B. Storgaard), pp. 58-63. Aarhus Universitetsforlag, Århus.
- Fischer, A. & Kristensen, K. (2002) *The Neolithisation of Denmark. 150 years of debate.* J.R.Collis, Sheffield, pp. 398.
- Fischer, A., Olsen, J., Richards, M., Heinemeier, J., Sveinbjörnsdóttir, A.E., & Bennike, P. (2007) Coast-inland mobility and diet in the Danish Mesolithic and Neolithic: evidence from stable isotope values of humans and dogs. *Journal of Archaeological Science*, **34**, 2125-2150.

- Flou, B., Kjær, B., & Thorndahl, J. (1980) Kolindsund et stykke Djursland fortæller danmarkshistorie. P. S. Johnsen, Grenaa, pp. 213.
- Flower, R.J. (1993) Diatom preservation: experiments and observations on dissolution and breakage in modern and fossil material. *Hydrobiologia*, **269/270**, 473-484.
- Fritz, S.C. (1990) Twentieth-century salinity and water level fluctuations in devils Lake, North dakota: Test of a diatom-based transfer function. *Limnology & Oceanography*, 35, 1771-1781.
- Funder, S., Demidov, I., & Yelovicheva, Y. (2002) Hydrography and mollusc faunas of the Baltic and the White Sea-North Sea seaway in the Eemian. *Palaeogeography Palaeoclimatology Palaeoecology*, **184**, 275-304.

Gabrielsen, S. (1953) Udgravningen på Flynderhage 1945-47. Århus stifts Årbøger, 46, 5-17.

- Gaillard, M.-J. (2007). Pollen methods and studies Archaeological implications. In Encyclopaedia of Quaternary Science (ed S.A. Elias), Vol. 3, pp. 2570-2595. Elsevier Science, Oxford.
- Gasse, F., Juggins, S., & Khelifa, L.B. (1995) Diatom-based transfer functions for inferring past hydrochemical characteristics of African lakes. *Palaeogeography, Palaeoclimatology, Palaeoecology.*, **117**, 31-54.
- Gehrels, W.R., Szkornik, K., Bartholdy, J., Kirby, J.R., Bradley, S.L., Marshall, W.A., Heinemeier, J., & Pedersen, J.B.T. (2006) Late Holocene sea-level changes and isostasy in western Denmark. *Quaternary Research*, 66, 288-302.
- Giesecke, T. & Bennett, K.D. (2004) The Holocene spread of Picea abies (L.) Karst. in Fennoscandia and adjacent areas. *Journal of Biogeography*, **31**, 1523-1548.
- Gjessing, J. (1980) The Aker moraines in southeast Norway. Norsk Geografisk Tidsskrift, 34, 9-34.
- Gjessing, J. & Spjeldnaes, N. (1979) Dating of the Grefsen moraine and remarks on the deglaciation of southeast Norway. *Norsk Geografisk Tidsskrift*, **33**, 71-81.
- Goodwin, T.W. (1980) The Biogeochemistry of the Carotenoids. Volume 1. Plants. Chapman & Hall, N.Y., 377 pp.
- Grimm, E.C. (1987) CONISS: A FORTRAN 77 program for stratigraphically constarined cluster analysis by the method of incremental sum of squares. *Computers & Geosciences*, **13**, 13-35.
- Grøn, O. (1995). The Maglemose Culture. The reconstruction of the social organisation of a Mesolithic culture in Northern Europe. In *BAR International Series*, Vol. 616, pp. 99. Oxford: Tempus Reparatum.
- Grossman, E.L. (1987) Stable isotopes in modern benthic foraminifera a study of vital effect. *Journal of Foraminiferal Research*, **17**, 48-61.
- Gustafsson, B.G. & Andersson, H.C. (2001) Modelling the exchange of the Baltic Sea from the meridional atmospheric pressure difference across the North Sea. *Journal of Geophysical Research*, **106**, 19731–19744.
- Gustafsson, B.G. & Westman, P. (2002) On the causes for salinity variations in the Baltic Sea during the last 8500 years. *Paleoceanography*, **17**, 12.1-12.4.
- Gyllencreutz, R. (2005) Late glacial and Holocene paleoceanography in the Skagerrak from high-resolution grain size records. *Palaeogeography, Palaeoclimatology, Palaeoecology*, **222**, 344-369.
- Gyllencreutz, R., Backman, J., Jakobsson, M., Kissel, C., & Arnold, E. (2006) Postgalcial palaeoceanography in the Skagerrak. *The Holocene*, **16**, 975-985.
- Gyllencreutz, R., Jakobsson, M., & Backman, J. (2005) Holocene sedimentation in the Skagerrak interpreted from chirp sonar and core data. *Journal of Quaternary Science*, 20, 21-32.
- Gyllencreutz, R. & Kissel, C. (2006) Lateglacial and Holocene sediment sources and transport patterns in the Skagerrak interpreted from high-resolution magnetic properties and grain size data. *Quaternary Science Reviews*, **25**, 1247-1364.

- Hafsten, U. (1983) Shore-level changes in south Norway during the last 13,000 years, traced by biostratigraphical methods and radiometric datings. *Norsk Geografisk Tidsskrift*, 37, 63–79.
- Håkansson, H. (1996) *Cyclotella striata* complex: typification and new combinations. *Diatom Research*, **11**, 241-260.
- Hanninen, J., Vuorinen, I., & Hjelt, P. (2000) Climatic factors in the Atlantic control the oceanographic and ecological changes in the Baltic Sea. *Limnology and Oceanography*, 45, 703-710.
- Hannon, G.E., Bradshaw, R.H.W., & Emborg, J. (2000) 6000 years of forest dynamics in Suserup Skov, a seminatural Danish woodland. *Global Ecology and Biogeography*, 9, 101-114.
- Harck, O. (1973) Eisenzeitliche Muschelhaufen an der schleswigschen Ost- und Westküste. *Offa*, **30**, 40-54.
- Hartz, S., Heinrich, D., & Lübke, H. (2002). Coastal Farmers the neolithisation of northernmost Germany. In *The Neolithisation of Denmark*. 150 years of debate (eds A. Fischer & K. Kristiansen), pp. 319-340. J.R. Collis, Sheffield.
- Hasle, G.R. & Syvertsen, E.E. (1996). Chapter 2. Marine Diatoms. In *Identifying Marine Phytoplankton* (ed C.R. Tomas), pp. 5-385. Academic Press, London, UK.
- Hebbeln, D., Knudsen, K.L., Gyllencreutz, R., Kristensen, P., Klitgaard-Kristensen, D.,
 Backman, J., Scheurle, C., Jiang, H., Gil, I., Smelror, M., Jones, P.D., & Sejrup, H.P. (2006) Late Holocene coastal hydrographic and climate changes in the eastern North Sea. *Holocene*, 16, 987-1001.
- Hedenström, A. & Possnert, G. (2001) Reservoir ages in Baltic sediments the Baltic Sea a case study of an isolation sequence from the Litorina Sea stage. *Quaternary Science Reviews*, **20**, 1779-1785.
- Hedges, R.E.M. (2004) Isotopes and red herrings: comments on Milner et al. 2004 and Lidén et al. 2004. *Antiquity*, **78**, 34-37.
- Hedges, R.E.M., Stevens, R.E., & Koch, P.L. (2005). Isotopes in bones and teeth. In *Isotopes in palaeoenvironmental research* (ed M.J. Leng), pp. 117-145. Springer, Dordrecht. The Netherlands.
- Heier-Nielsen, S. (1992) Foraminferanalyse, ¹⁴C-dateringer og stabil isotop analyse i kerne 95, Limfjorden. *Dansk Geologisk Forening, Årsskrift*, **1990-1991**, 39-45.
- Heier-Nielsen, S., Heinemeier, J., Nielsen, H.L., & Rud, N. (1995) Recent reservoir ages for Danish fjords and marine waters. *Radiocarbon*, **37**, 875-882.
- Hellesen, J.P. & Tuxen, O. (1988) Historisk Atlas Danmark. Copenhagen, pp. 303.
- Hendey, N.I. (1964) An Introductory Account of the Smaller Algae of British Coastal Waters. Koeltz Science Publishers, Koenigstein, Germany
- Hobson, L.A. & McQuoid, M.R. (1997) Temporal variations among planktonic diatom assemblages in a turbulent environment of southern Strait of Georgia, British Columbia, Canada. *Marine Ecology Progress Series*, **150**, 263-274.
- Hodgson, D.A., McMinn, A., Kirkup, H., Cremer, H., Gore, D., Melles, M., Roberts, D., & Montiel, P. (2003) Colonization, succession, and extinction of marine floras during a glacial cycle: A case study from the Windmill Islands (east Antarctica) using biomarkers. *Paleoceanography*, **18**, 1067.
- Hoffman, W. & Winn, K. (2000) The Littorina Transgression in the Western Baltic Sea as Indicated by Subfossil Chironomidae (Diptera) and Cladocera (Crustacea). *International Review of Hydrobiology*, **85**, 267-291.
- Holm, P. (1991). Kystfolk. Kontakter og sammenhænge over Kattegat og Skagerrak ca. 1550-1914 (English summary: Coastal people. Contacts across the Cattegat and Skagerrak, ca. 1550-1914. Fiskeri- og Søfartsmuseet, Esbjerg, pp. 418-421.
- Holm, P. & Bager, M. (2002). The Danish Fisheries, c. 1450-1800: Medieval and Early Modern Sources and Their Potential for Marine Environmental History. In *The*

exploited seas: new directions for marine environmental history (eds P. Holm, T.D. Smith & D.J. Starkey), Vol. Research in maritime history 21, pp. 97–122. International Maritime Economic History Association / Census of Marine Life, St. Johns, Newfoundland.

- Huckreide, H. & Meischener, D. (1996) Origin and environment of manganese-rich sediments within black shale basins. *Geochimica et Cosmochimica Acta*, **60**, 1399-1413.
- Huckriede, H., Clasen, S., & Meischener, D. (1996) Hydrographic and climatic changes recorded in Holocene sediments of the central Baltic Sea. *Baltica*, **9**, 76-91.
- Hunicke, B., Luterbacher, J., Pauling, A., & Zorita, E. (2008) Regional differences in winter sea level variations in the Baltic Sea for the past 200 yr. *Tellus Series a-Dynamic Meteorology and Oceanography*, **60**, 384-393.
- Hunicke, B. & Zorita, E. (2006) Influence of temperature and precipitation on decadal Baltic Sea level variations in the 20th century. *Tellus Series a-Dynamic Meteorology and Oceanography*, **58**, 141-153.
- Hurrell, J. (1995) Decadal trends in the North Atlantic Oscillation: regional temperatures and precipitation. *Science*, **269**, 676-679.
- Hustedt, F. (1957) Die Diatomeenflora des fluβsystems der Weser im Gebeit der Hansenstadt Bremen. Abhandlungen herausgegeben vom naturwissenschaftlichen Verein zu Bremen. *Abhanglungen Naturwissenschaftlichen Verein, Bremen*, **34**, 181-440.
- Hvass, S. & Storgaard, B. (1993) *Digging into the past. 25 years of archaeology in Denmark.* Aarhus University Press, Århus, pp. 312.
- Hyvärinen, H. (1988) Definition of the Baltic stages. *Annales Academiae Scientiarum Fennicae Ser. A III*, **148**, 7–11.
- Ingólfsson, Ó. & Hjort, C. (1999) The Antarctic contribution to global sea level rise. *Polar Research*, **18**, 323-330.
- Iversen, J. (1937) Undersøgelser over litorinatransgressioner i Danmark. *Meddelelser fra* Dansk Geologisk Forening, **9**, 223-232.
- Iversen, J. (1941) Landnam i Danmarks Stenalder. *Danmarks Geologiske Undersøgelse II*, **66**, 68 pp.
- Iversen, J. (1944) Viscum, Hedera and Ilex as climatic indicators: a contribution to the study of the post-glacial temperature climate. *Geologiska Föreningens I Stockholm Förhandlingar*, 66, 463–483.
- Iversen, J. (1969) Retrogressive development of a forest ecosystem demonstrated by pollen diagrams from fossil mor. *Oikos supplement*, **12**, 35-49.
- Iversen, J. (1973) The development of Denmark's nature since the last glacial. *Danmarks Geologiske Undersøgelse*, **V**, **7-C.**, 126 pp.
- Jackson, J.B.C., Kirby, M.X., Berger, W.H., Bjorndal, K.A., Botsford, L.W., Bourque, B.J., Bradbury, R.H., Cooke, R., Erlandson, J., Estes, J.A., Hughes, T.P., Kidwell, S., Lange, C.B., Lenihan, H.S., Pandolfi, J.M., Peterson, C.H., Steneck, R.S., Tegner, M.J., & Warner, R.R. (2001) Historical OverPshing and the Recent Collapse of Coastal Ecosystems. *Science*, **293**, 629-638.
- Jantunen, T. & Donner, J. (1996) The formation of raised beaches in southern Finland during the Ancylus and Litorina stages. *Bulletin of the Geological Society of Finland*, 68, 34– 39.
- Jeffrey, S., Mantoura, R., & Wright, S. (1997) *Phytoplankton Pigments in Oceanography: Guidelines to Modern Methods*. UNESCO publishing, Paris, pp. 661 pp.
- Jelgersma, S. (1979). Sea-level changes in the North Sea basin. In *The Quaternary History of the North Sea, Symposia Universitatis Upsaliensis Annum Quingentesimum Celebrantis* (ed E. Oerle, Shqttenhelm, R.T.E., Wiggers, A.J.), Vol. 2, pp. 223-248.
- Jennings, A.E. & Weiner, N.J. (1996) Environmental change in eastern Greenland during the last 1300 years: evidence from foraminifera and lithofacies in Nansen Fjord, 68 degrees N. *The Holocene*, **6**, 179-191.

Jensen, J. (2001) Danmarks Oldtid. Stenalder 13.000–2.000f. Kr. Gyldendal, Copenhagen

- Jensen, J.B. & Bennike, O. (2008) Geological setting as background for methane distribution in Holocene mud deposits, Århus Bay, Denmark. *Continental Shelf Research*, doi:10.1016/j.csr.2008.08.007.
- Jensen, J.P., Pedersen, A.R., Jeppesen, E., & Sondergaard, M. (2006) An empirical model describing the seasonal dynamics of phosphorus in 16 shallow eutrophic lakes after external loading reduction. *Limnology and Oceanography*, **51**, 791-800.
- Jensen, S. & Spärck, R. (1934) Bløddyr II. Saltvandsmuslinger. Danmarks Fauna, 40, 208.
- Jessen, A. (1920) Stenalderhavets Udbredelse i det nordlige Jylland. (with and English summary: The extension of the Stone-age Sea (Tapes-Litorina Sea) in Northern Jutland). *Danmarks Geologiske Undersøgelse II*, **Række**, **Nr. 35**, 112.
- Jevrejeva, S., Moore, J.C., Woodworth, P.L., & Grinsted, A. (2005) Influence of large-scale atmospheric circulation on European sea level: results based on the wavelet transform method. *Tellus Series a-Dynamic Meteorology and Oceanography*, **57**, 183-193.
- Jiang, H. (1996) Diatoms from the surface sediments of the Skagerrak and the Kattegat and their relationship to the spatial changes of environmental variables. *Journal of Biogeography*, 23, 129-137.
- Jiang, H., Björck, S., & Knudsen, K.L. (1997) A palaeoclimatic and palaeoceanographic record of the last 11 000 ¹⁴C years from the Skagerrak-Kattegat, northeastern Atlantic margin. *The Holocene*, **7**, 301-310.
- Jiang, H., Björck, S., & Svensson, N.-O. (1998) Reconstruction of Holocene sea-surface salinity in the Skagerrak-Kattegat: a climatic and environmental record of Scandanavia. *Journal of Quaternary Science*, **13**, 107-114.
- Johannesson, K., Rödström, E.-M., & Aase, H. (1989) Low genetic variability in Scandinavia populations of Ostrea edulis L. - possible causes and implications. Journal of Experimental Biology and Ecology, 128, 177-190.
- Johansen, K.L. (2006) Settlement and land use at the Mesolithic-Neolithic transition in southern Scandinavia. *Journal of Danish Archaeology*, **14**, 201-223.
- Jones, V. (2007). Diatom Introduction. In *Encyclopedia of Quaternary Science* (ed S.A. Elias), pp. 476-484. Elsevier, Oxford.
- Jørgensen, B.B. & Richardson, K., eds. (1996) *Eutrophication in coastal marine ecosystems*, pp 273. American Geophysical Union, Coastal and Estuarine Studies, 52, Washington, D.C.
- Jørgensen, B.B. & Sørensen, J. (1985) Seasonal cycles of O₂, NO₃- and SO²⁴- reduction in estuarine sediments: the significance of an NO3 reduction maximum in spring. *Marine Ecology Progress Series*, **24**, 65-74.
- Jowsey, P.C. (1966) An improved peat sampler. New Phytologist, 65, 245-248.
- Juggins, S. (1991-2009) C2 data anlysis.

<u>http://www.staff.ncl.ac.uk/staff/stephen.juggins/software/C2Home.htm</u>. Newcastle University, Newcastle.

- Juggins, S. (1992) Diatoms in the Thames Estuary, England: Ecology, Palaeoecology and Salinity Transfer Function. *Bibliotheca Diatomologica*, **25**, 216.
- Juggins, S. (2004). The Molten Diatom Database and Transfer Functions. User Guide. Newcastle University, Newcastle, UK, pp. 14.
- Juggins, S. & ter Braak, C.J.F. (1997) CALIBRATE A Computer Program for Species ± Environmental Calibration by [Weighted-Averaging] Partial Least Squares Regression (0.81). University of Newcastle, Newcastle.
- Karlén, W. (1988) Scandinavian glacial and climatic fluctuations during the Holocene. *Quaternary Science Reviews*, **7**, 199-209.
- Karlén, W. & Matthews, J.A. (1992) Reconstructing Holocene glacier variation from glacial lake sediments studies from Nordvestlandet and Jostedalsbreen-Jotunheimen, southern Norway. *Geografiska Annaler*, 74A, 327-348.

- Kilham, S.S., Theriot, E.C., & Fritz, S.C. (1996) Linking planktonic diatoms and climate change in large lakes of the Yellowstone ecosystem using resource theory. *Limnology and Oceanography*, **41**, 1052-1062.
- Kisum, O. (2007) Holocæn palæoøkologi i Horsens Fjord baseret på foraminiferer. Specialeopgave, Geologisk Institut, Aarhus Universitet,, 85 pp.
- Klassen, L. (1996-97) The debate on the Mesolithic-Neolithic transition in the western Baltic: a central European perspective. *Journal of Danish Archaeology*, **13**, 171-178.
- Knudsen, K.L. (1971) Late Quaternary Foraminifera from Vendsyssel Denmark and Sandnes Norway. *Bulletin of the Geological Society of Denmark*, **21**, 185-291.
- Knudsen, K.L. (1994) The marine Quaternary in Denmark: a review of new evidence from glacial-interglacial studies. *Bulletin of the Geological Society of Denmark*, **41**, 203-218.
- Knudsen, K.L. (1998) Foraminiferer i Kvartær stratigrafi: Laboratorie- og fremstillingsteknik samt udvalgte eksempler. *Geologisk Tidsskrift*, **3**, 1-25.
- Knudsen, K.L. & Konradi, P.B. (submitted). The marine transgression in Normsinde Fjord: foraminiferal evidence. In *Stone Age Settlment in the Coastal Fjord of Norsminde, Jutland, Denmark* (ed S. Andersen).
- Kolbe, R.W. (1927) Zur Ökologie, Morphologie, und Systematik der Brackwasser Diatomeen. *Pflanzenforschung, Jena*, **7**, 1-146.
- Korhola, A., Vasko, K., Toivonen, H.T.T., & Olander, H. (2002) Holocene temperature changes in northern Fennoscandia reconstructed from chironomids using Bayesian modelling. *Quaternary Science Reviews*, **21**, 1841-1860.
- Korrigna, P. (1952) Recent advances in oyster biology. *Quaterly Review of Biology*, **27**, 266-308.
- Köster, F.W., Möllmann, C., Hinrichsen, H.-H., Tomkiewicz, J., Wieland, K., Kraus, G., Voss, R., Mackenzie, B.R., Schnack, D., Makarchouk, A., Plikshs, M., & J.E., B. (2005) Baltic cod recruitment the impact of climate and species interaction. *ICES Journal of Marine Science*, 62, 1408–1425.
- Krammer, K. & Lange-Bertalot, H. (1986-1991). Bacillariophyceae. In Süsswasserflora von Mitteleuropa (eds H. Ettl, J. Gerloff, H. Heynig & D. Mollenhauer), Vol. Band 2/1-2/3. Gustav Fischer Verlag, Stuttgart and Jena, Germany.
- Kristensen, P., Heier-Nielsen, S., & Hylleberg, J. (1995) Late-Holocene salinity fluctuations in Bjørnsholm Bay, Limfjorden, Denmark, as deduced from micro- and macrofossil analysis. *The Holocene*, 5, 312-322.
- Kristensen, P.H. & Knudsen, K.L. (2006) Palaeoenvironments of a complete Eemian sequence at Mommark, South Denmark: foraminifera, ostracods and stable isotopes. *Boreas*, 35, 349-366.
- Kronvang, B., Ærtebjerg, G., Grant, R., Kristensen, P., Hov-mand, M., & Kirkegaard, J. (1993) Nationwide monitoring of nutrients and their ecological effects: state of the Danish aquatic environment. *Ambio*, **22**, 176-187.
- Kullenberg, G. (1981). Physical oceanography. In *The Baltic Sea. Elsevier Oceanography Series* (ed A. Voipio), Vol. 30, pp. 135-181. Elsevier, Amsterdam.
- Kurtz, J.C., Yates, D.F., Macauley, J.M., Quarles, R.L., Genthner, F.J., Chancy, C.A., & Devereux, R. (2003) Effects of light reduction on growth of the submerged macrophyte *Vallisneria americana* and the community of root-associated heterotrophic bacteria. *Journal of Experimental Marine Biology and Ecology*, 291, 199-218.
- Lamb, H. (1982) Climate, history and the modern world. Methuen, London, pp. 387.
- Lambeck, K. (1995) Late Devensian and Holocene shorelines of the British Isles and North Sea from models of glacio-hydroisostatic rebound. *Journal of the Geological Society* (*London*), **152**, 437-448.
- Lambeck, K. (1999) Shoreline displacements in southern–central Sweden and the evolution of the Baltic Sea since the last maximum glaciation. *Journal of the Geological Society* (*London*), **156**, 465-486.
- Lange-Bertalot, H. (2001). Navicula sensu stricto 10 genera separated from Navicula sensu lato Frustulia. In Diatoms of Europe (ed H. Lange-Bertalot), Vol. 2, pp. 526. A.R.G. Gantner Verlag K.G., Ruggell, Liechenstein.
- Lapointe, M. (2000) Modern diatom assemblages in surface sediments from the Maritime Estuary and the Gulf of St. Lawrence, Quebec (Canada). *Marine Micropaleontology*, 40, 43-65.
- Larsen, G., Eiriksson, J., Knudsen, K.L., & Heinemeier, J. (2002) Correlation of late Holocene terrestrial and marine markers, north Iceland: implications for reservoir age changes. *Polar Research*, 21, 283-290.
- Larsson, L. (1990) The Mesolithic of Southern Scandinavia. *Journal of World Prehistory*, **4**, 257-309.
- Larsson, L. (2007) Mistrust traditions, consider innovations? The Mesolihic-Neolithic in southern Scandinavia. *Proceedings of the British Academy*, **144**, 595-616.
- Lauridsen, T.L., Kairesalo, T., & Perrow, M.R. (1998). Impact of submerged macrophytes on fish-zooplankton interactions in lakes. In *The structuring role of submerged macrophytes in lakes* (eds E. Jeppesen, M. Søndergaard, M. Søndergaard & K. Christoffersen), Vol. 131, pp. 91-114. Springer Verlag, Dordrecht, The Netherlands.
- Leavitt, P.R. (1993) A review of factors that regulate carotenoid and chlorophyll deposition and fossil pigment abundance. *Journal of Paleolimnology*, **9**, 109-127.
- Leavitt, P.R. & Hodgson, D.A. (2001). Sedimentary Pigments. In *Traking Environmental Change Using Lake Sediments* (eds J.P. Smol, H.J.B. Birks & W.M. Last), Vol. 3: Terrestrial, Algal, and Siliceous indicators, pp. 295-325. Kluwer Academic Publishers, Dordrecht, The Netherlands.
- Leavitt, P.R., Vinebrooke, R.D., Donald, D.B., Smo, J.P., & Schindler, D.W. (1997) Past ultraviolet radiation enviornments in lakes derived from fossil pigments. *Nature*, **388**, 457-459.
- Lehmann, A. & Hinrichsen, H.H. (2000a) On the thermohaline variability of the Baltic Sea. *Journal of Marine Systems*, **25**, 333-357.
- Lehmann, A. & Hinrichsen, H.H. (2000b) On the wind driven and thermohaline circulation of the Baltic Sea. *Physics and Chemistry of the Earth Part B-Hydrology Oceans and Atmosphere*, 25, 183-189.
- Lehtonen, K.K. & Schiedek, D. (2006) Monitoring biological effects of pollution in the Baltic Sea: Neglected—but still wanted? *Marine Pollution Bulletin*, **53**, 377-386.
- Leśniewska, M. & Witak, M. (2008) Holocene diatom biostratigraphy of the SW Gulf of Gdańsk, Southern Baltic Sea (part III). *Oceanological and Hydrobiological Studies*. *International Journal of Oceanography and Hydrobiology*, **XXXVII**, 35-52.
- Leth, J.O. (1996) Late Quaternary geological development of the Jutland Bank and the initiation of the Jutland Current, NE North Sea. *Norges Geologiske Undersøkelse Bulletin*, **430**, 25-34.
- Leveneter, A., Domack, E.W., Ishman, S.E., Brachfeld, S., McClennen, C.E., & Manley, P. (1996) Productivity cycles of 200-300 years in the Antractic Peninsula region: understanding linkage among the sun, atmosphere, oceans, sea ice and biota. *Geological Society of America Bulletin*, **108**, 1626-1644.
- Levkov, Z. (2009). *Amphora* sensu lato. In *Diatoms of Europe* (ed H. Lange-Bertalot), Vol. 5, pp. 916. A.R.G. Gantner Verlag K.G., Ruggell, Liechenstein.
- Lindström, G. (1886) Om postglaciala sänkningar af Gotland. *Geologiska Föreningens I* Stockholm Förhandlingar, **102**, 251-281.
- Longva, O. & Bakkejord, K.J. (1990) Iceberg deformation and erosion in soft sediments, southeast Norway. *Marine Geology*, 1-2, 87-104.

- Longva, O. & Thoresen, M.K. (1991) Iceberg scours, iceberg gravity craters and current erosion marks from a gigantic Preboreal flood in southeastern Norway. *Boreas*, **20**, 47-62.
- Longva, O. & Thorsnes, T. (1997) Skagerrak in the past and present an integrated study of geology, chemistry, hydrography and microfossil ecology. *NGU Special publication*, 8, 1-100.
- Lowe, J.J., Hoek, W.Z., & INTIMATE Group (2001) Inter-regional correlation of palaeoclimatic records for the Last Glacial-Interglacial Transition: a protocol for improved precision recommended by the INTIMATE project group. *Quaternary Science Reviews*, **20**, 1175-1187.
- Lowe, J.J. & Walker, M.J.C. (1997) *Reconstructing Quaternary Environments*. 2nd edition edn. Longman, Essex, pp. 472.
- Lowe, J.J. & Walker, M.J.C. (2000) Radiocarbon dating the last glacial-interglacial transition (Ca. 14-9 14C Ka BP) in terrestrial and marine records: the need for new quality assurance protocols. *Radiocarbon*, **41**, 53-68.
- MacKenzie, B.R., Aleit, J., Conley, D.J., Holm, P., & Kinze, C.C. (2002) Ecological hypotheses for a historical reconstruction of the upper trophic level biomass in the Baltic Sea and Skagerrak. *Canadian Journal of Fisheries and Aquatic Sciences*, 59, 173-190.
- MacKenzie, B.R., Bager, M., Ojaveer, H., Awebro, K., Heino, U., Holm, P., & Must, A. (2007) Multi-decadal scale variability in the eastern Baltic cod fishery 1550-1860 -Evidence and causes. *Fisheries Research*, 87, 106-119.
- Madsen, A.P., Müller, S., Neergaard, C., Petersen, C.G.J., Rostrup, E., Steenstrup, K.J.V., & Winge, H. (1900) Affaldsdynger fra Stenalderen i Danmark. Undersøgte for Nationalmuseet C.A. Reitzel, Copenhagen
- Mandahl-Barth, G. & Bondesen, P. (1949) Bløddyr III Ferskvandsbløddyr. *Danmarks Fauna*, **54**, 1-249.
- Mangerud, J. (1972) Radio Carbon Dating of Marine Shells Including a Discussion of Apparent Age of Recent Shells from Norway. *Boreas (Oslo)*, **1**, 143-172.
- Mangerud, J., Jansen, E., & Landvik, J.Y. (1996) Late Cenozoic history of the Scandanavian and Barents Sea ice sheets. *Global and Planetary Change*, **12**, 11-26.
- Mannino, A. & Harvey, H.R. (2000) Biochemical composition of particles and dissolved organic matter along an estuarine gradient: Sources and implications for DOM reactivity. *Limnology & Oceanography*, **45**, 775-788.
- Maslin, M.A., Mikkelsen, C., Vilela, C., & Haq, B. (1998) Sea-level and gas-hydrate controlled sediment failures of the Amazon fan. *Geology*, **26**, 1107-1110.
- Maslin, M.A. & Swann, G. (2006). Isotopes in Marine Sediments. In *Isotopes in Palaeoenvironmental Research* (ed M. Leng), pp. 227-290. Springer, Dordrecht, The Netherlands.
- Matthäus, W. & Franck, H. (1992) Characteristics of Major Baltic Inflows a Statistical-Analysis. *Continental Shelf Research*, **12**, 1375-1400.
- Matthäus, W., Nehring, D., Feistel, R., Nausch, G., Mohrholz, V., & Lass, H.U. (2008). The inflow of highly saline water into the Baltic Sea. In *State and Evolution of the Baltic Sea*, 1952 2005. A Detailed 50-Year Survey of Meteorology and Climate, Physics, Chemistry, Biology, and Marine Environment (eds R. Feistel, G. Nausch & N. Wasmund), pp. 265-309. John Wiley & Sons, Hoboken.
- Matthews, J.A. & Karlén, W. (1992) Asynchronous neoglaciation and Holocene climatic change recosntructed from Norwegian lacustrine sedimentary sequences. *Geology*, **20**, 991-994.
- McGowan, S. (2007). Pigments in Sediments of Aquatic Environments. In *Encyclopedia of Quaternary Science* (ed S.A. Elias), pp. 2062-2074. Elsevier.

- McLusky, D.S. & Elliot, M. (2004) *The estuarine ecosystem: Ecology, threats and management.* 3rd edition edn. Oxford University Press
- McQuoid, M.R. & Nordberg, K. (2003) The diatom *Paralia sulcata* as an environmental indicator species in coastal sediments. *Estuarine Coastal and Shelf Science*, **56**, 339-354.
- Md. Islam, S. & Tanaka, M. (2004) Impacts of pollution on coastal and marine ecosystems including coastal and marine fisheries and approach for management: a review and synthesis *Marine Pollution Bulletin*, **48**, 624-649.
- Meier, H.E.M., Feistel, R., Piechura, J., Arneborg, L., Burchard, H., Fiekas, V., Golenko, N., Kuzmina, N., Mohrholz, V., Nohr, C., Paka, V.T., Sellschopp, J., Stips, A., & Zhurbas, V. (2006) Ventilation of the Baltic Sea deep water: A brief review of present knowledge from observations and models. *Oceanologia*, 48, 133-164.
- Mertens, K.N., Ribeiro, S., Bouimetarhan, I., Hulya Caner, H., Nebout, N.C., Dale, B., De Vernal, A., Ellegaard, M., Filipova, M., Godhe, A., Goubert, E., Grøsfjeld, K., Holzwarth, U., Kotthoff, U., Leroy, S.A.G., Londeix, L., Marret, F., Matsuoka, K., Mudie, P.J., Naudts, L., Peña-Manjarrez, J.L., Persson, A., Popescu, S.-P., Pospelova, V., Sangiorgi, F., van der Meer, M.T.J., Vink, A., Zonneveld, K.A.F., Vercauteren, D., Vlassenbroeck, J., & Louwye, S. (2009) Process length variation in cysts of a dinoflagellate, Lingulodinium machaerophorum, in surface sediments: Investigating its potential as salinity proxy. *Marine Micropaleontology*, 70, 54-69.
- Mertz, E.L. (1924) Late and post-glacial height changes in Denmark (In Danish). *Danish Geological Survey (DGU)* 2., **41**, 50.
- Middleboe, A.L. & Sand-Jensen, K. (2000) Long-term changes in macroalgal communities in a Danish estuary. *Estuaries*, **34**, 457-466.
- Miettinen, A. (2002) Relative sea level changes in the eastern part of the gulf of Finland during the last 8000 years. *Annales Academiae Scientiarum Fennicae, Geologica-Geographica*, **162**, 100 pp.
- Miettinen, A., Savelieva, L., Subetto, D.A., Dzhinoridze, R., Arslanov, K., & Hyvarinen, H. (2007) Palaeoenvironment of the Karelian Isthmus, the easternmost part of the Gulf of Finland, during the Litorina Sea stage of the Baltic Sea history. *Boreas*, **36**, 441-458.
- Mikkelsen, P.H. (1994) Arkæozoologiske og arkæobotaniske undersøgelser af bopladsmaterialet fra ældre jernalder. *LAG*, **5**, 73-114.
- Mikkelsen, V. (1949) Præstø Fjord. The development of the Post-Glacial vegetation and a contribution to the history of the Baltic Sea. *Dansk Botanisk Arkiv*, **13**, 1-171.
- Miller, U. (1982) Shore displacement and coastal dwelling in the Stockholm region during the past 5000 years. *Annales Academiae Scientiarum Fennicae*, **A134**, 185-211.
- Miller, U. & Hedin, K. (1988) The Holocene development of landscape and environment in the south-east Mälaren valley, with special reference to Helgö. *Excavations at Helgö XI, Kungliga Vitterhets Historie och Antikvitets Akademien Stockholm*, 72.
- Milner, N. (2001a) At the cutting edge: using thin sectioning to determine season of death of the European Oyster, *Ostrea edulis. Journal of Archaeological Science*, **28**, 861-873.
- Milner, N. (2001b) Seasonality information from the incremental growth of the European oyster for Ertebølle sites in Denmark. University of Durham and University of Newcastle Upon Tyne Archaeological Reports, Cambridge.
- Milner, N. (2002) Incremental growth of the European Oyster, *Ostrea edulis*: seasonality information from Danish kitchenmiddens. *British Archaeological Reports*.
- Milner, N., Craig, O.E., Bailey, G.N., Pedersen, K., & Andersen, S.H. (2004) Something fishy in the Neolithic? A re-evaluation of stable isotope analysis of Mesolithic and Neolithic coastal populations. *Antiquity*, **78**, 9-22.
- Milner, N., Craig, O.E., Bailey, G.N., Pedersen, K., & Andersen, S.H. (2006) A Response to Richards and Schulting. *Antiquity*, **308**, 456-457.

- Moeslund, B., Løjtnant, B., Mathiesen, H., Mathiesen, L., Pedersen, A., Thyssen, N., & Schou, J.C. (1990) *Danske vandplanter*. Miljøministeriet, Miljøstyrelsen, Copenhagen, pp. 192.
- Mörner, N.-A. (1976) Eustatic changes during the last 8,000 years in view of radiocarbon calibration and new information from the Kattegatt region and other northwestern European coastal areas. *Palaeogeography, Palaeoclimatology, Palaeoecology*, **19**, 63-85.
- Mörner, N.-A. (1979) The Fennoscandian uplift and Late Cenozoic geodynamics. *GeoJournal*, **3**, 287-318.
- Munthe, H. (1894) Preliminary report on the physical geography of the Littorina Sea. *Bulletin* of the Geological Society of the University of Uppsala, **2**, 1-38.
- Munthe, H. (1910) Studier oʻʻfver Gottlands senkvarta ra historia. Sveriges Geologiska Undersoʻkning, Ser. Ca, 4, 213 pp.
- Murray, J. (1991) *Ecology and Palaeoecology of benthic foraminifera*. Longman, Essex, pp. 397.
- Muus, B.J. (1967) The fauna of Danish estuaries and lagoons. Meddelelser fra Danmarks Fiskeri- og Havundersøgelser. pp. 316.
- Muylaert, K., Sabbe, K., & Vyverman, W. (2000) Spatial and temporal dynamics of phytoplankton communities in a freshwater tidal estuary (Schelde, Belgium). *Estuarine Coastal and Shelf Science*, **50**, 673-687.
- Nelson, D.A. & Smith, W.O. (1986) Phytoplankton bloom dynamics of the western Ross Sea Ice Edge II. Mesoscale cycling of nitrogen and silicon. *Deep Sea Research*, **33**, 1389-1412.
- Nelson, D.M., Treguer, P., Brzezinski, M.A., Leynaert, A., & Queguiner, B. (1995)
 Production and Dissolution of Biogenic Silica in the Ocean Revised Global
 Estimates, Comparison with Regional Data and Relationship to Biogenic
 Sedimentation. *Global Biogeochemical Cycles*, 9, 359-372.
- Nesje, A. & Dahl, S.O. (1994) Lateglacial and Holocene Glacier Fluctuations and Climate Variations in Western Norway a Review. *Quaternary Science Reviews*, **12**, 255-261.
- Nesje, A., Matthews, J.A., Dahl, S.O., Berrisford, M.S., & Andersson, C. (2001) Holocene glacier fluctuations of Flatebreen and winter precipitation changes in Jostedalsbreen region, Western Norway, based on glaciolacustrine sediment records. *Holocene*, **11**, 267-280.
- Nielsen, E.S. (1938) De danske farvandes hydrografi i Litorinatiden. *Dansk Geologisk Forening*, **9**, 337-350.
- Nielsen, K., Nielsen, L.P., & Rasmussen, P. (1995) Estuarine nitrogen retention independently estimated by the denitrification rates and mass balance methods: a study of Norsminde Fjord, Denmark. *Marine Ecology Progress Series*, **119**, 275-283.
- Nielsen, N. (2008). Marine Molluscs in Danish Stone Age Middens: A Case Study on Krabbesholm II. In *Early Human Impact on Megamolluscs. BAR International Series* (eds A. Antczak & R. Cipriani), Vol. 1865, pp. 157-167. Hadrian Books Ltd, Banbury, Oxford, UK.
- Nielsen, S.L., Sand-Jensen, K., Borum, J., & Geertz-Hansen, O. (2002) Depth colonisation of eelgrass (Zostera marina) and macroalgae as determined by water transparancy in Danish coastal waters. *Estuaries*, 25, 1025-1032.
- Nissling, A. & Westin, L. (1997) Salinity requirements for successful spawning of Baltic and Belt Sea cod and the potential for cod stock interactions in the Baltic Sea. *Marine Ecology Progress Series*, **152**, 261-271.
- Nixon, S.W. (1995) Coastal marine eutrophication: a definition, social causes and future concerns. *Ophelia*, **41**, 199-219.
- Nixon, S.W., Ammerman, J.W., Atkinson, L.P., Berounsky, V.M., Billen, G., Boicourt, W.C., Boynton, W.R., Church, T.M., Ditoro, D.M., Elmgren, R., Garber, J.H., Giblin, A.E.,

Jahnke, R.A., Owens, N.J.P., Pilson, M.E.Q., & Seitzinger, S.P. (1996) The fate of nitrogen and phosphorus at the land sea margin of the North Atlantic Ocean. *Biogeochemistry*, **35**, 141-180.

- Nordberg, K. (1991) Oceanography in the Kattegat & Skagerrak over the past 8,000 years. *Paleoceanography*, **6**, 461-484.
- Nordberg, K. & Bergsten, H. (1988) Biostratigraphic and sedimentological evidence of hydrographic changes in the Kattegat during the later part of the Holocene. *Marine Geology*, **83**, 135-158.
- Nordmann, V. (1903) Østerens (*Ostrea edulis* L.) udbredelse i nutiden og fortiden i havet omkring Danmark. *Meddelelser fra Dansk Geologisk Forening*, **9**, 45-60.
- Odgaard, B. (1989) Cultural landscape development through 5500 years at Lake Skånsø, North-western Jutland as reflected in a regional pollen diagram. *Journal of Danish Archaeology*, **8**, 200-210.
- Odgaard, B. (1994) The Holocene vegetation history of northern West Jutland. *Opera Botanica*, **123**, 171.
- Odgaard, B. (2006). Fra bondestenalder til nutid. In *Naturen i Danmark Geologien* (ed G. Larsen), pp. 333-359. Gyldendal.
- Odgaard, B.V. (1992) Wind-determined sediment distribution and Holocene sediment yield in a small Danish kettle lake. *Journal of Paleolimnology*, **8**, 3-13.
- Odgaard, B.V. & Rasmussen, P. (2000) Origin and temporal development of macro-scale vegetation patterns in the cultural landscape of Denmark. *Journal of Ecology*, **88**, 733-748.
- Olsen, J., Rasmussen, P., & Heinemeier, J. (2009) Holocene temporal and spatial variation in the radiocarbon reservoir age of three Danish fjords. *Boreas*, **38**, 458-470.
- Olsen, J., Rasmussen, P., & Heinemeier, J. (submitted). Norsminde Fjord: sediment core chronology. In *Stone Age Settlement in the coastal Fjord of Norsminde, Jutland, Denmark* (ed S. Andersen).
- Otto, L., Zimmerman, T.F., Furnes, G.K., Mork, M., Saetre, R., & Becker, G. (1990) Review of the physical oceanography of the North Sea. *Netherlands Journal of Sea Research*, **26**, 161-238.
- Page, M.J. & Trustrum, N.A. (1997) A late-Holocene lake sediment record of teh erosion repsonse to land use change in a steepland catchment, New Zealand. Z. *Geomorphologie N.F.*, **41**, 369-392.
- Parker, A.G., Goudie, A.S., Anderson, D.E., Robinson, M.A., & Bonsall, C. (2002) A review of the mid-Holocene elm decline in the British Isles. *Progress in Physical Geography*, 26.
- Parsons, M.L., Dortch, Q., Turner, R.E., & Rabalais, N.N. (1999) Salinity history of coastal marshes reconstructed from diatom remains. *Estuaries*, 22, 1078-1089.
- Pedersen, A., Fischer, A., & Aaby, B., eds. (1997) *The Danish Storebælt since the ice age man, sea and forest*, pp 339. The Storebælt Publications, Copenhagen.
- Pedersen, J.B.P., Svinth, S., & Bartholdy, J. (2009) Holocene evolution of a drowned meltwater valley in the Danish Wadden Sea. *Quaternary Research*, **72**, 68-79.
- Pedersen, S.A.S. & Petersen, K.S. (1997) Djurslands Geologi. Danmarks og Grønlands Geologiske Undersøgelse, 96.
- Peglar, S.M. (1993) The mid-Holocene *Ulmus* decline at Diss Mere, Norfolk, UK: a year-byyear pollen stratigraphy from annual laminations. *The Holocene*, **3**, 1-13.
- Peglar, S.M. & Birks, H.J.B. (1993) The mid-Holocene Ulmus fall at Diss Mere, South-East England - disease and human impact? Vegetation History and Archaeobotany, 2, 61-68.
- Peltier, W.R. (2002) On esutatic sea level history: Last Glacial Maximum to Holocene. *Quaternary Science Reviews*, **21**, 377-396.

- Pennington, W. (1986) Lags in adjustment of vegetation to climate caused by the pace of soil development. Evidence from Britain. *Plant Ecology*, **67**, 105-118.
- Perry, I. & Moore, P.D. (1987) Dutch elm disease as an analogue of Neolithic elm decline. *Nature*, **326**, 72-73.
- Petersen, K.-S. (1976) Om Limfjordens postglaciale marine udvikling og niveauforhold, belyst ved mollusc-faunaen og C-14 dateringer. *Danmarks Geologiske Undersøgelse*, Årbog 1975, **1975**, 75-103.
- Petersen, K.-S. (1980) Den Holocæne marine trangression og molluskfaunen i Hanherred beleyst ud fra en boring ved Vust. Dansk Geologisk Forening. Åsskrift for 1979, 15-17.
- Petersen, K.-S. (1981) The Holocene marine transgression and its molluscan fauna in the Skagerrak-Limfjord region, Denmark. *Special Publications International Association of Sedimentologists* **5**, 497-503.
- Petersen, K.-S. (1985) The Weichselian icesheets and land/sea configuration in the late Pleistocene and Holocene. *Journal of Danish Archaeology*, **4**, 7-22.
- Petersen, K.-S. (1986) An outline of the present stage of study on late Quaternary marine molluscs in the Nordic realm. Nordic late Quaternary biology and ecology; Eighteenth symposium. Societas Upsaliensis Pro Geologia Quaternaria (ed L.-K. Königsson), Vol. 24, pp. 39-45. *Striae*, Uppsala.
- Petersen, K.-S. (1993) Environmental changes recorded in the Holocene molluscan faunas from Djursland, Denmark. *Scripta Geologica Special Issue*, **2**, 359-369.
- Petersen, K.S. (2004) Late Quaternary environmental changes recorded in the Danish marine molluscan faunas. *Geological Survey of Denmark and Greenland Bulletin*, **3**, 268.
- Petersen, K.S., Rasmussen, K.L., Rasmussen, P., & von Platen-Hallermund, F. (2005) Main environmental changes since the Weichselian glaciation in the Danish waters between the North Sea and the Baltic Sea as reflected in the molluscan fauna. *Quaternary International*, **133**, 33-46.
- Petersson, H. (1996-97) Where did all the farmers come from? *Journal of Danish Archaeology*, **13**, 179-190.
- Pienitz, R., Smol, J.P., Last, W.M., Leavitt, P.R., & Cumming, B.F. (2000) Multi-proxy Holocene palaeoclimatic record from a saline lake in the Canadian Subarctic. *Holocene*, **10**, 673-686.
- Poulsen, B., Holm, P., & MacKenzie, B.R. (2007) A long-term (1667-1860) perspective on impacts of fishing and environmental variability for herring, eel. and whitefish in the Limfjord, Denmark. *Fisheries Research*, 2-3, 181-195.

Poulsen, K.L. (1978) Eisenzeitliche Muschelhaufen in Dänemark. Offa, 35, 64-85.

- Prasad, A.K.S.K., Nienow, J.A., & Livingston, R.J. (1990) The Genus Cyclotella (Bacillariophyta) in Choctawhatchee Bay, Florida, with Special Reference to C Striata and C Choctawhatcheeana Sp-Nov. Phycologia, 29, 418-436.
- Price, D.T. (1985). Affluent foragers of Mesotithic southern Scandinavia. In *Prehistoric Ihmter-Gatherers. The Emergence of Cultural Complexity* (eds D.T. Price & J.A. Brown), pp. 341-363. Academic Press, New York.
- Price, T.D. (1991) The Mesolithic of Northern Europe. *Annual Review of Anthropology*, **20**, 211-233.
- Punning, J.-M., Martma, T., Kessel, H., & Vaikme, R. (1988) The isotopic composition of oxygen and carbon in the subfossil mollusc shells of the Baltic Sea as an indicator of palaeosalinity. *Boreas*, **17**, 27-31.
- Rabalais, N.N., Turner, R.E, Sen Gupta, B.K., Platon, E., and Parsons, M.L. (2007) Sediments tell the history of eutrophication and hypoxia in the northern Gulf of Mexico. *Ecological Applications*, **17**, S129-143.
- Racca, J.M.J. & Prairie, Y.T. (2004) Apparent and real bias in numerical transfer function in palaeolimnology. *Journal of Paleolimnology*, **31**, 117-124.

Ramsey, C.B. (2008) Deposition models for chronological records. *Quaternary Science Reviews*, **27**, 42-60.

Ramsey, C.B. (2009) Bayesian analysis of radiocarbon dates. Radiocarbon, 51, 337-360.

Rasmussen, E. (1958) Past and Present Distribution of *Tapes* (Venerupis) *pullastra* (Montagu) in Danish Waters. *Oikos*, **9**, 77-93.

- Rasmussen, H. (1968). Limfjordsfiskeriet før 1825. Sædvane og centraldirigering. Folkelivs studier 2, Copenhagen
- Rasmussen, L.W. (1993a). Pitted Ware settlements. In *Digging into the past. 25 years of archaeology in Denmark* (eds S. Hvass & B. Storgaard), pp. 114-115. Aarhus Universitetsforlag, Aarhus.
- Rasmussen, L.W. & Richter, J. (1991) Kainsbakke. En kystboplads fra yngre stenalder. Aspects of the palaeoecology of neolithic man. Djurslands Museum/Dansk Fiskerimuseum, pp. 127.
- Rasmussen, M. (1993b) *Bopladskeramik i ældre bronzealder*. I kommission hos Aarhus universitetsforlag, Århus, pp. 148.
- Rasmussen, P. (2005) Mid- to late-Holocene land-use change and lake development at Dallund Sø, Denmark: vegetation and land-use history inferred from pollen data. *Holocene*, **15**, 1116-1129.
- Rasmussen, P. (submitted). The terrestrial and marine palaeoenvironment at Norsminde Fjord, East Jutland, c.6900-1700 cal. yr BC: study aims, site description, formations of the fjord, sedimentation history and hydrography. In *Stone Age Settlement in the coastal Fjord of Norsminde, Jutland, Denmark* (ed S.H. Andersen).
- Rasmussen, P. & Bradshaw, E.G. (2005) Mid- to late-Holocene land-use change and lake development at Dallund Sø, Denmark: study aims, natural and cultural setting, chronology and soil erosion history. *Holocene*, **15**, 1105-1115.
- Rasmussen, P. & Olsen, J. (2009) Soil erosion and land-use change during the last six millenia recorded in lake sediments from Gudme Sø, Fyn, Denmark. *Geological Survey of Denmark and Greenland Bulletin*, **17**, 37-40.
- Rasmussen, P., Petersen, K.S., & Ryves, D.B. (2007) Environmental change in Danish marine waters during the Roman Warm Period inferred from mollusc data. *Geological Survey of Denmark and Greenland Bulletin*, **13**, 21-24.
- Reed, J.M. (1998) A diatom-conductivity transfer function for Spanish salt lakes. *Journal of Paleolimnology*, **19**, 399-416.
- Reimer, P.J., Baillie, M.G.L., Bard, E., Bayliss, A., Beck, J.W., Bertrand, C.J.H., Blackwell, P.G., Buck, C.E., Burr, G.S., Cutler, K.B., Damon, P.E., Edwards, R.L., Fairbanks, R.G., Friedrich, M., Guilderson, T.P., Hogg, A.G., Hughen, K.A., Kromer, B., McCormac, F.G., Manning, S.W., Ramsey, C.B., Reimer, R.W., Remmele, S., Southon, J.R., Stuiver, M., Talamo, S., Taylor, F.W., van der Plicht, J., & Weyhenmeyer, C.E. (2004) IntCal04 Terrestrial radiocarbon age calibration, 26 0 ka BP. *Radiocarbon*, 46, 1029-1058.
- Renberg, I. (1990) A Procedure for Preparing Large Sets of Diatom Slides from Sediment Cores. *Journal of Paleolimnology*, **4**, 87-90.
- Renberg, I. (1991) The HON-Kajak sediment corer. Journal of Paleolimnology, 6, 167-170.
- Reuss, N. (2005a) Sediment pigments as biomarkers of environmental change. Ph.D. thesis. National Environmental Research Institute, Roskilde, Denmark.
- Reuss, N., and Conley, D.J. (2005b) Effects of sediment storage conditions on pigment analyses. *Limnology and Oceanography: Methods*, **3**, 477-487.
- Reuss, N., Conley, D.J., & Bianchi, T.S. (2005) Preservation conditions and the use of sediment pigments as a tool for recent ecological reconstruction in four Northern European estuaries. *Marine Chemistry*, 95, 283-302.

- Richards, M.P. & Hedges, R.E.M. (1999) Stable Isotope Evidence for Similarities in the Types of Marine Foods Used by Late Mesolithic Humans at Sites Along the Atlantic Coast of Europe. *Journal of Archaeological Science*, **26**, 717-722.
- Richards, M.P. & Schulting, R.J. (2006a) Against the grain? A response to Milner et al (2004). *Antiquity*, **80**, 444-458.
- Richards, M.P. & Schulting, R.J. (2006b) Touch not the fish: the Mesolithic-Neolithic change of diet and its significance. *Antiquity*, **80**, 444-456.
- Rines, J.E.B. & Hargraves, P.E. (1988) *The Chaetoceros Ehrenberg (Bacillariophyceae) flora* of Narragansett Bay, Rhode Island, U.S.A. J. Crammer, Berlin, Stuttgart, pp. 196.
- Ringkjøbing Amtskommune (1988). Kilen. Tilstand 1972-1986 samt udviklingsmuligheder. Ringkjøbning Amtskommune, Teknik- og Miljøforvaltningen, Recipientafdelingen, pp. 26.
- Ringkjøbning Amtskommune (1988). Kilen. Tilstand 1972-1986 samt udviklingsmuligheder, pp. 26.
- Ringtved, J. (1992). Fiskeri i jernalderen (500 f.K.r.-1050 e.K.r.) et overset erhvervsaspekt? Limfjordsprojektet. Rapport nr. 4: Limfjordsfiskeri i fortid og nutid, pp. 107-124.
- Rippey, B., Anderson, N.J., & Foy, R.H. (1997) Accuracy of diatom inferred total phosphorus concentrations and the accelerated eutrophication of a lake due to reduced flushing and increased internal loading. *Canadian Journal of Fisheries and Aquatic Sciences*, 54, 2637-2646.
- Rodhe, J. (1987) The large-scale circulation in the Skagerrak; interpretation of some observations. *Tellus*, **39A**, 245-253.
- Rodhe, J. (1992) Studies of currents and mixing in the Skagerrak. Ph.D Thesis, University of Göteborg, Göteborg, Sweden.
- Rodhe, J. (1996) On the dynamics of the large-scale circulation of the Skagerrak. *Journal of Sea Research*, **35**, 9-21.
- Rohling, E.J. & Pälike, H. (2005) Centennial-scale climate cooling with a sudden cold event around 8,200 years ago. *Nature*, **434**, 975-979.
- Rosenberg, R., Elmgren, R., Flesicher, S., Jonsson, P., Persson, G., & Dahlin, H. (1990) Marine eutrophication in Sweden. *Environmetrics*, **2**, 263-278.
- Round, F.E., Crawford, R.M., & Mann, D.G. (1990) *The diatoms. Biology and morphology of the genera*. Cambridge University Press, Cambridge, pp. 760.
- Rowley-Conwy, P. (1984) The Laziness of the Short-Distance Hunter: The Origins of Agriculture in Western Denmark. *Journal of Anthropological Archaeology*, **3**, 300-324.
- Rowley-Conwy, P. (1985) The origin of agriculture in Denmark; a review of some theories. *Journal of Danish Archaeology*, **4**, 188-195.
- Ryves, D.B., Battarbee, R.W., & Fritz, S.C. (2009) The dilemma of disappearing diatoms: Incorporating diatom dissolution data into palaeoenvironmental modelling and reconstruction. *Quaternary Science Reviews*, **28**, 120-136.
- Ryves, D.B., Battarbee, R.W., Juggins, S., Fritz, S.C., & Anderson, N.J. (2006) Physical and chemical predictors of diatom dissolution in freshwater and saline lake sediments in North America and West Greenland. *Limnology and Oceanography*, **51**, 1355-1368.
- Ryves, D.B., Clarke, A.L., Appleby, P.G., Amsinck, S.L., Jeppesen, E., Landkildehus, F., & Anderson, N.J. (2004) Reconstructing the salinity and environment of the Limfjord and Vejlerne Nature Reserve, Denmark, using a diatom model for brackish lakes and fjords. *Canadian Journal of Fisheries and Aquatic Sciences*, **61**, 1988-2006.
- Ryves, D.B., Juggins, S., Fritz, S.C., & Battarbee, R.W. (2001) Experimental diatom dissolution and the quantification of microfossil preservation in sediments. *Palaeogeography Palaeoclimatology Palaeoecology*, **172**, 99-113.

- Ryves, D.B., McGowan, S., & Anderson, N.J. (2002) Development and evaluation of a diatom-conductivity model from lakes in West Greenland. *Freshwater Biology*, 47, 995-1014.
- Saarse, L., Heinsalu, A., & Veski, S. (2009) Litorina Sea sediments of ancient Vaana Lagoon, northwestern Estonia. *Estonian Journal of Earth Sciences*, **58**, 85-93.
- Sabbe, K. & Vyverman, W. (1995) Taxonomy, morphology and ecology of some widespread representatives of the diatom genus Opephora. *European Journal of Phycology*, 30, 235-249.
- Scheffera, M., Carpenter, S., & de Young, B. (2005) Cascading effects of overfishing marine systems. *Trends in Ecology & Evolution*, 20, 579-581.
- Scheurle, C. & Hebbeln, D. (2003) Stable oxygen isotopes as recorders of salinity and river discharge in the German Bight, North Sea. *Geo-Marine Letters*, **23**, 130-136.
- Schmölcke, U., Endtmann, E., Klooss, S., Meyer, M., Michaelis, D., Rickert, B.H., & Rossler, D. (2006) Changes of sea level, landscape and culture: A review of the south-western Baltic area between 8800 and 4000BC. *Palaeogeography Palaeoclimatology Palaeoecology*, 240, 423-438.
- Schoeninger, M.J., DeNiro, M.J., & Tauber, H. (1983) Stable Nitrogen Isotope Ratios of Bone Collagen Reflect Marine and Terrestrial Components of Prehistoric Human Diet. *Science*, 220, 1381-1383.
- Schweikhardt, P., Doris Sloan, D., & Lynn Ingram, B. (2010) Early Holocene Evolution of San Francisco Estuary, Northern California. *Journal of Coastal Research*, 26, 704-713.
- Scourse, J.D., Kennedy, H., Scott, G.A., & Austin, W.E.N. (2004) Stable isotopic analyses of modern benthic foraminifera from seasonally stratified shelf seas: disequilibria and 'the seasonal' effect. *The Holocene*, 14, 747-758.
- Seppä, H. & Birks, H.J.B. (2001) July mean temperature and annual precipitation trends during the Holocene in the Fennoscandian tree-line area: pollen-based climate resonstructions. *Holocene*, **11**, 527-539.
- Seppä, H., Bjune, A.E., Telford, R.J., Birks, H.J.B., & Veski, S. (2009) Last nine-thousand years of temperature variability in Northern Europe. *Climate of the Past*, **5**, 523–535.
- Seppä, H., Tikkanen, M., & Shemeikka, P. (2000) Late-Holocene shore displacement of the Finnish south coast: diatom, litho- and chemostratigraphic evidence from three isolation basins. *Boreas*, **29**, 219-231.
- Shennan, I., Innes, J.B., Long, A.J., & Zong, Y. (1993) Late Devensian and Holocene relative sea-level changes at Rumach, near Arisaig, northwest Scotland. Norsk. Geologisk. *Tidsskrift*, **73**, 161-174.
- Siddall, M., Rohling, E.J., Almogi-Labin, A., Hemleben, C., Meischner, D., Schmelzer, I., & Smeed, D.A. (2003) Sea-level fluctuations during the last glacial cycle. *Nature*, 423, 853-858.
- Sinninghe Damste, J.S., Muyzer, G., Abbas, B., Rampen, S.W., Masse, G., Allard, W.G., Belt, S.T., Robert, J.-M., Rowland, S.J., Moldowan, J.M., Barbanti, S.M., Fago, F.J., Denisevich, P., Dahl, J., Trindade, L.A.F., & Schouten, S.S., 304:584-587 (2004) The rise of the rhizosolenid diatoms. *Science*, **304**, 584-587.
- Sloss, C.R., Jones, B.G., Murray-Wallace, C.V., & McClennen, C.E. (2005) Holocene Sea Level Fluctuations and the Sedimentary Evolution of a Barrier Estuary: Lake Illawarra, New South Wales, Australia. *Journal of Coastal Research*, 21, 943-959.
- Smith, V.H. (2006) Eutrophication of freshwater and coastal marine ecosystems a global problem. *Environmental Science and Pollution Research*, **10**, 126-139.
- Snoeijs, P., Vilbaste, S., Potapova, M., Kasperoviciene, J., & Balashova, N. (1993-1998) Intercalibration and distribution of diatom species in the Baltic Sea. Opulus Press, Uppsala

- Snowball, I., Korhola, A., Briffa, K.R., & Koç, N. (2004). Holocene climate dynamics in Fennoscandia and the North Atlantic. In *Past climate variability through Europe and Africa* (eds R.W. Batterbee, F. Gasse & C.E. Stickely), pp. 364-397. Kluwer Academic Publishers, Dordrecht, The Netherlands.
- Sohlenius, G., Emeis, K.C., Andrén, E., Andrén, T., & Kohly, A. (2001) Development of anoxia during the Holocene fresh-brackish water transition in the Baltic Sea. *Marine Geology*, **177**, 221-242.
- Sohlenius, G., Sternbeck, J., Andrén, E., & Westman, P. (1996) Holocene history of the Baltic Sea as recorded in a sediment core from the Gotland Deep. *Marine Geology*, **134**, 183-201.
- Sohlenius, G. & Westman, P. (1998) Salinity and redox alternations in the northwestern Baltic proper during the late Holocene. *Boreas*, **27**, 101-114.
- Solomon, S., Qin, D., Manning, M., Chen, Z., Marquis, M., Averyt, K.B., Tignor, M., & Miller, H.L., eds. (2007) Climate Change 2007: The Physical Science Basis. Contribution of Working Group I to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change, pp 996. Cambridge University Press, Cambridge, New York.
- Sommer, R.S., Persson, A., Wieseke, N., & Fritz, U. (2007) Holocene recolonization and extinction of the pond turtle, Emys orbicularis (L., 1758), in Europe. *Quaternary Science Reviews*, **26**, 3099-3107.
- Sørensen, R. (1979) Late Weichselian deglaciation in the Oslofjord area, south Norway. *Boreas*, **8**, 241–246.
- Sorgenfrei, T. (1958) Molluscan assemblages from the marine middle Miocene of South Jutland and their environment. *Danmarks Geologiske Undersøgelse, II*, **79**, 356-503.
- Spaulding, S. & Edlund, M. (2009) *Ellerbeckia*. In Diatoms of the United States. <u>http://westerndiatoms.colorado.edu/taxa/genus/Ellerbeckia</u>.
- Stabell, B. (1985) The development and succession of taxa within the diatom genus *Fragilaria* Lyngbye as a response to basin isolation from the sea. *Boreas*, **14**, 273-286.
- Stabell, B. & Thiede, J. (1986) Paleobathymetry and paleogeography of southern Scandinavia in the late Quaternary. *Meyniana*, **38**, 43-59.
- Stafford, M. (1999) From forager to farmer in flint. A lithic analysis of the prehistoric transition to agriculture in southern Scandinavia. Aarhus University Press, Århus, Denmark, pp. 147.
- Stanley, D. & Warne, A. (1997) Holocene sea level change and early human utilization of deltas. GSA Today, 7, 1-8.
- Stanley, D.J. & Chen, Z. (1996) Neolithic settlement distributions as a function of sea levelcontrolled topography in the Yangtze delta, China. *Geology*, **24**, 1083–1086.
- Stanley, D.J. & Warne, A.G. (1993) Sea level and initiation of Predynastic culture in the Nile delta. *Nature*, 363, 435–438.
- Stuiver, M. & Braziunas, T.F. (1993) Modeling Atmospheric C-14 Influences and C-14 Ages of Marine Samples to 10,000 BC. *Radiocarbon*, **35**, 137-189.
- Stuiver, M. & Polach, H.A. (1977) Reporting of C-14 data. Radiocarbon, 19, 355-363.
- Sundelin, U. (1922) Några ord angående förläggningen av L.G. i de av transgression ej drabbade delarna av det baltiska området samt angående tidpunkten för Litorinahavets inträde. *Geologiska Föreningen i Stockholm Förhandlingar*, **44**, 543–544.
- Svansson, A. (1975). Physical and Chemical Oceanography of the Skagerrak and the Kattegat. Report No. 1, Fishery Board of Sweden. Institute of Marine Research, Lysekil, Sweden, pp. 1-77.
- Tanner, W.F. (1993) An 8000-year record of sea-level change from grain size parameters: data from beach ridges in Denmark. *The Holocene*, **3**, 220-231.
- Tauber, H. (1981) ¹³C evidence for dietary habits of prehistoric man in Denmark. *Nature*, **292**, 332-333.

Telford, R.J., Andersson, C., Birks, H.J.B., & Juggins, S. (2004) Biases in the estimation of transfer function prediction errors. *Paleoceanography*, **19**, PA4014.

- Telford, R.J. & Birks, H.J.B. (2005) The secret assumption of transfer functions: problems with spatial autocorrelation in evaluating model performance. *Quaternary Science Reviews*, **24**, 2173-2179.
- Telford, R.J. & Birks, H.J.B. (2009) Evaluation of transfer functions in spatially structured environments. *Quaternary Science Reviews*, **28**, 1309-1316.

Ten Hove, H.A. (1968) The Ulmus fall at the transition Atlanticum–Subboreal. *Palaeogeography, Palaeoclimatology, Palaeoecology*, **5**, 359–369.

ter Braak, C.J.F. & Juggins, S. (1993) Weighted averaging partial least squares regression (WA-PLS): an improved method for reconstructing evironmental variable from species assemblages. *Hydrobiologia*, **269/270**, 485-502.

ter Braak, C.J.F. & Šmilauer, P. (2002) CANOCO Reference manual and CanoDraw for Windows User's guide: Software for Canonical Community Ordination (version 4.5). Microcomputer Power, Ithaca, NY, USA, pp. 500.

Thulin, B., Possnert, G., & Vuorela, I. (1992) Stratigraphy and age of two postglacial sediment cores from the Baltic Sea. *Geologiska Föreningen i Stockholm Förhandlingar*, **114**, 165–179.

Treguer, P., Nelson, D.M., Van Bennekom, A.J., DeMaster, D.J., Leynaert, A., & Queguiner, B. (1995) The silica balance in the world ocean: a reestimate. *Science*, **268**, 375-379.

Troels-Smith, J. (1942). Geologisk datering af Dyrholm-fundet. In Dyrholmen. Det Kongelige Danske Videnskaberens Selskab, Arkæologiske-Kunsthistoriske Skrifter (eds T. Mathiassen, M. Degerbøl & J. Troels-Smith), Vol. I (1), pp. 139-212.

Troels-Smith, J. (1955) Karakterisering af løse jordater (Characterisation of unconsolidated sediments). *Denmarks Geologiske Undersogelse, Series IV/3*, **10**, 1-73.

- Troels-Smith, J. (1960) Ivy, mistletoe and elm. Climate indicators fodder plants. *Danmarks Geologiske Undersogelse, Series IV/3*, **10**, 1-73.
- Troels-Smith, J. (1995). Claudi-kiler, østersbanker og tidevand. In *Stavns Fjord et natur- og kulturhistorisk forskningsområ på Samsø* (eds H.H. Hansen & B. Aaby), pp. 59-67. Carlsbergfondet og Nationalmuseet, Copenhagen.

Turney, C.S.M. & Lowe, J.J. (2001). Tephrochronology. In *Tracking Environmental Change* in Lake Sediments (eds W.M. Last & J.P. Smol), Vol. 1, pp. 451-471. Kluwer Academic Publishers, Dordrecht, The Netherlands.

Tutin, T.G., Heywood, V.H., Burges, N.A., Moore, D.M., Valentine, D.H., Walters, S.M., & Webb, D.A. (1964-1980) *Flora Europaea*. Cambridge University Press, Cambridge, pp. 2415.

Uehara, K., Scourse, J.D., Horsburgh, K.J., Lambeck, K., & Purcell, A.P. (2006) Tidal evolution of the northwest European shelf seas from the Last Glacial Maximum to the present. *Journal of Geophysical Research-Oceans*, **111**, C09025.

Underwood, G.J.C., Phillips, J., & Saunders, K. (1998) Distribution of estuarine benthic diatom species along salinity and nutrient gradients. *European Journal of Phycology*, 33, 173-183.

Underwood, G.J.C. & Provot, L. (2000) Determining the environmental preferences of four estuarine epipelic diatom taxa: growth across a range of salinity, nitrate and ammonium conditions. *European Journal of Phycology*, **35**, 173-182.

van der Werff, A. & Huls, H. (1958-1974) *Diatomeënflora van Nederland*. Koeltz Science Publishers, Koenigstein

van Weering, T.C.E. (1982a) Recent sediments and sediment transport in the northern North Sea; pistoncores from Skagerrak. *Proceedings of the Koninklijke Nederlandse Akademie van Wetenschappen. Series B*, **85**, 155-201. van Weering, T.C.E. (1982b) Shallow seismic and acoustic reflection profiles from the Skagerrak; implications for recent sedimentation. *Proceedings of the Koninklijke Nederlandse Akademie van Wetenschappen. Series B*, **85**, 129-154.

Vang Petersen, P. (1999) Flint fra Denmarks Oldtid. Høst and Søn, Copenhagen

- Veihe, A., Hasholt, B., & Schiøtz, I.G. (2003) Soil erosion in Denmark: processes and politics. *Environmental Science & Policy*, 6, 37-50.
- Velle, G., Brodersen, K.P., Birks, H.J.B., & Willassen, E. (2010) Midges as quantitative temperature indicator species: Lessons for palaeoecology. *Holocene*, 20, 989-1002.
- von Tynni, R. (1978) Über Finnlands Rezente Und Subfossile Diatomeen, X. *Geological Survey of Finland Bulletin*, **296**, 55.
- Wachnicka, A., Gaiser, E., Collins, L., Frankovich, T., & Boyer, J. (2010) Distribution of Diatoms and Development of Diatom-Based Models for Inferring Salinity and Nutrient Concentrations in Florida Bay and Adjacent Coastal Wetlands of South Florida (USA). *Estuaries and Coasts*, **33**, 1080–1098.
- Weckström, K. (2005) Recent eutrophication of coastal waters of southern Finland A palaeolimnological assessment. Ph.D, University of Helsinki, Helsinki, Finland.
- Weckström, K. (2006) Assessing recent eutrophication in coastal waters of the Gulf of Finland (Baltic Sea) using subfossil diatoms. *Journal of Paleolimnology*, **35**, 571-592.
- Weckström, K. & Juggins, S. (2006) Coastal diatom-environment relationships from the Gulf of Finland, Baltic Sea. *Journal of Phycology*, 42, 21-35.
- Weckström, K., Juggins, S., & Korhola, A. (2004) Quantifying background nutrient concentrations in coastal waters: A case study from an urban embayment of the Baltic Sea. *Ambio*, **33**, 324-327.
- Weckström, K., Lewis, J.P., Andrén, E., Ellegaard, M., Rasmussen, P., Ryves, D.B., & Telford, R.J. (in review). Palaeoenvironmental history of the Baltic Sea - one of the largest brackish-water ecosystems in the world. In *Applications of paleoenvironmental techniques to estuarine systems* (eds J. Gibson, P. Gell, J. Tibby & K. Saunders).
- Westman, P. & Sohlenius, G. (1999) Diatom stratigraphy in five offshore sediment cores from the northwestern Baltic proper implying large-scale circulation changes during the last 8500 years. *Journal of Paleolimnology*, 22, 53-69.
- Widerlund, A. & Andersson, P.S. (2006) Strontium isotopic composition of modem and Holocene mollusc shells as a palaeosalinity indicator for the Baltic Sea. *Chemical Geology*, 232, 54-66.
- Wiley, M., ed. (1976) *Estuarine processes v.1: Uses, stresses and adaptation to the estuary*, pp 541. Academic Press, New York and London.
- Williams, J.W., Post, D.M., Cwynar, L.C., Lotter, A.F., & Levesque, A.J. (2002) Rapid and widespread vegetation responses to past climate change in the North Atlantic region. *Geology*, **30**, 971-974.
- Windolf, J., Jeppesen, E., Jensen, J.P., & Kristensen, P. (1996) Modelling of seasonal variation in nitrogen retention and in-lake concentration: A four-year mass balance study in 16 shallow Danish lakes. *Biogeochemistry*, **33**, 25-44.
- Winn, K., Averdieck, F.-R., Erlenkeuser, H., & Werner, F. (1988) Hydrography of the Kiel Bay, Western Baltic, during the Littorina transgression. *Meyniana*, **40**, 61-80.
- Winn, K., Erlenkeuser, H., Nordberg, K., & Gustafsson, M. (1998) Paleohydrography of the Great Belt, Denmark, during the Littorina transgression: the isotope signal. *Meyniana*, 50, 237-251.
- Winsor, P., Rodhe, J., & Omstedt, A. (2001) Baltic Sea ocean climate: an analysis of 100 yr of hydrographic data with focus on the freshwater budget. *Climate Research*, **18**, 5-15.
- Witkowski, A. (1994) Recent and fossil diatom flora of the Gulf of Gdańsk, Southen Baltic Sea. Origin, composition and changes of diatom assemblages during the Holocene. J. Crammer, Berlin, Stuttgart, pp. 312.

- Witkowski, A., Cedro, B., Kierzek, A., & Baranowski, D. (2009) Diatoms as a proxy in reconstructing the Holocene environmental changes in the south-western Baltic Sea: the lower Rega River Valley sedimentary record. *Hydrobiologia*, **631**, 155-172.
- Witkowski, A., Lange-Bertalot, H., & Metzeltin, D. (2000) *Diatom flora of marine coasts I*. A.R.G. Gantner Verlag K.G., Ruggell, pp. 925.
- Woolf, D.K., Shaw, A.G.P., & Tsimplis, M.N. (2003) The influence of the North Atlantic Oscillation on sea-level variability in the North Atlantic region. *Journal of Atmospheric & Ocean Science*, 9, 145 - 167.
- Wulff, A., Vilbaste, S., & Truu, J. (2005) Depth distribution of photosynthetic pigments and diatoms in the sediments of a microtidal fjord. *Hydrobiologia*, **534**, 117-130.
- Wulff, F., Stigebrandt, A., & Rahm, L. (1990) Nutrient dynamics of the Baltic Sea. *Ambio*, **19**, 126–33.
- Yonge, C.M. (1960) Oysters. Collins, London, pp. 209.
- Yu, S.Y., Andren, E., Barnekow, L., Berglund, B.E., & Sandgren, P. (2003) Holocene palaeoecology and shoreline displacement on the Biskopsmala Peninsula, southeastern Sweden. *Boreas*, 32, 578-589.
- Yu, S.Y., Berglund, B.E., Sandgren, P., & Lambeck, K. (2007) Evidence for a rapid sea-level rise 7600 yr ago. *Geology*, 35, 891-894.
- Yuan, L.L. (2005) Sources of bias in weighted average inferences of environmental conditions. *Journal of Paleolimnology*, 34, 245–255
- Zhang, Q., Zhu, C., Liu, C.L., & Jiang, T. (2005) Environmental change and its impacts on human settlement in the Yangtze Delta, P.R. China. *Catena*, **60**, 267–277.
- Zillén, L., Conley, D.J., Andrén, T., Andrén, E., & Björck, S. (2008) Past occurrences of hypoxia in the Baltic Sea and the role of climate variability, environmental change and human impact. *Earth-Science Reviews*, **91**, 77-92.
- Zong, Y. & Horton, B.P. (1999) Diatom-based tidal-level transfer functions as an aid reconstructing Quaternary history of sea-level movements in the UK. *Journal of Quaternary Science*, 14, 153-167.
- Zorita, E. & Laine, A. (2000) Dependence of salinity and oxygen concentrations in the Baltic Sea on large-scale atmospheric circulation. *Climate Research*, **14**, 25-41.

Appendix A. Paralia sulcata dissolution stages



Stage 1 - Valves in pristine condition showing very little or no signs of dissolution.



Stage 2 - Valves clearly showing all of the main identification features visible under light microscopy but exhibiting some preliminary signs of dissolution. This might include slight wearing of the marginal rim and/or signs of the radial markings beginning to dissolve.



Stage 3 - Valves showing clear advanced signs of dissolution being in a relatively poor state. Marginal rim showing substantial dissolution. Radial markings generally visible, but in some specimens can be poorly preserved appearing as irregularly shaped silica blotches on the valve surface.



Stage 4 - Valves in very poor state with outer margin (almost) completely dissolved away and radial markings commonly showing clear signs of dissolution, more regularly appearing as silica blotches on the valve surface (e.g. S4a). In some valves radial markings might even be absent. (NB. Scale unavailable for photo S4a).

*All *Paralia sulcata* dissolution images are taken from fossil material present in Danish coastal sediments.